## **Domain Perturbation for Parabolic Equations**

Parinya Sa Ngiamsunthorn

A thesis submitted in fulfillment of the requirements for the degree of Doctor of Philosophy



School of Mathematics and Statistics

July 2011

## Abstract

We study the effect of domain perturbation on the behaviour of parabolic equations. The first aspect considered in this thesis is the behaviour of solutions under changes of the domain. We show how solutions of linear and semilinear parabolic equations behave as a sequence of domains  $\Omega_n$  converges to an open set  $\Omega$  in a certain sense. In particular, we are interested in singular domain perturbations so that a change of variables is not possible on these domains. For autonomous linear equations, it is known that convergence of solutions under domain perturbation is closely related to the corresponding elliptic equations via a standard semigroup theory. We show that there is also a relation between domain perturbation for *non-autonomous* linear parabolic equations and domain perturbation for elliptic equations. The key result for this is the equivalence of Mosco convergences between various closed and convex subsets of Banach spaces. An important consequence is that the same conditions for a sequence of domains imply convergence of solutions under domain perturbation for both parabolic and elliptic equations. By applying variational methods, we obtain the convergence of solutions of initial value problems under Dirichlet or Neumann boundary conditions. A similar technique can be applied to obtain the convergence of weak solutions of parabolic variational inequalities when the underlying convex set is perturbed. Using the linear theory, we then study domain perturbation for initial boundary value problems of semilinear type. We are also interested in the behaviour of bounded entire solutions of parabolic equations defined on the whole real line. We establish a convergence result for bounded entire solutions of linear parabolic equations under  $L^2$  and  $L^p$ -norms. For the  $L^p$ -theory, we also prove Hölder regularity of bounded entire solutions with respect to time. In addition, the persistence of some classes of bounded entire solutions is given for semilinear equations using the Leray-Schauder degree theory.

The second aspect is to study the dynamics of parabolic equations under domain perturbation. In this part, we consider parabolic equation as a dynamical system in an

## ABSTRACT

 $L^2$  space and study the stability of invariant manifolds near a stationary solution. In particular, we prove the continuity (upper and lower semicontinuity) of both, the local stable invariant manifolds and the local unstable invariant manifolds under domain perturbation.

# Acknowledgements

This thesis would not have been possible without the support and the guidance of several individuals who in one way or another contributed and extended their valuable assistance in the preparation and completion of this work. First and foremost, I would like to express my utmost gratitude to my supervisor, Dr. Daniel Daners, for his support, warm encouragement and thoughtful guidance. I sincerely thank him for always taking the time to help me whenever I visited his office.

I would also like to acknowledge my associate supervisor, Prof. E. N. Dancer, for his help and advice in the later stage of this work. Although the problem remains unsolved, his advice is definitely invaluable to my future research career.

I would like to thank my family, especially my parents who have been a constant source of support. The formal and informal support from other staffs and fellow postgraduate students in the School of Mathematics and Statistics is also appreciated. Finally, I gratefully acknowledge the Institute for the Promotion of Teaching Science and Technology (IPST), Thailand, for financial support.

# Statement of originality

This thesis contains no material which has been accepted for the award of any other degree or diploma. All work in this thesis, except where due acknowledgement is made, is believed to be original.

> Parinya Sa Ngiamsunthorn July 2011

# Contents

Α	Abstract						
A	ckno	wledgements	iii				
St	aten	nent of originality	iv				
In	trod	uction	1				
1	Dor	nain perturbation for linear non-autonomous parabolic equations	9				
	1.1	Abstract parabolic equations	10				
	1.2	Mosco Convergence	12				
	1.3	Equivalence of Mosco convergences for parabolic and elliptic equations .	14				
	1.4	Convergence of solutions of initial value problems	21				
		1.4.1 Dirichlet problems	22				
		1.4.2 Neumann problems	32				
	1.5	Convergence of solutions of final value problems	42				
	1.6	Domain perturbation for parabolic variational inequalities	45				
2	Domain perturbation for semilinear parabolic equations						
	2.1	Evolution Systems	52				
	2.2	Existence of solutions of semilinear evolution equations	54				
	2.3	Convergence of solutions of semilinear evolution equations	58				
	2.4	Convergence of solutions of semilinear parabolic equations	64				
		2.4.1 Non-autonomous semilinear parabolic equations	64				
		2.4.2 Autonomous semilinear parabolic equations	69				

## CONTENTS

3	Invariant manifolds for parabolic equations under domain perturba-						
	tion	tion 7					
	3.1	I Introduction					
	3.2	2 Invariant manifolds for parabolic equations					
		3.2.1	Semiflows induced by abstract parabolic equations	79			
		3.2.2	Existence of invariant manifolds for parabolic equations	80			
	3.3	3 Domain perturbation for invariant manifolds					
	3.4	4 Some technical results towards the proof of semicontinuity					
		3.4.1	Convergence of sequence in finite dimensional subspaces	91			
		3.4.2	Characterisation of upper and lower semicontinuity	96			
	3.5	5 Convergence of unstable invariant manifolds					
		3.5.1	Convergence of global unstable manifolds	98			
		3.5.2	Upper and lower semicontinuity of local unstable manifolds $\ldots$	105			
	3.6	Conve	rgence of stable invariant manifolds	107			
4	Persistence of bounded entire solutions of parabolic equations under						
	domain perturbation						
	4.1 Bounded entire solutions of linear parabolic equations			118			
	4.2	.2 Convergence of bounded entire solutions of linear equations: $L^2$ -Theorem					
	4.3	3 $L^p$ -Theory for bounded entire solutions of linear parabolic equations .					
	4.4	Domain Perturbation for bounded entire solutions of semilinear					
		parabolic equations					
		4.4.1	Persistence of $C_0(\mathbb{R}, L^2(\Omega))$ solutions	140			
		4.4.2	Some remarks on almost periodic solutions	148			
Bibliography 151							

#### vi

# Introduction

The study of domain perturbation or sometimes referred to as "perturbation of the boundary" for boundary value problems is a special topic in perturbation problems. The main characteristic is that the differential operators and the solutions of the perturbed problems live in different spaces (depending on the domain of a boundary value problem). Domain perturbation appears to be a simple problem if we are only interested in smooth perturbation of the domain. This is because we could perform a change of variables to consider the perturbed problems in a fixed domain and only perturb the coefficients. Hence, it turns back to a standard perturbation problem and we may apply standard techniques such as the implicit function theorem, the Liapunov-Schmidt method and the transversality theorem. Nevertheless, difficulties arrive when we perform a change of variables is not possible, domain perturbation is even more challenging.

The fundamental question in domain perturbation is to look at how solutions behave upon varying domains. In particular, we would like to know when the solutions converge and what the limit problem is. This topic has been extensively studied for elliptic equations (see for example [37, 5, 39, 34, 30, 21, 22, 79] and references therein). Typically, the work of [37] provides a characterisation of domains  $\Omega_n, \Omega \subset \mathbb{R}^N$  for which solutions of the perturbed problem subject to Dirichlet boundary condition

$$-\Delta u = f_n \qquad \text{in } \Omega_n \tag{0.1}$$
$$u = 0 \qquad \text{on } \partial \Omega_n$$

converge (in some sense) to a solution of the limit problem

$$-\Delta u = f \qquad \text{in } \Omega$$
  

$$u = 0 \qquad \text{on } \partial \Omega.$$
(0.2)

Older techniques for the convergence of solutions of the Dirichlet problem for harmonic functions can be found in [58]. In the above work, the convergence result was referred to

#### INTRODUCTION

as "stability of solutions" under variation of the boundary of the domain. In [21, 22], a sufficient condition on domains for which the solutions converge is obtained for Neumann problems (u = 0 is replaced  $\frac{\partial u}{\partial \nu} = 0$  on  $\partial\Omega$ ). A more complicated behaviour occurs in the case of Robin problems ( $\frac{\partial u}{\partial \nu} + \alpha u = 0$ ,  $\alpha > 0$  on  $\partial\Omega$ ). Dancer and Daners [34] show that both convergence of solutions and boundary homogenisation (the limit problem prescribed by a different boundary condition) are possible depending largely on how domains  $\Omega_n$  converge to  $\Omega$ . A variation to elliptic equations with nonlinear boundary condition is studied in [7, 8, 9]. There are also results on nonlinear equations investigated in [31, 32]. On the other hand, an explicit estimate for solutions under domain perturbation in terms of a certain "distance" between  $\Omega_n$  and  $\Omega$  is the subject of [72].

Another aspect of domain perturbation for elliptic equations is the dependence of the spectrum of an elliptic operator on domains. This involves the study of the eigenvalue problem

$$-\Delta u + \lambda u = 0 \qquad \text{in } \Omega$$
  
$$Bu = 0 \qquad \text{on } \partial \Omega$$
 (0.3)

under domain perturbation where B is a boundary condition operator. For Dirichlet problem, it is known that the spectrum behaves continuously under a large class of domain perturbation as characterised in [37]. In contrast, the spectrum for Neumann problems is not well-behaved as seen in [29, page 420] or [6, 56]. A sufficient condition on domains for which the spectrum behaves continuously is given in [10]. Besides, an explicit estimate for the variation of eigenvalues on domains is considered in [24, 25, 26, 12].

The motivation to study domain perturbation comes from various sources. The main ones include shape optimization, solutions structure of nonlinear problems and numerical analysis. A classical shape optimisation is to find minimisers of shape functionals (functions where the unknown variable runs over a class of domains). For functionals arising from differential equations, the problem involves solving and extracting convergent subsequences of solutions of differential equations defined on a sequence of domains (see [27, 20, 23, 79]). This problem leads to the concept of  $\Gamma$ -convergence for a sequence of open sets and *Mosco convergence* for a sequence of function spaces related to differential equations. In particular, the question to establish a sufficient condition for the  $\Gamma$ -convergence in terms of geometrical properties of the domains has attracted interest in [23, 79, 19]. The solutions structure of nonlinear elliptic equations is the main concern in [31, 32]. It is shown that the number of positive solutions and solutions structure of nonlinear elliptic equations depends on geometric properties of the domain. Examples of domains are constructed to show that there can be arbitrarily many positive solutions. Lastly, in numerical analysis, the concept of *discrete approximation* of normed spaces leads to the study of perturbation of linear operators defined on a sequence of normed spaces to another sequence of normed spaces (see [48, 78, 75, 76, 77]). The generalised perturbation results (discrete convergence, discrete compactness of linear operators) give a general framework for approximation of solutions of linear equations and eigenvalue problems. A more recent work on numerical approximation of eigenvalues of domains with multiple cracks is studied in [16].

Much less result on domain perturbation seems to be known for parabolic equations. Rauch and Taylor [70] studied domain perturbation for autonomous equations. In addition, Hale and Vegas [51] investigated the persistence of stationary (equilibrium) solutions of nonlinear autonomous parabolic equations subject to Neumann boundary condition. Dancer [32, Section 5] briefly discussed non-autonomous equations but explicit time dependent is not allowed in higher order terms (equations can be reduced to autonomous problems by a change of time variable). The methods in these works are based on domain perturbation results for linear elliptic equations. In [70] and [32], convergence of solutions under domain perturbation is proved by means of convergence of resolvents or semigroups of the corresponding elliptic operators. The persistence of equilibrium solutions in [51] is obtained by applying the Liapunov-Schmidt method and the implicit function theorem to a bifurcation equation derived from the corresponding elliptic equations. A crucial result to study domain perturbation for autonomous parabolic equations from the corresponding elliptic equations is the convergence of *pseudo resol*vents established by Arendt [4, Theorem 5.2] (see its application in domian perturbation in [4, Section 6] and [38, Section 6]). For quasilinear parabolic equations, there are convergence results in [45] and [73] using the concept of Mosco convergence. The main work on domain perturbation for non-autonomous parabolic equations is provided by Daners [35]. The results of [35] include convergence of solutions for both linear and semilinear parabolic initial value problems subject to Dirichlet boundary condition as well as persistence of periodic solutions under domain perturbation. There is a different perspective and theory for non-autonomous parabolic equations in the monograph by Mierczyński and Shen [64]. In particular, fundamental properties of the principal spectrum and principal Liapunov exponents for non-autonomous (as well as random) parabolic equations

#### INTRODUCTION

are developed. They also study the effect of the shape of the domain on the principal spectrum and principal Liapunov exponents and extend the Faber-Krahn inequalities for elliptic and periodic parabolic problems to general non-autonomous and random parabolic equations in [64, Section 5.4].

The aim of this thesis is to develop a general theory for domain perturbation for both autonomous and non-autonomous parabolic equations subject to either Dirichlet or Neumann boundary conditions. We are particularly interested in singular domain perturbation so that change of variables is not possible on these domains. Moreover, we mostly do not assume any smoothness of the domains.

Let  $\Omega_n$  and  $\Omega$  be bounded domains in  $\mathbb{R}^N$ ,  $N \ge 2$  such that  $\Omega_n, \Omega \subset D$  for some ball  $D \in \mathbb{R}^N$ . We consider the following parabolic equation

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}(t)u = f(x, t, u) & \text{in } \Omega \times (0, T] \\ \mathcal{B}(t)u = 0 & \text{on } \partial\Omega \times (0, T] \\ u(\cdot, 0) = u_0 & \text{in } \Omega, \end{cases}$$
(0.4)

where  $\mathcal{A}$  is an elliptic operator of the form

$$\mathcal{A}(t)u := -\partial_i [a_{ij}(x,t)\partial_j u + a_i(x,t)u] + b_i(x,t)\partial_i u + c_0(x,t)u \tag{0.5}$$

and  $\mathcal{B}(t)$  is one of the following boundary operators

$$\mathcal{B}(t)u := u \qquad \text{Dirichlet boundary condition} \\ \mathcal{B}(t)u := [a_{ij}(x,t)\partial_j u + a_i(x,t)u] \nu_i \qquad \text{Neumann boundary condition.}$$
(0.6)

In the above, we use summation convention with i, j running from 1 to N. The boundary condition is considered in a weak sense as the (outer) unit normal vector  $\nu = (\nu_1, \ldots, \nu_N)$ may not exist. Also, we assume that  $a_{ij}, a_i, b_i$  and  $c_0$  are functions in  $L^{\infty}(D \times (0, T))$ and that there exists a constant  $\alpha > 0$  independent of  $(x, t) \in D \times (0, T)$  such that

$$a_{ij}(x,t)\xi_i\xi_j \ge \alpha |\xi|^2,$$

for all  $\xi \in \mathbb{R}^N$ . We study the perturbed equation

$$\begin{cases}
\frac{\partial u}{\partial t} + \mathcal{A}_n(t)u = f_n(x, t, u) & \text{in } \Omega_n \times (0, T] \\
\mathcal{B}_n(t)u = 0 & \text{on } \partial \Omega_n \times (0, T] \\
u(\cdot, 0) = u_{0,n} & \text{in } \Omega_n,
\end{cases}$$
(0.7)

where  $\mathcal{A}_n$  and  $\mathcal{B}_n$  are defined similarly as in (0.5) and (0.6), respectively.

In this thesis, we investigate two aspects of the above parabolic equations under domain perturbation. The first aspect is the convergence of solutions of linear and semilinear parabolic equations under domain perturbation. Most of the time, we follow a standard bilinear form setting so that existence and uniqueness of solutions for (0.4) and (0.7) (for linear problems) are available. Our main concern is to investigate conditions on domains  $\Omega_n$  and  $\Omega$  for which solutions of the perturbed problems converge to a solution of the limit problem. We also study the persistence of bounded entire solutions (solutions that are defined for all time  $t \in (-\infty, \infty)$  and are bounded in a suitable function space) of parabolic equations. The second aspect is to study the dynamics of parabolic equations under domain perturbation. In this part we consider parabolic equation as a dynamical system in an  $L^2$  space and study the persistence of invariant manifolds near a stationary solution of (0.4) when the domain  $\Omega$  is perturbed.

An overview of this thesis is given below.

We start our domain perturbation analysis for non-autonomous linear parabolic equations in Chapter 1. This is inspired by the work of Daners [35] for Dirichlet problems. Our main focus is hence the Neumann problem. In the case of autonomous linear parabolic equations, convergence of solutions under domain perturbation is closely related to the corresponding elliptic equations via the semigroups generated by the corresponding elliptic operators. Our results show that this also applies to non-autonomous problems using variational approach. The link between elliptic and parabolic equations is obtained by the equivalence of *Mosco convergence* established in Section 1.3. This result is an abstract generalisation of a similar equivalence of Mosco convergence of certain function spaces in [73]. The abstract approach here can be applied to both Dirichlet problem and Neumann problem. Section 1.4 presents convergence results for both boundary value problems. We note that Mosco convergence assumption for Dirichlet problem is weaker than the notion of convergence of domains used in [35]. Hence, our result allows a larger class of domain perturbation. We also include a direct application of convergence results to study final value problems in Section 1.5. In Section 1.6, we use a similar technique to obtain convergence of solutions of parabolic variational inequalities under perturbation of the underlying convex sets.

Chapter 2 provides some background and results on domain perturbation for semilinear parabolic equations. While most of the material is based on [38], our version contains an improvement. We prove in Theorem 2.3.5 that the convergence of solutions under perturbation is also possible when the nonlinear terms *converge pointwise* only. This convergence result was originally proved in [38, Theorem 4.6] when the uniform convergence of nonlinear terms is assumed. Using this improved result and the work in Chapter 1, we give domain perturbation theory for non-autonomous semilinear parabolic equations in Section 2.4. This is complementary to [38] where the application is done for autonomous semilinear parabolic equations. Nevertheless, we also include the results for autonomous problems in Section 2.4.2. In particular, we collect known results on convergence of the spectrum and resolvent operators for the corresponding elliptic equations under domain perturbation.

In Chapter 3 we study the dynamical behaviour of parabolic equations under domain perturbation. Similar studies on dynamical behaviour under domain perturbation include, for example, the works of [42] for Dirichlet problems and [10, 68] for Neumann problems. Our main focus in this thesis is the persistence of invariant manifolds near a stationary point for semilinear parabolic equations under Dirichlet boundary condition. Section 3.2 introduces the dynamical system in terms of *semiflows* induced by solutions of parabolic equations, and outlines the construction of the local stable and the local unstable invariant manifolds by the method of [14]. The main results on the persistence of unstable and stable invariant manifolds are stated in Theorem 3.3.3 and Theorem 3.3.4, respectively. The proofs are given in Section 3.5 for unstable invariant manifolds and in Section 3.6 for stable invariant manifolds. Our continuity (upper and lower semicontinuity) results for the local stable and the local unstable invariant manifolds are given in the  $L^2$  setting using the semiflows induced by mild solutions. In [10, 68] and [42], they are only interested in continuity of local unstable invariant manifolds (and consequently continuity of attractors), but the convergence is obtained in  $H^1(\mathbb{R}^N)$ . Moreover, we keep track of the dependence of invariant manifolds constructed by the method of [14], whereas the existence of invariant manifolds in [10] follows from a standard construction in [52].

Finally, in Chapter 4, we turn back to study the behaviour of solutions under domain perturbation. However, we give attention to a class of *bounded entire solutions* of autonomous parabolic equations on the whole real line rather than solutions of initial value problems on a bounded interval considered in Chapter 1 and Chapter 2. This investigation is rather new in the literature. Under an *exponential dichotomy* assumption, we collect some results on the existence and uniqueness of bounded entire solutions of linear parabolic equations in Section 4.1. Roughly, the bounded entire solution can be

#### INTRODUCTION

represented in terms of semigroups (evolution systems) and projections onto stable and unstable subspaces. In Section 4.2, we give domain perturbation results for bounded entire solutions of linear parabolic equations in an  $L^2$  setting. The results are then used in Section 4.3 to establish the convergence of solutions in  $L^p$  for a certain range of p > 2. The highlight of this section is the Hölder regularity (with respect to time) of bounded entire solutions in  $L^p$  proved in Proposition 4.3.7. Although a similar regularity result can be found in [40, Corollary 5.6] for evolution equations in interpolation spaces or [52, Section 3.3] for evolution equations in fractional power spaces, the same argument cannot be applied when we work in the  $L^p$  scales. In Section 4.4, we look at semilinear parabolic equations. We discuss the cases where the *Leray-Schauder degree* can or cannot be applied to establish the persistence of a known bounded entire solution of semilinear problems under domain perturbation. The technique of using the Leray-Schauder degree is not new. It appeared in [35] in the context of periodic solutions and in [31] in the context of nonlinear elliptic equations. We give a remark on the difficulty of using this technique for almost periodic solutions.

## INTRODUCTION

## Chapter 1

# Domain perturbation for linear non-autonomous parabolic equations

In this chapter, we study the behaviour of solutions of linear non-autonomous parabolic equations. We first collect preliminary results on the existence and uniqueness of solutions of non-autonomous linear parabolic equations by variational methods. In Section 1.2, we discuss a useful notion of *Mosco convergence* of closed and convex sets. It will be the main tool to deal with a sequence of functions belonging to different function spaces. Section 1.3 presents a key result that enables us to study domain perturbation for non-autonomous parabolic equations via the corresponding elliptic equations. We show the equivalence between Mosco convergences of various closed and convex subsets of a Banach space. This result is an abstract generalisation of the equivalence of Mosco convergences in  $W_0^{1,r}(\mathbb{R}^N)$  and in  $L^p((0,T), W_0^{1,r}(\mathbb{R}^N))$  proved in [73]. Section 1.4 gives some applications of the key result. In particular, we prove the convergence of solutions under domain perturbation for non-autonomous parabolic equations under Dirichlet or Neumann boundary conditions. Moreover, we include convergence results for final value problems in Section 1.5. Finally, in Section 1.6, we obtain a similar convergence of solutions for parabolic variational inequalities under perturbation of the underlying convex set.

## **1.1** Abstract parabolic equations

In this section, we introduce basic notations and collect standard results on linear parabolic equations.

Suppose V is a real separable and reflexive Banach space and H is a separable Hilbert space such that V is dense in H. By identifying H with its dual space H', we consider the following evolution triple

$$V \stackrel{d}{\hookrightarrow} H \stackrel{d}{\hookrightarrow} V'.$$

For an interval  $(a, b) \subset \mathbb{R}$ , we denote by  $L^2((a, b), V)$ , the Bochner-Lebesgue space. We define the Bochner-Sobolev space

$$W((a,b),V,V') := \{ u \in L^2((a,b),V) : u' \in L^2((a,b),V') \},\$$

where u' is the derivative in the sense of distributions taking values in V'. The space W((a, b), V, V') is a Banach space when equipped with the following norm

$$||u||_W := \left(\int_a^b ||u(t)||_V^2 dt + \int_a^b ||u'(t)||_{V'}^2 dt\right)^{1/2}.$$

It is well known that  $W((a,b), V, V') \hookrightarrow C([a,b], H)$ , where the space of *H*-valued continuous functions C([a,b], H) is equipped with the uniform norm ([41, Theorem I1.3.1]). More precisely, if  $u \in W((a,b), V, V')$ , then there exists a uniquely determined *H*-valued continuous functions on [a,b] which coincides almost everywhere on (a,b) with the function u. Hereafter, we use this uniquely determined function in C([a,b], H) as a representative of a function  $u \in W((a,b), V, V')$ . In this sense, there exists a constant C > 0 such that

$$\sup_{t \in [a,b]} \|u(t)\|_H \le C \|u\|_W,$$

for all  $u \in W((a, b), V, V')$ . Moreover, for  $u, v \in W((a, b), V, V')$  and  $a_0, b_0 \in [a, b]$  with  $a_0 < b_0$ , we have the integration by parts formula

$$(u(b_0)|v(b_0)) - (u(a_0)|v(a_0)) = \int_{a_0}^{b_0} \langle u'(t), v(t) \rangle + \langle v'(t), u(t) \rangle \, dt. \tag{1.1}$$

Here and throughout the thesis, we denote by  $(\cdot|\cdot)$ , the scalar product in H and by  $\langle \cdot, \cdot \rangle$ , the duality paring between V' and V. Let I and J be two sets, we write  $J \subset \subset I$  if  $\overline{J} \subset \operatorname{Int}(I)$ , where  $\operatorname{Int}(I)$  denotes the interior of I. For a subset X of a Banach space V, we define the closed convex hull by

$$\overline{\operatorname{conv}}(X) := \overline{\left\{\sum_{i=1}^{k} \alpha_{i} x_{i} \mid x_{i} \in X, \alpha_{i} \in \mathbb{R}, \alpha_{i} \ge 0, \sum_{i=1}^{k} \alpha_{i} = 1, k = 1, 2, \ldots\right\}}.$$

The framework of parabolic equations studied throughout this thesis follows a standard bilinear form setting. For each  $t \in [0, T]$ , suppose  $a(t; \cdot, \cdot)$  is a continuous bilinear form on V satisfying the following hypothesis:

- For every  $u, v \in V$ , the map  $t \mapsto a(t; u, v)$  is measurable.
- There exists a constant M > 0 independent of  $t \in [0, T]$  such that

$$|a(t; u, v)| \le M ||u||_V ||v||_V, \tag{1.2}$$

for all  $u, v \in V$  and  $t \in [0, T]$ .

• There exist  $\alpha > 0$  and  $\lambda \in \mathbb{R}$  such that

$$a(t; u, u) + \lambda \|u\|_{H}^{2} \ge \alpha \|u\|_{V}^{2}, \tag{1.3}$$

for all  $u \in V$  and  $t \in [0, T]$ .

For each  $t \in [0, T]$  and  $u \in V$ , the bilinear form  $a(t; \cdot, \cdot)$  induces a continuous linear operator  $A(t) \in \mathscr{L}(V, V')$  with

$$\langle A(t)u, v \rangle = a(t; u, v), \tag{1.4}$$

for all  $u, v \in V$ . We easily see from (1.2) that  $\sup_{t \in [0,T]} ||A(t)||_{\mathscr{L}(V,V')} \leq M$ .

Let us consider the abstract linear parabolic equation

$$\begin{cases} u'(t) + A(t)u = f(t) & \text{for } t \in (0,T] \\ u(0) = u_0, \end{cases}$$
(1.5)

where  $u_0 \in H$  and  $f \in L^2((0,T), V')$ .

**Definition 1.1.1.** A function  $u \in W(0, T, V, V')$  satisfying (1.5) is called a *weak solution* of (1.5).

It is well known that u is a weak solution of (1.5) if and only if  $u \in L^2((0,T),V)$ and

$$-\int_{0}^{T} (u(t)|v)\phi'(t) dt + \int_{0}^{T} a(t;u(t),v)\phi(t) dt$$
  
=  $(u_{0}|v)\phi(0) + \int_{0}^{T} \langle f(t),v\rangle\phi(t) dt,$  (1.6)

for all  $v \in V$  and for all  $\phi \in \mathscr{D}([0,T])$ . The existence and uniqueness of solution is given in the following theorem (see, for example, [41, XVIII §3] and [81, §23.7]).

**Theorem 1.1.2.** Given  $f \in L^2((0,T), V')$  and  $u_0 \in H$ , there exists a unique weak solution of (1.5) satisfying

$$\|u\|_{W(0,T,V,V')} \le C\Big(\|u_0\|_H + \|f\|_{L^2((0,T),V')}\Big)$$
(1.7)

with C > 0 independent of f and  $u_0$ . Moreover, if  $\lambda = 0$  in (1.3), then the weak solution satisfies

$$\|u(t)\|_{H}^{2} + \alpha \int_{0}^{t} \|u(s)\|_{V}^{2} ds \leq \|u_{0}\|_{H}^{2} + \alpha^{-1} \int_{0}^{t} \|f(s)\|_{V'}^{2} ds,$$
(1.8)

for all  $t \in [0,T]$ .

Note that  $v(t) := e^{-\lambda t}u(t)$  is a weak solution of (1.5) with A(t) replaced by  $A(t) + \lambda$ . Hence, we can assume without loss of generality that  $\lambda = 0$  in (1.3).

The remaining part of this section is devoted to an abstract formulation of linear non-autonomous parabolic equations. In other words, we consider the equation (0.4) in Introduction when the inhomogeneous term f(t, x, u) is independent of u as written below

$$\begin{cases}
\frac{\partial u}{\partial t} + \mathcal{A}(t)u = f(x, t) & \text{in } \Omega \times (0, T] \\
\mathcal{B}(t)u = 0 & \text{on } \partial\Omega \times (0, T] \\
u(\cdot, 0) = u_0 & \text{in } \Omega.
\end{cases}$$
(1.9)

For  $t \in [0, T]$ , we consider a bilinear form  $a(t; \cdot, \cdot)$  given by

$$a(t;u,v) := \int_{\Omega} [a_{ij}(x,t)\partial_j u + a_i(x,t)u]\partial_i v + b_i(x,t)\partial_i uv + c_0(x,t)uv \, dx, \qquad (1.10)$$

for  $u, v \in V$ . By assumptions on  $a_{ij}, a_i, b_i$  and  $c_0$ , it is clear that the map  $t \mapsto a(t; u, v)$ is measurable for all  $u, v \in V$ . Moreover, it can be verified that the form  $a(t; \cdot, \cdot)$  defined above satisfies (1.2) and (1.3) (see [41]).

It is well known that (1.9) can be written as an abstract equation (1.5) by taking the evolution triple  $V \stackrel{d}{\hookrightarrow} H \stackrel{d}{\hookrightarrow} V'$  with

$$V = H_0^1(\Omega)$$
 and  $H = L^2(\Omega)$  for Dirichlet boundary value problem or,  
 $V = H^1(\Omega)$  and  $H = L^2(\Omega)$  for Neumann boundary value problem (1.11)

(see [81, Corollary 23.24]).

## 1.2 Mosco Convergence

Mosco convergence was originally introduced in [65] for a sequence of convex sets and was the main tool to establish convergence properties of solutions of elliptic variational inequalities when the underlying convex set is perturbed. We assume that V is a reflexive and separable Banach space, and  $K_n, K$  are closed and convex subsets of V. We start by giving a definition of Mosco convergence in various spaces including  $V, L^2((0,T), V)$ and W((0,T), V, V').

**Definition 1.2.1.** We say that  $K_n$  converges to K in the sense of Mosco if the following conditions hold:

- (M1) For every  $u \in K$ , there exists a sequence  $u_n \in K_n$  such that  $u_n \to u$  in V strongly.
- (M2) If  $(n_k)$  is a sequence of indices converging to  $\infty$ ,  $(u_k)$  is a sequence such that  $u_k \in K_{n_k}$  for every k and  $u_k \rightharpoonup u$  in V weakly, then  $u \in K$ .

There is an alternative definition of Mosco convergence defined in terms of Kuratowski limits. A general result on Mosco convergence and equivalence of these definitions can be found in [11, Chapter 3]. We give a trivial example of Mosco convergence from our domain perturbation point of view below.

Example 1.2.2. Let  $\Omega_n$  and  $\Omega$  be bounded open sets in  $\mathbb{R}^N$  such that  $\Omega_n, \Omega \subset D$  for some ball D for all  $n \in \mathbb{N}$ . By identifying a function  $u \in H_0^1(\Omega)$  with its trivial extension (extension by zero on  $D \setminus \Omega$ ), we regard u as a function in  $H^1(D)$ . By a similar identification for functions in  $H_0^1(\Omega_n)$ , we regard  $H_0^1(\Omega)$  and  $H_0^1(\Omega_n)$  as subsets of  $H^1(D)$ . It is easy to see that they are closed subspaces (hence, closed and convex subsets) of  $H^1(D)$ . If  $\Omega_n \subset \Omega_{n+1} \subset \Omega$  for all  $n \in \mathbb{N}$  and  $\Omega = \bigcup_{n \in \mathbb{N}} \Omega_n$ , then  $H_0^1(\Omega_n)$ converges to  $H_0^1(\Omega)$  in the sense of Mosco.

As seen in Section 1.1, solutions of parabolic equation (1.5) is in  $L^2((0,T), V)$ . Thus, it is worthwhile to study Mosco convergence in  $L^2((0,T), V)$ . We let

$$L^{2}((0,T),K) := \{ u \in L^{2}((0,T),V) \mid u(t) \in K \text{ a.e.} \},\$$

and

$$C([0,T],K) := \{ u \in C([0,T],V) \mid u(t) \in K \quad \forall t \in [0,T] \}.$$

It can be verified that  $L^2((0,T), K)$  is a closed and convex subset of  $L^2((0,T), V)$ . We next state Mosco convergence of function spaces for parabolic problems.

**Definition 1.2.3.** We say that  $L^2((0,T), K_n)$  converges to  $L^2((0,T), K)$  in the sense of Mosco if the following conditions hold:

(M1') For every  $u \in L^2((0,T), K)$ , there exists a sequence  $u_n \in L^2((0,T), K_n)$  such that  $u_n \to u$  in  $L^2((0,T), V)$  strongly.

(M2') If  $(n_k)$  is a sequence of indices converging to  $\infty$ ,  $(u_k)$  is a sequence such that  $u_k \in L^2((0,T), K_{n_k})$  for every k and  $u_k \rightharpoonup u$  in  $L^2((0,T), V)$  weakly, then  $u \in L^2((0,T), K)$ .

Similarly, we can state Mosco convergence in W((0,T), V, V').

**Definition 1.2.4.** We say that  $W((0,T), V, V') \cap L^2((0,T), K_n)$  converges to  $W((0,T), V, V') \cap L^2((0,T), K)$  in the sense of Mosco if the following conditions hold:

- (M1'') For every  $u \in W((0,T), V, V') \cap L^2((0,T), K)$ , there exists a sequence  $u_n$  in  $W((0,T), V, V') \cap L^2((0,T), K_n)$  such that  $u_n \to u$  in W((0,T), V, V') strongly.
- (M2'') If  $(n_k)$  is a sequence of indices converging to  $\infty$ ,  $(u_k)$  is a sequence such that  $u_k \in W((0,T), V, V') \cap L^2((0,T), K_{n_k})$  for every  $k, u_k \rightharpoonup u$  in  $L^2((0,T), V)$  weakly and  $u'_k \rightharpoonup w$  in  $L^2((0,T), V')$  weakly, then u' = w and  $u \in W((0,T), V, V') \cap L^2((0,T), K)$ .

# **1.3** Equivalence of Mosco convergences for parabolic and elliptic equations

In this section, we prove a key result that relates domain perturbation for nonautonomous parabolic equations and domain perturbation for elliptic equations. We remark that for autonomous parabolic equations, the convergence of solutions of parabolic equations can be deduced from the convergence of solutions of the corresponding elliptic equations under domain perturbation. This is simply because we can apply semigroup methods together with the convergence result of degenerate semigroups due to Arendt [4, Theorem 5.2]. In Section 6 of the same paper, the convergence of solutions of Dirichlet heat equations is given as an application. Further examples on other boundary conditions including Neumann and Robin boundary conditions can be found in [38, Section 6].

We state our key result below.

**Theorem 1.3.1.** The following assertions are equivalent:

- (i)  $K_n$  converges to K in the sense of Mosco.
- (ii)  $L^2((0,T), K_n)$  converges to  $L^2((0,T), K)$  in the sense of Mosco.

(iii)  $W((0,T), V, V') \cap L^2((0,T), K_n)$  converges to  $W((0,T), V, V') \cap L^2((0,T), K)$  in the sense of Mosco.

Before proving the equivalence of Mosco convergences in Theorem 1.3.1, we require some technical lemmas.

**Lemma 1.3.2.** For a bounded open interval  $(a,b) \subset \mathbb{R}$ , let  $u \in L^2((a,b),K)$ . If  $\phi \in \mathscr{D}((a,b))$  such that  $\phi \geq 0$  and  $\int_a^b \phi(t) dt = 1$ , then  $\int_a^b u(t)\phi(t) dt \in K$ .

*Proof.* Since K is closed and convex, we have  $\int_a^b u(t)\phi(t) dt \in \overline{\operatorname{conv}}\{u(t) \mid t \in (a,b)\} \subset K$  for all  $u \in L^2((a,b), K)$ .

**Lemma 1.3.3.** Let I = (a, b) be a bounded open interval in  $\mathbb{R}$ . If  $u \in L^2(I, V)$  and  $\int_I u(t)\phi(t) dt \in K$  for all  $\phi \in \mathscr{D}(I)$  with  $\phi \ge 0$  and  $\int_I \phi(t) dt = 1$ , then  $u \in L^2(J, K)$  for all  $J = (c, d) \subset \subset I$ .

*Proof.* Let  $\eta \in \mathscr{D}(\mathbb{R})$  be the standard mollifier. For  $\varepsilon > 0$ , we define  $\eta_{\varepsilon}(t) = \frac{1}{\varepsilon}\eta(\frac{t}{\varepsilon})$  so that  $\eta_{\varepsilon} \in \mathscr{D}(\mathbb{R})$  with  $\int_{\mathbb{R}} \eta_{\varepsilon}(t) dt = 1$  and  $\operatorname{supp}(\eta_{\varepsilon}) \subset (-\varepsilon, \varepsilon)$ . Consider the mollified function  $u_{\varepsilon} := \eta_{\varepsilon} * u$ . For a.e.  $t \in I$ , we have

$$\begin{aligned} \|u_{\varepsilon}(t) - u(t)\|_{V} &= \left\| \int_{t-\varepsilon}^{t+\varepsilon} \eta_{\varepsilon}(t-s)[u(s) - u(t)] \, ds \right\|_{V} \\ &\leq \frac{1}{\varepsilon} \int_{t-\varepsilon}^{t+\varepsilon} \eta\Big(\frac{t-s}{\varepsilon}\Big) \|u(s) - u(t)\|_{V} \, ds \\ &\leq C \frac{1}{\varepsilon} \int_{t-\varepsilon}^{t+\varepsilon} \|u(s) - u(t)\|_{V} \, ds. \end{aligned}$$

By Lebesgue's differentiation theorem for vector valued functions (Theorem III.12.8 of [44]),  $u_{\varepsilon}(t) \to u(t)$  in V a.e.  $t \in I$ . By the definition of  $u_{\varepsilon}$ ,

$$u_{\varepsilon}(t) = \int_{I} \eta_{\varepsilon}(t-s)u(s) \ ds =: \int_{I} u(s)\phi_{\varepsilon}(s) \ ds,$$

where we set  $\phi_{\varepsilon}(s) := \eta_{\varepsilon}(t-s)$ . Let  $J \subset \subset I$ . For  $t \in J$ , we can choose  $\varepsilon$  sufficiently small so that  $\operatorname{supp}(\phi_{\varepsilon}) \subset (t-\varepsilon, t+\varepsilon) \subset I$ . It follows from the assumption that  $u_{\varepsilon}(t) \in K$ for all  $t \in J$ . Since K is a closed subset of V, the limit point  $u(t) \in K$  a.e.  $t \in J$ . Hence, u belongs to  $L^2(J, K)$  as required.  $\Box$ 

**Lemma 1.3.4.** The set C([0,T],K) is dense in  $L^2((0,T),K)$ .

*Proof.* Note first that the lemma is trivial if K is a subspace of V (i.e. K is a Banach space) [81, Theorem 23.2 (c)]. Let  $u \in L^2((0,T), K)$  be arbitrary. We choose

a function  $\phi \in \mathscr{D}((0,T))$  with  $\int_0^T \phi(t) dt = 1$ . It follows from Lemma 1.3.2 that  $\xi := \int_0^T u(t)\phi(t) dt \in K$ . Define the extended function  $\tilde{u} \in L^2((-1,T+1),K)$  by

$$\tilde{u}(t) := \begin{cases} \xi & \text{on } (-1,0) \cup (T,T+1) \\ u(t) & \text{on } (0,T). \end{cases}$$

By a mollification argument, the function  $u_{\varepsilon} := \eta_{\varepsilon} * \tilde{u}$  belongs to  $C(\mathbb{R}, V)$ . Moreover,  $u_{\varepsilon}$  converges to  $\tilde{u}$  in  $L^2((-1, T+1), V)$ . By choosing  $0 < \varepsilon < 1$ , we have  $u_{\varepsilon}(t) \in K$ for all  $t \in [0, T]$ . Therefore, the restriction of  $u_{\varepsilon}$  to [0, T] belongs to C([0, T], K) and converges to u in  $L^2((0, T), V)$  as  $\varepsilon \to 0$ .

**Lemma 1.3.5.** The set  $C^{\infty}([0,T],V) \cap C([0,T],K)$  is dense in  $W((0,T),V,V') \cap L^2((0,T),K)$ .

*Proof.* Let  $u \in W((0,T), V, V') \cap L^2((0,T), K)$  be arbitrary. For  $\delta > 0$ , we define the stretching map  $S_{\delta} : [0,T] \to [-\delta, T+\delta]$  by

$$S_{\delta}(t) := \left(\frac{T+2\delta}{T}\right)t - \delta.$$
(1.12)

We define  $u_{\delta} \in W((-\delta, T+\delta), V, V') \cap L^2((-\delta, T+\delta), K)$  by  $u_{\delta} := u \circ S_{\delta}^{-1}$ . It can be easily seen from the dominated convergence theorem and a chain rule that the restriction of  $u_{\delta}$ to (0, T) converges to u in W((0, T), V, V') as  $\delta \to 0$ . Let  $\eta_{\varepsilon}$  be a mollifier. For  $t \in [0, T]$ and  $\varepsilon < \delta$ , the translation of  $\eta_{\varepsilon}$  by t (denoted by  $\eta_{\varepsilon,t}$ ) belongs to  $\mathscr{D}((-\delta, T+\delta))$ . Hence, if  $\varepsilon < \delta$ , then  $\eta_{\varepsilon} * u_{\delta}$  belongs to  $C^{\infty}([0, T], V) \cap C([0, T], K)$ . Moreover, a mollification argument shows that  $\eta_{\varepsilon} * u_{\delta}$  converges to  $u_{\delta}$  in W((0, T), V, V') as  $\varepsilon \to 0$ . The result then follows.

**Proposition 1.3.6.** Suppose that the Mosco condition (M1) is satisfied. For  $\delta \ge 0$ , let  $A_{\delta,n} := \left\{ \sum_{i=1}^{m} \phi_i(t) v_i, m \in \mathbb{N} \right\}$ , where

$$\begin{cases} v_i \in K_n, \phi_i \in C^{\infty}([-\delta, T+\delta]) & \text{for all } i = 1, \dots, m, \\ 0 \le \phi_i(t) \le 1 & \text{for all } t \in [-\delta, T+\delta] \text{ and for all } i = 1, \dots, m, \\ \sum_{i=1}^m \phi_i(t) = 1 & \text{for all } t \in [-\delta, T+\delta]. \end{cases}$$

$$(1.13)$$

If  $u_{\delta} \in C([-\delta, T+\delta], K)$ , then there exists a sequence of functions  $u_{\delta,n} \in A_{\delta,n}$  such that  $u_{\delta,n}(t) \to u_{\delta}(t)$  in V uniformly on  $[-\delta, T+\delta]$  as  $n \to \infty$ .

*Proof.* Let  $u_{\delta}$  be a function in  $C([-\delta, T+\delta], K)$ . We extend  $u_{\delta}$  to  $\tilde{u}_{\delta} \in C(\mathbb{R}, K)$  by

$$\tilde{u}_{\delta}(t) := \begin{cases} u_{\delta}(-\delta) & \text{ on } (-\infty, -\delta) \\ u_{\delta}(t) & \text{ on } [-\delta, T+\delta] \\ u_{\delta}(T+\delta) & \text{ on } (T+\delta, \infty) \end{cases}$$

Let  $\varepsilon > 0$  be fixed. We denote by  $B(t) := B_V(\tilde{u}_{\delta}(t), \varepsilon/2)$  the open ball in V about  $\tilde{u}_{\delta}(t)$ of radius  $\varepsilon/2$ . Let us construct an open covering  $\mathscr{O}$  of  $(-\delta - 1, T + \delta + 1)$  by

$$\mathscr{O} = \{ \widetilde{u}_{\delta}^{-1}(B(t)) \cap (-\delta - 1, T + \delta + 1) \}_{t \in [-\delta, T + \delta]}$$

Since  $\mathscr{O}$  is also an open covering of the compact set  $[-\delta, T + \delta]$ , there exists a finite subcovering

$$\tilde{\mathscr{O}} = \{\tilde{u}_{\delta}^{-1}(B(t_i)) \cap (-\delta - 1, T + \delta + 1)\}_{i=1,\dots,m},$$

where  $t_i \in [-\delta, T + \delta]$  for all i = 1, ..., m. We can assume that  $t_1 < t_2 < ... < t_m$ and  $t_1 = -\delta$ ,  $t_m = T + \delta$  (add them if required) so that  $\tilde{\mathcal{O}}$  is an open covering of  $[-\delta - 1/2, T + \delta + 1/2]$ . For each  $i \in \{1, ..., m\}$ , we have  $u_{\delta}(t_i) \in K$ . Thus, by the Mosco condition (M1), there exist  $v_{i,n} \in K_n$  and  $N_i \in \mathbb{N}$  such that  $||v_{i,n} - u_{\delta}(t_i)||_V < \varepsilon/2$  if  $n > N_i$  for each i = 1, ..., m. Let  $N := \max_{i=1,...,m} N_i$ . It follows that  $||v_{i,n} - u_{\delta}(t_i)||_V < \varepsilon/2$ if n > N for all  $i \in \{1, ..., m\}$ .

Choose a smooth partition of unity  $\{\phi_i\}_{i=1,\dots m}$  for  $[-\delta - 1/2, T + \delta + 1/2]$  subordinate to  $\tilde{\mathscr{O}}$ . More precisely, we choose  $\phi_i$  such that  $\phi_i \in C_0^{\infty}(\tilde{u}_{\delta}^{-1}(B(t_i)) \cap (-\delta - 1, T + \delta + 1))$ and  $\sum_{i=1}^m \phi_i(t) = 1$  for all  $t \in [-\delta - 1/2, T + \delta + 1/2]$ . Define a function  $u_{\delta,n}$  on  $(-\delta - 1, T + \delta + 1)$  by

$$u_{\delta,n}(t) := \sum_{i=1}^{m} \phi_i(t) v_{i,n}.$$

It is clear that the restriction of  $u_{\delta,n}$  to  $[-\delta, T+\delta]$  belongs to  $A_{\delta,n}$  if n > N. Moreover, for  $t \in [-\delta, T+\delta]$ ,

$$\begin{aligned} \|u_{\delta,n}(t) - u_{\delta}(t)\|_{V} &\leq \sum_{i=1}^{m} \phi_{i}(t) \|v_{i,n} - u_{\delta}(t)\|_{V} \\ &\leq \sum_{i=1}^{m} \phi_{i}(t) \|v_{i,n} - u_{\delta}(t_{i})\|_{V} + \sum_{i=1}^{m} \phi_{i}(t) \|u_{\delta}(t_{i}) - u_{\delta}(t)\|_{V} \\ &< \varepsilon/2 + \varepsilon/2 = \varepsilon, \end{aligned}$$

if n > N. Note that m and N chosen above depend on  $\varepsilon$ . As the above argument holds for each fixed  $\varepsilon$ , we conclude that for every  $\varepsilon > 0$ , there exist a sequence  $u_{\delta,n}^{\varepsilon} \in A_{\delta,n}$  and  $N(\varepsilon) \in \mathbb{N}$  such that

$$\|u_{\delta,n}^{\varepsilon}(t) - u_{\delta}(t)\|_{V} \le \varepsilon,$$

for all  $t \in [-\delta, T + \delta]$  if  $n > N(\varepsilon)$ .

In particular, for every  $k \in \mathbb{N}$  we can find a sequence  $u_{\delta,n}^k \in A_{\delta,n}$  and  $N_k \in \mathbb{N}$  such that

$$\|u_{\delta,n}^{k}(t) - u_{\delta}(t)\|_{V} \le \frac{1}{k},$$
(1.14)

for all  $t \in [-\delta, T+\delta]$  if  $n > N_k$ . By choosing inductively we can assume that  $N_k < N_{k+1}$  for all  $k \in \mathbb{N}$ . We extract a sequence of the form

$$u_{\delta,1}^1, u_{\delta,2}^1, \dots, u_{\delta,(N_1+1)}^1, \dots, u_{\delta,N_2}^1, u_{\delta,(N_2+1)}^2, \dots, u_{\delta,N_3}^2, u_{\delta,(N_3+1)}^3, \dots, u_{\delta,N_4}^3, \dots$$

so that the *n*-th element of this sequence belongs to  $A_{\delta,n}$  for all  $n \in \mathbb{N}$ . Moreover, by (1.14), we see that this sequence converges to  $u_{\delta}$  uniformly with respect to  $t \in [-\delta, T+\delta]$  as  $n \to \infty$ . This proves the statement of the proposition.

We are now in a position to prove our main result.

Proof of Theorem 1.3.1. The proof is divided into four parts including  $(i) \Rightarrow (ii), (ii) \Rightarrow (i), (i) \Rightarrow (iii)$  and  $(iii) \Rightarrow (i)$ . For  $(i) \Rightarrow (ii)$ , we actually show that  $(M1) \Rightarrow (M1')$  and  $(M2) \Rightarrow (M2')$ . The other three directions are proved in a similar way.

 $(i) \Rightarrow (ii)$ : Let  $u \in L^2((0,T), K)$  be arbitrary. By the density of C([0,T], K) in  $L^2((0,T), K)$  (Lemma 1.3.4), we may assume that  $u \in C([0,T], K)$ . We apply Proposition 1.3.6 with  $\delta = 0$  to obtain a sequence of functions  $u_n \in L^2((0,T), K_n)$  such that  $u_n(t) \rightarrow u(t)$  in V uniformly on [0,T]. The uniform convergence on [0,T] implies that  $u_n \rightarrow u$  in  $L^2((0,T), V)$ , showing (M1'). To verify condition (M2'), suppose  $(n_k)$  is a sequence of indices converging to  $\infty$ ,  $(u_k)$  is a sequence such that  $u_k \in L^2((0,T), K_{n_k})$  for every k and  $u_k \rightarrow u$  in  $L^2((0,T), V)$ . By the definition of weak convergence,

$$\int_0^T \langle w(t), u_k(t) \rangle \ dt \to \int_0^T \langle w(t), u(t) \rangle \ dt, \tag{1.15}$$

for all  $w \in L^2((0,T), V')$ . By taking w of the form  $w = \xi \phi(t)$  where  $\xi \in V'$  and  $\phi \in \mathscr{D}((0,T))$  in (1.15) and applying a basic property of Bochner-Lebesgue space [81, Proposition 23.9(a)], it follows that

$$\int_0^T u_k(t)\phi(t) dt \rightharpoonup \int_0^T u(t)\phi(t) dt$$
(1.16)

in V weakly for all  $\phi \in \mathscr{D}((0,T))$ . Let  $\phi_0 \in \mathscr{D}((0,T))$  with  $\int_0^T \phi_0(t) dt = 1$  and define  $\zeta_k := \int_0^T u_k(t)\phi_0(t) dt$ . Lemma 1.3.2 implies that  $\zeta_k \in K_{n_k}$  for all  $k \in \mathbb{N}$ . Since  $\zeta_k \rightharpoonup \zeta := \int_0^T u(t)\phi_0(t) dt$  by (1.16), the Mosco condition (M2) implies that  $\zeta \in K$ . We now extend  $u_k$  to  $\tilde{u}_k \in L^2((-1,T+1), K_{n_k})$  by

$$\tilde{u}_k(t) := \begin{cases} \zeta_k & \text{ on } (-1,0) \cup (T,T+1) \\ u_k(t) & \text{ on } (0,T). \end{cases}$$
(1.17)

It can be easily seen that  $\tilde{u}_k \rightarrow \tilde{u}$  in  $L^2((-1, T+1), V)$  weakly, where  $\tilde{u}$  is defined as (1.17) with k deleted. Using the definition of weak convergence in  $L^2((-1, T+1), V)$  and a similar argument as above, we obtain

$$\int_{-1}^{T+1} \tilde{u}_k(t)\phi(t) \ dt \rightharpoonup \int_{-1}^{T+1} \tilde{u}(t)\phi(t) \ dt$$

in V weakly for all  $\phi \in \mathscr{D}((-1, T+1))$ . In particular, taking  $\phi \in \mathscr{D}((-1, T+1))$  with  $\int_{-1}^{T+1} \phi(t) dt = 1$ , we have that  $\int_{-1}^{T+1} \tilde{u}_k(t)\phi(t) dt \in K_{n_k}$  converges to  $\int_{-1}^{T+1} \tilde{u}(t)\phi(t) dt$  in V weakly. Thus, the Mosco condition (M2) implies  $\int_{-1}^{T+1} \tilde{u}(t)\phi(t) dt \in K$  for all  $\phi \in \mathscr{D}((-1, T+1))$  with  $\int_{-1}^{T+1} \phi(t) dt = 1$ . By Lemma 1.3.3, we conclude that  $u \in L^2((0,T), K)$  and the Mosco condition (M2') follows.

 $(ii) \Rightarrow (i)$ : Let  $u \in K$  be arbitrary. Define  $v \in L^2((0,T), K)$  by the constant function v(t) := u for  $t \in (0,T)$ . By condition (M1'), there exists  $(v_n)_{n \in \mathbb{N}}$  with  $v_n \in L^2((0,T), K_n)$  such that  $v_n \to v$  in  $L^2((0,T), V)$ . Let  $\phi_0 \in \mathscr{D}((0,T))$  with  $\int_0^T \phi_0(t) dt =$ 1. We show that the sequence  $(u_n)_{n \in \mathbb{N}}$  defined by  $u_n := \int_0^T v_n(t)\phi_0(t) dt$  gives the Mosco condition (M1). First note that  $u_n \in K_n$  for all  $n \in \mathbb{N}$  by Lemma 1.3.2. Moreover,

$$\|u_n - u\|_V = \left\| \int_0^T v_n(t)\phi_0(t) \, dt - u \right\|_V$$
  
=  $\left\| \int_0^T [v_n(t)\phi_0(t) - v(t)\phi_0(t)] \, dt \right\|_V$   
 $\leq \int_0^T |\phi_0(t)| \|v_n(t) - v(t)\|_V \, dt$   
 $\leq \sqrt{T} \Big( \int_0^T \|v_n(t) - v(t)\|_V^2 \, dt \Big)^{\frac{1}{2}} \|\phi_0\|_{\infty}$   
 $\to 0.$ 

as  $n \to \infty$ . To verify condition (M2), suppose  $(n_k)$  is a sequence of indices converging to  $\infty$ ,  $(u_k)$  is a sequence such that  $u_k \in K_{n_k}$  for every k and  $u_k \to u$  in V. Define  $v_k \in L^2((0,T), K_{n_k})$  by the constant function  $v_k(t) := u_k$  for  $t \in (0,T)$ . It can be easily verified that  $v_k \rightharpoonup v$  in  $L^2((0,T), V)$ , where v is the constant function v(t) := u for  $t \in (0,T)$ . It follows from the Mosco condition (M2') that  $v \in L^2((0,T), K)$ . Hence, u belongs to K as required.

 $(i) \Rightarrow (iii)$ : Let  $u \in W((0,T), V, V') \cap L^2((0,T), K)$  be arbitrary. By Lemma 1.3.5, we may assume that  $u \in C^{\infty}([0,T], V) \cap C([0,T], K)$ . For  $\delta > 0$ , we define the stretched function  $u_{\delta} \in C^{\infty}([-\delta, T + \delta], V) \cap C([-\delta, T + \delta], K)$  by  $u_{\delta} = u \circ S_{\delta}^{-1}$ , where  $S_{\delta}$  is the stretching map given by (1.12). As before, it can be easily seen from the dominated convergence theorem that the restriction of  $u_{\delta}$  to [0, T] converges to u in W((0, T), V, V')as  $\delta \to 0$ . By Proposition 1.3.6, there exists a sequence of functions  $u_{\delta,n} \in A_{\delta,n}$  such that  $u_{\delta,n}(t) \to u_{\delta}(t)$  uniformly on  $[-\delta, T + \delta]$  as  $n \to \infty$ . Let  $\eta_{1/j}$  be a mollifier. For  $t \in [0, T]$ and  $j > 1/\delta$ , the translation of  $\eta_{1/j}$  by t (denoted by  $\eta_{1/j,t}$ ) belongs to  $\mathscr{D}((-\delta, T + \delta))$ . Hence, if  $j > 1/\delta$ , we have  $\eta_{1/j} * u_{\delta,n} \in C^{\infty}([0,T], V) \cap C([0,T], K_n)$ . By the continuity of convolution and the well known fact on the r-th order derivative that

$$\frac{d^r}{dt^r}(\eta_{1/j} * u_{\delta,n}) = \frac{d^r}{dt^r}\eta_{1/j} * u_{\delta,n} = \eta_{1/j} * \frac{d^r}{dt}u_{\delta,n},$$

we deduce that  $\eta_{1/j} * u_{\delta,n} \to \eta_{1/j} * u_{\delta}$  in  $C^{\infty}([0,T], V)$  as  $n \to \infty$ . Similarly, we have  $\eta_{1/j} * u_{\delta} \to u_{\delta}$  in  $C^{\infty}([0,T], V)$  as  $j \to \infty$ . The above argument shows that we can construct a function of the form  $\eta_{1/j} * u_{\delta,n} \in W((0,T), V, V') \cap L^2((0,T), K_n)$ converging to u in W((0,T), V, V'). Hence, the Mosco condition (M1'') follows. To verify condition (M2''), suppose  $(n_k)$  is a sequence of indices converging to  $\infty$ ,  $(u_k)$ is a sequence such that  $u_k \in W((0,T), V, V') \cap L^2((0,T), K_{n_k})$  for every  $k, u_k \to u$ in  $L^2((0,T), V)$  and  $u'_k \to w$  in  $L^2((0,T), V')$ . Since V is continuously embedded in V', it follows immediately that u' = w and hence we have  $u \in W((0,T), V, V')$  (see [81, Proposition 23.19]). Using  $(i) \Rightarrow (ii)$ , specifically the Mosco condition (M2'), we conclude that  $u \in W((0,T), V, V') \cap L^2((0,T), K)$ .

 $(iii) \Rightarrow (i)$ : Let  $u \in K$  be arbitrary. Define  $v \in W((0,T), V, V') \cap L^2((0,T), K)$  by the constant function v(t) := u for  $t \in (0,T)$ . By condition (M1''), there exists  $(v_n)_{n \in \mathbb{N}}$ with  $v_n \in W((0,T), V, V') \cap L^2((0,T), K_n)$  such that  $v_n \to v$  in W((0,T), V, V'). In particular,  $v_n$  converges to v in  $L^2((0,T), V)$  strongly. By the same argument as in the proof of  $(ii) \Rightarrow (i)$ , we can show that  $u_n := \int_0^T v_n(t)\phi_0(t) dt$ , for some  $\phi_0 \in \mathscr{D}((0,T))$ with  $\int_0^T \phi_0(t) dt = 1$  establishes the Mosco condition (M1). To verify condition (M2), suppose  $(n_k)$  is a sequence of indices converging to  $\infty$ ,  $(u_k)$  is a sequence such that  $u_k \in K_{n_k}$  for every k and  $u_k \to u$  in V. Define  $v_k \in W((0,T), V, V') \cap L^2((0,T), K_{n_k})$ by the constant function  $v_k(t) := u_k$  for  $t \in (0,T)$ . By the same argument as in the proof of  $(ii) \Rightarrow (i)$ , we have  $v_k \to v$  in  $L^2((0,T), V)$ , where v(t) := u for  $t \in (0,T)$ . Moreover, it is clear that  $v'_k = 0$  for all  $k \in \mathbb{N}$  and hence  $v'_k \rightharpoonup v' = 0$  in  $L^2((0,T), V')$ . We apply (M2'') to deduce that  $v \in W((0,T), V, V') \cap L^2((0,T), K)$ . Hence, we have  $u \in K$ .

### 1.4 Convergence of solutions of initial value problems

In this section, we give an application of the main theorem in Section 1.3. We consider a non-autonomous linear parabolic equation subject to Dirichlet or Neumann boundary conditions under domain perturbation.

Assumption 1.4.1. We assume that  $\Omega_n$  and  $\Omega$  are bounded open sets in  $\mathbb{R}^N$  and that  $D \subset \mathbb{R}^N$  is a ball such that  $\Omega_n, \Omega \subset D$  for all  $n \in \mathbb{N}$ .

Suppose  $a_{ij}, a_i, b_i$  and  $c_0$  are functions in  $L^{\infty}(D \times (0, T))$  and  $a_{ij}$  satisfies an ellipticity condition. More precisely, there exists  $\alpha > 0$  such that  $a_{ij}(x, t)\xi_i\xi_j \ge \alpha |\xi|^2$  for all  $\xi \in \mathbb{R}^N$ . We consider the evolution triple  $V_n \stackrel{d}{\hookrightarrow} H_n \stackrel{d}{\hookrightarrow} V'_n$ , where we choose

- $V_n = H_0^1(\Omega_n)$  and  $H_n = L^2(\Omega_n)$  for the Dirichlet problem
- $V_n = H^1(\Omega_n)$  and  $H_n = L^2(\Omega_n)$  for the Neumann problem.

For  $t \in (0,T)$ , suppose  $a_n(t; \cdot, \cdot)$  is a bilinear form on  $V_n$  defined by

$$a_n(t;u,v) := \int_{\Omega_n} [a_{ij}(x,t)\partial_j u + a_i(x,t)u]\partial_i v + b_i(x,t)\partial_i uv + c_0(x,t)uv \, dx.$$
(1.18)

It follows that for all  $n \in \mathbb{N}$ , there exist three constants  $M > 0, \alpha > 0$  and  $\lambda \in \mathbb{R}$ independent of  $t \in [0, T]$  such that

$$|a_n(t; u, v)| \le M ||u||_{V_n} ||v||_{V_n}, \tag{1.19}$$

for all  $u, v \in V_n$  and

$$a_n(t; u, u) + \lambda \|u\|_{H_n}^2 \ge \alpha \|u\|_{V_n}^2, \qquad (1.20)$$

for all  $u \in V_n$ . Given  $u_{0,n} \in L^2(D)$  and  $f_n \in L^2(D \times (0,T))$ , let us consider the following boundary value problem in  $\Omega_n \times (0,T)$ .

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}_n(t)u = f_n(x,t) & \text{in } \Omega_n \times (0,T] \\ \mathcal{B}_n(t)u = 0 & \text{on } \partial \Omega_n \times (0,T] \\ u(\cdot,0) = u_{0,n} & \text{in } \Omega_n, \end{cases}$$
(1.21)

where  $\mathcal{A}_n$  and  $\mathcal{B}_n$  are operators on  $V_n$  given by

$$\mathcal{A}_n(t)u := -\partial_i [a_{ij}(x,t)\partial_j u + a_i(x,t)u] + b_i(x,t)\partial_i u + c_0(x,t)u$$
(1.22)

and  $\mathcal{B}_n$  is one of the following

$$\mathcal{B}_{n}(t)u := u \qquad \text{Dirichlet boundary condition} \\ \mathcal{B}_{n}(t)u := [a_{ij}(x,t)\partial_{j}u + a_{i}(x,t)u] \nu_{i} \qquad \text{Neumann boundary condition.}$$
(1.23)

We wish to show that a sequence of solutions of the above parabolic equations on  $\Omega_n \times (0,T)$  converges to the solution of the following limit problem

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}(t)u = f(x, t) & \text{in } \Omega \times (0, T] \\ \mathcal{B}(t)u = 0 & \text{on } \partial \Omega \times (0, T] \\ u(\cdot, 0) = u_0 & \text{in } \Omega. \end{cases}$$
(1.24)

However, we will consider the boundary value problems (1.21) and (1.24) in the abstract form. As discussed in Section 1.1, we can write (1.21) as

$$\begin{cases} u'(t) + A_n(t)u = f_n(t) & \text{ for } t \in (0,T] \\ u(0) = u_{0,n}, \end{cases}$$
(1.25)

where  $A_n(t) \in \mathscr{L}(V_n, V'_n)$  is the operator induced by the bilinear form  $a_n(t; \cdot, \cdot)$ . Similarly, we write (1.24) as

$$\begin{cases} u'(t) + A(t)u = f(t) & \text{ for } t \in (0,T] \\ u(0) = u_0. \end{cases}$$
(1.26)

Throughout this section, we denote the weak solution of (1.25) by  $u_n$  and the weak solution of (1.26) by u.

#### 1.4.1 Dirichlet problems

The convergence of solutions of Dirichlet problems under domain perturbation has been extensively studied in the literature. In particular, Daners [35] provides convergence results for solutions of non-autonomous parabolic equations under domain perturbation. In [35], it is assumed that a sequence of domains  $\Omega_n$  converges to  $\Omega$  in the following sense:

There exists a compact set  $K \subset \overline{\Omega}$  of capacity zero and a compact set  $K' \subset \mathbb{R}^N$  of measure zero such that

- if  $\Omega' \subset \subset \Omega \setminus K$ , then  $\Omega' \subset \subset \Omega_n$  for large  $n \in \mathbb{N}$ ,
- if U is an open set with  $\overline{\Omega} \cup K' \subset U$ , then  $\Omega_n \subset U$  for large  $n \in \mathbb{N}$ .

A weak regularity assumption on the limit domain  $\Omega$  is also imposed as follows:

$$u \in H^1(\mathbb{R}^N)$$
 and supp  $u \subset \overline{\Omega}$  imply  $u \in H^1_0(\Omega)$ .

In this section, we prove the convergence of solutions under domain perturbation using a technique adapted from [35]. However, we replace the notion of convergence of the domains mentioned above by Mosco convergence. In fact, it is not difficult to see that the assumptions on domains in [35] implies Mosco convergence of  $H_0^1(\Omega_n)$  to  $H_0^1(\Omega)$ . We explain a precise concept of Mosco convergence for these Sobolev spaces below.

When the domain is perturbed, the weak solutions belong to different function spaces. We often extend functions by zero outside the domain. We embed the spaces  $H_0^1(\Omega_n)$  into  $H^1(D)$  by the inclusion map

$$i_n(v) := \begin{cases} v & \text{on } \Omega_n \\ 0 & \text{on } D \backslash \Omega_n, \end{cases}$$
(1.27)

for all  $v \in H_0^1(\Omega_n)$ . The map  $i_n$  also maps  $L^p(\Omega_n)$  into  $L^p(D)$ . We may consider the embedding  $L^2((0,T), H_0^1(\Omega_n)) \hookrightarrow L^2((0,T), H^1(D))$  by a similar inclusion  $i_n$  for a.e.  $t \in (0,T)$ . In the same way, we consider  $H_0^1(\Omega)$  and  $L^2((0,T), H_0^1(\Omega))$  as subspaces of  $H^1(D)$  and  $L^2((0,T), H^1(D))$ , respectively by the trivial extension

$$i(v) := \begin{cases} v & \text{on } \Omega\\ 0 & \text{on } D \backslash \Omega, \end{cases}$$
(1.28)

for all  $v \in H_0^1(\Omega)$ . On the other hand, we have the restriction map  $r_n : L^p(D) \to L^p(\Omega_n)$ defined by

$$r_n(v) := v|_{\Omega_n}.\tag{1.29}$$

We denote again by  $r_n$  the restriction map  $L^2((0,T), L^p(D)) \to L^2((0,T), L^p(\Omega_n))$ . Similarly, we define the restriction map

$$r(v) := v|_{\Omega} \tag{1.30}$$

when considered as a map  $L^p(D) \to L^p(\Omega)$  or  $L^2((0,T), L^p(D)) \to L^2((0,T), L^p(\Omega))$ .

Remark 1.4.2. It is clear that  $r_n i_n(v) = v$  for all  $v \in L^p(\Omega_n)$ . However,  $i_n r_n(v) \neq v$  for most  $v \in L^p(D)$  (v such that  $v|_{(D \setminus \Omega_n)} \neq 0$ ). A similar conclusion is valid for the inclusion i and the restriction r.

Let us take  $V := H^1(D)$ ,  $K_n := H_0^1(\Omega_n)$  and  $K := H_0^1(\Omega)$ , and consider Mosco convergence of  $K_n$  to K. In this case,  $K_n$  and K are closed and convex subsets of V in the sense of the above embedding. In fact,  $K_n$  and K are actually closed subspaces of V.

The main application of Theorem 1.3.1 is to show that the weak solution  $u_n$  of (1.25) converges to the weak solution u of (1.26) by applying various Mosco conditions.

**Theorem 1.4.3.** Suppose that  $i_n f_n$  is uniformly bounded in  $L^2((0,T), L^2(D))$  and  $i_n u_{0,n}$  is uniformly bounded in  $L^2(D)$ . Assume that  $ri_n f_n \rightharpoonup f$  in  $L^2((0,T), L^2(\Omega))$  weakly and  $ri_n u_{0,n} \rightharpoonup u_0$  in  $L^2(\Omega)$  weakly. If  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco, then  $i_n u_n$  converges to in  $L^2((0,T), H^1(D))$  weakly.

Proof. Since  $i_n f_n$  is uniformly bounded in  $L^2((0,T), L^2(D))$  and  $i_n u_{0,n}$  is uniformly bounded in  $L^2(D)$ , it follows from (1.7) that  $||u_n||_{W(0,T,V_n,V'_n)}$  is uniformly bounded. Hence  $i_n u_n$  is uniformly bounded in  $L^2((0,T), H^1(D))$ . We can extract a subsequence (denoted again by  $u_n$ ) such that  $i_n u_n \to w$  in  $L^2((0,T), H^1(D))$  weakly. The Mosco condition (M2') (from Theorem 1.3.1) implies that  $w \in L^2((0,T), H_0^1(\Omega))$ . It remains to show that w = u in  $L^2((0,T), H_0^1(\Omega))$ .

Let  $\xi \in H_0^1(\Omega)$  and  $\phi \in \mathscr{D}([0,T))$  be arbitrary. The Mosco condition (M1) implies that there exists  $\xi_n \in H_0^1(\Omega_n)$  such that  $i_n\xi_n \to i\xi$  in  $H^1(D)$ . As  $u_n$  is the weak solution of (1.25), we get from (1.6) that

$$-\int_{0}^{T} (u_{n}(t)|\xi_{n})\phi'(t) dt + \int_{0}^{T} a_{n}(t;u_{n}(t),\xi_{n})\phi(t) dt$$
  
=  $(u_{0,n}|\xi_{n})\phi(0) + \int_{0}^{T} \langle f_{n}(t),\xi_{n}\rangle\phi(t) dt.$  (1.31)

Now

$$(i_n u_{0,n})|i_n \xi_n)_{L^2(D)} = (ri_n u_{0,n}|ri_n \xi_n)_{L^2(\Omega)} + (i_n u_{0,n}|i_n \xi_n)_{L^2(D\setminus\Omega)}$$

Since  $i_n\xi_n \to i\xi$  in  $L^2(D)$ , we have  $ri_n\xi_n \to ri\xi = \xi$  in  $L^2(\Omega)$  and  $i_n\xi_n|_{(D\setminus\Omega)} \to 0$  in  $L^2(D\setminus\Omega)$ . Applying the dominated convergence theorem in the second term above and using the weak convergence of initial condition  $ri_nu_{0,n}$  in the first term above, we see that

$$(u_{0,n}|\xi_n)_{L^2(\Omega_n)} \to (u_0|\xi)_{L^2(\Omega)}.$$
 (1.32)

By letting  $n \to \infty$  in (1.31), we get

$$-\int_{0}^{T} (w(t)|\xi)\phi'(t) dt + \int_{0}^{T} a(t;w(t),\xi)\phi(t) dt$$
  
=  $(u_{0}|\xi)\phi(0) + \int_{0}^{T} \langle f(t),\xi\rangle\phi(t) dt.$  (1.33)

Hence, w is a weak solution of (1.26). By the uniqueness of solution, we conclude that w = u in  $L^2((0,T), H_0^1(\Omega))$  and the whole sequence converges weakly.

In fact, we can expect a better convergence result by using the following compactness from [35].

**Lemma 1.4.4** ([35, Lemma 2.1]). Let D be a ball in  $\mathbb{R}^N$ . Suppose that  $\Omega_n \subset D$ for all  $n \in \mathbb{N}$  and J is a bounded interval. If  $\{v_n\}_{n\in\mathbb{N}}$  is a sequence with  $v_n \in W(J, H_0^1(\Omega_n), H^{-1}(\Omega_n))$  for each  $n \in \mathbb{N}$  and

$$||v_n||_{W(J,H^1_0(\Omega_n),H^{-1}(\Omega_n))} \le M$$

for some constant M > 0, then  $\{i_n v_n\}_{n \in \mathbb{N}}$  is relatively compact in  $L^2(J, L^q(D))$  for all  $q \in [1, 2N(N-2)^{-1})$ .

**Theorem 1.4.5.** Suppose that the hypothesis of Theorem 1.4.3 holds. Then  $i_n u_n \rightarrow iu$  in  $L^2((0,T), L^2(D))$  strongly.

Proof. By Lemma 1.4.4, we can extract a subsequence  $u_{n_k}$  such that  $i_{n_k}u_{n_k} \to v$  in  $L^2((0,T), L^2(D))$  strongly. Since we know from Theorem 1.4.3 that the whole sequence  $i_n u_n$  converges to iu in  $L^2((0,T), H^1(D))$  weakly, we conclude that v = iu and the whole sequence converges in  $L^2((0,T), L^2(D))$  strongly.  $\Box$ 

**Lemma 1.4.6.** Suppose that the hypothesis of Theorem 1.4.3 holds. Then for each  $t \in [0,T]$ , we have  $ri_n u_n(t) \rightharpoonup u(t)$  in  $L^2(\Omega)$ .

*Proof.* Since  $i_n f_n$  is uniformly bounded in  $L^2((0,T), L^2(D))$  and  $i_n u_{0,n}$  is uniformly bounded in  $L^2(D)$ , we have from (1.8) that

$$\sup_{n \in \mathbb{N}} \max_{t \in [0,T]} \|i_n u_n(t)\|_{L^2(D)} \le M,$$

for some M > 0. Hence, for a subsequence denoted again by  $u_n(t)$ , there exists  $w \in L^2(\Omega)$  such that  $ri_n u_n(t) \rightharpoonup w$  in  $L^2(\Omega)$ . Let  $\xi \in H_0^1(\Omega)$  and  $\phi \in \mathscr{D}((0,t])$  be arbitrary.

The Mosco condition (M1) implies that there exists  $\xi_n \in H_0^1(\Omega_n)$  such that  $i_n \xi_n \to i\xi$ in  $H^1(D)$ . As  $u_n$  is the weak solution of (1.25), we have

$$-\int_{0}^{t} (u_{n}(s)|\xi_{n})\phi'(s) \, ds + \int_{0}^{t} a_{n}(s;u_{n}(s),\xi_{n})\phi(s) \, ds$$
$$= -(u_{n}(t)|\xi_{n})\phi(t) + \int_{0}^{t} \langle f_{n}(s),\xi_{n}\rangle\phi(s) \, ds.$$

We get in the same way as in (1.32) that

$$(i_n u_n(t)|i_n \xi_n)_{L^2(D)} = (ri_n u_n(t)|ri_n \xi_n)_{L^2(\Omega)} + (i_n u_n(t)|i_n \xi_n)_{L^2(D\setminus\Omega)} \to (w|\xi)_{L^2(\Omega)},$$

as  $n \to \infty$ . Hence,

$$-\int_{0}^{t} (u(s)|\xi)\phi'(s) \, ds + \int_{0}^{t} a(s; u(s), \xi)\phi(s) \, ds$$
  
=  $-(w|\xi)\phi(t) + \int_{0}^{t} \langle f(s), \xi \rangle \phi(s) \, ds,$  (1.34)

as  $n \to \infty$ . As u is the weak solution of (1.26), a similar equation holds with  $(w|\xi)_{L^2(\Omega)}$ replaced by  $(u(t)|\xi)_{L^2(\Omega)}$ . Therefore,  $(w|\xi)_{L^2(\Omega)} = (u(t)|\xi)_{L^2(\Omega)}$  for all  $\xi \in H_0^1(\Omega)$ . By the density of  $H_0^1(\Omega)$  in  $L^2(\Omega)$ , we deduce that w = u(t). Hence, for subsequences,  $ri_n u_n(t) \to u(t)$  in  $L^2(\Omega)$ . By the uniqueness of solutions, the whole sequence  $ri_n u_n(t)$ converges to u(t) in  $L^2(\Omega)$  weakly.

Next we show the strong convergence of solutions in  $L^2((0,T), H^1(D))$ .

**Theorem 1.4.7.** Suppose that the hypothesis of Theorem 1.4.3 holds. Then for any  $\delta \in (0,T]$ , the solution  $i_n u_n$  converges to iu in  $L^2((\delta,T), H^1(D))$  strongly. In addition, if  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$  strongly, then the above assertion holds for  $\delta = 0$ .

Proof. Recall that we have the weak convergence  $i_n u_n \rightarrow iu$  in  $L^2((0,T), H^1(D))$  from Theorem 1.4.3 and the strong convergence  $i_n u_n \rightarrow iu$  in  $L^2((0,T), L^2(D))$  from Theorem 1.4.5. Hence, we can extract a subsequence (denoted again by  $u_n$ ) so that  $i_n u_n(t, \cdot)$ converges in  $L^2(D)$  for almost every  $t \in (0,T)$ . Choosing  $\delta \in (0,T]$  arbitrarily close to zero such that  $i_n u_n(\delta) \rightarrow iu(\delta)$  in  $L^2(D)$  strongly. By the Mosco condition (M1')(from Theorem 1.3.1), we can find  $w_n \in L^2((0,T), H_0^1(\Omega_n))$  such that  $i_n w_n \rightarrow iu$  in  $L^2((0,T), H^1(D))$ . For  $t \in (\delta, T]$ , we consider

$$d_n(t) = \frac{1}{2} \|i_n u_n(t) - iu(t)\|_{L^2(D)}^2 + \alpha \int_{\delta}^t \|i_n u_n(s) - i_n w_n(s)\|_{H^1(D)}^2 ds.$$
(1.35)

By (1.20) (with  $\lambda = 0$ ), we have

$$d_{n}(t) \leq \frac{1}{2} \|i_{n}u_{n}(t)\|_{L^{2}(D)}^{2} + \int_{\delta}^{t} a_{n}(s; u_{n}(s), u_{n}(s)) ds + \frac{1}{2} \|iu(t)\|_{L^{2}(D)}^{2} + \int_{\delta}^{t} a_{n}(s; w_{n}(s), w_{n}(s)) ds - (i_{n}u_{n}(t)|iu(t))_{L^{2}(D)} - \int_{\delta}^{t} a_{n}(s; u_{n}(s), w_{n}(s)) ds - \int_{\delta}^{t} a_{n}(s; w_{n}(s), u_{n}(s)) ds,$$

$$(1.36)$$

for all  $n \in \mathbb{N}$ . It can be easily seen from the weak convergence of  $i_n u_n$  and the strong convergence of  $i_n w_n$  to iu in  $L^2((0,T), H^1(D))$  that

$$\lim_{n \to \infty} \left[ \int_{\delta}^{t} a_{n}(s; u_{n}(s), w_{n}(s)) \, ds + \int_{\delta}^{t} a_{n}(s; w_{n}(s), u_{n}(s)) \, ds \right]$$
  
=  $2 \int_{\delta}^{t} a(s; u(s), u(s)) \, ds,$  (1.37)

and

$$\lim_{n \to \infty} \int_{\delta}^{t} a_n(s; w_n(s), w_n(s)) \, ds = \int_{\delta}^{t} a(s; u(s), u(s)) \, ds.$$
(1.38)

Also, by Lemma 1.4.6, we have

$$\lim_{n \to \infty} (i_n u_n(t) | i u(t))_{L^2(D)} = \lim_{n \to \infty} (r i_n u_n(t) | u(t))_{L^2(\Omega)} = \| u(t) \|_{L^2(\Omega)}^2.$$
(1.39)

Finally, as  $u_n$  is the weak solution of (1.25) it satisfies

$$\frac{1}{2} \|u_n(t)\|_{L^2(\Omega_n)}^2 + \int_{\delta}^t a_n(s; u_n(s), u_n(s)) \, ds$$
$$= \frac{1}{2} \|u_n(\delta)\|_{L^2(\Omega_n)}^2 + \int_{\delta}^t \langle f_n(s), u_n(s) \rangle \, ds.$$

Using that  $i_n u_n(\delta) \to iu(\delta)$  in  $L^2(D)$  strongly,  $ri_n f_n \rightharpoonup f$  in  $L^2((0,T), L^2(\Omega))$  weakly and the strong convergence of solution  $i_n u_n \to iu$  in  $L^2((0,T), L^2(D))$ , we get

$$\lim_{n \to \infty} \left[ \frac{1}{2} \| u_n(t) \|_{L^2(\Omega_n)}^2 + \int_{\delta}^t a_n(s; u_n(s), u_n(s)) \, ds \right] \\
= \frac{1}{2} \| u(\delta) \|_{L^2(\Omega)}^2 + \int_{\delta}^t \langle f(s), u(s) \rangle \, ds \qquad (1.40) \\
= \frac{1}{2} \| u(t) \|_{L^2(\Omega)}^2 + \int_{\delta}^t a(s; u(s), u(s)) \, ds.$$

Hence, it follows from (1.36) – (1.40) that  $d_n(t) \to 0$  for all  $t \in [\delta, T]$ . As  $\delta > 0$  was arbitrarily chosen, this implies pointwise convergence  $i_n u_n(t) \to iu(t)$  in  $L^2(D)$  for all

 $t \in (0, T]$ . Moreover, by taking t = T we get

$$\begin{split} \int_{\delta}^{T} \|i_{n}u_{n}(s) - iu(s)\|_{H^{1}(D)}^{2} ds \\ &\leq \int_{\delta}^{T} \|i_{n}u_{n}(s) - i_{n}w_{n}(s)\|_{H^{1}(D)}^{2} ds + \int_{\delta}^{T} \|i_{n}w_{n}(s) - iu(s)\|_{H^{1}(D)}^{2} ds \\ &\to 0, \end{split}$$

as  $n \to \infty$ . This proves the strong convergence  $i_n u_n \to iu$  in  $L^2((\delta, T), H^1(D))$  for all  $\delta \in (0, T]$ .

If  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$  strongly, we repeat the above arguments with  $\delta = 0$ . It is trivial that all arguments remain valid. Hence, the statement of the theorem follows.  $\Box$ 

In the next theorem, we prove convergence of solutions in a stronger norm. We show that Mosco convergence is sufficient for uniform convergence of solutions in  $L^2(D)$  with respect to  $t \in [0, T]$ . We require the following result on Mosco convergence of  $H_0^1(\Omega_n)$ to  $H_0^1(\Omega)$  [37, Proposition 6.3].

Lemma 1.4.8. The following statements are equivalent.

- 1. Mosco condition (M1): for every  $w \in H_0^1(\Omega)$ , there exists a sequence  $w_n \in H_0^1(\Omega_n)$ such that  $i_n w_n \to iw$  in  $H^1(D)$ .
- 2.  $\operatorname{cap}(K \cap \Omega_n^c) \to 0$  as  $n \to \infty$  for all compact set  $K \subset \Omega$ .

**Theorem 1.4.9.** Suppose that the hypothesis of Theorem 1.4.3 holds. Then for any  $\delta \in (0,T]$ , the solution  $i_n u_n$  converges to iu in  $C([\delta,T], L^2(D))$  strongly. In addition, if  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$  strongly, then the above assertion holds for  $\delta = 0$ .

*Proof.* Let  $w_n$  be a sequence chosen as in the proof of Theorem 1.4.7. We notice from the proof of Theorem 1.4.7 that

$$\int_{\delta}^{t} \|i_n u_n(s) - i_n w_n(s)\|_{H^1(D)}^2 \, ds \to 0$$

uniformly with respect to  $t \in [\delta, T]$ . Indeed, by (1.35), we have

$$\int_{\delta}^{t} \|i_n u_n(s) - i_n w_n(s)\|_{H^1(D)}^2 \, ds \le \alpha^{-1} d_n(T),$$

for all  $n \in \mathbb{N}$  and for all  $t \in [\delta, T]$ . Moreover, it is clear that (1.37), (1.38) and (1.40) hold uniformly on  $[\delta, T]$ . It remains to show uniform convergence of (1.39).

Fix  $s \in [\delta, T]$ . For  $\varepsilon > 0$  arbitrary, we choose a compact set  $K \subset \Omega$  such that  $\|u(s)\|_{L^2(\Omega\setminus K)} \leq \varepsilon/2$ . Since  $u \in C([0, T], L^2(\Omega))$ , there exists  $\eta > 0$  only depending on  $\varepsilon$  such that  $\|u(t) - u(s)\|_{L^2(\Omega)} \leq \varepsilon/2$  for all  $t \in (s - \eta, s + \eta) \cap [\delta, T]$ . It follows that

$$\|u(t)\|_{L^2(\Omega\setminus K)} \le \|u(t) - u(s)\|_{L^2(\Omega)} + \|u(s)\|_{L^2(\Omega\setminus K)} \le \varepsilon/2 + \varepsilon/2 = \varepsilon, \tag{1.41}$$

for all  $t \in (s - \eta, s + \eta) \cap [\delta, T]$ . We next choose a cut-off function  $\phi \in C_0^{\infty}(\Omega)$  such that  $0 \leq \phi \leq 1$  and  $\phi = 1$  on K. Since  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco, we have from Lemma 1.4.8 that  $\operatorname{cap}(\operatorname{supp}(\phi) \cap \Omega_n^c) \to 0$ . By the definition of capacity, there exists a sequence  $\xi_n \in C_0^{\infty}(\Omega)$  such that  $0 \leq \xi_n \leq 1$ ,  $\xi_n = 1$  on a neighborhood of  $\operatorname{supp}(\phi) \cap \Omega_n^c$  and  $\|\xi_n\|_{H^1(\mathbb{R}^N)} \leq \operatorname{cap}(\operatorname{supp}(\phi) \cap \Omega_n^c) + 1/n$ . Define  $\phi_n := (1 - \xi_n)\phi$ . We have that  $\phi_n \in C_0^{\infty}(\Omega_n)$  and  $\phi_n \to \phi$  in  $L^2(D)$ . Consider

$$\begin{aligned} \left| (i_{n}u_{n}(t) - iu(t)|iu(t))_{L^{2}(D)} \right| \\ &\leq \left| (i_{n}u_{n}(t)|\phi_{n}iu(t))_{L^{2}(D)} - (iu(t)|\phi_{iu}(t))_{L^{2}(D)} \right| \\ &+ \left| (i_{n}u_{n}(t)|(1 - \phi_{n})iu(t))_{L^{2}(D)} - (iu(t)|(1 - \phi)iu(t))_{L^{2}(D)} \right| \\ &\leq \left| (i_{n}u_{n}(t)|\phi_{n}iu(t))_{L^{2}(D)} - (iu(t)|\phi_{iu}(t))_{L^{2}(D)} \right| \\ &+ \left| (i_{n}u_{n}(t) - iu(t)|(1 - \phi_{n})iu(t))_{L^{2}(D)} \right| + \left| (iu(t)|(\phi - \phi_{n})iu(t))_{L^{2}(D)} \right|. \end{aligned}$$
(1.42)

We prove that each term on the right of (1.42) is uniformly small for  $t \in (s - \eta, s + \eta) \cap [\delta, T]$  if *n* is sufficiently large. For the first term, applying the integration by parts formula (1.1) and the definition of weak solutions, we obtain

$$\begin{aligned} &(i_n u_n(t)|\phi_n iu(t))_{L^2(D)} \\ &= (u_n(\delta)|\phi_n u(\delta))_{L^2(\Omega_n)} + \int_{\delta}^t \langle u'_n(s), \phi_n iu(s) \rangle \ ds + \int_{\delta}^t \langle u'(s), \phi_n i_n u_n(s) \rangle \ ds \\ &= (u_n(\delta)|\phi_n u(\delta))_{L^2(\Omega_n)} + \int_{\delta}^t \langle f_n(s), \phi_n iu(s) \rangle \ ds - \int_{\delta}^t a_n(s; u_n(s), \phi_n iu(s)) \ ds \\ &+ \int_{\delta}^t \langle f(s), \phi_n i_n u_n(s) \rangle \ ds - \int_{\delta}^t a(s; u(s), \phi_n i_n u_n(s)) \ ds. \end{aligned}$$
(1.43)

It can be easily verified using the dominated convergence theorem that  $\phi_n iu \to \phi iu$  in  $L^2((0,T), L^2(D))$ . Moreover,

$$\begin{split} &\int_{\delta}^{T} \|\phi_{n}i_{n}u_{n}(t) - \phi iu(t)\|_{L^{2}(D)}^{2} dt \\ &\leq \int_{\delta}^{T} \|\phi_{n}iu(t) - \phi iu(t)\|_{L^{2}(D)}^{2} dt + \int_{\delta}^{T} \|\phi_{n}i_{n}u_{n}(t) - \phi_{n}iu(t)\|_{L^{2}(D)}^{2} dt \\ &\leq \int_{\delta}^{T} \|\phi_{n}iu(t) - \phi iu(t)\|_{L^{2}(D)}^{2} dt + \|\phi_{n}\|_{\infty}^{2} \int_{\delta}^{T} \|i_{n}u_{n}(t) - iu(t)\|_{L^{2}(D)}^{2} dt \end{split}$$
Hence,  $\phi_n i_n u_n \to \phi i u$  in  $L^2((0,T), L^2(D))$ . Taking into consideration that  $i_n u_n(\delta) \to i u(\delta)$  in  $L^2(D)$  and  $r i_n f_n \rightharpoonup f$  in  $L^2((0,T), L^2(\Omega))$ , we conclude form (1.43) that

$$(i_n u_n(t)|\phi_n iu(t))_{L^2(D)} \to (iu(t)|\phi iu(t))_{L^2(D)}$$
 (1.44)

uniformly with respect to  $t \in [\delta, T]$ . For the last term on the right of (1.42), applying a similar argument as above, we write

$$(iu(t)|\phi_n iu(t))_{L^2(D)} = (u(\delta)|\phi_n u(\delta))_{L^2(\Omega)} + 2\int_{\delta}^{t} \langle u'(s), \phi_n \tilde{u}(s) \rangle \ ds.$$

We conclude that

$$(iu(t)|\phi_n iu(t))_{L^2(D)} \to (iu(t)|\phi iu(t))_{L^2(D)}$$
 (1.45)

uniformly with respect to  $t \in [\delta, T]$ . Finally, for the second term on the right of (1.42), we notice that  $0 \le 1 - \phi_n \le 1$  on  $\Omega$  and  $1 - \phi_n = 1 - (1 - \xi_n)\phi = \xi_n$  on K. Moreover, by the uniform boundedness of  $i_n u_{0,n}$  in  $L^2(D)$  and the uniform boundedness of  $i_n f_n$ in  $L^2((0,T), L^2(D))$ , we see from (1.8) that there exists a constant  $M_0 > 0$  such that

$$\|i_n u_n(t)\|_{L^2(D)}, \|iu(t)\|_{L^2(D)} \le M_0, \tag{1.46}$$

for all  $t \in [0, T]$ . Hence, by the Cauchy-Schwarz inequality and (1.41),

$$\begin{aligned} \left| (i_n u_n(t) - iu(t) | (1 - \phi_n) iu(t))_{L^2(D)} \right| &\leq \| i_n u_n(t) - iu(t) \|_{L^2(D)} \| (1 - \phi_n) iu(t) \|_{L^2(D)} \\ &\leq 2M_0 \left( \| u(t) \|_{L^2(\Omega \setminus K)}^2 + \| \xi_n u(t) \|_{L^2(K)} \right) \\ &\leq 2M_0 \left( \varepsilon + \| \xi_n u(t) \|_{L^2(K)} \right), \end{aligned}$$

$$(1.47)$$

for all  $t \in (s - \eta, s + \eta) \cap [\delta, T]$  and for all  $n \in \mathbb{N}$ . Since  $\xi_n \to 0$  in  $L^2(D)$ , a standard argument using the dominated convergence theorem implies that  $\xi_n u(s) \to 0$  in  $L^2(\Omega)$ . Hence, there exists  $N_{s,\varepsilon} \in \mathbb{N}$  such that  $\|\xi_n u(s)\|_{L^2(\Omega)} \leq \varepsilon/2$  for all  $n \geq N_{s,\varepsilon}$ . Therefore,

$$\begin{aligned} \|\xi_n u(t)\|_{L^2(K)} &\leq \|\xi_n u(s)\|_{L^2(K)} + \|\xi_n u(t) - \xi_n u(s)\|_{L^2(K)} \\ &\leq \|\xi_n u(s)\|_{L^2(K)} + \|u(t) - u(s)\|_{L^2(K)} \\ &\leq \varepsilon/2 + \varepsilon/2 = \varepsilon, \end{aligned}$$

for all  $t \in (s - \eta, s + \eta) \cap [\delta, T]$  and for all  $n \ge N_{s,\varepsilon}$ . It follows from (1.47) that

$$\left| (i_n u_n(t) - i u(t)) | (1 - \phi_n) i u(t))_{L^2(D)} \right| \le 2M_0(\varepsilon + \varepsilon) = 4M_0\varepsilon, \tag{1.48}$$

for all  $t \in (s - \eta, s + \eta) \cap [\delta, T]$  and for all  $n \ge N_{s,\varepsilon}$ . Therefore, by (1.42), (1.44), (1.45), and (1.48), we conclude that there exist  $\tilde{N}_{s,\varepsilon} \in \mathbb{N}$  and a positive constant C such that

$$\left| (i_n u_n(t) - i u(t) | i u(t))_{L^2(D)} \right| \le C\varepsilon,$$

for all  $t \in (s - \eta, s + \eta) \cap [\delta, T]$  and for all  $n \ge \tilde{N}_{s,\varepsilon}$ .

Finally, as  $[\delta, T]$  is a compact interval and  $\eta$  only depends on  $\varepsilon$ , it follows that  $(i_n u_n(t)|iu(t))_{L^2(D)} \to (iu(t)|iu(t))_{L^2(D)}$  uniformly with respect to  $t \in [\delta, T]$ . This proves the convergence  $i_n u_n \to iu$  in  $C([\delta, T], L^2(D))$  for all  $\delta \in (0, T]$ . If  $i_n u_{0,n} \to iu_0$  in  $L^2(D)$  strongly, we take  $\delta = 0$  in the above arguments. The last assertion of the theorem follows.

It is known that the convergence of solution of elliptic equations subject to Dirichlet boundary condition under domain perturbation can be obtained from Mosco convergence of  $H_0^1(\Omega_n)$  to  $H_0^1(\Omega)$  [37]. Hence, we can use the same criterion on  $\Omega_n$  and  $\Omega$  to conclude the convergence of solutions of parabolic equations. In particular, the conditions on domains given in [37, Theorem 7.5] implies convergence of solutions of non-autonomous parabolic equations (1.21) subject to Dirichlet boundary condition under domain perturbation. For an open set  $U \subset \mathbb{R}^N$ , we define

$$\lambda_1(U) = \inf_{u \in H_0^1(U), u \neq 0} \frac{\|\nabla u\|^2}{\|u\|^2}$$

and  $\lambda_1(\emptyset) = \infty$ . As a consequence of [37, Theorem 7.5], we can state the following corollary.

**Corollary 1.4.10.** Suppose that  $i_n f_n$  is uniformly bounded in  $L^2((0,T), L^2(D))$  and  $i_n u_{0,n}$  is uniformly bounded in  $L^2(D)$ . Assume that  $ri_n f_n \rightharpoonup f$  in  $L^2((0,T), L^2(\Omega))$  weakly and  $ri_n u_{0,n} \rightharpoonup u_0$  in  $L^2(\Omega)$  weakly. If  $\Omega_n$  and  $\Omega$  are open and bounded sets in  $\mathbb{R}^N$  satisfying

- 1.  $\operatorname{cap}(K \cap \Omega_n^c) \to 0$  as  $n \to \infty$  for all compact set  $K \subset \Omega$ ;
- 2. There exists an open covering  $\mathscr{O}$  of  $\mathbb{R}^N \setminus \overline{\Omega}$  such that  $\lambda_1(U \cap \Omega_n) \to \infty$  as  $n \to \infty$ for all  $U \in \mathscr{O}$ ;
- 3.  $H_0^1(\Omega) = H_0^1(\Omega \cup \Gamma)$ , where

$$\Gamma := \bigcap_{n \in \mathbb{N}} \left( \overline{\bigcup_{k \ge n} (\Omega_k \cap \partial \Omega)} \right),$$

then for any  $\delta \in (0,T]$ , the solution  $i_n u_n$  converges strongly to in  $L^2((\delta,T), H^1(D))$ and in  $C([\delta,T], L^2(D))$ . In addition, if  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$  strongly, then the above assertion holds for  $\delta = 0$ . Examples of domains satisfying the above conditions include the following (see [37] for details and further examples):

- (i) a standard dumbbell-shaped domain with shrinking handle (possibly also with a hole converging to zero in capacity) as shown in Figure 1.1;
- (ii) a square with "fingers" attached in Figure 1.2. Here we increase the number of fingers and reduce their width so that the Lebesgue measure of  $\Omega_n$  is preserved. This means  $|\Omega_n| \neq |\Omega|$  as  $n \to \infty$ .



Figure 1.1: Dumbbell-shaped domains with shrinking handle.



Figure 1.2: A square with fingers attached.

#### 1.4.2 Neumann problems

It is more difficult to handle Neumann boundary condition than Dirichlet boundary condition. Convergence results for Neumann problems often require a certain regularity of the domains. For autonomous problems, it is proved in [10, Proposition 2.7] that linear semigroups converge under the following assumptions on domains (referred to as "admissible family of domains" in [10, Definition 2.5]):

•  $\Omega_n$  is bounded and Lipschitz for all  $n \in \mathbb{N}$ ;

- For every  $K \subset \subset \Omega$ , there exists  $N_K \in \mathbb{N}$  such that  $K \subset \Omega_n$  if  $n > N_K$ ;
- Spectra of the corresponding elliptic operators behave continuously (one of the conditions (i)–(iv) of [10, Proposition 2.3]).

Our main emphasis in this work is to study *non-autonomous* problems without any smoothness of domains. We cannot simply consider the trivial extension by zero outside the domain for functions in  $H^1(\Omega)$  as in the case of Dirichlet problems, because the extended function does not belong to  $H^1(\mathbb{R}^N)$ . Also, there is no smooth extension from  $H^1(\Omega)$  to  $H^1(D)$  as we do not impose any regularity of the domain. Furthermore, the compactness result in [35, Lemma 2.1] cannot be applied in the case of Neumann problems.

In order to study the limit of  $u_n \in H^1(\Omega_n)$  when the domain is perturbed, we embed the space  $H^1(\Omega_n)$  into the following space

$$H^1(\Omega_n) \hookrightarrow L^2(D) \times L^2(D, \mathbb{R}^N)$$

by

$$v_n \mapsto (i_n v_n, i_n \nabla v_n),$$

where  $i_n$  is the inclusion map defined in (1.27). Note that  $i_n \nabla v_n$  is not the gradient of  $i_n v_n$  in the sense of distribution, that is  $i_n \nabla v_n \neq \nabla(i_n v_n)$ . By a similar embedding for  $H^1(\Omega)$ , we can consider Mosco convergence of

$$K_n := \{ (i_n v_n, i_n \nabla v_n) \in L^2(D) \times L^2(D, \mathbb{R}^N) \mid v_n \in H^1(\Omega_n) \}$$

 $\operatorname{to}$ 

$$K := \{(iv, i\nabla v) \in L^2(D) \times L^2(D, \mathbb{R}^N) \mid v \in H^1(\Omega)\}$$

in  $V := L^2(D) \times L^2(D, \mathbb{R}^N)$ . In this case,  $K_n$  and K are closed subspaces of V. For simplicity, we say  $H^1(\Omega_n)$  converges to  $H^1(\Omega)$  in the sense of Mosco for  $K_n$  and K chosen above.

When dealing with solutions of parabolic equations, we identify the space

$$L^{2}((0,T),V) \equiv L^{2}((0,T),L^{2}(D)) \times L^{2}((0,T),L^{2}(D,\mathbb{R}^{N}))$$

via the isomorphism between them. Hence,

$$L^{2}((0,T), K_{n}) \equiv \{(i_{n}w_{n}, i_{n}\nabla w_{n}) \mid w_{n} \in L^{2}((0,T), H^{1}(\Omega_{n})\} \\ \subset L^{2}((0,T), L^{2}(D)) \times L^{2}((0,T), L^{2}(D, \mathbb{R}^{N})),$$

and

$$L^{2}((0,T),K) \equiv \{(iw,i\nabla w) \mid w \in L^{2}((0,T),H^{1}(\Omega))\}$$
$$\subset L^{2}((0,T),L^{2}(D)) \times L^{2}((0,T),L^{2}(D,\mathbb{R}^{N})).$$

We point out that the assumptions on domains in [10, Definition 2.5]) imply Mosco convergence of  $H^1(\Omega_n)$  to  $H^1(\Omega)$ .

**Lemma 1.4.11.** Suppose that  $\Omega_n$  and  $\Omega$  satisfy the assumptions in [10, Definiton 2.5]. Then  $H^1(\Omega_n)$  converges to  $H^1(\Omega)$  in the sense of Mosco.

Proof. We show that the Mosco conditions (M1) and (M2) are satisfied. Let  $v \in H^1(\Omega)$ be arbitrary. Since  $\Omega$  is Lipschitz, there exits an extension operator  $\mathscr{E} : H^1(\Omega) \to H^1(\mathbb{R}^N)$ . We extend v to a function  $\mathscr{E}v$  in  $H^1(\mathbb{R}^N)$ . By restriction to  $\Omega_n$ , we have  $v_n := r_n \mathscr{E}v \in H^1(\Omega_n)$ . Clearly,  $\|i_n v_n - iv\|_{H^1(\Omega \cap \Omega_n)} = 0$  for all  $n \in \mathbb{N}$ . The condition [10, (1.4)] implies that there exists a non-increasing sequence  $\rho_n$  converging to zero such that the set

$$K_n := \{ x \in \Omega : \operatorname{dist}(x, \partial \Omega) > \rho_n \} \subset \Omega_n,$$

for all  $n \in \mathbb{N}$  (see also [10, (2.1)]). As  $\Omega$  is bounded, this yields  $|\Omega \setminus \Omega_n| \leq |\Omega \setminus K_n| \to 0$ as  $n \to \infty$ . Hence,

$$\|i_n v_n - iv\|_{L^2(\Omega \setminus \Omega_n)} + \|i_n \nabla v_n - i \nabla v\|_{L^2((\Omega \setminus \Omega_n), \mathbb{R}^N)} = \|v\|_{H^1(\Omega \setminus \Omega_n)} \to 0$$

as  $n \to \infty$ . Using that  $|\Omega_n \setminus K_n| \to 0$  (condition (v) in [10, Proposition 2.3]), we easily see that

$$\|v_n\|_{H^1(\Omega_n \setminus \Omega)} = \|r_n \mathscr{E} v\|_{H^1(\Omega_n \setminus \Omega)} \to 0$$

as  $n \to \infty$ . Therefore,

$$\|i_n v_n - iv\|_{L^2(\Omega_n \setminus \Omega)} + \|i_n \nabla v_n - i \nabla v\|_{L^2((\Omega_n \setminus \Omega), \mathbb{R}^N)} = \|v_n\|_{H^1(\Omega_n \setminus \Omega)}$$
  
$$\to 0$$

as  $n \to \infty$ . The above implies

$$\|i_n v_n - iv\|_{L^2(\mathbb{R}^N)} + \|i_n \nabla v_n - i\nabla v\|_{L^2(\mathbb{R}^N,\mathbb{R}^N)} \to 0$$

as  $n \to \infty$ . Hence, condition (M1) holds.

Next, suppose  $(n_k)$  is a sequence of indices converging to  $\infty$ ,  $(v_k)$  is a sequence such that  $v_k \in H^1(\Omega_{n_k})$  for every k and  $(i_{n_k}v_k, i_{n_k}\nabla v_k) \rightharpoonup (v, w)$  in  $L^2(\mathbb{R}^N) \times L^2(\mathbb{R}^N, \mathbb{R}^N)$ 

weakly. By condition (iv) in [10, Proposition 2.3], we can extract a subsequence so that there exists  $\phi \in H^1(\Omega)$  such that  $i_{n_{k_j}}v_{k_j} \to i\phi$  in  $L^2(\mathbb{R}^N)$  as  $j \to \infty$  and

$$\int_{\Omega_{n_{k_j}}} \nabla v_{k_j} \nabla \chi dx \to \int_{\Omega} \nabla \phi \nabla \chi dx,$$

as  $j \to \infty$  for all  $\chi \in H^1(\mathbb{R}^N)$ . In fact, we can show by modifying the proof in [10] that

$$\int_{\Omega_{n_{k_j}}} \nabla v_{k_j} \psi dx \to \int_{\Omega} \nabla \phi \psi dx,$$

as  $j \to \infty$  for all  $\psi \in L^2(\mathbb{R}^N, \mathbb{R}^N)$ . To see this, we replace  $\chi \in H^1(\mathbb{R}^N)$  by  $\psi \in L^2(\mathbb{R}^N, \mathbb{R}^N)$  in the second part of (iii)  $\Rightarrow$  (iv) in [10, Proposition 2.3]. Since  $|\Omega \setminus K_n|$  and  $|\Omega_n \setminus \Omega|$  converge to zero (condition (iv) in [10, Proposition 2.3]), for any given  $\eta > 0$ , we can find  $N \in \mathbb{N}$  sufficiently large so that  $\|\psi\|_{L^2((\Omega_{n_{k_j}} \cup \Omega) \setminus K_N)} \leq \eta$  for all  $j \in \mathbb{N}$  with  $n_{k_j} > N$ . Then

$$\begin{split} \left| \int_{\Omega_{n_{k_j}}} \nabla v_{k_j} \psi dx - \int_{\Omega} \nabla \phi \psi dx \right| \\ &\leq \left| \int_{K_N} (\nabla v_{k_j} - \nabla \phi) \psi dx \right| + \int_{\Omega_{n_{k_j}} \setminus K_N} |\nabla v_{k_j}| |\psi| dx + \int_{\Omega \setminus K_N} |\nabla \phi| |\psi| dx \\ &\leq \left| \int_{K_N} (\nabla v_{k_j} - \nabla \phi) \psi dx \right| + 2\eta \|\psi\|_{L^2(\mathbb{R}^N, \mathbb{R}^N)} \\ &\to 2\eta \|\psi\|_{L^2(\mathbb{R}^N, \mathbb{R}^N)}, \end{split}$$

as  $j \to \infty$ . As  $\eta > 0$  was arbitrary, this means  $i_{n_{k_j}} \nabla v_{k_j} \rightharpoonup i \nabla \phi$  in  $L^2(\mathbb{R}^N, \mathbb{R}^N)$ . Therefore, we conclude that  $(v, w) = (i\phi, i\nabla\phi)$  and the whole sequence  $i_{n_k}v_k \to v = i\phi$ in  $L^2(\mathbb{R}^N)$  and  $i_{n_k}\nabla v_k \rightharpoonup w = i\nabla\phi$  in  $L^2(\mathbb{R}^N, \mathbb{R}^N)$  as  $k \to \infty$ . This proves the Mosco condition (M2).

As in the case of Dirichlet problems, we apply various Mosco conditions from Theorem 1.3.1 to prove that the weak solution  $u_n$  of (1.25) converges to the weak solution u of (1.26). We emphasis that we do not explicitly impose any regularity of the domains and only assume the abstract Mosco convergence of  $H^1(\Omega_n)$  to  $H^1(\Omega)$ . However, the Mosco convergence assumption requires some topological (and/or regularity) conditions on domains. We shall discuss the conditions on domains after we establish the convergence results for solutions.

**Theorem 1.4.12.** Suppose that  $i_n f_n$  is uniformly bounded in  $L^2((0,T), L^2(D))$  and  $i_n u_{0,n}$  is uniformly bounded in  $L^2(D)$ . Assume that  $ri_n f_n \rightharpoonup f$  in  $L^2((0,T), L^2(\Omega))$ 

weakly and  $ri_n u_{0,n} \rightarrow u_0$  in  $L^2(\Omega)$  weakly. If  $H^1(\Omega_n)$  converges to  $H^1(\Omega)$  in the sense of Mosco, then the solution  $i_n u_n$  converges to iu in  $L^2((0,T), L^2(D))$  weakly and  $i_n \nabla u_n$ converges to  $i \nabla u$  in  $L^2((0,T), L^2(D, \mathbb{R}^N))$  weakly.

Proof. By a similar argument as in the proof of Theorem 1.4.3, we have the uniform boundedness of  $(i_n u_n, i_n \nabla u_n)$  in  $L^2((0,T), L^2(D)) \times L^2((0,T), L^2(D, \mathbb{R}^N))$ . We can extract a subsequence (denoted again by  $u_n$ ), such that  $i_n u_n \to w$  in  $L^2((0,T), L^2(D))$ and  $i_n \nabla u_n \to (v_1, \ldots, v_N)$  in  $L^2((0,T), L^2(D, \mathbb{R}^N))$ . The Mosco condition (M2') (from Theorem 1.3.1) implies that  $w \in L^2((0,T), H^1(\Omega))$ .

To show that w = u, we let  $\xi \in H^1(\Omega)$  and  $\phi \in \mathscr{D}([0,T))$  and then use Mosco convergence of  $H^1(\Omega_n)$  to  $H^1(\Omega)$ . In the same way as the proof of Theorem 1.4.3, we get (1.33) holds for all  $\xi \in H^1(\Omega)$  and all  $\phi \in \mathscr{D}([0,T))$ . Hence, by the uniqueness of solution, w = u in  $L^2((0,T), H^1(\Omega))$  and the whole sequence converges weakly.  $\Box$ 

**Lemma 1.4.13.** Suppose that the hypothesis of Theorem 1.4.12 holds. Then for each  $t \in [0,T]$ , we have  $ri_n u_n(t) \rightarrow u(t)$  in  $L^2(\Omega)$ .

*Proof.* We use the same argument as in the proof of Lemma 1.4.6 with Mosco convergence of  $H_0^1(\Omega_n)$  to  $H_0^1(\Omega)$  replaced by Mosco convegence of  $H^1(\Omega_n)$  to  $H^1(\Omega)$  and the fact that  $H^1(\Omega)$  is also dense in  $L^2(\Omega)$ .

In the remainder, we impose the strong convergences of inhomogeneous terms and initial data to obtain the strong convergence of solutions.

**Theorem 1.4.14.** Assume that  $i_n f_n \to if$  in  $L^2((0,T), L^2(D))$  strongly and  $i_n u_{0,n} \to iu_0$  in  $L^2(D)$  strongly. If  $H^1(\Omega_n)$  converges to  $H^1(\Omega)$  in the sense of Mosco, then the solution  $i_n u_n$  converges to iu in  $L^2((0,T), L^2(D))$  strongly and  $i \nabla u_n$  converges to  $i \nabla u$  in  $L^2((0,T), L^2(D, \mathbb{R}^N))$  strongly.

Proof. The proof is similar to the one in Theorem 1.4.7. We show some details here for the sake of completeness. By Theorem 1.4.12,  $(i_n u_n, i_n \nabla u_n)$  converges to  $(iu, i \nabla u)$  in  $L^2((0,T), L^2(D)) \times L^2((0,T), L^2(D, \mathbb{R}^2))$  weakly. Since  $u \in L^2((0,T), H^1(\Omega))$ , we can find  $w_n \in L^2((0,T), H^1(\Omega_n))$  from the Mosco condition (M1') in Theorem 1.3.1 such that  $i_n w_n \to iu$  in  $L^2((0,T), L^2(D))$  and  $i_n \nabla w_n \to i \nabla u$  in  $L^2((0,T), L^2(D, \mathbb{R}^N))$ . For  $t \in [0, T]$ , we consider

$$d_{n}(t) = \frac{1}{2} \|i_{n}u_{n}(t) - iu(t)\|_{L^{2}(D)}^{2} + \alpha \int_{0}^{t} \|u_{n}(s) - w_{n}(s)\|_{L^{2}(\Omega_{n})}^{2} ds + \alpha \int_{0}^{t} \|\nabla u_{n}(s) - \nabla w_{n}(s)\|_{L^{2}(\Omega_{n},\mathbb{R}^{N})}^{2} ds$$
(1.49)  
$$= \frac{1}{2} \|i_{n}u_{n}(t) - iu(t)\|_{L^{2}(D)}^{2} + \alpha \int_{0}^{t} \|u_{n}(s) - w_{n}(s)\|_{H^{1}(\Omega_{n})}^{2} ds.$$

By (1.20) (with  $\lambda = 0$ ), we can show that  $d_n$  satisfies (1.36) with  $\delta = 0$  for all  $n \in \mathbb{N}$ . It can be easily seen from the weak convergence of  $(i_n u_n, i_n \nabla u_n)$  and the strong convergence of  $(i_n w_n, i_n \nabla w_n)$  to  $(iu, i \nabla u)$  in  $L^2((0, T), L^2(D)) \times L^2((0, T), L^2(D, \mathbb{R}^N))$  that (1.37) and (1.38) also hold with  $\delta = 0$ . By using Lemma 1.4.13 instead of Lemma 1.4.6, we obtain (1.39). Finally, using that  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$  strongly,  $i_n f_n \to if$  in  $L^2((0,T), L^2(D))$  strongly and the weak convergence of solution  $i_n u_n \to i u$  in  $L^2((0,T), L^2(D))$ , we get (1.40) with  $\delta = 0$ . Hence,  $d_n(t) \to 0$  for all  $t \in [0,T]$ . This shows pointwise convergence of  $i_n u_n(t)$  to iu(t) in  $L^2(D)$  for all  $t \in [0,T]$ . Moreover, by taking t = T we get

$$\begin{split} \int_0^T \|i_n u_n(s) - iu(s)\|_{L^2(D)}^2 \, ds \\ &\leq \int_0^T \|i_n u_n(s) - i_n w_n(s)\|_{L^2(D)}^2 \, ds + \int_0^T \|i_n w_n(s) - iu(s)\|_{L^2(D)}^2 \, ds \\ &\to 0, \end{split}$$

and

$$\int_{0}^{T} \|i_{n} \nabla u_{n}(s) - i \nabla u(s)\|_{L^{2}(D,\mathbb{R}^{N})}^{2} ds$$
  

$$\leq \int_{0}^{T} \|i_{n} \nabla u_{n}(s) - i_{n} \nabla w_{n}(s)\|_{L^{2}(D,\mathbb{R}^{N})}^{2} ds + \int_{0}^{T} \|i_{n} \nabla w_{n}(s) - i \nabla u(s)\|_{L^{2}(D,\mathbb{R}^{N})}^{2} ds$$
  

$$\to 0.$$

This proves the strong convergence  $i_n u_n \to iu$  in  $L^2((0,T), L^2(D))$  and  $i_n \nabla u_n \to i \nabla u$ in  $L^2((0,T), L^2(D, \mathbb{R}^N))$ .

Recall that we embed the space  $K = H^1(\Omega)$  in  $V = L^2(D) \times L^2(D, \mathbb{R}^N)$ . If v is a function in  $W((0,T), H^1(\Omega), H^1(\Omega)')$ , then  $v' \in L^2((0,T), H^1(\Omega)')$ . It is not always true that we can embed v'(t) in  $V' = L^2(D) \times L^2(D, \mathbb{R}^N)$  a.e.  $t \in (0,T)$  and claim that  $v \in W((0,T), V, V') \cap L^2((0,T), K)$ . However, a similar argument as in the proof of Theorem 1.3.1  $(i) \Rightarrow (iii)$  for the Mosco condition (M1'') gives the following result. **Lemma 1.4.15.** Suppose that  $H^1(\Omega_n)$  converges to  $H^1(\Omega)$  in the sense of Mosco, If  $w \in C^{\infty}([0,T], H^1(\Omega))$ , then there exists  $w_n \in C^{\infty}([0,T], H^1(\Omega_n))$  such that  $i_n w_n$  converges to iw in  $C^{\infty}([0,T], L^2(D))$  strongly.

Proof. We note that Proposition 1.3.6 gives uniform convergence of the approximation sequence in  $V = L^2(D) \times L^2(D, \mathbb{R}^N)$ . The proof follows from the same arguments as in the proof of Theorem 1.3.1  $(i) \Rightarrow (iii)$ . The only difference is that we assume here  $w \in C^{\infty}([0,T], H^1(\Omega))$ . Hence, the stretched function  $w_{\delta} = w \circ S_{\delta}^{-1}$  belongs to  $C^{\infty}([-\delta, T + \delta], H^1(\Omega))$ . We point out that, by using uniform continuity of the *k*-th order derivative  $w^{(k)}$  on [0,T], the restriction of  $w_{\delta}$  to [0,T] converges to w in  $C^{\infty}([0,T], H^1(\Omega))$ . This gives the required convergence in  $C^{\infty}([0,T], L^2(D))$ .

Using the above lemma, we show in the next theorem that the solution  $u_n$  of (1.25) indeed converges uniformly with respect to  $t \in [0, T]$ .

**Theorem 1.4.16.** Suppose that the hypothesis of Theorem 1.4.14 holds. Then the solution  $i_n u_n$  converges to iu in  $C([0,T], L^2(D))$  strongly.

*Proof.* As in the proof of Theorem 1.4.9, it requires to show the uniform convergence of  $(i_n u_n(t)|iu(t))_{L^2(D)} \rightarrow (iu(t)|iu(t))_{L^2(D)}$ .

Let  $\varepsilon > 0$  arbitrary. By a similar argument as in the proof Lemma 1.3.5, we have the density of  $C^{\infty}([0,T], H^1(\Omega))$  in  $W((0,T), H^1(\Omega), H^1(\Omega)')$ . Since the solution u is in the space  $W((0,T), H^1(\Omega), H^1(\Omega)')$ , there exists  $w \in C^{\infty}([0,T], H^1(\Omega))$  such that

$$\|w - u\|_{W((0,T),H^1(\Omega),H^1(\Omega)')} \le \varepsilon.$$

As  $W((0,T), H^1(\Omega), H^1(\Omega)')$  is continuously embedded in  $C([0,T], L^2(\Omega))$ , we can indeed choose  $w \in C^{\infty}([0,T], H^1(\Omega))$  such that

$$\|w(t) - u(t)\|_{L^2(\Omega)} \le \varepsilon, \tag{1.50}$$

for all  $t \in [0,T]$ . By Lemma 1.4.15, there exists  $w_n \in C^{\infty}([0,T], H^1(\Omega_n))$  such that  $i_n w_n \to iw$  in  $C^{\infty}([0,T], L^2(D))$ . We can write

$$\begin{aligned} \left| (i_n u_n(t) - iu(t)|iu(t))_{L^2(D)} \right| &\leq \left| (i_n u_n(t)|i_n w_n(t))_{L^2(D)} - (iu(t)|iw(t))_{L^2(D)} \right| \\ &+ \left| (i_n u_n(t) - iu(t)|iu(t) - iw(t))_{L^2(D)} \right| \\ &+ \left| (i_n u_n(t)|iw(t) - i_n w_n(t))_{L^2(D)} \right|, \end{aligned}$$
(1.51)

for all  $n \in \mathbb{N}$ . Since  $u_n$  is the solution of (1.25),

$$\begin{aligned} (i_n u_n(t)|i_n w_n(t))_{L^2(D)} &= (i_n u_{0,n}|i_n w_n(0))_{L^2(D)} + \int_0^t \langle f_n(s), w_n(s) \rangle \ ds \\ &+ \int_0^t \langle w'_n(s), u_n(s) \rangle \ ds - \int_0^t a_n(s; u_n(s), w_n(s)) \rangle \ ds, \end{aligned}$$

for all  $n \in \mathbb{N}$ . Taking into consideration that  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$ ,  $i_n f_n \to i f$  in  $L^2((0,T), L^2(D))$  and  $i_n w'_n \to i w'$  in  $C([0,T], L^2(D))$ , we conclude that

$$(i_n u_n(t)|i_n w_n(t))_{L^2(D)} \to (iu(t)|iw(t))_{L^2(D)}$$
 (1.52)

uniformly with respect to  $t \in [0, T]$ . Moreover, by (1.50) and the uniform boundedness of solutions as in (1.46), we have

$$\begin{aligned} \left| \left( i_n u_n(t) - i u(t) | i u(t) - \tilde{w}(t) \right)_{L^2(D)} \right| \\ \leq \| i_n u_n(t) - i u(t) \|_{L^2(D)} \| i u(t) - i w(t) \|_{L^2(D)} \\ \leq 2M_0 \varepsilon, \end{aligned}$$
(1.53)

for all  $t \in [0,T]$  and for all  $n \in \mathbb{N}$ . Finally, as  $i_n w_n \to iw$  in  $C^{\infty}([0,T], L^2(D))$ , there exists  $N_{\varepsilon} \in \mathbb{N}$  such that

$$\|i_n w_n(t) - iw(t)\|_{L^2(D)} \le \varepsilon,$$

for all  $t \in [0, T]$  and for all  $n \ge N_{\epsilon}$ . Hence,

$$\left| \left( i_n u_n(t) | i_n w_n(t) - i w(t) \right)_{L^2(D)} \right| \leq \| i_n u_n(t) \|_{L^2(D)} \| \| i_n w_n(t) - i w(t) \|_{L^2(D)}$$

$$\leq M_0 \varepsilon,$$
(1.54)

for all  $t \in [0, T]$  and for all  $n \ge N_{\varepsilon}$ . Therefore, by (1.51) – (1.54), there exist  $\tilde{N}_{\varepsilon} \in \mathbb{N}$ and a positive constant C such that

$$\left| \left( i_n u_n(t) - i u(t) | i u(t) \right)_{L^2(D)} \right| \le C \varepsilon,$$

for all  $t \in [0,T]$  and for all  $n \ge \tilde{N}_{\varepsilon}$ . As  $\varepsilon > 0$  was arbitrary, this proves the required uniform convergence of  $(i_n u_n(t)|iu(t))_{L^2(D)} \rightarrow (iu(t)|iu(t))_{L^2(D)}$  with respect to  $t \in [0,T]$ .

We can use the same criterion on  $\Omega_n$  and  $\Omega$  for elliptic equations with Neumann boundary condition to conclude the convergence of solutions of parabolic equations subject to Neumann boundary condition under domain perturbation. In particular, in two dimensional space, conditions on domains given in [21, Theorem 3.1] imply the convergence of solutions of non-autonomous parabolic equations (1.21) subject to Neumann boundary condition. **Corollary 1.4.17.** Assume that  $i_n f_n \to if$  in  $L^2((0,T), L^2(D))$  strongly and  $i_n u_{0,n} \to iu_0$  in  $L^2(D)$  strongly. Let  $\Omega_n$  and  $\Omega$  be uniformly bounded open sets in  $\mathbb{R}^2$  such that  $\Omega_n$  converges in the Hausdorff complementary topology to  $\Omega$  and the number of connected components of  $\Omega_n^c$  and  $\Omega^c$  are uniformly bounded. If the Lebesgue measure  $|\Omega_n| \to |\Omega|$  as  $n \to \infty$ , then the solution  $i_n u_n$  converges to iu in  $C([0,T], L^2(D))$  strongly and  $i_n \nabla u_n$  converges to  $i\nabla u$  in  $L^2((0,T), L^2(D, \mathbb{R}^2))$  strongly.

Examples of domains in  $\mathbb{R}^2$  satisfying the conditions of [21] include the following:

- (i) a domain obtained by cutting a line segment and then rotating the segment about a point as shown in Figure 1.3;
- (ii) a square with fingers attached in Figure 1.2. In contrast to Dirichlet problems, here we require the measure of fingers converges to zero.



Figure 1.3: Rotating a line segment inside a domain

Remark 1.4.18. The assumptions that  $i_n f_n \to if$  in  $L^2((0,T), L^2(D))$  strongly and that  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$  strongly were imposed to overcome the lack of compactness result similar to Lemma 1.4.4 for a sequence in  $W((0,T), H^1(\Omega), H^1(\Omega)')$ . In particular, this assumption is required to obtain the convergence  $\int_0^T \langle f_n(s), u_n(s) \rangle ds \to \int_0^T \langle f(s), u(s) \rangle ds$  in the proof of Theorem 1.4.14. However, if we impose some regularity on the domains, we could only assume weak convergences of the inhomogeneous terms and the initial data to obtain the convergence of solutions in  $C([\delta, T], L^2(D))$  for all  $\delta \in (0, T]$  as stated in Dirichlet problems.

We give an example of domains satisfying the *cone condition* (see [1, Section 4.3]) below. Clearly, the example below is excluded from [10] as the domains are not Lipschitz. *Example* 1.4.19. Let N = 2 and let

$$\Omega := \{ x \in \mathbb{R}^2 : |x| < 1 \} \setminus \{ (x_1, 0) : 0 \le x_1 < 1 \},\$$
$$\Omega_n := \{ x \in \mathbb{R}^2 : |x| < 1 \} \setminus \{ (x_1, 0) : \delta_n \le x_1 < 1 \},\$$

where  $\delta_n \searrow 0$  (shown in Figure 1.4). This example is an exterior perturbation of the



Figure 1.4: Cutting into a unit disk.

domain, that is  $\Omega \subset \Omega_{n+1} \subset \Omega_n$  for all  $n \in \mathbb{N}$ . It is easy to see that  $\Omega$  and  $\Omega_n$  satisfy the cone condition uniformly with respect to  $n \in \mathbb{N}$ , but  $H^1(\Omega)$  and  $H^1(\Omega_n)$  do not have the extension property. Moreover, these domains satisfy the conditions in [21] (i.e. conditions listed in Corollary 1.4.17). Hence,  $H^1(\Omega_n)$  converges to  $H^1(\Omega)$  in the sense of Mosco. Note that here we take D to be the open unit disk center at 0 in  $\mathbb{R}^2$ . In this example, we only need that  $ri_n f_n \to f$  in  $L^2((0,T), L^2(\Omega))$  and  $ri_n u_{0,n} \to u_0$  in  $L^2(\Omega)$ to conclude the convergence of solutions  $i_n u_n \to iu$  in  $C([\delta, T], L^2(D))$  for all  $\delta \in (0, T]$ . In addition, if  $i_n u_{0,n} \to iu_0$  in  $L^2(D)$ , then the assertion holds for  $\delta = 0$ .

To see this, we note from Lemma 1.4.13 that  $ri_n u_n(t) \rightharpoonup u(t)$  in  $L^2(\Omega)$  weakly for all  $t \in [0,T]$ . Since  $u \in L^2((0,T), H^1(\Omega))$ , we have  $u(t) \in H^1(\Omega)$  for almost everywhere  $t \in (0,T)$ . Fix now such  $t \in (0,T)$ . By the continuity of solutions  $u_n \in C([0,T], L^2(\Omega_n))$ , for each  $n \in \mathbb{N}$  we can choose  $\rho_n > 0$  such that

$$||u_n(s) - u_n(t)||_{L^2(\Omega_n)} \le \frac{1}{n}$$

for all  $s \in (t - \rho_n, t + \rho_n) \cap (0, T)$ . As  $u_n \in L^2((0, T), H^1(\Omega_n))$ , we can choose  $t_n \in (t - \rho_n, t + \rho_n) \cap (0, T)$  such that  $u_n(t_n) \in H^1(\Omega_n)$  for all  $n \in \mathbb{N}$ . For these choices of  $t_n$ , we have  $\|u_n(t_n) - u_n(t)\|_{L^2(\Omega_n)} \to 0$  as  $n \to \infty$ . It follows that  $ri_nu_n(t_n) \rightharpoonup u(t)$  in  $L^2(\Omega)$  weakly. Since  $\Omega \subset \Omega_n$  for all  $n \in \mathbb{N}$ , the restriction  $ru_n(t_n)$  belongs to  $H^1(\Omega)$  for all  $n \in \mathbb{N}$ . Hence, it follows from the weak convergence of  $ri_nu_n(t_n) = ru_n(t_n) \rightharpoonup u(t)$  in  $L^2(\Omega)$  that

$$\int_{\Omega} \partial_j (r u_n(t_n)) \phi dx = -\int_{\Omega} r u_n(t_n) \partial_j \phi dx \to -\int_{\Omega} u(t) \partial_j \phi dx = \int_{\Omega} \partial_j u(t) \phi dx,$$

for all  $\phi \in C_c^{\infty}(\Omega)$  for j = 1, 2. This means  $\nabla r u_n(t_n) \rightharpoonup \nabla u(t)$  in  $L^2(\Omega, \mathbb{R}^2)$ . Thus,  $r u_n(t_n)$  is bounded in  $H^1(\Omega)$ . As  $\Omega$  is bounded and satisfies the cone condition, we have from the Rellich-Kondrachov theorem that the embedding  $H^1(\Omega) \hookrightarrow L^2(\Omega)$  is compact (see [1, Theorem 6.2]). Therefore,  $ru_n(t_n)$  has a subsequence which converges in  $L^2(\Omega)$ strongly. Since we have a prior knowledge of weak convergence  $ru_n(t_n) \rightarrow u(t)$  in  $L^2(\Omega)$ , we conclude that the whole sequence  $ru_n(t_n) \rightarrow u(t)$  in  $L^2(\Omega)$  strongly. By the choices of  $t_n$ , we conclude that  $ru_n(t) \rightarrow u(t)$  in  $L^2(\Omega)$  strongly. Since the above argument works for almost everywhere  $t \in (0,T)$ , we deduce from the dominated convergence theorem that  $ri_nu_n = ru_n \rightarrow u$  in  $L^2((0,T), L^2(\Omega))$  strongly. As the cutting line is a set of measure zero in  $\mathbb{R}^2$ , we have  $i_nu_n \rightarrow iu$  in  $L^2((0,T), L^2(D))$ . By extracting a subsequence (indexed again by n), we can choose  $\delta$  arbitrarily closed to zero in (0,T]such that  $i_nu_n(\delta) \rightarrow iu(\delta)$  in  $L^2(D)$ . We can now follow the proof of Theorem 1.4.14 and Theorem 1.4.16 with the integration taken over  $[\delta, T]$  instead of [0, T] to conclude the claim.

### 1.5 Convergence of solutions of final value problems

In this short section, we give some remarks on domain perturbation for final value problems. It is a direct application from initial value problems. Let T > 0. We consider the final value problem

$$\begin{cases} -\frac{\partial u}{\partial t} + \mathcal{A}(t)u = g(x, t) & \text{in } \Omega \times [0, T) \\ \mathcal{B}(t)u = 0 & \text{on } \partial\Omega \times [0, T) \\ u(\cdot, T) = u_T & \text{in } \Omega, \end{cases}$$
(1.55)

where  $\mathcal{A}$  and  $\mathcal{B}$  are defined by (0.5) and (0.6) in Introduction, respectively. As in the case of initial value problem, we choose the evolution triple  $V \stackrel{d}{\hookrightarrow} H \stackrel{d}{\to} V'$  as in (1.11). Let  $a(t, \cdot, \cdot)$  be the corresponding bilinear form on V given by (1.10) and  $A \in \mathscr{L}(V, V')$  be the operator defined by (1.4). Then we can consider the final value problem in an abstract form

$$\begin{cases} -u'(t) + A(t)u = g(t) & \text{ for } t \in [0, T) \\ u(T) = u_T \end{cases}$$
(1.56)

in V' for a given final condition  $u_T \in H$  and  $g \in L^2((0,T), V')$ .

**Definition 1.5.1.** A function  $u \in W(0, T, V, V')$  is called a weak solution of (1.55) if u is a solution of (1.56), that is u satisfies (1.56) a.e.  $t \in [0, T)$ .

As in [2, Chapter V 2.5], we may associate (1.56) with the initial value problem

$$\begin{cases} v'(t) + A(T-t)v = g(T-t) & \text{for } t \in (0,T] \\ v(0) = u_T \end{cases}$$
(1.57)

in V'. Let u(t) := v(T - t). Then it is trivial that u is a solution of the final value problem (1.56) if and only if v is a solution of the initial value problem (1.57). Moreover, u is a weak solution of (1.55) if and only if  $u \in L^2((0,T), V)$  and

$$-\int_{0}^{T} (u(t)|v)\phi'(t)dt + \int_{0}^{T} a(t, u(t), v)\phi(t)dt$$
  
=  $(u_{T}|v)\phi(T) + \int_{0}^{T} \langle g(t), v \rangle \phi(t)dt,$  (1.58)

for all  $v \in V$  and for all  $\phi \in \mathscr{D}((0,T])$ .

**Theorem 1.5.2.** Given  $g \in L^2((0,T),V')$  and  $u_T \in H$ , there exists a unique weak solution u of (1.55) satisfying

$$||u||_{W(0,T,V,V')} \le C \Big( ||u_T||_H + ||g||_{L^2((0,T),V')} \Big).$$
(1.59)

Moreover, if  $\lambda = 0$  in (1.3), then

$$\|u(t)\|_{H}^{2} + \alpha \int_{t}^{T} \|u(s)\|_{V}^{2} ds \le \|u_{T}\|_{H}^{2} + \alpha^{-1} \int_{t}^{T} \|g(s)\|_{V}^{2} ds, \qquad (1.60)$$

for all  $t \in [0,T]$ .

*Proof.* By the above observation and Theorem 1.1.2, there exists a unique weak solution  $v \in W(0, T, V, V')$  of the associated initial value problem satisfying the estimates (1.7) and (1.8) with  $u_0$ , f and f(s) replaced by  $u_T$ , g and g(T-s), respectively. Taking into account that u(t) = v(T-t) is a weak solution of the corresponding final value problem, we obtain the estimates (1.59) and (1.60).

We now consider domain perturbation for final value problems with Dirichlet or Neumann boundary conditions. We assume the same assumptions on the domains  $\Omega_n$ and  $\Omega$ , and on the bilinear forms  $a_n(\cdot, u, v)$  for  $u, v \in V_n$  and  $a(\cdot, u, v)$  for  $u, v \in V$ as in Section 1.4. Also, the operators  $\mathcal{A}_n, \mathcal{A}, \mathcal{B}_n$  and  $\mathcal{B}$ , and the induced operators  $A_n \in \mathscr{L}(V_n, V'_n)$  and  $A \in \mathscr{L}(V, V')$  are defined in the same way as in Section 1.4. Similar to the initial value problems, we study the following perturbation of the final value problem

$$-\frac{\partial u}{\partial t} + \mathcal{A}_n(t)u = g_n(x,t) \qquad \text{in } \Omega_n \times [0,T)$$
$$\mathcal{B}_n(t)u = 0 \qquad \text{on } \partial \Omega_n \times [0,T)$$
$$u(\cdot,T) = u_{T,n} \qquad \text{in } \Omega,$$
$$(1.61)$$

where  $\mathcal{B}_n$  is either Dirichlet boundary condition or Neumann boundary condition. As usual, we write (1.61) in the abstract form as

$$\begin{cases} -u'(t) + A_n(t)u = g_n(t) & \text{ for } t \in [0,T) \\ u(T) = u_{T,n}. \end{cases}$$
(1.62)

As a consequence of the continuity of solutions of initial value problems under domain perturbation, we can state a convergence result for final value problems.

**Theorem 1.5.3.** Suppose that  $i_ng_n$  is uniformly bounded in  $L^2((0,T), L^2(D))$  and  $i_nu_{T,n}$  is uniformly bounded in  $L^2(D)$ . Let  $u_n$  and u be the (unique) weak solutions of (1.62) and (1.56), respectively.

- (i) Dirichlet problems: Assume that ringn → g in L<sup>2</sup>((0,T), L<sup>2</sup>(Ω)) weakly and rinu<sub>T,n</sub> → u<sub>T</sub> in L<sup>2</sup>(Ω) weakly. If H<sup>1</sup><sub>0</sub>(Ω<sub>n</sub>) converges to H<sup>1</sup><sub>0</sub>(Ω) in the sense of Mosco, then i<sub>n</sub>u<sub>n</sub> converges to iu in L<sup>2</sup>((0,T), H<sup>1</sup>(D)) and in C([0,δ], L<sup>2</sup>(D)) for every δ ∈ [0,T). In addition, if i<sub>n</sub>u<sub>T,n</sub> → iu<sub>T</sub> in L<sup>2</sup>(D) strongly, the above assertion holds for δ = T.
- (ii) Neumann problems: Assume that  $i_ng_n \to ig$  in  $L^2((0,T), L^2(D))$  strongly and  $i_nu_{T,n} \to iu_T$  in  $L^2(D)$  strongly. If  $H^1(\Omega_n)$  converges to  $H^1(\Omega)$  in the sense of Mosco, then  $(i_nu_n, i_n\nabla u_n)$  converges to  $(iu, i\nabla u)$  in  $L^2((0,T), L^2(D)) \times$  $L^2((0,T), L^2(D, \mathbb{R}^N))$  and  $i_nu_n$  converges to iu in  $C([0,T], L^2(D))$ .

*Proof.* This immediately follows from the above observation that  $v_n(t) = u_n(T-t)$  and v(t) = u(T-t) are weak solutions of the corresponding initial value problems and the results in Theorems 1.4.7, 1.4.9, 1.4.14 and 1.4.16.

We remark that if -u'(t) is replaced by u'(t) in the final value problem (1.56), then the associated initial value problem is v'(t) - A(T - t)v(t) = -g(T - t) with  $v(0) = u_T$ . We cannot apply the results for parabolic initial value problems in Section 1.4 as -A(T - t) no longer satisfies the ellipticity assumption. We will discuss final value problems for autonomous parabolic equations of the form u'(t) + Au(t) = g(t) for  $t \in [0, T)$  with  $u(T) = u_T$  where A is a generator of a group later. In particular, we prove convergence of the component of solutions of autonomous parabolic equations in the unstable direction in Lemma 4.2.4 (ii).

### 1.6 Domain perturbation for parabolic variational inequalities

In Section 1.4, the applications of the equivalence of Mosco convergences established in Theorem 1.3.1 were given when  $K_n$  and K are closed subspaces of V. The purpose of this section is to give an application when  $K_n$  and K are just closed and convex subsets of V. We show here a similar convergence of solutions of parabolic variational inequalities.

Suppose that K is a closed and convex subset of V. We denote by

$$L^{2}((0,T),K) := \{ u \in L^{2}((0,T),V) \mid u(t) \in K \text{ a.e.} \}.$$

For each  $t \in (0, T)$ , suppose  $a(t; \cdot, \cdot)$  is a continuous bilinear form on V satisfying (1.2) and (1.3). As before, we denote the induced linear operator by A(t). Given  $u_0 \in K$  and  $f \in L^2((0, T), V')$ , we wish to find u such that for a.e.  $t \in (0, T)$ ,  $u(t) \in K$  and

$$\begin{cases} \langle u'(t), v - u(t) \rangle + \langle A(t)u(t), v - u(t) \rangle - \langle f(t), v - u(t) \rangle \ge 0, \quad \forall v \in K \\ u(0) = u_0. \end{cases}$$
(1.63)

A function  $u \in W((0,T), V, V')$  satisfying (1.63) is called a *strong solution* of parabolic variational inequality (1.63). There are various (slightly different) definitions of weak solution of parabolic variational inequalities (see e.g.[18, 55, 60, 61]). We shall define a weak notion of solution similar to the one in [55] as follows.

**Definition 1.6.1.** A function u is a *weak solution* of parabolic variational inequality (1.63) if  $u \in L^2((0,T), K)$  and

$$\int_{0}^{T} \langle v'(t), v(t) - u(t) \rangle + \langle A(t)u(t), v(t) - u(t) \rangle - \langle f(t), v(t) - u(t) \rangle dt + \frac{1}{2} \|v(0) - u_0\|_{H}^{2} \ge 0,$$
(1.64)

for all  $v \in W((0,T), V, V') \cap L^2((0,T), K)$ .

The existence and uniqueness of weak solutions of parabolic variational inequalities have been studied by various authors according to their definitions. In our case, we can state the result in the following theorem.

**Theorem 1.6.2.** Given  $u_0 \in K$  and  $f \in L^2((0,T), V')$ . There exists a unique weak solution u of the parabolic variational inequality (1.63) satisfying  $u \in L^{\infty}((0,T), H)$ .

Note that the existence of our weak solution follows immediately from the existence results in [60, Theorem 6.2]. The uniqueness can be proved in the same way as in [61, Theorem 2.3].

Similar to domain perturbation for parabolic equations, we examine the behaviour of weak solutions of parabolic variational inequalities when the underlying convex set K is perturbed.

Let  $K_n$  and K be closed and convex subsets of V. For each  $t \in (0, T)$ , suppose  $a(t; \cdot, \cdot)$  is a continuous bilinear form on V satisfying (1.2) and (1.3). For simplicity, we assume that  $\lambda = 0$  in (1.3). We denote by A(t) the linear operator induced by  $a(t; \cdot, \cdot)$ . Let us consider the following parabolic variational inequalities. Given  $u_{0,n} \in K_n$  and  $f_n \in L^2((0,T), V')$ , we want to find  $u_n$  such that for a.e.  $t \in (0,T)$ ,  $u_n(t) \in K_n$  and

$$\begin{cases} \langle u'(t), v - u(t) \rangle + \langle A(t)u(t), v - u(t) \rangle - \langle f_n(t), v - u(t) \rangle \ge 0, \quad \forall v \in K_n \\ u(0) = u_{0,n}. \end{cases}$$
(1.65)

When  $K_n$ ,  $f_n$  and  $u_{0,n}$  converge to K, f and  $u_0$  respectively, we wish to obtain convergence results of the weak solution of (1.65) to that of the limit inequalities (1.63). In the remainder of this section, we denote the weak solution of (1.65) by  $u_n$  and the weak solution of (1.63) by  $u_n$ .

**Theorem 1.6.3.** Let  $u_{0,n} \in K_n$ ,  $u_0 \in K$  and  $f_n, f \in L^2((0,T), V')$ . Suppose that  $f_n \to f$  in  $L^2((0,T), V')$  strongly,  $u_{0,n} \to u_0$  in V weakly and  $u_{0,n} \to u_0$  in H strongly. Then the sequence of weak solutions  $u_n$  is bounded in  $L^2((0,T), V)$ .

Proof. Let  $v \in W((0,T), V, V') \cap L^2((0,T), K)$  be the constant function defined by  $v(t) := u_0$  for  $t \in [0,T]$ . Similarly,  $v_n \in W((0,T), V, V') \cap L^2((0,T), K_n)$  defined by  $v_n(t) := u_{0,n}$  for  $t \in [0,T]$ . It follows that  $v_n \rightharpoonup v$  in  $L^2((0,T), V)$ . Since  $u_n$  is the weak solution of (1.65), we have

$$\int_{0}^{T} \langle A(t)u_{n}(t), u_{n}(t) - v_{n}(t) \rangle dt$$
  

$$\leq \int_{0}^{T} \langle v_{n}'(t), v_{n}(t) - u_{n}(t) \rangle - \langle f_{n}(t), v_{n}(t) - u_{n}(t) \rangle dt + \frac{1}{2} \|v_{n}(0) - u_{0,n}\|_{H}^{2}$$
  

$$= -\int_{0}^{T} \langle f_{n}(t), v_{n}(t) - u_{n}(t) \rangle dt.$$

It follows that

$$\int_{0}^{T} \langle A(t)u_{n}(t) - A(t)v_{n}(t), u_{n}(t) - v_{n}(t) \rangle dt$$
  

$$\leq \int_{0}^{T} \langle A(t)v_{n}(t), v_{n}(t) - u_{n}(t) \rangle - \langle f_{n}(t), v_{n}(t) - u_{n}(t) \rangle dt$$
  

$$\leq \|A(t)v_{n} - f_{n}\|_{L^{2}((0,T),V')} \|v_{n} - u_{n}\|_{L^{2}((0,T),V)}.$$

By the coercivity of A(t), we get

$$\alpha \|u_n - v_n\|_{L^2((0,T),V)} \le \|A(t)v_n - f_n\|_{L^2((0,T),V')}.$$

By the weak convergences of  $v_n$  and  $f_n$ , we conclude that  $u_n$  is bounded in  $L^2((0,T), V)$ .

**Theorem 1.6.4.** Let  $u_{0,n} \in K_n$ ,  $u_0 \in K$  and  $f_n, f \in L^2((0,T), V')$ . Suppose that  $f_n \to f$  in  $L^2((0,T), V')$  strongly,  $u_{0,n} \rightharpoonup u_0$  in V weakly and  $u_{0,n} \to u_0$  in H strongly. If  $K_n$  converges to K in the sense of Mosco, then the sequence of weak solutions  $u_n$  converges to u in  $L^2((0,T), V)$  weakly.

Proof. By Theorem 1.6.3, we can extract a subsequence of  $u_n$  (denoted again by  $u_n$ ) such that  $u_n \to \kappa$  in  $L^2((0,T), V)$  weakly. Since  $u_n \in L^2((0,T), K_n)$ , we apply the Mosco condition (M2') (from Theorem 1.3.1) to deduce that the weak limit  $\kappa$  belongs to  $L^2((0,T), K)$ . By the uniqueness of weak solutions, it suffices to prove that  $\kappa$  satisfies (1.64) (with u replaced by  $\kappa$ ) in the definition of weak solution.

By the Mosco condition (M1'), there exists  $w_n \in L^2((0,T), K_n)$  such that  $w_n \to \kappa$ in  $L^2((0,T), V)$ . Let  $v \in W((0,T), V, V') \cap L^2((0,T), K)$  be arbitrary. We again apply Theorem 1.3.1 (Mosco condition (M1'')) to get a sequence of functions  $v \in$  $W((0,T), V, V') \cap L^2((0,T), K_n)$  such that  $v_n \to v$  in W((0,T), V, V'). For each  $n \in \mathbb{N}$ , we can write

$$\begin{aligned} \langle A(t)w_n(t), v_n(t) - u_n(t) \rangle \\ &= \langle A(t)u_n(t), v_n(t) - u_n(t) \rangle + \langle A(t)w_n(t) - A(t)u_n(t), v_n(t) - u_n(t) \rangle \\ &= \langle A(t)u_n(t), v_n(t) - u_n(t) \rangle + \langle A(t)w_n(t) - A(t)u_n(t), w_n(t) - u_n(t) \rangle \\ &+ \langle A(t)w_n(t) - A(t)u_n(t), v_n(t) - w_n(t) \rangle. \end{aligned}$$

Hence, by the definition of weak solution on  $K_n$  and the coercivity of A(t), we have

$$\int_{0}^{T} \langle v_{n}'(t), v_{n}(t) - u_{n}(t) \rangle + \langle A(t)w_{n}(t), v_{n}(t) - u_{n}(t) \rangle dt$$
  
$$- \int_{0}^{T} \langle f_{n}(t), v_{n}(t) - u_{n}(t) \rangle dt + \frac{1}{2} \|v_{n}(0) - u_{0,n}\|_{H}^{2}$$
  
$$= \int_{0}^{T} \langle v_{n}'(t), v_{n}(t) - u_{n}(t) \rangle + \langle A(t)u_{n}(t), v_{n}(t) - u_{n}(t) \rangle dt$$
  
$$- \int_{0}^{T} \langle f_{n}(t), v_{n}(t) - u_{n}(t) \rangle dt + \frac{1}{2} \|v_{n}(0) - u_{0,n}\|_{H}^{2}$$
  
$$+ \int_{0}^{T} \langle A(t)w_{n}(t) - A(t)u_{n}(t), w_{n}(t) - u_{n}(t) \rangle dt$$
  
$$+ \int_{0}^{T} \langle A(t)w_{n}(t) - A(t)u_{n}(t), v_{n}(t) - w_{n}(t) \rangle dt$$
  
$$\geq \int_{0}^{T} \langle A(t)w_{n}(t) - A(t)u_{n}(t), v_{n}(t) - w_{n}(t) \rangle dt,$$

for all  $n \in \mathbb{N}$ . Letting  $n \to \infty$ , we get

$$\int_0^T \langle v'(t), v(t) - \kappa(t) \rangle + \langle A(t)\kappa(t), v(t) - \kappa(t) \rangle dt$$
$$- \int_0^T \langle f(t), v(t) - \kappa(t) \rangle dt + \frac{1}{2} \|v(0) - u_0\|_H^2 \ge 0.$$

This implies that  $\kappa$  is a weak solution of (1.63) as required.

We finally prove the strong convergence of solutions.

**Theorem 1.6.5.** Let  $u_{0,n} \in K_n$ ,  $u_0 \in K$  and  $f_n, f \in L^2((0,T), V')$ . Suppose that  $f_n \to f$  in  $L^2((0,T), V')$  strongly,  $u_{0,n} \rightharpoonup u_0$  in V weakly and  $u_{0,n} \to u_0$  in H strongly. If  $K_n$  converges to K in the sense of Mosco, then the sequence of weak solutions  $u_n$  converges to u in  $L^2((0,T), V)$  strongly.

*Proof.* By the coercivity of A(t), we have

$$\liminf_{n \to \infty} \int_0^T \langle A(t)u_n(t) - A(t)u(t), u_n(t) - u(t) \rangle \ dt \ge 0.$$
(1.66)

For each  $\epsilon > 0$ , we define  $u_{\epsilon}$  by

$$\epsilon u'_{\epsilon} + u_{\epsilon} = u$$
$$u_{\epsilon}(0) = u_0$$

Then  $u_{\epsilon} \in W((0,T), V, V') \cap L^2((0,T), K)$  and  $u_{\epsilon} \to u$  in  $L^2((0,T), V)$  as  $\epsilon \to 0$  (see in the proof of [61, Theorem 2.3]). For each  $\epsilon > 0$ , the Mosco condition (M1'') (from

Theorem 1.3.1) implies that there exists  $u_{\epsilon,n} \in W((0,T), V, V') \cap L^2((0,T), K_n)$  such that  $u_{\epsilon,n} \to u_{\epsilon}$  in W((0,T), V, V') as  $n \to \infty$ . Since  $u_n$  is the weak solution of (1.65), we have

$$\int_{0}^{T} \langle A(t)u_{n}(t), u_{n}(t) - u(t) \rangle dt$$
  

$$\leq \int_{0}^{T} \langle v_{n}'(t), v_{n}(t) - u_{n}(t) \rangle dt - \int_{0}^{T} \langle f_{n}(t), v_{n}(t) - u_{n}(t) \rangle dt$$
  

$$+ \frac{1}{2} \|v_{n}(0) - u_{0,n}\|_{H}^{2} + \int_{0}^{T} \langle A(t)u_{n}(t), v_{n}(t) - u(t) \rangle dt,$$

for all  $v \in W((0,T), V, V') \cap L^2((0,T), K)$ . In particular, taking  $v_n = u_{\epsilon,n}$ , we get

$$\int_{0}^{T} \langle A(t)u_{n}(t), u_{n}(t) - u(t) \rangle dt$$
  

$$\leq \int_{0}^{T} \langle u_{\epsilon,n}'(t), u_{\epsilon,n}(t) - u_{n}(t) \rangle dt - \int_{0}^{T} \langle f_{n}(t), u_{\epsilon,n}(t) - u_{n}(t) \rangle dt$$
  

$$+ \frac{1}{2} \|u_{\epsilon,n}(0) - u_{0,n}\|_{H}^{2} + \int_{0}^{T} \langle A(t)u_{n}(t), u_{\epsilon,n}(t) - u(t) \rangle dt.$$

Letting  $n \to \infty$ , we obtain

$$\begin{split} \limsup_{n \to \infty} \int_0^T \langle A(t) u_n(t), u_n(t) - u(t) \rangle \, dt \\ &\leq \int_0^T \langle u_\epsilon'(t), u_\epsilon(t) - u(t) \rangle \, dt - \int_0^T \langle f(t), u_\epsilon(t) - u(t) \rangle \, dt \\ &\quad + \frac{1}{2} \| u_\epsilon(0) - u_0 \|_H^2 + \int_0^T \langle A(t) u(t), u_\epsilon(t) - u(t) \rangle \, dt \\ &= -\epsilon \int_0^T \| u_\epsilon'(t) \|_H^2 \, dt - \int_0^T \langle f(t), u_\epsilon(t) - u(t) \rangle \, dt \\ &\quad + \int_0^T \langle A(t) u(t), u_\epsilon(t) - u(t) \rangle \, dt \\ &\leq -\int_0^T \langle f(t), u_\epsilon(t) - u(t) \rangle \, dt + \int_0^T \langle A(t) u(t), u_\epsilon(t) - u(t) \rangle \, dt. \end{split}$$

This is true for any  $\epsilon > 0$ . Hence, by letting  $\epsilon \to 0$ , we conclude that

$$\limsup_{n \to \infty} \int_0^T \langle A(t)u_n(t), u_n(t) - u(t) \rangle \, dt \le 0.$$

On the other hand, the weak convergence of  $u_n$  in Theorem 1.6.4 implies

$$\lim_{n \to 0} \int_0^T \langle A(t)u(t), u_n(t) - u(t) \rangle \, dt = 0.$$

Thus, we have

$$\limsup_{n \to \infty} \int_0^T \langle A(t)u_n(t) - A(t)u(t), u_n(t) - u(t) \rangle dt \le 0.$$
(1.67)

### CHAPTER 1. LINEAR NON-AUTONOMOUS PROBLEMS

It follows from the coercivity of A(t), (1.66) and (1.67) that

$$\alpha \|u_n - u\|_{L^2((0,T),V)}^2 \le \int_0^T \langle A(t)u_n(t) - A(t)u(t), u_n(t) - u(t) \rangle \ dt \to 0,$$

as  $n \to \infty$ . Therefore,  $u_n$  converges to u in  $L^2((0,T),V)$  strongly.

50

# Chapter 2

# Domain perturbation for semilinear parabolic equations

The main purpose of this chapter is to present some preliminary results on domain perturbation for semilinear parabolic equations which are required for the study of invariant manifolds in Chapter 3. Although we will only consider the dynamics of autonomous semilinear equations in  $L^2(\mathbb{R}^N)$  in Chapter 3, we give an abstract domain perturbation result for non-autonomous semilinear evolution equations. Most of the material presented in this chapter is based on [38]. However, we point out that the convergence result in [38, Theorem 4.6] only requires a weaker assumption on convergence of the nonlinear terms. We show that the same result holds if we replace the local uniform convergence of the nonlinear terms by pointwise convergence (see Theorem 2.3.5 below). Using this improved convergence result, we consider domain perturbation for both autonomous and non-autonomous semilinear parabolic equations under pointwise convergence of the nonlinear terms in Section 2.4. We note that in [35] and [38] the nonlinear terms are fixed for the perturbed problems. As in Chapter 1, we use Mosco convergence to establish the stability of solutions under domain perturbation. The convergence results are obtained for parabolic equations under Dirichlet or Neumann boundary conditions, but stronger assumptions are required for Neumann problems. For autonomous problems, we also include some known results on convergence of the spectrum and resolvent operators under domain perturbation.

For the sake of completeness and mathematical necessity, we include the proof of some background materials from [38].

### 2.1 Evolution Systems

We collect some preliminary results for evolution systems arising from (abstract) nonautonomous parabolic equations. Fix T > 0 and  $s \in [0, T)$ . We denote by

$$\Delta_T := \{(s,t) \in \mathbb{R}^2 : 0 \le s \le t \le T\}$$

and

$$\dot{\Delta}_T := \{ (s, t) \in \mathbb{R}^2 : 0 \le s < t \le T \} \subset \Delta_T.$$

Consider the following homogeneous parabolic equation

$$\begin{cases}
\frac{\partial u}{\partial t} + \mathcal{A}(t)u = 0 & \text{in } \Omega \times (s, T] \\
\mathcal{B}(t)u = 0 & \text{on } \partial \Omega \times (s, T] \\
u(\cdot, s) = u_0 & \text{in } \Omega,
\end{cases}$$
(2.1)

where  $\mathcal{A}(t)$  and  $\mathcal{B}(t)$  are defined by (0.5) and (0.6), respectively. By taking the evolution triple as in (1.11), we may write (2.1) as an abstract equation of the form

$$\begin{cases} u'(t) + A(t)u(t) = 0 & \text{for } t \in (s, T] \\ u(s) = u_0, \end{cases}$$
(2.2)

where  $A(t) \in \mathscr{L}(V, V')$  is the operator induced by  $a(t, \cdot, \cdot)$  as in (1.10).

By Theorem 1.1.2, for every given  $u_0 \in H$ , there exists a unique (weak) solution  $u \in W((s,T), V, V')$  of the homogeneous non-autonomous parabolic equation.

**Definition 2.1.1** (Evolution systems). Denote the solution u of (2.2) by  $U(\cdot, s)u_0$ . Then  $U(\cdot, \cdot)$  defined on  $\Delta_T$  is called the *evolution system* corresponding to the family  $(A(t))_{t \in [0,T]}$ .

It is known that the operator  $U(\cdot, \cdot)$  is in  $\mathscr{L}(H)$  ([41, Section XVIII.3.4]). Since  $\mathcal{A}(\cdot)$ is uniform strongly elliptic, a standard heat kernel estimates ([36, Section 7]) shows that for Dirichlet problems,  $U(\cdot, \cdot) \in \mathscr{L}(L^p(\Omega), L^q(\Omega))$  satisfies the estimate

$$\|U(t,s)\|_{\mathscr{L}(L^{p}(\Omega),L^{q}(\Omega))} \leq C(t-s)^{-\frac{N}{2}\left(\frac{1}{p}-\frac{1}{q}\right)},$$
(2.3)

for all  $1 \le p \le q \le \infty$  where the constant C is independent of  $\Omega$ . If  $\Omega$  satisfies the interior cone condition or has the extension property, then the estimate (2.3) is valid for Neumann problems with C depending on the cone or the norm of the extension

operator, respectively. In that case, the boundary condition is considered in a weak sense as the outer unit normal may not exist.

Throughout this thesis, we write  $\mathscr{L}(E, F)$  for the space of bounded linear operators from a Banach space E into a Banach space F equipped with the operator norm. We simply write  $\mathscr{L}(E)$  for  $\mathscr{L}(E, E)$ . The notation  $\mathcal{L}_s(E, F)$  represents the space of bounded linear operators under the strong operator topology. More precisely,  $S_n \to S$ in  $\mathcal{L}_s(E, F)$  if  $S_n x \to S x$  in F for all  $x \in E$  as  $n \to \infty$ . Again, we simply denote by  $\mathcal{L}_s(E)$  for  $\mathcal{L}_s(E, E)$ .

As mentioned earlier, we will give the perturbation results for evolution equations. We make the following assumptions for evolution systems.

Assumption 2.1.2. The evolution system  $U(t,s), (t,s) \in \Delta_T$  on the Banach space E satisfies the following properties.

- (i)  $U(t,s) = U(t,\tau)U(\tau,s)$  for all  $0 \le s < \tau < t \le T$ ;
- (ii) There exists a Banach space F such that

$$U(\cdot, s) \in C((s, T], \mathcal{L}_s(E, F)) \cap C((s, T], \mathcal{L}_s(E)) \cap C((s, T], \mathcal{L}_s(F)),$$

for all  $s \in [0,T)$ ;

(iii) There exist C > 0 and  $\gamma \in [0, 1)$  such that

$$\|U(t,s)\|_{\mathscr{L}(E)} + \|U(t,s)\|_{\mathscr{L}(F)} + (t-s)^{\gamma}\|U(t,s)\|_{\mathscr{L}(E,F)} \le C, \qquad (2.4)$$

for all  $(t,s) \in \dot{\Delta}_T$ .

There are several problems for which the above assumptions are satisfied (see [38, Example 2.2, 2.3]). In particular, this includes the evolution systems for parabolic equations discussed above. To see this, we take  $E := L^p(\Omega)$  and  $F := L^q(\Omega)$  for 1 satisfying

$$\gamma := \frac{N}{2} \left( \frac{1}{p} - \frac{1}{q} \right) \in [0, 1).$$
(2.5)

Then by the estimate (2.3),  $U(\cdot, \cdot)$  satisfies Assumption 2.1.2.

## 2.2 Existence of solutions of semilinear evolution equations

In this section, we prove the existence and uniqueness of solutions of the following non-autonomous semilinear evolution equation

$$\begin{cases} u'(t) + A(t)u(t) = f(t, u(t)) & \text{ for } t \in (s, T] \\ u(s) = u_0, \end{cases}$$
(2.6)

in a Banach space E whose evolution system satisfies Assumption 2.1.2.

**Definition 2.2.1.** A function u is called a *mild solution* of (2.6) with an initial value  $u_0 \in F$  if u satisfies the variation of constants formula

$$u(t) = U(t,s)u_0 + \int_s^t U(t,\tau)f(\tau,u(\tau))d\tau,$$

for all  $t \in (s, T]$ , where  $U(t, s), 0 \le s \le t \le T$  is the evolution system corresponding to the family  $(A(t))_{t \in [0,T]}$ .

Before studying the existence of solutions of semilinear evolution equations, we give a basic property of solutions of the inhomogeneous linear equation

$$\begin{cases} u'(t) + A(t)u(t) = g(t) & \text{ for } t \in (s, T] \\ u(s) = u_0, \end{cases}$$
(2.7)

where  $g \in L^{\infty}((s,T],E)$  and  $u_0 \in F$ . Define

$$L_s(u_0,g)(t) := U(t,s)u_0 + \int_s^t U(t,\tau)g(\tau)d\tau.$$
 (2.8)

Then  $u = L_s(u_0, g)$  is called a mild solution of (2.7). The next proposition shows that  $L_s(u_0, g)$  belongs to the space

$$BC((s,T],F) := C((s,T],F) \cap L^{\infty}([s,T],F).$$

**Proposition 2.2.2** ([38, Proposition 2.4]). For every  $s \in [0, T)$ , we have

$$L_s \in \mathscr{L}(F \times L^{\infty}((s,T],E), BC((s,T],F))).$$

*Proof.* See [38, Proposition 2.4] and also [40, Corollary 5.6].

It can easily be seen that u is a mild solution of (2.6) if u is a fixed point of the map  $G := G_{u_0}$  given by

$$G(u)(t) = U(t,s)u_0 + \int_s^t U(t,\tau)f(\tau,u(\tau))d\tau$$

in the space  $L^{\infty}([s,t],F)$ .

To get the existence of solutions of semilinear equations, the following Lipschitz assumption on the nonlinearity is usually imposed.

Assumption 2.2.3. We assume that  $f \in C([0,T] \times F, E)$  and that for every R > 0, there exists  $k_R > 0$  such that

$$||f(t,u) - f(t,v)||_E \le k_R ||u - v||_F,$$

for all  $u, v \in F$  with  $||u||_F$ ,  $||v||_F \leq R$  and all  $t \in [0, T]$ .

By the assumptions on f, we have that  $f(\cdot, u(\cdot)) \in L^{\infty}((s, T], E)$  for all  $u \in L^{\infty}([s, t], F)$ . Hence, by Proposition 2.2.2, mild solutions of (2.6) are always in the space BC((s, T], F).

We next determine an invariant subset of the space BC((s,T],F) such that G is a contraction.

**Proposition 2.2.4** ([38, Proposition 3.3]). For all  $\rho > 0$ , there exist h > 0 and R > 0 such that if  $||u_0||_F \le \rho$  and  $(t_0, s) \in \dot{\Delta}_T$  with  $0 < t_0 - s \le h$ , then

$$G_{u_0}: M_R(t_0, s) \to M_R(t_0, s),$$

where

$$M_R(t_0, s) := \{ u \in BC((s, t_0], F) : ||u(\tau)||_F \le R \text{ for all } \tau \in (s, t_0] \}.$$

Indeed, R can be chosen by  $R := C\rho + 1$  where C is the constant from (2.4). Moreover, h depends only on  $\gamma$  and upper bounds for  $C, k_R$  and  $M := \sup\{\|f(\tau, 0)\|_E : \tau \in [0, T]\}$ .

*Proof.* Suppose that  $u_0$  is in F with  $||u_0||_F \leq \rho$ . By the assumptions on  $U(\cdot, \cdot)$ , we have

$$\begin{aligned} \|G(u)(t)\|_{F} &\leq \|U(t,s)\|_{\mathscr{L}(F)} \|u_{0}\|_{F} + \int_{s}^{t} \|U(t,\tau)\|_{\mathscr{L}(E,F)} \|f(\tau,u(\tau))\|_{E} d\tau \\ &\leq C\rho + C \sup_{\tau \in (s,T]} \|f(\tau,u(\tau))\|_{E} \int_{s}^{t} (t-\tau)^{-\gamma} d\tau \\ &= C\rho + C \frac{(t-s)^{1-\gamma}}{1-\gamma} \sup_{\tau \in (s,t_{0}]} \|f(\tau,u(\tau))\|_{E}, \end{aligned}$$
(2.9)

for all  $u \in BC((s, t_0], F)$  and for all  $t \in (s, t_0]$ . In particular, for  $u \in M_R(t_0, s)$ , we have from Assumption 2.2.3 that

$$\|f(\tau, u(\tau))\|_{E} \leq \|f(\tau, u(\tau)) - f(\tau, 0)\|_{E} + \|f(\tau, 0)\|_{E}$$
$$\leq Rk_{R} + \sup_{\tau \in (s, t_{0}]} \|f(\tau, 0)\|_{E}$$
$$\leq Rk_{R} + M,$$

for all  $t \in (s, t_0]$  where we set  $M := \sup\{||F(\tau, 0)||_E : \tau \in [0, T]\}$ . It follows from (2.9) that

$$||G(u)(t)||_F \le C\rho + C\frac{(t-s)^{1-\gamma}}{1-\gamma}(Rk_R + M)$$

for all  $u \in M_R(t_0, s)$  and  $t \in (s, t_0)$ . By setting  $R := C\rho + 1$  and choosing h > 0 such that

$$\frac{C}{1-\gamma}(Rk_R+M)h^{1-\gamma} \le 1,$$

we conclude that if  $(t_0, s) \in \dot{\Delta}_T$  with  $t_0 - s \le h$ , then

$$||G(u)||_{L^{\infty}((s,t_0],F)} \le C\rho + 1 = R,$$

for all  $u \in M_R(t_0, s)$ .

Usually, we use the metric  $d_0$  on  $BC((s, t_0], F)$  induced by the norm, that is,

$$d_0(u,v) = \sup_{\tau \in (s,t_0]} \|u(\tau) - v(\tau)\|_F.$$
(2.10)

In domain perturbation problems, it is useful to consider a weaker metric. We consider a metric  $d_1$  induced by the Fréchet space  $L^{\infty}_{loc}((s, t_0], F)$  or a metric inducing the topology of uniform convergence on compact subsets of  $(s, t_0]$ . Let  $\varepsilon_j := 2^{-j}$  and  $h := t_0 - s$ . For  $j \in \mathbb{N}$ , we define the seminorms

$$q_j(u) := \sup_{\tau \in [s + \varepsilon_j h, t_0]} \|u(\tau)\|_F.$$

It is well known that for  $\beta_j := \varepsilon_j^{1-\gamma}$ ,

$$d_1(u,v) := \sum_{j=1}^{\infty} \beta_j \frac{q_j(u-v)}{1+q_j(u-v)}$$
(2.11)

defines a metric on  $L^{\infty}_{\text{loc}}((s, t_0], F)$  (see [71, Section 1.44]). Note that the set  $M_R(t_0, s)$  is a complete metric space with respect to the metrics  $d_0$  and  $d_1$  (see [38, Remark 3.4]).

**Proposition 2.2.5** ([38, Proposition 3.5]). For every  $R > 0, (t_0, s) \in \Delta_T$  and  $u_0 \in F$ , the map

$$G_{u_0}: M_R(t_0, s) \to BC((s, t_0], F)$$

is Lipschitz with respect to  $d_0$  and  $d_1$ . Indeed,

$$d_0(G(u), G(v)) \le \frac{Ck_R}{1 - \gamma} (t_0 - s)^{1 - \gamma} d_0(u, v)$$
  
$$d_1(G(u), G(v)) \le \frac{2^{1 - \gamma} (1 + 2R)}{(2^{1 - \gamma} - 1)} \frac{Ck_R}{1 - \gamma} (t_0 - s)^{1 - \gamma} d_1(u, v),$$

for all  $u, v \in M_R(t_0, s)$  where C is the constant from (2.4).

*Proof.* We give a proof here for the metric  $d_0$  only. By the definition of  $G := G_{u_0}$ , we have

$$\|G(u)(t) - G(v)(t)\|_F \le \int_s^t \|U(t,\tau)\|_{\mathscr{L}(E,F)} \|f(\tau,u(\tau)) - f(\tau,v(\tau))\|_E d\tau,$$

for all  $t \in (s, t_0]$ . Since  $u, v \in M_R(t_0, s)$ , Assumption 2.2.3 and (2.4) imply

$$\begin{aligned} \|G(u)(t) - G(v)(t)\|_F &\leq Ck_R \int_s^t (t-\tau)^{-\gamma} d\tau \sup_{\tau \in (s,t_0]} \|u(\tau) - v(\tau)\|_E \\ &\leq \frac{Ck_R}{1-\gamma} (t_0 - s)^{1-\gamma} d_0(u,v), \end{aligned}$$

for all  $t \in (s, t_0]$  and for all  $u, v \in M_R(t_0, s)$ . Hence, the proposition follows for the metric  $d_0$ . The proof for the metric  $d_1$  is a bit more complicated. We refer to [38] for full details.

We can now prove the (local) existence and uniqueness of solutions of (2.6).

**Theorem 2.2.6** ([38, Theorem 3.6, Corollary 3.7]). Suppose that Assumption 2.1.2 and Assumption 2.2.3 are satisfied. Given  $\rho > 0$ , there exists h > 0 such that the equation (2.6) has a unique mild solution  $u \in BC((s, \min\{s + h, T\}], F)$  for every  $u_0 \in F$  with  $\|u_0\|_F \leq \rho$ . Moreover, h can be chosen depending only on  $\gamma$  and upper bounds for  $C, k_{C\rho+1}$  and  $\sup\{\|f(\tau, 0)\|_E : \tau \in [0, T]\}$ .

Proof. We can choose R and h so that  $M_R(t_0, s)$  is invariant under  $G_{u_0}$  for all  $u_0 \in F$ with  $||u_0||_F \leq \rho$ . (Proposition 2.2.4). In addition, by Proposition 2.2.5, we can make sure that  $G_{u_0}$  is a contraction map on  $M_R(t_0, s)$  with Lipschitz constant less than one by possibly decreasing h (this is possible for both metrics  $d_0$  and  $d_1$ ). It follows from the contraction mapping theorem that there exists a unique fixed point  $u \in M_R(t_0, s)$ of  $G_{u_0}$ . Since mild solutions of (2.6) correspond to fixed points of  $G_{u_0}$ , the assertion follows. **Theorem 2.2.7** ([38, Theorem 3.8]). Suppose that Assumption 2.1.2 and Assumption 2.2.3 are satisfied. For every  $u_0 \in F$  and  $s \in [0,T)$ , there exists  $t^+(s,u_0) \in (s,T]$  such that the equation (2.6) has a unique mild solution  $u \in BC((s,t^+(s,u_0)),F)$  and u cannot be extended further from  $t^+(s,u_0)$ .

*Proof.* By extending a local solution in Theorem 2.2.6 using a standard application of Zorn's lemma, we obtain the maximal solution  $u \in BC((s, t^+(s, u_0)), F)$ .

We call  $t^+(s, u_0)$  the maximal existence time or the positive escape time of the solution.

## 2.3 Convergence of solutions of semilinear evolution equations

In this section, we study perturbations of semilinear evolution equation (2.6) in a fixed Banach space E of the form

$$\begin{cases} u'(t) + A_n(t)u(t) = f_n(t, u(t)) & \text{ for } t \in (s, T] \\ u(s) = u_{0,n}. \end{cases}$$
(2.12)

We make the following assumptions on the corresponding evolution systems  $U_n(s,t)$  on E.

Assumption 2.3.1. Suppose that the evolution system  $U_n(t,s)$  for  $(t,s) \in \dot{\Delta}$  satisfies the conditions below:

- (i)  $U_n(t,s) = U_n(t,\tau)U_n(\tau,s)$  for all  $0 \le s < \tau < t \le T$  and for all  $n \in \mathbb{N}$ ;
- (ii) There exists a Banach space F such that

$$U_n(\cdot, s) \in C((s, T], \mathcal{L}_s(E, F)),$$

for all  $s \in [0, T)$  and for all  $n \in \mathbb{N}$ ;

(iii) There exist C > 0 and  $\gamma \in [0, 1)$  such that

$$||U_n(t,s)||_{\mathscr{L}(E)} + ||U_n(t,s)||_{\mathscr{L}(F)} + (t-s)^{\gamma} ||U_n(t,s)||_{\mathscr{L}(E,F)} \le C,$$
(2.13)

for all  $(t,s) \in \dot{\Delta}_T$  and for all  $n \in \mathbb{N}$ ;

(iv)  $U_n(t,s)$  converges to U(t,s) in  $\mathcal{L}_s(E,F)$  as  $n \to \infty$  for all  $(t,s) \in \dot{\Delta}_T$  with the limit  $U(\cdot,s) \in C((s,T], \mathcal{L}_s(E,F))$ .

#### CHAPTER 2. SEMILINEAR PROBLEMS

Furthermore, we impose the following conditions on the nonlinearity  $f_n$ .

Assumption 2.3.2. We assume that  $f_n \in C([0,T] \times F, E)$  and that for every R > 0, there exists  $k_R > 0$  such that

$$||f_n(t,u) - f_n(t,v)||_E \le k_R ||u - v||_F,$$

for all  $u, v \in F$  with  $||u||_F, ||v||_F \leq R$  and all  $t \in [0, T]$  and  $n \in \mathbb{N}$ . In addition, we assume that

$$\sup\{\|f_n(t,0)\| : t \in [0,T], n \in \mathbb{N}\} < \infty.$$

We now state a convergent result under perturbation.

**Theorem 2.3.3** ([38, Theorem 4.4]). Suppose that Assumptions 2.3.1 and 2.3.2 are satisfied. Let  $u_0, u_{0,n}$  be in F with  $||u_0||_F, ||u_{0,n}||_F \leq \rho$  for some  $\rho > 0$ . For  $n \in \mathbb{N}$ , let  $G_n, G: BC((s,T],F) \rightarrow BC((s,T],F)$  be the maps defined by

$$G_n(u)(t) := U_n(t,s)u_{0,n} + \int_s^t U_n(t,\tau)f_n(\tau,u(\tau))d\tau$$
$$G(u)(t) := U(t,s)u_0 + \int_s^t U(t,\tau)f(\tau,u(\tau))d\tau,$$

for all  $u \in BC((s,T],F)$ . Suppose that  $G_n(u) \to G(u)$  in BC((s,T],F) for all  $u \in BC((s,T],F)$  with respect to the metric  $d_0$  (or  $d_1$ ). Then there exist h > 0 and R > 0 such that (2.23) and (2.12) have unique mild solutions  $u_n, u \in BC((s,\min\{s+h,T\}], B_F(0,R))$  for all  $s \in [0,T)$ . Moreover,  $u_n \to u$  in  $BC((s,\min\{s+h,T\}],F)$  with respect to  $d_0$  (or  $d_1$ ).

Proof. By the assumptions on  $U_n$  and  $f_n$ , we have from Theorem 2.2.5 that there exist h > 0 and R > 0 such that  $G_n, G$  are contractions on  $M_R(t, s)$  for all  $(t, s) \in \dot{\Delta}$  with  $t-s \leq h$  for all  $n \in \mathbb{N}$ . Moreover,  $G_n$  are uniform contractions with a Lipschitz constant independent of  $n \in \mathbb{N}$ . As the fixed points  $u_n$  of  $G_n$  and u of G are the unique mild solutions of (2.23) and (2.12) respectively, it remains to show that the corresponding fixed points converge. Since  $G_n(u) \to G(u)$  in  $M_R(t, s)$  for all  $u \in M_R(t, s)$  with respect to  $d_0$  (or  $d_1$ ), a parameter dependent contraction mapping principle (see [52, Section 1.2.6]) implies that the corresponding fixed points converge with respect to  $d_0$  (or  $d_1$ ). Thus, the statement of the theorem follows.

**Theorem 2.3.4** ([38, Theorem 4.5]). Suppose that the assumptions in Theorem 2.3.3 are satisfied. Let  $t_n^+(s, u_{0,n})$  be the positive escape time for (2.12). Then

$$t^+(s, u_0) \le \liminf_{n \to \infty} t_n^+(s, u_{0,n}).$$
 (2.14)

Moreover, if  $G_n(u) \to G(u)$  in BC((s,T],F) for all  $u \in BC((s,T],F)$  with respect to the metric  $d_0$  (or  $d_1$ ), then  $u_n \to u$  in  $BC(s,t_0],F$ ) for all  $t_0 \in (s,t^+(s,u_0))$  with respect to the metric  $d_0$  (or  $d_1$ ).

*Proof.* See [38, Theorem 4.5].

Our contribution in this chapter is that we weaken a local uniform convergence assumption of the inhomogeneous terms in [38, Theorem 4.6]:

$$f_n(\tau, v) \to f(\tau, v)$$
 in E uniformly with respect to  $(\tau, v) \in [0, T] \times B_F(0, R)$   
for all  $R > 0$ 

by using only pointwise convergence. We obtain the same convergence result as follow.

**Theorem 2.3.5** (c.f. [38, Theorem 4.6]). Let Assumptions 2.3.1 and 2.3.2 be satisfied. Assume that

$$U_n(\cdot, s)u_{0n} \to U(\cdot, s)u_0 \quad in \ BC((s, T], F)$$

$$(2.15)$$

with respect to  $d_0$  (or  $d_1$ ) and

$$U_n(\cdot, \cdot) \to U(\cdot, \cdot) \quad in \ C(\dot{\Delta}_T, \mathcal{L}_s(E, F)).$$
 (2.16)

Finally, assume that  $f_n(\tau, v) \to f(\tau, v)$  in E pointwise on  $(\tau, v) \in [0, T] \times F$ . Then

$$t^+(s, u_0) \le \liminf_{n \to \infty} t_n^+(s, u_{0,n}).$$

Moreover, for every  $t < t^+(s, u_0)$ , we have  $u_n \to u$  in BC((s, t], F) with respect to the metric  $d_0$  (or  $d_1$ ).

*Proof.* We only need to verify the assumptions of Theorem 2.3.3 that  $G_n(u) \to G(u)$ in BC((s,T],F) for all  $u \in BC((s,T],F)$  with respect to the metric  $d_0$  (or  $d_1$ ). The statement of the theorem then follows from Theorem 2.3.4.

Fix  $s \in [0,T)$  and  $u \in BC((s,T],F)$ . By our assumptions, we have  $U_n(\cdot,s)u_{0n} \to U(\cdot,s)u_0$  in BC((s,T],F) with respect to  $d_0$  (or  $d_1$ ). Hence, we only need to show that

$$H_n(u)(t) := \int_s^t U_n(t,\tau) f_n(\tau, u(\tau)) d\tau \to H(u)(t) := \int_s^t U(t,\tau) f(\tau, u(\tau)) d\tau$$

in F uniformly with respect to  $t \in [s, T]$ , that is with respect to the metric  $d_0$  and hence  $d_1$ . If we set  $V_n(t, \tau) := U_n(t, \tau) - U(t, \tau)$ ,  $g_n(\tau) := f_n(\tau, u(\tau))$  and  $g(\tau) := f(\tau, u(\tau))$ , then

$$H_n(u)(t) - H(u)(t) = \int_s^t U_n(t,\tau)(g_n(\tau) - g(\tau))d\tau + \int_s^t V_n(t,\tau)g(\tau)d\tau.$$

By the heat kernel estimates on  $U_n$  and U, we know that  $||U_n(t,\tau)||_{\mathcal{L}(E,F)} \leq C(t-\tau)^{-\gamma}$ and  $||V_n(t,\tau)||_{\mathcal{L}(E,F)} \leq 2C(t-\tau)^{-\gamma}$  for all  $(t,\tau) \in \dot{\Delta}_T$ . Since u is a bounded function, we conclude from the assumptions on  $f_n$  and f that  $g_n(\tau) \to g(\tau)$  in E pointwise on  $\tau \in (s,T]$ . Moreover, for every  $\tau \in (s,T]$ , we have

$$\begin{split} \|g_n(\tau)\|_E &\leq \|f_n(\tau, u(\tau)) - f_n(\tau, 0)\|_E + \|f_n(\tau, 0)\|_E \\ &\leq k_R \|u(\tau) - 0\|_F + \|f_n(\tau, 0)\|_E \\ &\leq k_R \sup_{\tau \in (s,T]} \|u(\tau)\|_F + \sup\{\|f_n(\tau, 0)\|_E : \tau \in [0, T], n \in \mathbb{N}\} \end{split}$$

If we set  $\|g\|_{\infty} := \sup_{\tau \in (s,T]} \|g(\tau)\|_E$  and  $\|g_n\|_{\infty} := \sup_{\tau \in (s,T]} \|g_n(\tau)\|_E$ , then it follows from Assumption 4.3 that  $\sup_{n \in \mathbb{N}} \|g_n\|_{\infty} < \infty$ . We first show that

$$\int_{s}^{t} U_{n}(t,\tau)(g_{n}(\tau) - g(\tau))d\tau \to 0$$

in F uniformly with respect to  $t \in [s, T]$ . To do so, it suffices to show that for every sequence  $(t_n)_{n \in \mathbb{N}}$  in [s, T], we have that

$$\left\|\int_{s}^{t_{n}} U_{n}(t_{n},\tau)(g_{n}(\tau)-g(\tau))d\tau\right\|_{F} \to 0$$

as  $n \to \infty$ . Indeed, we only need to show that for every sequence  $(t_n)_{n \in \mathbb{N}}$  in [s, T], if  $(t_{n_k})$  is an arbitrary subsequence, then we can find a further subsequence (denoted again by  $t_{n_k}$ ) such that

$$\left\| \int_{s}^{t_{n_{k}}} U_{n_{k}}(t_{n_{k}},\tau)(g_{n_{k}}(\tau) - g(\tau))d\tau \right\|_{F} \to 0$$
(2.17)

as  $k \to \infty$ . Let  $(t_n)_{n \in \mathbb{N}}$  be a sequence in [s, T] and  $(t_{n_k})_{k \in \mathbb{N}}$  be an arbitrary subsequence. Since  $t_{n_k}$  is a bounded sequence and [s, T] is compact, there exists a convergent subsequence (denoted again by  $t_{n_k}$ ) such that  $t_{n_k} \to t_0$  as  $k \to \infty$  for some  $t_0 \in [s, T]$ . If  $t_0 = s$ , then

$$\begin{aligned} \left\| \int_{s}^{t_{n_{k}}} U_{n_{k}}(t_{n_{k}},\tau)(g_{n_{k}}(\tau)-g(\tau))d\tau \right\|_{F} \\ &\leq \int_{t_{0}}^{t_{n_{k}}} C(t_{n_{k}}-\tau)^{-\gamma}d\tau \left( \sup_{n\in\mathbb{N}} \|g_{n}\|_{\infty} + \|g\|_{\infty} \right) \\ &= C \frac{(t_{n_{k}}-t_{0})^{1-\gamma}}{1-\gamma} \left( \sup_{n\in\mathbb{N}} \|g_{n}\|_{\infty} + \|g\|_{\infty} \right) \\ &\to 0, \end{aligned}$$

as  $k \to \infty$ . Hence, we assume  $t_0 \in (s, T]$ . Fix  $\varepsilon > 0$  arbitrary and choose  $\eta \in (0, t_0 - s)$  such that

$$C\frac{(2\eta)^{1-\gamma}}{1-\gamma}\left(\sup_{n\in\mathbb{N}}\|g_n\|_{\infty}+\|g\|_{\infty}\right)<\frac{\varepsilon}{2}$$

By the convergence of  $t_{n_k}$ , we can find  $K_1 \in \mathbb{N}$  such that  $t_0 - \eta < t_{n_k} < t_0 + \eta$  for all  $k > K_1$ . We can write

$$\int_{s}^{t_{n_{k}}} U_{n_{k}}(t_{n_{k}},\tau)(g_{n_{k}}(\tau)-g(\tau))d\tau = \int_{s}^{t_{0}-\eta} U_{n_{k}}(t_{n_{k}},\tau)(g_{n_{k}}(\tau)-g(\tau))d\tau + \int_{t_{0}-\eta}^{t_{n_{k}}} U_{n_{k}}(t_{n_{k}},\tau)(g_{n_{k}}(\tau)-g(\tau))d\tau,$$

for all  $k > K_1$ . By the choice of  $\eta$ , we have

$$\begin{aligned} \left\| \int_{t_0-\eta}^{t_{n_k}} U_{n_k}(t_{n_k},\tau) (g_{n_k}(\tau) - g(\tau)) d\tau \right\|_F \\ &\leq \int_{t_0-\eta}^{t_{n_k}} C(t_{n_k}-\tau)^{-\gamma} d\tau \left( \sup_{n\in\mathbb{N}} \|g_n\|_{\infty} + \|g\|_{\infty} \right) \\ &= C \frac{(t_{n_k}-(t_0-\eta))^{1-\gamma}}{1-\gamma} \left( \sup_{n\in\mathbb{N}} \|g_n\|_{\infty} + \|g\|_{\infty} \right) \\ &\leq C \frac{(2\eta)^{1-\gamma}}{1-\gamma} \left( \sup_{n\in\mathbb{N}} \|g_n\|_{\infty} + \|g\|_{\infty} \right) \\ &< \frac{\varepsilon}{2}, \end{aligned}$$

$$(2.18)$$

for all  $k > K_1$ . In addition, for all  $\tau \in (s, t_0 - \eta)$ , we have

$$\begin{aligned} \|U_{n_k}(t_{n_k},\tau)(g_{n_k}(\tau) - g(\tau))\|_F &\leq C(t_{n_k} - \tau)^{-\gamma} \|g_{n_k}(\tau) - g(\tau)\|_E \\ &\leq C(t_{n_k} - (t_0 - \eta))^{-\gamma} \|g_{n_k}(\tau) - g(\tau)\|_E \end{aligned}$$
(2.19)

for all  $k > K_1$ . By the pointwise convergence of  $g_n$  and the convergence of  $t_{n_k}$ , we get

$$C(t_{n_k} - (t_0 - \eta))^{-\gamma} \|g_{n_k}(\tau) - g(\tau)\|_E \to C\eta^{-\gamma}(0) = 0,$$

for all  $\tau \in (s, t_0 - \eta)$  as  $k \to \infty$ . Hence,  $U_{n_k}(t_{n_k}, \tau)(g_{n_k}(\tau) - g(\tau)) \to 0$  in F pointwise on  $\tau \in (s, t_0 - \eta)$ . We can also estimate (2.19) for  $\tau \in (s, t_0 - \eta)$  by

$$\begin{aligned} \|U_{n_k}(t_{n_k},\tau)(g_{n_k}(\tau) - g(\tau))\|_F &\leq C(t_{n_k} - \tau)^{-\gamma} \|g_{n_k}(\tau) - g(\tau)\|_E \\ &\leq C((t_0 - \eta) - \tau)^{-\gamma} \left( \sup_{n \in \mathbb{N}} \|g_n\|_{\infty} + \|g\|_{\infty} \right), \end{aligned}$$

for all  $k > K_1$ . Since  $\int_s^{t_0-\eta} C((t_0-\eta)-\tau)^{-\gamma} (\sup_{n\in\mathbb{N}} \|g_n\|_{\infty} + \|g\|_{\infty}) d\tau < \infty$ , we conclude from the Dominated Convergence Theorem that

$$\int_{s}^{t_{0}-\eta} U_{n_{k}}(t_{n_{k}},\tau)(g_{n_{k}}(\tau)-g(\tau))d\tau \to 0$$

in F as  $k \to \infty$ . Thus, we can find  $K_2 > K_1$  in  $\mathbb{N}$  such that

$$\left\| \int_{s}^{t_{0}-\eta} U_{n_{k}}(t_{n_{k}},\tau)(g_{n_{k}}(\tau)-g(\tau))d\tau \right\|_{F} < \frac{\varepsilon}{2},$$
(2.20)

#### CHAPTER 2. SEMILINEAR PROBLEMS

for all  $k > K_2$ . By (2.18) and (2.20), we get

$$\left\|\int_{s}^{t_{n_{k}}} U_{n_{k}}(t_{n_{k}},\tau)(g_{n_{k}}(\tau)-g(\tau))d\tau\right\|_{F} \leq \varepsilon,$$

for all  $k > K_2$ . As  $\varepsilon$  was arbitrary, we obtain (2.17) and thus the required uniform convergence holds.

It remains to show  $\int_s^t V_n(t,\tau)g(\tau)d\tau \to 0$  in F uniformly with respect to  $t \in [s,T]$ . We follow the argument in [38]. Fix  $\varepsilon > 0$  and choose  $\eta \in (0, T - s)$  such that

$$\frac{2C\|g\|_{\infty}}{1-\gamma}\eta^{1-\gamma} < \frac{\varepsilon}{2}.$$
(2.21)

For  $t \in [s + \eta, T]$ , we have

$$\int_{s}^{t} \|V_{n}(t,\tau)g(\tau)\|_{F} d\tau \leq \int_{s}^{t-\eta} \|V_{n}(t,\tau)g(\tau)\|_{F} d\tau + \int_{t-\eta}^{t} \|V_{n}(t,\tau)g(\tau)\|_{F} d\tau.$$

By the definition of  $V_n$ , the heat kernel estimates and (2.21), we have

$$\int_{t-\eta}^{t} \|V_n(t,\tau)g(\tau)\|_F d\tau \le 2C \|g\|_{\infty} \int_{t-\eta}^{t} (t-\tau)^{-\gamma} d\tau$$
$$\le \frac{2C \|g\|_{\infty}}{1-\gamma} \eta^{1-\gamma}$$
$$< \frac{\varepsilon}{2},$$

for all  $t \in [s + \eta, T]$ . Hence,

$$\int_{s}^{t} \|V_{n}(t,\tau)g(\tau)\|_{F}d\tau \leq \int_{s}^{t-\eta} \|V_{n}(t,\tau)g(\tau)\|_{F}d\tau + \frac{\varepsilon}{2},$$

for all  $t \in [s + \eta, T]$ . Since  $u \in BC((s, T], F)$ , we have  $g \in C([s + \eta, T], E)$ . It follows that the set  $\{g(\tau) : \tau \in [s + \eta, T]\}$  is compact in E. We also have that the set

$$K := \{(t,\tau) : s + \eta \le t \le T \text{ and } s \le \tau \le t\}$$

is a compact subset of  $\dot{\Delta}_T$ . By our assumptions,  $V_n(t,\tau) \to 0$  in  $\mathcal{L}_s(E,F)$  uniformly with respect to  $(t,\tau) \in K$ . Since strongly converging operators converge uniformly on compact sets, we can find  $N_0 \in \mathbb{N}$  such that

$$\|V_n(t,\tau)g(\tau)\|_F \le \frac{\varepsilon}{2T},$$

for all  $(t, \tau) \in K$  and for all  $n > N_0$ . It follows that

$$\int_{s}^{t} \|V_{n}(t,\tau)g(\tau)\|_{F} d\tau \leq \int_{s}^{t-\eta} \frac{\varepsilon}{2T} d\tau + \frac{\varepsilon}{2} \leq \frac{\varepsilon}{2} + \frac{\varepsilon}{2} = \varepsilon,$$

for all  $t \in [s + \eta, T]$  and for all  $n > N_0$ . By the choice of  $\eta$  in (2.21), we obtain from the heat kernel estimates that

$$\int_{s}^{t} \|V_{n}(t,\tau)g(\tau)\|_{F}d\tau \leq 2C\|g\|_{\infty} \int_{s}^{t} (t-\tau)^{-\gamma}d\tau$$
$$\leq \frac{2C\|g\|_{\infty}}{1-\gamma} \eta^{1-\gamma}$$
$$< \frac{\varepsilon}{2},$$

for all  $t \in [s, s + \eta]$  for all  $n \in \mathbb{N}$ . Therefore,

$$\int_{s}^{t} \|V_{n}(t,\tau)g(\tau)\|_{F} d\tau \leq \varepsilon,$$

for all  $t \in [s,T]$  and for all  $n > N_0$ . Since  $\varepsilon > 0$  was arbitrary, we conclude that

$$\int_{s}^{t} \|V_{n}(t,\tau)g(\tau)\|_{F} d\tau \to 0$$

uniformly with respect to  $t \in [s, T]$ . This completes the proof of the theorem.

#### 

# 2.4 Convergence of solutions of semilinear parabolic equations

In this section, we concentrate on semilinear parabolic equations. We apply the convergence results under the perturbation of evolution equations studied in Section 2.3 to deal with domain perturbation for semilinear parabolic equations.

#### 2.4.1 Non-autonomous semilinear parabolic equations

Let T > 0 and  $s \in [0, T)$ . As before, we assume that  $\Omega_n$  and  $\Omega$  satisfy Assumption 1.4.1. Consider the following non-autonomous parabolic equation

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}(t)u = g(x, t, u) & \text{in } \Omega \times (s, T] \\ \mathcal{B}(t)u = 0 & \text{on } \partial\Omega \times (s, T] \\ u(\cdot, s) = u_0 & \text{in } \Omega, \end{cases}$$
(2.22)

where  $\mathcal{A}(t)$  and  $\mathcal{B}(t)$  are defined by (0.5) and (0.6), respectively. By taking the evolution triple as in (1.11), we may write (2.22) as an abstract equation of the form

$$\begin{cases} u'(t) + A(t)u(t) = f(t, u(t)) & \text{ for } t \in (s, T] \\ u(s) = u_0, \end{cases}$$
(2.23)

where f(t, u)(x) := g(t, x, u(x)) is the substitution operator induced by g and  $A(t) \in \mathscr{L}(V, V')$  is the operator induced by  $a(t, \cdot, \cdot)$  defined in (1.10).

Since we have an improved convergence result in Theorem 2.3.5, we can consider a more general perturbation result. We study the following perturbation of (2.22)

$$\begin{cases}
\frac{\partial u}{\partial t} + \mathcal{A}_n(t)u = g_n(x, t, u) & \text{in } \Omega_n \times (s, T] \\
\mathcal{B}_n(t)u = 0 & \text{on } \partial \Omega_n \times (s, T] \\
u(\cdot, s) = u_{0,n} & \text{in } \Omega_n,
\end{cases}$$
(2.24)

where  $\mathcal{A}_n(t)$  and  $\mathcal{B}_n(t)$  are defined by (1.22) and (1.23), respectively. As usual, we write (2.24) in the abstract form as

$$\begin{cases} u'(t) + A_n(t)u(t) = f_n(t, u(t)) & \text{ for } t \in (s, T] \\ u(s) = u_{0,n}, \end{cases}$$
(2.25)

where  $f_n(t, u)(x) := g_n(t, x, u(x))$  is the substitution operator induced by  $g_n$  and  $A_n(t) \in \mathscr{L}(V_n, V'_n)$  is the operator induced by  $a_n(t, \cdot, \cdot)$  defined in (1.18).

Note that here we allow a perturbation  $g_n$  of the nonlinear term g, whereas in [35] the nonlinear term is fixed (see [35, Equation (6.17)]). Also, the application in domain perturbation in [38] is given for *autonomous* equations with a fixed nonlinear term (see [38, Equation (6.1)]). Possibly, the nonlinear terms are fixed under perturbation in [35] and [38] in order to have a local *uniform convergence* of the substitution operators without any further conditions on  $g_n$  and g. Here, we only need that  $g_n$  converges to g pointwise to get the pointwise convergence of the substitution operators.

We see from Theorem 1.1.2 that the evolution system  $U_n(\cdot, \cdot)$  corresponding to the family  $(A_n(t))_{t\in[0,T]}$  exists for all  $n \in \mathbb{N}$  and  $U_n(\cdot, \cdot) \in \mathscr{L}(L^2(\Omega_n))$ . Similarly, the evolution system  $U(\cdot, \cdot)$  corresponding to the family  $(A(t))_{t\in[0,T]}$  exists and  $U(\cdot, \cdot) \in$  $\mathscr{L}(L^2(\Omega))$ . Since the perturbation result in Section 2.3 is proved for evolution systems in a fixed Banach space E, we consider  $i_n U_n(\cdot, \cdot)r_n \in \mathscr{L}(L^2(D))$  and  $iU(\cdot, \cdot)r \in \mathscr{L}(L^2(D))$ where  $i_n$  and i are inclusion maps defined by (1.27) and (1.28) respectively, and  $r_n$  and r are restriction maps defined by (1.29) and (1.30), respectively. In particular, we take  $E := L^2(D)$  and  $F := L^2(D)$ . Clearly,  $iU(\cdot, \cdot)r$  satisfies Assumption 2.1.2 with  $\gamma = 0$  in (2.4). Moreover,  $i_n U_n(\cdot, \cdot)r_n$  satisfies Assumption 2.3.1 (i) – (iii) with  $\gamma = 0$  in (2.13). We show that Assumption 2.3.1 (iv) is satisfied in the following lemma.

**Lemma 2.4.1.** Suppose that  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco for the case of Dirichlet problems; or  $H^1(\Omega_n)$  converges to  $H^1(\Omega)$  in the sense of Mosco
with  $|\Omega_n| \to |\Omega|$  for the case of Neumann problems. Then the corresponding evolution systems satisfy

$$i_n U_n(t,s) r_n \to i U(t,s) r_n$$

in  $\mathcal{L}_s(L^2(D))$  for all  $(t,s) \in \dot{\Delta}_T$ .

Indeed,  $i_n U_n(\cdot, \cdot) r_n \to iU(\cdot, \cdot)r$  in  $\mathcal{L}_s(L^2(D))$  uniformly with respect to  $(t, s) \in \dot{\Delta}_T$ , that is

$$i_n U_n(\cdot, \cdot) r_n \to i U(\cdot, \cdot) r$$

in  $C(\dot{\Delta}_T, \mathcal{L}_s(L^2(D))).$ 

Proof. Let  $u_0 \in L^2(D)$  be arbitrary. By the definition of evolution systems  $u_n(t) := U_n(t,s)r_nu_0$  is the unique (weak) solution of (1.25) with  $f_n = 0 \in L^2((0,T), L^2(\Omega_n))$  satisfying the initial condition  $r_nu_0$  at time s. Similarly,  $u(t) := U(t,s)ru_0$  is the unique (weak) solution of (1.26) with  $f = 0 \in L^2((0,T), L^2(\Omega))$  satisfying the initial condition  $ru_0$  at time s. Clearly, we have  $i_n f_n \to if$  in  $L^2((0,T), L^2(D))$  strongly as  $n \to \infty$ . We consider first the case of Dirichlet problems. Since we have  $\operatorname{cap}(K \cap \Omega_n^c) \to 0$  for all  $K \subset \subset \Omega$  from Mosco convergence (Lemma 1.4.8), it is easy to see that  $ri_n r_n u_0 \to ru_0$  in  $L^2(\Omega)$  strongly. As the initial time that we start with is irrelevant, we conclude from Theorem 1.4.9 that  $i_n u_n(t) \to iu(t)$  in  $L^2(D)$  uniformly on (s,T]. Since the argument holds for all  $u_0 \in L^2(D)$  and  $s \in [0,T)$ , this implies  $i_n U_n(t,s)r_n \to iU(t,s)r$  in  $\mathcal{L}_s(L^2(D))$  for all  $(t,s) \in \dot{\Delta}_T$  and  $i_n U_n(\cdot,s)r_n \to iU(\cdot,s)r$  in  $C((s,T], \mathcal{L}_s(L^2(D)))$  for all  $s \in [0,T)$ . Again, as the initial time that we start with is irrelevant, we also have  $i_n U_n(t, \cdot)r_n \to iU(t, \cdot)r$  in  $C([0,t), \mathcal{L}_s(L^2(D)))$  for all  $t \in (0,T]$ . Hence, the assertion of the lemma for Dirichlet problems follows.

Similarly, we use the assumption that  $|\Omega_n| \to |\Omega|$  for Neumann problems to get  $i_n r_n u_0 \to i r u_0$  in  $L^2(D)$  strongly. We conclude from Theorem 1.4.16 that  $i_n u_n(t) \to i u(t)$  in  $L^2(D)$  uniformly on (s, T]. By a similar argument as in the case of Dirichlet problems, we can deduce the assertion of the lemma for Neumann problems.  $\Box$ 

We require the following conditions on the nonlinearity.

Assumption 2.4.2. We assume that  $g_n, g \in C(\mathbb{R}^N \times [0, T] \times \mathbb{R})$  satisfy the following conditions:

(i) There exist c > 0 and  $\psi \in L^2(\mathbb{R}^N)$  such that

$$|g_n(t, x, \xi)| \le \psi(x) + c|\xi|$$
  

$$|g(t, x, \xi)| \le \psi(x) + c|\xi|,$$
(2.26)

#### CHAPTER 2. SEMILINEAR PROBLEMS

for all  $x \in \mathbb{R}^N$ ,  $\xi \in \mathbb{R}$  and  $n \in \mathbb{N}$ .

(ii) There exists an essentially bounded function  $\phi$  such that

$$\begin{aligned} |g_n(x,t,\xi_1) - g_n(x,t,\xi_2)| &\le \phi(x,\zeta) |\xi_1 - \xi_2| \\ |g(x,t,\xi_1) - g(x,t,\xi_2)| &\le \phi(x,\zeta) |\xi_1 - \xi_2|, \end{aligned}$$
(2.27)

for all  $|\xi_1|, |\xi_2| \leq \zeta$  and all  $n \in \mathbb{N}$ .

(iii)  $g_n$  converges to g pointwise on  $\mathbb{R}^N \times [0, T] \times \mathbb{R}$ .

It is well known that the condition (2.26) is a necessary and sufficient condition for the substitution operators  $f_n(t, \cdot)$  defined by  $f_n(t, u)(x) := g_n(x, t, u(x))$  and  $f(t, \cdot)$ defined by f(t, u)(x) := g(x, t, u(x)) to be in  $C(L^2(\mathbb{R}^N), L^2(\mathbb{R}^N))$  ([3, Chapter 3]). Moreover, by condition (2.27), we have  $f_n(t, \cdot)$  and  $f(t, \cdot)$  are locally Lipschitz in uuniformly with respect to  $t \in [0, T]$  and the Lipschitz constants are independent of  $n \in \mathbb{N}$  (see [3, Theorem 3.10]). Therefore, the substitution operators  $f_n(t, \cdot)$  and  $f(t, \cdot)$ satisfy Assumption 2.3.2 and Assumption 2.2.3 respectively with the Lipschitz constant  $k_R$  independent of  $n \in \mathbb{N}$ .

Let  $u_{0,n} \in L^2(\Omega_n)$  and  $u_0 \in L^2(\Omega)$ . By Theorem 2.2.7, there exist  $t_n^+(s, u_{0,n})$ and  $t^+(s, u_0)$  such that the semilinear equation (2.25) has a unique maximal solution  $u_n$  in  $BC((s, t_n^+(s, u_{0,n})), L^2(\Omega_n))$  and (2.23) has a unique maximal solution uin  $BC((s, t^+(s, u_0)), L^2(\Omega))$ . In fact, the condition (2.26) implies that the substitution operators  $f_n$  and f are linearly bounded. More precisely, for any  $u_n \in L^2(\Omega_n)$  and  $u \in L^2(\Omega)$ , we have

$$||f_n(t, u_n)||_{L^2(\Omega_n)} \le ||\psi||_{L^2(\Omega_n)} + c||u_n||_{L^2(\Omega_n)}$$
$$||f(t, u)||_{L^2(\Omega)} \le ||\psi||_{L^2(\Omega)} + c||u||_{L^2(\Omega)}.$$

Hence, the solutions exist globally, that is  $t_n^+(s, u_{0,n}) = t^+(s, u_0) = \infty$  for all  $n \in \mathbb{N}$ .

We are now in the position to state the convergence of solutions of semilinear parabolic equations under domain perturbation.

**Theorem 2.4.3.** Let  $\Omega_n$  and  $\Omega$  be uniformly bounded domains in  $\mathbb{R}^N$  with  $\Omega_n, \Omega \subset D$ . Suppose that  $g_n$  and g satisfy Assumption 2.4.2. Suppose also that  $i_n u_{0,n}$  is uniformly bounded in  $L^2(D)$ .

(i) **Dirichlet problems:** Assume that  $ri_nu_{0,n} \rightharpoonup u_0$  in  $L^2(\Omega)$  weakly. If  $H^1_0(\Omega_n)$  converges to  $H^1_0(\Omega)$  in the sense of Mosco, then  $i_nu_n \rightarrow iu$  in  $C((s, t_0], L^2(D))$  for all  $t_0 \in (s, \infty)$ .

#### CHAPTER 2. SEMILINEAR PROBLEMS

In addition, if 
$$i_n u_{0,n} \rightarrow i u_0$$
 in  $L^2(D)$  strongly, then  $i_n u_n \rightarrow i u$  in  $C([s, t_0], L^2(D))$ .

(i) Neumann problems: Assume that  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$  strongly. If  $H^1(\Omega_n)$ converges to  $H^1(\Omega)$  in the sense of Mosco and  $|\Omega_n| \to |\Omega|$  as  $n \to \infty$ , then  $i_n u_n \to i u$  in  $C([s, t_0], L^2(D))$  for all  $t_0 \in (s, \infty)$ .

*Proof.* By Assumption 2.4.2 (iii), it is easy to conclude from the dominated convergence theorem that the substitution operator  $f_n(\tau, v)$  converges pointwise to  $f(\tau, v)$  in  $L^2(D)$ for all  $(\tau, v) \in [0, T] \times L^2(D)$ . For Dirichlet problems, we can see from the convergence of solutions under domain perturbation in Theorem 1.4.9 that

$$i_n U_n(\cdot, s) u_{0,n} \to i U(\cdot, s) u_0 \tag{2.28}$$

in  $C((s,T], L^2(D))$  if  $ri_n u_{0,n} \rightarrow u_0$  in  $L^2(\Omega)$  weakly. The convergence (2.28) above holds in  $C([s,T], L^2(D))$  if  $i_n u_{0,n} \rightarrow i u_0$  in  $L^2(D)$  strongly. Taking Lemma 2.4.1 into account, we have that all assumptions in Theorem 2.3.5 are satisfied. Hence, the assertion of the theorem follows from Theorem 2.3.5 with  $E = F = L^2(D)$ . The statement for Neumann problems follows in the same way by using Theorem 1.4.16 instead of Theorem 1.4.9.  $\Box$ 

Remark 2.4.4. We could study the convergence results in the  $L^p$  space as done in [35] and [38]. In the case of Dirichlet problems, the evolution system  $U_n(\cdot, \cdot)$  corresponding to the family  $(A_n(t))_{t\in[0,T]}$  exists for all  $n \in \mathbb{N}$ . Moreover,  $U_n(\cdot, \cdot) \in \mathscr{L}(L^p(\Omega_n), L^q(\Omega_n))$ satisfies the same estimate as in (2.3) uniformly with respect to  $n \in \mathbb{N}$ , that is

$$\|U_n(t,s)\|_{\mathscr{L}(L^p(\Omega_n),L^q(\Omega_n))} \le C(t-s)^{-\frac{N}{2}\left(\frac{1}{p}-\frac{1}{q}\right)},$$
(2.29)

for all  $1 \leq p \leq q \leq \infty$  where *C* is independent of  $n \in \mathbb{N}$ . For Neumann problems, the estimate (2.3) holds if  $\Omega$  satisfies the cone condition. However, the constant *C* depends on the cone (see [36]). Thus, in order to have the uniform estimate (2.29) with respect to  $n \in \mathbb{N}$ , a sufficient condition is that  $\Omega_n$  satisfies the cone condition for all  $n \in \mathbb{N}$ , where the same cone works for all domains  $\Omega_n$  and  $\Omega$ . Examples of domains satisfying this condition include the domains obtained by rotating a line segment inside an open set and the domains obtained by cutting a line into a unit disk as previously seen in Figure 1.3 and Figure 1.4, respectively. By taking  $E := L^p(D)$  and  $F := L^q(D)$  with  $1 satisfying (2.5), we see that Assumptions 2.3.1 (i) – (iii) are satisfied. To verify Assumption 2.3.1 (iv), (2.15) and (2.16), we need the <math>L^p$ -theory of domain perturbation for linear parabolic equations. This can be done in the same way as in [35, Section 4] by using the interpolation arguments and the convergence results in

 $L^2$ . Note also that we need to adjust conditions (2.26) and (2.27) in order to have the substitution operators acting on  $L^p(\mathbb{R}^N)$  into  $L^q(\mathbb{R}^N)$ . We refer to [3] for a necessary and sufficient condition. In this case, the maximal existence time  $t^+(s, u_0)$  may be strictly finite. An application of Theorem 2.3.5 gives us a lower semicontinuity of the maximal existence time, that is  $t^+(s, u_0) \leq \liminf_{n \to \infty} t_n^+(s, u_{0,n})$  and the convergence of solutions in  $C((s, t_0], L^q(D))$  for all  $t_0 \in (s, t^+(s, u_0))$ .

#### 2.4.2 Autonomous semilinear parabolic equations

Although autonomous problems can be regarded as a special case of non-autonomous problems with the equations (operators) do not depend on the time variable and the corresponding evolutions systems are just *semigroups*. We spend some time in this section to study domain perturbation for autonomous problems. In particular, we intend to discuss the convergence of semigroups and the spectrum of the corresponding elliptic operators. These results are discussed in [38, Section 6] for the heat equations. We give here domain perturbation results for parabolic equations in the abstract form using some results from [39].

We consider the following autonomous semilinear initial boundary value problem

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}u = g(x, u) & \text{in } \Omega \times (s, T] \\ \mathcal{B}u = 0 & \text{on } \partial\Omega \times (s, T] \\ u(\cdot, 0) = u_0 & \text{in } \Omega, \end{cases}$$
(2.30)

where  $g \in C(\mathbb{R}^N \times \mathbb{R})$ ,  $\mathcal{A}$  is an elliptic operator of the form

$$\mathcal{A}u := -\partial_i [a_{ij}(x)\partial_j u + a_i(x)u] + b_i(x)\partial_i u + c_0(x)u$$
(2.31)

and  $\mathcal{B}$  is one of the following boundary conditions

$$\mathcal{B}u := u \qquad \text{Dirichlet boundary condition} \\ \mathcal{B}u := [a_{ij}(x)\partial_j u + a_i(x)u] \nu_i \qquad \text{Neumann boundary condition.}$$
(2.32)

We assume that  $a_{ij}, a_i, b_i$  and  $c_0$  are functions in  $L^{\infty}(\mathbb{R}^N)$  and that there exists a constant  $\alpha > 0$  independent of  $x \in \Omega$  such that

$$a_{ij}(x)\xi_i\xi_j \ge \alpha_0|\xi|^2, \tag{2.33}$$

for all  $\xi \in \mathbb{R}^N$ . We consider the evolution triple  $V \stackrel{d}{\hookrightarrow} H \stackrel{d}{\hookrightarrow} V'$  chosen as in 1.11. Define a bilinear form  $a(\cdot, \cdot)$  associated with  $\mathcal{A}$  on V by

$$a(u,v) := \int_{\Omega} [a_{ij}(x)\partial_j u + a_i(x)u]\partial_i v + b_i(x)\partial_i uv + c_0(x)uvdx, \qquad (2.34)$$

for  $u, v \in V$ . Let

$$\lambda_{\mathcal{A}} := \|c_0^-\|_{\infty} + \frac{1}{2\alpha_0} \sum_{i=1}^N \|a_i + b_i\|_{\infty},$$

where  $c_0^- := \max(-c_0, 0)$  is the negative part of  $c_0$ . We set

$$\lambda_0 := \lambda_{\mathcal{A}} + \frac{\alpha_0}{2}.\tag{2.35}$$

It can be verified that  $a(\cdot, \cdot)$  is continuous and

$$a(u,u) + \lambda \|u\|_{H}^{2} \ge \frac{\alpha_{0}}{2} \|u\|_{V}^{2}, \qquad (2.36)$$

for all  $u \in V$  and for all  $\lambda \geq \lambda_0$  (see [39, Proposition 2.1.6]). By the Lax-Milgram theorem, there exists  $A_{\Omega} \in \mathscr{L}(V, V')$  such that

$$a(u,v) = \langle A_{\Omega}u, v \rangle, \qquad (2.37)$$

for all  $u, v \in V$ . We may consider  $A_{\Omega}$  as an operator on V' with the domain V. A standard argument shows that  $[\lambda_0, \infty) \subset \varrho(-A_{\Omega})$  (see [39, Theorem 2.2.2]). Let A be the maximal restriction of the operator  $A_{\Omega}$  on  $H = L^2(\Omega)$ . We have that  $[\lambda_0, \infty) \subset \varrho(-A_{\Omega}) \subset \varrho(-A)$ . It is well known that -A generates a strongly continuous analytic semigroup  $S(t), t \geq 0$  on  $L^2(\Omega)$  (see [41, Proposition 3, XVII §6]). Moreover, there exist C > 0 and  $\omega \in \mathbb{R}$  such that

$$\|S(t)\|_{\mathscr{L}(L^2(\Omega))} \le Ce^{\omega t},$$

for  $t \ge 0$ . We consider (2.30) as an abstract equation of the form

$$\begin{cases} \dot{u}(t) + Au(t) = f(u(t)) & t \in (0, \infty) \\ u(0) = u_0 \end{cases}$$
(2.38)

in  $L^2(\Omega)$ , where f(u)(x) := g(x, u(x)) is the substitution operator induced by g.

We study perturbation of (2.30) given by

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}_n u = g_n(x, u) & \text{in } \Omega_n \times (s, T] \\ \mathcal{B}_n u = 0 & \text{on } \partial \Omega_n \times (s, T] \\ u(\cdot, 0) = u_{0,n} & \text{in } \Omega_n, \end{cases}$$
(2.39)

where  $g_n \in C(\mathbb{R}^N \times \mathbb{R})$ . As usual, we assume that the perturbed domains  $\Omega_n$  and  $\Omega$  satisfy Assumption 1.4.1. The operator  $\mathcal{A}_n$  above is an elliptic operator of the form

$$\mathcal{A}_n u := -\partial_i [a_{ij,n}(x)\partial_j u + a_{i,n}(x)u] + b_{i,n}(x)\partial_i u + c_{0,n}(x)u, \qquad (2.40)$$

where  $a_{ij,n}, a_{i,n}, b_{i,n}$  and  $c_{0,n}$  are functions in  $L^{\infty}(\mathbb{R}^N)$  and  $a_{ij,n}$  satisfies the ellipticity condition as in (2.33) with the *uniform ellipticity constant* for all  $n \in \mathbb{N}$ . The operator  $\mathcal{B}_n$  is either Dirichlet boundary condition or Neumann boundary condition defined similarly to (2.32). We assume that the coefficients of  $\mathcal{A}_n$  converges to the corresponding coefficient of  $\mathcal{A}$ .

Assumption 2.4.5. For all i, j = 1, ..., N, we assume that  $\lim_{n\to\infty} a_{ij,n} = a_{ij}$ ,  $\lim_{n\to\infty} a_{i,n} = a_i, \lim_{n\to\infty} b_{i,n} = b_i$  and  $\lim_{n\to\infty} c_{0,n} = c_0$  in  $L^{\infty}(\mathbb{R}^N)$ .

As before, the evolution triple  $V_n \stackrel{d}{\hookrightarrow} H_n \stackrel{d}{\hookrightarrow} V'_n$  is chosen according to the boundary condition (either  $V_n = H_0^1(\Omega_n)$  or  $V_n = H^1(\Omega_n)$ ). In the abstract form, we can write (2.39) as

$$\begin{cases} \dot{u}(t) + A_n u(t) = f_n(u(t)) & t \in (s, T] \\ u(0) = u_{0,n} \end{cases}$$
(2.41)

in  $L^2(\Omega_n)$ , where  $f_n(u)(x) := g_n(x, u(x))$  is the substitution operator induced by  $g_n$ . In the above,  $A_n$  is the maximal restriction operator on  $H_n = L^2(\Omega_n)$  of the operator induced by the associated bilinear form. By Assumption 2.4.5 and the assumption that  $a_{ij,n}$  satisfies (2.33) uniformly with respect to  $n \in \mathbb{N}$ , we can choose  $\lambda_0 := \lambda_A + \alpha_0/2$ where

$$\lambda_{\mathcal{A}} := \max\left\{\sup_{n \in \mathbb{N}} \left\{ \|c_{0,n}^{-}\|_{\infty} + \frac{1}{2\alpha_{0}} \sum_{i=1}^{N} \|a_{i,n} + b_{i,n}\|_{\infty} \right\}, \left( \|c_{0}^{-}\|_{\infty} + \frac{1}{2\alpha_{0}} \sum_{i=1}^{N} \|a_{i} + b_{i}\|_{\infty} \right) \right\}$$
  
< \infty (2.42)

so that

$$a_n(u,u) + \lambda \|u\|_{H_n}^2 \ge \frac{\alpha_0}{2} \|u\|_{V_n}^2, \tag{2.43}$$

for all  $\lambda \geq \lambda_0$ , for all  $u \in V_n$  and for all  $n \in \mathbb{N}$ . It follows from [39, Theorem 2.2.2] that  $[\lambda_0, \infty) \subset \varrho(-A_n) \cap \varrho(-A)$  for all  $n \in \mathbb{N}$ . Moreover, the operator  $-A_n$  is the generator of a strongly continuous analytic semigroup  $S_n(t), t \geq 0$  on  $L^2(\Omega_n)$  with the uniform exponential estimate

$$||S_n(t)||_{\mathscr{L}(L^2(\Omega_n))} \le Ce^{\omega t}, \tag{2.44}$$

for all  $t \ge 0$  and for all  $n \in \mathbb{N}$ .

We impose Assumption 2.4.2 on the nonlinearities  $g_n$  and g in  $C(\mathbb{R}^N \times [0, T] \times \mathbb{R})$ (where we disregard the time variable). We note that this means the substitution operators induced by  $g_n$  and g are in  $C(L^2(\mathbb{R}^N), L^2(\mathbb{R}^N))$  and locally Lipschitz with the Lipschitz constant independent of  $n \in \mathbb{N}$ . Indeed, for every R > 0 there exists  $k_R$  independent of  $n \in \mathbb{N}$  such that

$$||f_n(u) - f_n(v)||_{L^2(\Omega_n)} \le k_R ||u - v||_{L^2(\Omega_n)},$$

for all  $u, v \in L^2(\Omega_n)$  with  $||u||_{L^2(\Omega_n)}, ||v||_{L^2(\Omega_n)} \leq R$  and for all  $n \in \mathbb{N}$ . In addition, we have

$$f_n(v) \to f(v)$$

in  $L^2(D)$  pointwise for all  $v \in L^2(D)$ .

By Theorem 2.2.7 (or simply using the existence and uniqueness results for autonomous problems in [66]), we have that for every  $u_0 \in L^2(\Omega)$  the equation (2.38) has a unique maximal solution  $u \in C((s, t^+(s, u_0)), L^2(\Omega))$ . Moreover, u can be represented by the variation of constants formula

$$u(t) = S(t)u_0 + \int_0^t S(t-\tau)f(u(\tau))d\tau.$$
 (2.45)

Similarly, for every  $u_{0,n} \in L^2(\Omega_n)$  the equation (2.41) has a unique maximal solution  $u_n \in C((s, t_n^+(s, u_{0,n})), L^2(\Omega_n))$  which can be represented by the variation of constant formula

$$u_n(t) = S_n(t)u_{0,n} + \int_0^t S_n(t-\tau)f_n(u_n(\tau))d\tau.$$
 (2.46)

As noted in the case of non-autonomous equations, the conditions on  $f_n$  and f imply that  $u_n$  and u are indeed global solutions, that is  $t_n^+(s, u_{0,n}) = t^+(s, u_0) = \infty$  for all  $n \in \mathbb{N}$ .

To deal with domain perturbation, we make use of the inclusion operators  $i_n$  and i, and the restriction operators  $r_n$  an r as in Section 2.4.1. We define a *degenerate* semigroup  $S_n(t)$  on  $L^2(D)$  for  $t \ge 0$  by

$$\mathcal{S}_n(t) := i_n S_n(t) r_n.$$

The corresponding *pseudo resolvent* is given by

$$\mathcal{R}_n(\lambda) := i_n (\lambda + A_n)^{-1} r_n$$

whenever it is defined. We define  $\mathcal{S}(t)$  and  $\mathcal{R}$  similarly (with *n* deleted).

For Dirichlet problems, the strong convergence of pseudo resolvent operators is equivalent to Mosco convergence (see [39, Theorem 5.2.4] or [37, Theorem 3.3]). For Neumann problems, we also have that Mosco convergence implies the strong convergence of pseudo resolvent operators (see [21, Proposition 3.2] for the case  $A_n = -\Delta$ ). It is not clear that the converse is true for the Neumann case. However, as remarked in [21, Remark 3.3], the converse follows if  $\Omega_n$  converges to  $\Omega$  in the Hausdorff complementary topology. We state the convergence of pseudo resolvent with respect to the strong operator topology in the following theorem.

**Theorem 2.4.6.** Let  $\Omega_n$  and  $\Omega$  be domains satisfying Assumption 1.4.1. Suppose that  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco for the case of Dirichlet problems; or  $H^1(\Omega_n)$  converges to  $H^1(\Omega)$  in the sense of Mosco for the case of Neumann problems. If  $\lambda \geq \lambda_0$ , then

$$\mathcal{R}_n(\lambda)f \to \mathcal{R}(\lambda)f$$

in  $L^2(D)$  as  $n \to \infty$  for all  $f \in L^2(D)$ .

*Proof.* For Dirichlet problems, the statement is given in [39, Theorem 5.2.4] or [37, Theorem 3.3]. For Neumann problems, a similar argument as in the proof of [21, Proposition 3.2] for  $-\Delta$  can be easily modified to a general elliptic operator in divergence form. The only nontrivial part is to prove the strong convergence. For this, we use arguments similar to the proof of [39, Proposition 5.5.2] to obtain

$$\frac{\alpha_0}{2} \|u_n - v_n\|_{H^1(\Omega_n)}^2 \le a_n(u_n - v_n, u_n - v_n) + \lambda \|u_n - v_n\|_{L^2(\Omega_n)}^2 \to 0,$$

where  $u_n := \mathcal{R}_n(\lambda)f$  and  $v_n \in H_0^1(\Omega_n)$  is a sequence from Mosco convergence converging to  $u := \mathcal{R}(\lambda)f$  (that is,  $(i_nv_n, i_n\nabla v_n) \to (iu, i\nabla u)$  in  $L^2(D) \times L^2(D, \mathbb{R}^N)$ ). Note that here we need to consider  $a_n(u_n - v_n, u_n - v_n)$  in stead of  $a_n(u_n - u, u_n - u)$  in the original proof because the bilinear form  $a_n(\cdot, \cdot)$  is only defined on  $H^1(\Omega_n)$ . It is now clear that  $\mathcal{R}_n(\lambda)f \to \mathcal{R}(\lambda)f$  in  $L^2(D)$ .

The key result to obtain the convergence of degenerate semigroup  $S_n(t)$  is the following theorem.

**Theorem 2.4.7** ([4, Theorem 5.2]). Suppose there exists M > 0 such that

$$\|\lambda \mathcal{R}_n(\lambda)\|_{\mathscr{L}(L^2(D))} \le M,$$

for all  $\lambda \in \mathbb{C}$  with  $\operatorname{Re}(\lambda) > 0$ . If there exists  $\lambda > 0$  such that  $\mathcal{R}_n(\lambda) \to \mathcal{R}(\lambda)$  in the strong operator topology, then  $\mathcal{S}_n(t) \to \mathcal{S}(t)$  in the strong operator topology uniformly with respect to t in compact subsets of  $(0, \infty)$ .

We cannot directly apply Theorem 2.4.7 to obtain the convergence of degenerate semigroups  $S_n(t)$  in our framework because the condition  $\|\lambda \mathcal{R}_n(\lambda)\|_{\mathscr{L}(L^2(D))} \leq M$  for all  $\lambda \in \mathbb{C}$  with  $\operatorname{Re}(\lambda) > 0$  does not hold for a general elliptic operator  $\mathcal{A}_n$ . However, we can consider the coercive form  $a_{\lambda_0}(\cdot, \cdot)$  on V defined by

$$a_{\lambda_0}(u,v) := a(u,v) + \lambda_0(u|v)_{L^2(\Omega)}, \qquad (2.47)$$

for all  $u, v \in V$  where  $a(\cdot, \cdot)$  is given by (2.34). We define a similar coercive form  $a_{\lambda_0,n}(\cdot, \cdot)$  on  $V_n$  for each  $n \in \mathbb{N}$ . The coercive forms  $a_{\lambda_0}(\cdot, \cdot)$  and  $a_{\lambda_0,n}(\cdot, \cdot)$  associate with the operators  $A + \lambda_0$  and  $A_n + \lambda_0$  respectively for which their negatives are generators of analytic  $C_0$ -semigroups on  $L^2(\Omega)$  and  $L^2(\Omega_n)$  (denoted by T(t) and  $T_n(t)$ , respectively). Moreover,  $S(t) = e^{\lambda_0 t} T(t)$  and  $S_n(t) = e^{\lambda_0 t} T_n(t)$ . Since the resolvent of  $-(A + \lambda_0)$  satisfies the assumption of Theorem 2.4.7, we get  $i_n T_n(t)r_n \to iT(t)r$  in the strong operator topology uniformly with respect to t on compact subsets of  $(0, \infty)$ . Hence, we obtain the strong convergence of degenerate semigroups  $\mathcal{S}_n(t)$  and the convergence is uniform with respect to t on compact subsets of  $(0, \infty)$ . From the above discussion, we have the following result.

**Theorem 2.4.8.** Suppose that the assumptions of Theorem 2.4.6 are satisfied. Then  $S_n \to S$  in  $C((0,T], \mathcal{L}_s(L^2(D)))$  for all T > 0.

We state convergence of solutions under domain perturbation below.

**Theorem 2.4.9.** Suppose that the assumptions of Theorem 2.4.6, Assumption 2.4.5 and Assumption 2.4.2 (with the time variable disregarded) are satisfied. Suppose also that  $i_n u_{0,n}$  is uniformly bounded in  $L^2(D)$ . We denote the solutions of (2.38) and (2.41) by u and  $u_n$  respectively.

(i) **Dirichlet problems:** Assume that  $ri_n u_{0,n} \rightarrow u_0$  in  $L^2(\Omega)$  weakly. If  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco, then  $i_n u_n \rightarrow iu$  in  $C((s, t_0], L^2(D))$  for all  $t_0 \in (s, \infty)$ .

In addition, if  $i_n u_{0,n} \rightarrow i u_0$  in  $L^2(D)$  strongly, then  $i_n u_n \rightarrow i u$  in  $C([s, t_0], L^2(D))$ .

(ii) Neumann problems: Assume that  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$  strongly. If  $H^1(\Omega_n)$ converges to  $H^1(\Omega)$  in the sense of Mosco, then  $i_n u_n \to i u$  in  $C([s, t_0], L^2(D))$  for all  $t_0 \in (s, \infty)$ .

*Proof.* By Theorem 2.4.8,  $S_n \to S$  in  $C((0,T], \mathcal{L}_s(L^2(D)))$ . For Dirichlet problems, we get from the weak convergence of  $u_{0,n}$  that  $S_n(t)u_{0,n} \to S(t)u_0$  in  $L^2(D)$  uniformly with respect to t in compact subsets of  $(0, \infty)$ . This can be seen by considering  $S_n(t)u_{0,n}$ 

and  $S(t)u_0$  as solutions of homogeneous linear parabolic equations and applying the convergence results for linear parabolic equations under domain perturbation as done in the proof of Theorem 2.4.3. Therefore, the statement of theorem follows directly from Theorem 2.3.5 with  $U_n(t,s) = S_n(t-s)$  and U(t,s) = S(t-s). The assertion for Neumann problems follows in the same way.

Remark 2.4.10. (i) We do not explicitly assume that  $|\Omega_n| \to |\Omega|$  for autonomous equations under Neumann boundary condition in the above theorem. In this case, we make use of [4, Theorem 5.2] and the strong convergence of resolvent operators to get  $S_n \to S$ in  $C((0,T], \mathcal{L}_s(L^2(D)))$ . However, in Theorem 2.4.3 for non-autonomous equations we require  $|\Omega_n| \to |\Omega|$  to obtain (2.16) (see Lemma 2.4.1). This is because we need a strong convergence  $i_n r_n u_0 \to ir u_0$  in  $L^2(D)$  for a fixed function  $u_0 \in L^2(D)$  in order to apply our convergence result for non-autonomous linear equations with Neumann boundary condition in Theorem 1.4.16.

(ii) Let  $1 be such that <math>\frac{N}{2}(\frac{1}{p} - \frac{1}{2}) \in [0, 1)$ . Assume that all assumptions in Theorem 2.4.9 are satisfied. If we replace (2.26) (with the time variable disregarded) by

$$|g_n(x,\xi)| \le \psi(x) + c|\xi|^{2/p}$$
  
 $|g(x,\xi)| \le \psi(x) + c|\xi|^{2/p}$ 

where  $\psi \in L^p(\mathbb{R}^N)$ , then the substitution operators  $f_n(t, \cdot)$  defined by  $f_n(u)(x) := g_n(x, u(x))$  and  $f(t, \cdot)$  defined by f(u)(x) := g(x, u(x)) are in  $C(L^2(\mathbb{R}^N), L^p(\mathbb{R}^N))$  ([3, Chapter 3]). For Dirichlet problems, a standard heat kernel estimate (2.29) with  $U_n(t, s)$  replaced by  $S_n(t-s)$  holds for all  $n \in \mathbb{N}$  and the constant C > 0 is independent of  $n \in \mathbb{N}$ . Taking  $E := L^p(D)$  and  $F := L^2(D)$ , we have that  $U_n(t, s) := i_n S_n(t-s)r_n$  and U(t,s) := iS(t-s)r satisfy assumptions in Section 2.3 with  $\gamma := \frac{N}{2}(\frac{1}{p} - \frac{1}{2})$ . Unlike Theorem 2.4.9, the maximal existence time  $t^+(s, u_0)$  may be strictly finite. In this case, an application of Theorem 2.3.5 implies that  $t^+(s, u_0) \leq \liminf_{n\to\infty} t_n^+(s, u_{0,n})$  and the weak solution  $i_n u_n$  converges to iu in  $C((s, t_0], L^2(D))$  for all  $t_0 \in (s, t^+(s, u_0))$ . In general, we could also study the convergence of solutions in an  $L^p$  space as noted in Remark 2.4.4 (see also [38, Section 6]).

We only mention above that the pseudo resolvent  $\mathcal{R}_n(\lambda)$  converges to  $\mathcal{R}(\lambda)$  in the strong operator topology. Since  $\Omega_n, \Omega \subset D$  for all  $n \in \mathbb{N}$ , resolvent operators for Dirichlet problems are compact on  $L^2(D)$  (from compactness of  $H_0^1(\Omega_n)$  in  $L^2(\Omega_n)$  by Rellich theorem). It turns out that the pseudo resolvent for Dirichlet problems converges in the operator norm as stated below. **Theorem 2.4.11** ([37, Corollary 4.7]). Suppose that  $\Omega_n$  and  $\Omega$  satisfy Assumption 1.4.1 and  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco. If  $\lambda \in \rho(-A)$ , then  $\lambda \in \rho(-A_n)$ for n sufficiently large and  $\mathcal{R}_n(\lambda) \to \mathcal{R}(\lambda)$  in  $\mathscr{L}(L^2(D))$ .

An important consequence of Theorem 2.4.11 and [57, Theorem IV.3.16] is the following corollary.

**Corollary 2.4.12** ([37, Corollary 4.2]). Suppose that assumptions in Theorem 2.4.11 are satisfied. If  $\Sigma \subset \sigma(-A)$  is a compact spectral set and  $\Gamma$  is a rectifiable closed curve enclosing  $\Sigma$  and separating it from the remaining of spectrum, then  $\sigma(-A_n)$  is separated by  $\Gamma$  into a compact spectral set  $\Sigma_n$  and the rest of the spectrum for n sufficiently large. Moreover, for the corresponding spectral projections P and  $P_n$ , we have that the images of P and  $P_n$  have the same dimension and  $i_n P_n r_n$  converges to iPr in norm.

For Neumann problems, we do not have the uniform convergence (convergence in the operator norm) for pseudo resolvent operators if we use the corresponding Mosco convergence  $H^1(\Omega_n)$  to  $H^1(\Omega)$  in Theorem 2.4.11. In fact, there are several examples including a sequence of dumbbell shaped domains converging to two isolated balls (see [56] and references therein) or a more general exterior domain perturbation in Arrieta [6] show that eigenvalues do not behave continuously under domain perturbation. By [57, Theorem IV.3.16], pseudo resolvent operators cannot converge in the operator norm. Stronger assumptions on domain are required to get the uniform convergence of pseudo resolvent operators for Neumann problems. For examples, if we replace a standard dumbbell shaped domains by two touching balls opened up slightly near the touching point (see [39, Section 7] for a general assumptions) or a nonstandard dumbbell type domains constructed with a certain channel in [10], then we obtain the convergence of pseudo resolvent operators in the operator norm and consequently the convergence of spectral projections.

# Chapter 3

# Invariant manifolds for parabolic equations under domain perturbation

In this chapter, we consider parabolic equations as dynamical systems and study their behaviour under domain perturbation. In particular, we study stable and unstable invariant manifolds for parabolic equations near a stationary point. We only consider autonomous parabolic equations with Dirichlet boundary condition because the most complete results are known. Similar results for Neumann problems could be obtained by the same technique. However, stronger assumptions on domains are needed because convergence of spectral projections for Neumann problems requires stronger assumptions as seen in Section 2.4.2.

## 3.1 Introduction

The study of invariant manifolds is an important tool to understand the behaviour of a dynamical system near a stationary point. The first step is to construct invariant manifolds of various types which are characterised by their behaviour (grow, decay or neither) under the flow. For nonlinear differential equations, a fundamental way to construct invariant manifolds is done by considering the corresponding linearised equations. The main methods refer back to Hadamard [49], Liapunov [59] and Perron [67]. The Hadamard approach involves the use of splitting between various subspaces to estimate projections of the flow in the different directions. The Liapunov-Perron approach uses the fixed point argument for a certain integral equation to obtain invariant manifolds. For parabolic equations, Henry [52] obtains the existence of invariant manifolds under the assumption that the differential operator of the linearised equation generates an analytic semigroup. This allows the use of fractional power of the operator. Later, Bates and Jones [14] consider semilinear equations with the assumption that the linearised differential operator generates only a  $C_0$ -semigroup, and also allow infinite dimensional center manifolds. Once the existence of invariant manifolds is established, the fundamental problem is the *persistence* of invariant manifolds under perturbation. For finite dimensional systems, normal hyperbolicity is the main concept to deal with persistence invariant manifolds (see for example [46], [54], and [63]). For infinite dimensional systems, existence and persistence of invariant manifolds for semiflows in Banach space are obtained in [15] using normal hyperbolicity assumption.

The purpose of our work is to study the behaviour of invariant manifolds for semilinear parabolic equations under domain perturbation. There are similar results on the effect of domain variation on the dynamics of parabolic equations. In [68], upper semicontinuity of attractors is obtained for reaction-diffusion equations with Neumann boundary condition when the domain  $\Omega \subset \mathbb{R}^M \times \mathbb{R}^N$  is squeezed in the  $\mathbb{R}^N$ -direction. Arrieta and Carvalho [10] consider a similar problem on a sequence of bounded and Lipschitz perturbed domains  $\Omega_n$ . They give necessary and sufficient conditions on domains for spectral convergence of the linearised problem and obtain continuity (upper and lower semicontinuity) of local unstable manifolds and consequently continuity of attractors. For results under Dirichlet boundary condition, we refer to [42]. In [10], continuity of local unstable invariant manifolds is proved by keeping track of the construction adapted from Henry [52]. Although our framework on semilinear parabolic equations fits into [52], we use the existence of invariant manifolds from [14] to prove continuity of both stable and unstable invariant manifolds under domain perturbation.

### **3.2** Invariant manifolds for parabolic equations

#### 3.2.1 Semiflows induced by abstract parabolic equations

Let  $\Omega$  be a bounded open set in  $\mathbb{R}^N$ ,  $N \geq 2$ . We consider the autonomous semilinear initial boundary value problem

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}u = g(x, u) & \text{in } \Omega \times (0, \infty) \\ u = 0 & \text{on } \partial\Omega \times (0, \infty) \\ u(\cdot, 0) = u_0 & \text{in } \Omega, \end{cases}$$
(3.1)

where  $g \in C(\mathbb{R}^N \times \mathbb{R})$ , and  $\mathcal{A}$  is an elliptic operator defined by (2.31). As seen in Section 2.4.2, we can write (3.1) in the abstract form as

$$\begin{cases} \dot{u}(t) + Au(t) = f(u(t)) & t \in (0, \infty) \\ u(0) = u_0 \end{cases}$$
(3.2)

in  $L^2(\Omega)$ , where f(u)(x) := g(x, u(x)) is the substitution operator induced by g. We make the following assumption on the nonlinearity f as in the framework of [14].

Assumption 3.2.1. We assume that  $f: L^2(\Omega) \to L^2(\Omega)$  is locally Lipschitz and f(0) = 0. Moreover, for every  $\varepsilon > 0$  there exists a neighbourhood  $U = U(\varepsilon)$  of 0 such that f has a Lipschitz constant  $\varepsilon$  in U.

The above assumption is satisfied for g satisfying conditions in Assumption 2.4.2 (with the time variable disregarded) and g(x,0) = 0 for almost all  $x \in \mathbb{R}^N$ . Note also that Assumption 3.2.1 means f(u) is a higher order term. We could think of (3.2) as a linearised problem near a fixed point.

As discussed in Section 2.4.2, for any given initial condition  $u_0 \in L^2(\Omega)$  the abstract equation (3.2) has a unique maximal solution  $u \in C([0, t^+(u_0)), L^2(\Omega))$ . In addition, the solution u can be represented by the variation of constants formula (2.45). By our assumption that the substitution operator f belongs to  $C(L^2(\Omega), L^2(\Omega))$ , the nonlinearity f is linearly bounded. Thus, by a direct application of Gronwall's inequality, the estimate  $||u(t)||_{L^2(\Omega)} \leq M$  holds for  $t \in [0, T] \cap [0, t^+(u_0))$ , where M is a positive constant depending on T > 0 and  $||u_0||_{L^2(\Omega)}$ . This implies that  $[0, T] \subset [0, t^+(u_0))$  (see [40, Lemma 17.1, Corollary 17.2]). Since the above consideration holds for all T > 0, we have  $t^+(u_0) = \infty$ . To study abstract parabolic equations as dynamical systems, we consider a *semiflow*  $\Phi_t : L^2(\Omega) \to L^2(\Omega)$  defined by

$$\Phi_t(u_0) := u(t), \tag{3.3}$$

for all  $t \in [0, t^+(u_0))$ . Sometimes we would like to study the backwards behaviour of solutions. We call a continuous curve  $u : [-t, 0] \to L^2(\Omega)$  for some t > 0 a backwards solution branch for  $u_0 \in L^2(\Omega)$  if

$$\Phi_s(u(-s)) = u_0$$

for all  $s \in [0, t]$ . We write  $\Phi_{-s}(u_0) = u(-s)$  when we look at a particular backwards solution branch.

#### 3.2.2 Existence of invariant manifolds for parabolic equations

The purpose of this subsection is to summarise the construction of invariant manifolds in [14], which applies to our framework. Note that it only requires that -A generates a  $C_0$ -semigroup in [14].

**Definition 3.2.2.** Let V be a subset of U.

1. V is called *positively invariant* relative to U if for each  $v \in V$ 

$$\bigcup_{s \in [0,t]} \Phi_s(v) \subset U \quad \text{implies} \quad \bigcup_{s \in [0,t]} \Phi_s(v) \subset V$$

for all t > 0.

2. V is called *negatively invariant* relative to U if for each  $v \in V$  we have that if a backwards solution branch for v exists then there exists  $t_v > 0$  and a backwards branch  $\Phi_{-s}(v)$  for  $0 \leq s < t_v$  with  $t_v$  maximal such that

$$\bigcup_{s \in [0,t]} \Phi_{-s}(v) \subset U \quad \text{implies} \quad \bigcup_{s \in [0,t]} \Phi_{-s}(v) \subset V$$

for all  $0 \leq t < t_v$ .

3. V is *invariant* relative to U if it is positively and negatively invariant relative to U.

If the condition that orbit lies in U is removed then the set is called positively invariant, negatively invariant or invariant, respectively.

**Definition 3.2.3.** Let U be a neighbourhood of 0. We define

 $W^{s} = \{u \in U : \Phi_{t}(u) \in U \text{ for all } t \geq 0 \text{ and } \Phi_{t}(u) \to 0 \text{ exponentially as } t \to \infty \}$  $W^{u} = \{u \in U : \text{ some backwards branch } \Phi_{t}(u) \text{ exists for all } t < 0 \text{ and lies in } U, \text{ and } \Phi_{t}(u) \to 0 \text{ exponentially as } t \to -\infty \}$ 

These sets  $W^s$  and  $W^u$  are invariant relative to U and are called stable and unstable sets, respectively. It is proved in [14] that they are indeed invariant manifolds. We sometimes write  $W^s(U)$  and  $W^u(U)$  to indicate their dependence on the neighbourhood U.

Recall from Section 2.4.2 that -A is a generator of an analytic  $C_0$ -semigroup  $S(t), t \ge 0$  on  $L^2(\Omega)$ . We denote the spectrum of -A by  $\sigma(-A)$  and decompose it as

$$\sigma(-A) = \sigma^s \cup \sigma^c \cup \sigma^u$$

where

$$\sigma^{s} = \{\lambda \in \sigma(-A) : \operatorname{Re}(\lambda) < 0\}$$
  

$$\sigma^{c} = \{\lambda \in \sigma(-A) : \operatorname{Re}(\lambda) = 0\}$$
  

$$\sigma^{u} = \{\lambda \in \sigma(-A) : \operatorname{Re}(\lambda) > 0\}.$$
  
(3.4)

Since  $\Omega$  is bounded, Rellich's theorem implies that the embedding  $H_0^1(\Omega) \hookrightarrow L^2(\Omega)$  is compact. Hence the resolvent  $(\lambda + A)^{-1} : L^2(\Omega) \to L^2(\Omega)$  is also compact when it is defined. This implies that  $\sigma(-A)$  consists of eigenvalues with finite multiplicities (see [57]). It is easily seen from [41, Theorem 3, XVII §6] that  $\sigma^c$  and  $\sigma^u$  are finite sets. Let  $\Gamma^c$  and  $\Gamma^u$  be rectifiable closed curves separating  $\sigma^c$  and  $\sigma^s$  respectively from the remaining spectrum. There are invariant subspaces of  $L^2(\Omega)$  associated to  $\sigma^s, \sigma^c$  and  $\sigma^u$  via the spectral projections (see [57])

$$P^{c} = \frac{1}{2\pi i} \int_{\Gamma^{c}} (\lambda + A)^{-1} d\lambda$$
$$P^{u} = \frac{1}{2\pi i} \int_{\Gamma^{u}} (\lambda + A)^{-1} d\lambda.$$

Indeed, we decompose  $L^2(\Omega) = X^s \oplus X^c \oplus X^u$  where  $X^s = (1 - P^c - P^u)L^2(\Omega)$ ,  $X^c = P^c L^2(\Omega)$  and  $X^u = P^u L^2(\Omega)$ . Note that  $\dim(X^c)$  and  $\dim(X^u)$  are finite. We set  $X^{cs} = X^c \oplus X^s$  and  $X^{cu} = X^c \oplus X^u$ . For \* = s, c, u, cs, cu, we have that  $-A^* = -A|_{X^*}$ is a generator of  $S^*(t) = S(t)|_{X^*}$ . Since S(t) is an analytic semigroup, there exist M > 0and  $\sigma > 0$  such that

$$\|S^s(t)\| \le M e^{-\sigma t},$$

for all t > 0.

To obtain the existence of local stable and unstable invariant manifolds, we decompose  $L^2(\Omega) = X^- \oplus X^+$  with dim  $X^+ < \infty$  in two different ways; either  $X^- = X^s$  and  $X^+ = X^{cu}$  or  $X^- = X^{cs}$  and  $X^+ = X^u$ . We denote a natural projection (via spectral projection) onto  $X^+$  by  $P^+$ , a natural projection on  $X^-$  by  $P^- := 1 - P^+$  and write  $-A^{\pm} = -A|_{X^{\pm}}$ . In both cases, we have that  $-A^-$  generates a  $C_0$ -semigroup  $S^-(t)$  on  $X^-$  satisfying

$$|S^{-}(t)|| \le M_1 e^{\alpha t},\tag{3.5}$$

for all  $t \ge 0$  where  $M_1 > 0$  and  $\alpha \in \mathbb{R}$ . Similarly,  $-A^+$  generates a  $C_0$ -group  $S^+(t)$  on  $X^+$  satisfying

$$\|S^+(t)\| \le M_2 e^{\beta t},\tag{3.6}$$

for all  $t \leq 0$  where  $M_2 > 0$  and  $\beta > \alpha$ . The parameters  $\alpha$  and  $\beta$  can be chosen as follows (see proof of Theorem 1.1 case (D) and proof of Theorem 1.2 case (D) in [14]).

- If  $X^- = X^s$  and  $X^+ = X^{cu}$ , we take  $\alpha = -\sigma$  and fix  $\beta$  such that  $-\sigma < \beta < 0$ .
- If  $X^- = X^{cs}$  and  $X^+ = X^u$ , we take  $\beta > 0$  such that  $\beta < \min\{\operatorname{Re}(\lambda) : \lambda \in \sigma^u\}$ and fix  $\alpha$  such that  $0 < \alpha < \beta$ .

The main techniques in Bates and Jones [14] are a renorming of  $X^-$  and  $X^+$  and a modification of nonlinearity f. Since we decompose  $L^2(\Omega) = X^- \oplus X^+$ , norms on  $X^$ and  $X^+$  are originally inherited from  $L^2(\Omega)$ . Indeed, if  $u = v \oplus w \in L^2(\Omega)$  where  $v \in X^$ and  $w \in X^+$ , then

$$\frac{1}{\|P^-\| + \|P^+\|} (\|v\|_{L^2(\Omega)} + \|w\|_{L^2(\Omega)}) \le \|u\|_{L^2(\Omega)} \le (\|v\|_{L^2(\Omega)} + \|w\|_{L^2(\Omega)}).$$
(3.7)

However, we can renorm  $X^-$  and  $X^+$  by

$$\|v\|_{X^{-}} := \sup_{t \ge 0} e^{-\alpha t} \|S^{-}(t)v\|_{L^{2}(\Omega)},$$
(3.8)

for  $v \in X^-$  and similarly

$$\|w\|_{X^+} := \sup_{t \le 0} e^{-\beta t} \|S^+(t)w\|_{L^2(\Omega)},$$
(3.9)

for  $w \in X^+$ . These norms are equivalent on  $X^-$  and  $X^+$ , respectively. It is easy to see that (see also [14, Lemma 2.1] )

$$\|v\|_{L^{2}(\Omega)} \leq \|v\|_{X^{-}} \leq M_{1}\|v\|_{L^{2}(\Omega)}, \qquad (3.10)$$

for all  $v \in X^-$  and

$$\|w\|_{L^{2}(\Omega)} \leq \|w\|_{X^{+}} \leq M_{2}\|w\|_{L^{2}(\Omega)}, \qquad (3.11)$$

for all  $w \in X^+$ . The modification of nonlinearity f is done by cutting off arguments so that we obtain a globally Lipschitz function  $\tilde{f}$ . Let  $\eta > 0$  be arbitrary. By Assumption 3.2.1, we can choose  $\delta > 0$  such that f has a Lipschitz constant less than  $\eta/12$  in  $B_{L^2(\Omega)}(0, 2\delta)$ . Let  $\Psi : L^2(\Omega) \to \mathbb{R}$  be a function defined by

$$\Psi(u) = \begin{cases} 1 & \text{if } \|u\|_{L^{2}(\Omega)} \le \delta \\ 2 - \frac{\|u\|_{L^{2}(\Omega)}}{\delta} & \text{if } \delta \le \|u\|_{L^{2}(\Omega)} \le 2\delta \\ 0 & \text{if } \|u\|_{L^{2}(\Omega)} \ge 2\delta. \end{cases}$$

By setting  $\tilde{f}(u) := \Psi(u)f(u)$  for all  $u \in L^2(\Omega)$ , we have that  $\tilde{f}$  is globally Lipschitz continuous with constant  $\varepsilon < \eta/4$ . This Lipschitz constant  $\varepsilon$  can be chosen as small as we require by shrinking  $\delta$ .

With this modified system  $\dot{u}(t) + Au(t) = \tilde{f}(u(t))$ , the solution to an initial value parabolic equation u(t) also exists for  $t \ge 0$ , that is the maximal existence time  $t^+(u_0) = \infty$  for all  $u_0 \in L^2(\Omega)$ . Moreover, the modified system agrees with the original system (3.2) inside  $B_{L^2(\Omega)}(0, \delta)$ . Hence, the modification gives us a local behaviour of the original system. Although our assumptions imply that the original system has a global solution for any initial data  $u_0$ , we modify the nonlinear term f in order to gain some useful estimates of the projections of semiflows in various directions (subspaces).

In [14], invariant manifolds for the modified system are constructed as follows. We choose the Lipschitz constant  $\varepsilon$  of  $\tilde{f}$  so that  $\varepsilon < (\beta - \alpha)/4$  and there exists  $\gamma$  such that

$$-\beta + 2\varepsilon < \gamma < -\alpha - 2\varepsilon. \tag{3.12}$$

By abuse of notations, we denote again by  $\Phi_t(u_0)$  the solution u(t) of the modified system with the initial condition  $u_0$ . Let

$$W^{-} = \{ u \in L^{2}(\Omega) : e^{\gamma t} \Phi_{t}(u) \to 0 \text{ as } t \to \infty \}$$
$$W^{+} = \{ u \in L^{2}(\Omega) : \text{ a backward branch } \Phi_{t}(u) \text{ exists for all } t \leq 0$$
and  $e^{\gamma t} \Phi_{t}(u) \to 0 \text{ as } t \to -\infty \}.$ 

The main idea to show that  $W^-$  and  $W^+$  are invariant manifolds is that certain cones and moving cones are positively invariant, which can be determined by the difference in the growth rates on  $X^-$  and  $X^+$ . For  $\lambda > 0$ , we define a cone

$$K_{\lambda} = \{ (v, w) \in X^{-} \times X^{+} : \lambda \| v \|_{X^{-}} \le \| w \|_{X^{+}} \}.$$
(3.13)

It is shown in [14, Lemma 2.4] that  $K_{\lambda}$  is positively invariant if  $\lambda \in [\mu, \nu]$  where  $\mu$  and  $\nu$  are positive parameters with  $\mu < 1 < \nu$  satisfying

$$\varepsilon < (\beta - \alpha)/(2 + \nu + \mu^{-1}).$$
 (3.14)

Indeed,  $\mu$  and  $\nu$  can be further restricted so that

$$\varepsilon(1+\mu^{-1}) - \beta < \gamma < -\varepsilon(1+\nu) - \alpha.$$
(3.15)

The next two theorems give the existence of global stable and global unstable invariant manifolds for the modified system.

**Theorem 3.2.4** ([14, Theorem 2.1]). There exists a Lipschitz function  $h^-: X^- \to X^+$ such that  $W^- = \operatorname{graph}(h^-)$  and  $h^-(0) = 0$ .

Sketch of the proof. Fix  $v_0 \in X^-$  and let

$$B = \{ w_0 \in X^+ : \|w_0\|_{X^+} \le \mu \|v_0\|_{X^-} \}.$$

We write  $\Phi_t(u_0) = u(t)$  as  $u(t) = v(t) \oplus w(t)$  where  $v(t) \in X^-$  and  $w(t) \in X^+$ . Define

$$G_t = \{ w_0 \in B : \|w(t)\|_{X^+} \le \mu \|v(t)\|_{X^-} \}.$$

It can be shown that  $G_{\infty} := \bigcap_{t \ge 0} G_t$  contains exactly one element. A function  $h^-$  defined by  $h^-(v_0) = G_{\infty}$  for  $v_0 \in X^-$  is a Lipschitz function with  $h^-(0) = 0$  and  $\operatorname{graph}(h^-) = W^-$ .

**Theorem 3.2.5** ([14, Theorem 2.2]). There exists a Lipschitz function  $h^+: X^+ \to X^$ such that  $W^+ = \operatorname{graph}(h^+)$  and  $h^+(0) = 0$ .

*Sketch of the proof.* The proof is based on a standard contraction mapping argument. Let

$$Y = \{h \in C(X^+, X^-) : h(0) = 0 \text{ and } h \text{ is } \nu^{-1}\text{-Lipschitz } \}.$$

Then Y is a complete metric space with the norm

$$\|h\|_{\text{Lip}} = \sup_{w \neq 0} \frac{\|h(w)\|_{X^{-}}}{\|w\|_{X^{+}}}.$$
(3.16)

For an arbitrary  $h \in Y$ , it can be shown that  $P^+\Phi_t(\operatorname{graph}(h)) = X^+$  and that  $\Phi_t(\operatorname{graph}(h))$  is the graph of a  $\nu^{-1}$ -Lipschitz function for all  $t \ge 0$ . Hence, the map  $T_t: Y \to Y$  for  $t \ge 0$  given by

$$T_t(h) = \tilde{h}$$

where  $\tilde{h} \in Y$  with graph $(\tilde{h}) = \Phi_t(\text{graph}(h))$  is well-defined. Furthermore,  $T_t$  is a contraction on Y for t sufficiently large. Indeed,

$$||T_t(h_2) - T_t(h_1)||_{\text{Lip}} \le \nu(\nu - \mu)^{-1} \exp\left((\alpha - \beta + \varepsilon(2 + \mu + \nu^{-1}))t\right)||h_2 - h_1||_{\text{Lip}}.$$

Hence, there exists a unique fixed point  $h_t \in Y$  for t sufficiently large. We can show that  $h_t$  is a fixed point of  $T_{\tau}$  for all  $\tau \ge 0$  and  $h^+ := h_t$  is the required Lipschitz function with graph $(h^+) = W^+$  and  $h^+(0) = 0$ .

Remark 3.2.6. Let  $Y_0 = \{h \in Y : h \text{ is differentiable at } 0 \text{ and } Dh(0) = 0\}$ . Then  $Y_0$  is closed in Y. As  $D\tilde{f}(0) = 0$  (in fact Df(0) = 0 from Assumption 3.2.1), it can be shown that  $T_t: Y_0 \to Y_0$  for all t > 0. Hence, the fixed point  $h^+$  in Theorem 3.2.5 lies on  $Y_0$  (see the proposition after the proof of Theorem 2.2 in [14]).

The next two theorems give the existence of the local stable and the local unstable invariant manifolds for (3.2).

**Theorem 3.2.7** ([14, Theorem 1.1(i)]). Under the assumptions given above, there exists an open neighbourhood U of 0 in  $L^2(\Omega)$  such that  $W^s$  is a Lipschitz manifold which is tangent to  $X^s$  at 0, that is, there exists a Lipschitz function  $h^s : P^s(U) \to X^{cu}$  such that graph $(h^s) = W^s$ ,  $h^s(0) = 0$  and  $h^s$  is differentiable at 0 with  $Dh^s(0) = 0$ .

Sketch of the proof. Set  $X^- = X^s$  and  $X^+ = X^{cu}$ . We take  $\alpha = -\sigma$  and fix  $\beta$  such that  $-\sigma < \beta < 0$ . Renorm  $X^-$  and  $X^+$  by (3.8) and (3.9), respectively. By Assumption 3.2.1, there exists  $\delta > 0$  such that the modification  $\tilde{f}$  has a Lipschitz constant  $\varepsilon < (\beta - \alpha)/4$  and the modified system agrees with the original system on  $B_{L^2(\Omega)}(0, \delta)$ . By applying Theorem 3.2.4, we can find a product neighbourhood  $U \subset B_{L^2(\Omega)}(0, \delta)$  and prove that  $W^s = W^- \cap U$  is a local stable invariant manifold. It can be shown that any local stable manifold constructed using another renorming and modification agrees on a neighbourhood on which the manifolds are both defined. The tangency condition  $Dh^s(0) = 0$  follows by making  $\mu \to 0$  (by letting  $\varepsilon \to 0$  and possibly shrinking U).

**Theorem 3.2.8** ([14, Theorem 1.2(i)]). Under the assumptions given above, there exists an open neighbourhood U of 0 in  $L^2(\Omega)$  such that  $W^u$  is a Lipschitz manifold which is tangent to  $X^u$  at 0, that is, there exists a Lipschitz function  $h^u : P^u(U) \to X^{cs}$  such that graph $(h^u) = W^u$ ,  $h^u(0) = 0$  and  $h^u$  is differentiable at 0 with  $Dh^u(0) = 0$ .

Sketch of the proof. Set  $X^- = X^{cs}$  and  $X^+ = X^u$ . We take  $\beta > 0$  such that  $\beta < \min\{\operatorname{Re}(\lambda) : \lambda \in \sigma^u\}$  and fix  $\alpha$  such that  $0 < \alpha < \beta$ . Renorm  $X^-$  and  $X^+$  and modify

the nonlinearity f as in the proof of Theorem 3.2.7. Applying Theorem 3.2.5, we can find a product neighbourhood  $U \subset B_{L^2(\Omega)}(0, \delta)$  and prove that  $W^u = W^+ \cap U$  is a local unstable invariant manifold. It can be shown that any local unstable manifold constructed using another renorming and modification agrees on a neighbourhood on which the manifolds are both defined. The tangency condition  $Dh^s(0) = 0$  follows from Remark 3.2.6.

The product neighbourhood U in Theorem 3.2.7 and Theorem 3.2.8 can be chosen to be  $U = V_1 \times V_2$  where  $V_1 \subset X^-$  is a ball of radius  $\delta_1$  and  $V_2 \subset X^+$  is a ball of radius  $\delta_2$ such that  $\delta_1 < \delta_2$  for the local stable manifold and  $\delta_1 > \delta_2$  for the local unstable manifold. In fact, with these choices of product neighbourhoods,  $W^s$  is positively invariant and  $W^u$  is negatively invariant (see property (P4) in [14]).

# 3.3 Domain perturbation for invariant manifolds

In this section, we introduce domain perturbation for invariant manifolds. We state the main result on convergence of the local stable and the local unstable invariant manifolds under changes of the domain.

Let  $\Omega_n$  and  $\Omega$  be bounded open sets in  $\mathbb{R}^N$  satisfying Assumption 1.4.1. We consider the perturbation of (3.1) given by

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}_n u = g_n(x, u) & \text{in } \Omega_n \times (0, \infty) \\ u = 0 & \text{on } \partial \Omega_n \times (0, \infty) \\ u(\cdot, 0) = u_{0,n} & \text{in } \Omega_n, \end{cases}$$
(3.17)

where  $g_n \in C(\mathbb{R}^N \times \mathbb{R})$  and  $\mathcal{A}_n$  is the operator defined by (2.40). As in Section 2.4.2, we can consider the corresponding abstract equation of (3.17) given by

$$\begin{cases} \dot{u}(t) + A_n u(t) = f_n(u(t)) & t \in (0, \infty) \\ u(0) = u_{0,n} \end{cases}$$
(3.18)

in  $L^2(\Omega_n)$ , where  $f_n(u)(x) := g_n(x, u(x))$  is the substitution operator induced by  $g_n$ . As previously seen,  $-A_n$  is a generator of a strongly continuous analytic semigroup  $S_n(t), t \ge 0$  on  $L^2(\Omega_n)$ . When well-posed, the mild solution  $u_n$  of (3.18) can be represented by the variation of constants formula (2.46). We denote the local semiflow induced by parabolic equation (3.18) by  $\Phi_{t,n}: L^2(\Omega_n) \to L^2(\Omega_n)$  where

$$\Phi_{t,n}(u_{0,n}) := u_n(t), \tag{3.19}$$

for  $t \in (0, t^+(u_{0,n}))$ .

As discussed in Section 3.1, one of the main subjects in invariant manifold theory is to study the persistence of invariant manifolds under perturbation. The focus of this work is to study the dependence of invariant manifolds on domain variation. We could think of  $\Phi_{t,n}$  as a perturbation of  $\Phi_t$ . However, the sequence of semiflows  $\Phi_{t,n}$ is defined on different spaces for each  $n \in \mathbb{N}$ , namely  $L^2(\Omega_n)$ . As usual, we make use of the inclusion operators  $i_n$  and i (defined by (1.27) and (1.28), respectively) and the restriction operators  $r_n$  and r (defined by (1.29) and (1.30), respectively) when dealing with domain perturbation.

By the above inclusion operators, we consider the stable and unstable invariant manifolds as subsets of  $L^2(D)$ . The main difficulty to establish convergence of invariant manifolds under domain perturbation in this framework is that invariant manifolds are constructed using a special renorming of  $X^-$  and  $X^+$  (see (3.8) and (3.9)). Although we have seen in Theorem 2.4.8 that under suitable assumptions on domains the degenerate semigroup  $i_n S_n(t)r_n$  converges to iS(t)r in the strong operator topology uniformly with respect to t in compact subsets of  $(0, \infty)$ , we cannot expect convergence of the projection of solutions  $P_n^-u_n$  and  $P_n^+u_n$  with respect to a sequence of the norms  $\|\cdot\|_{X_n^-}$  and  $\|\cdot\|_{X_n^+}$ , respectively. This is because norms on  $X_n^-$  and  $X_n^+$  (see (3.8) and (3.9)) involve taking the supremum of  $e^{-\alpha t} \|S_n^-(t)v\|_{L^2(\Omega_n)}$  on  $[0,\infty)$  and of  $e^{-\beta t} \|S_n^+(t)v\|_{L^2(\Omega_n)}$  on  $(-\infty, 0]$ , respectively.

We next state our assumptions for the perturbed equations.

Assumption 3.3.1. We assume the following assumptions.

- (i) The coefficients of the operator  $\mathcal{A}_n$ converge to the corresponding coefficients of  $\mathcal{A}$  as  $n \to \infty$  (see Assumption 2.4.5).
- (ii) The stationary point  $0 \in L^2(\Omega)$  of (3.2) is hyperbolic, that is  $\sigma^c = \emptyset$ .

Assumption 3.3.2. We impose the following conditions on the nonlinearity.

- (i)  $f_n : L^2(\Omega_n) \to L^2(\Omega_n)$  is locally Lipschitz and  $f_n(0) = 0$ . In addition, for every  $\varepsilon > 0$  there exists a neighbourhood  $U_n = U_n(\varepsilon)$  of 0 such that  $f_n$  has a Lipschitz constant  $\varepsilon$  in  $U_n$ . Moreover,  $U_n$  can be chosen uniformly with respect to  $n \in \mathbb{N}$  in the sense that we can take  $U_n$  to be a ball centered at 0 in  $L^2(\Omega_n)$  of the same radius for all  $n \in \mathbb{N}$ .
- (ii)  $i_n f_n r_n(u) \to i f r(u)$  in  $L^2(D)$  pointwise for  $u \in L^2(D)$ .

Note that if  $g_n$  and g satisfy conditions in Assumption 2.4.2 (with the time variable disregarded) and  $g_n(x,0) = 0$  for almost all  $x \in \mathbb{R}^N$ , then the substitution operators  $f_n$  and f satisfy the above assumption.

Our main result can be stated as follows.

**Theorem 3.3.3** (Continuity of local unstable manifolds). Suppose that Assumption 3.3.1 and Assumption 3.3.2 are satisfied. If  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco, then there exists a local unstable invariant manifold  $W_n^u$  for parabolic equation (3.18) for each  $n \in \mathbb{N}$ . Moreover, there exists  $\delta > 0$  such that if we denote by  $B_n = B_{L^2(\Omega_n)}(0, \delta)$  and  $B = B_{L^2(\Omega)}(0, \delta)$ , then we have

(i) Upper semicontinuity:

$$\sup_{v \in W_n^n \cap B_n} \inf_{u \in W^u \cap B} \|i_n(v) - i(u)\|_{L^2(D)} \to 0$$

as  $n \to \infty$ .

(ii) Lower semicontinuity:

$$\sup_{u \in W^u \cap B} \inf_{v \in W^u_n \cap B_n} \|i_n(v) - i(u)\|_{L^2(D)} \to 0$$

as  $n \to \infty$ .

A similar result can be stated for local stable invariant manifolds with an additional assumption of the convergence in measure of the domains.

**Theorem 3.3.4** (Continuity of local stable manifolds). Suppose that Assumption 3.3.1 and Assumption 3.3.2 are satisfied. If  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco, then there exists a local stable invariant manifold  $W_n^s$  for parabolic equation (3.18) for each  $n \in \mathbb{N}$ .

In addition, assume that the Lebesgue measure  $|\Omega_n| \to |\Omega|$  as  $n \to \infty$ . Then there exists  $\delta > 0$  such that if we denote by  $B_n = B_{L^2(\Omega_n)}(0, \delta)$  and  $B = B_{L^2(\Omega)}(0, \delta)$ , then we have

(i) Upper semicontinuity:

$$\sup_{v \in W_n^s \cap B_n} \inf_{u \in W^s \cap B} \|i_n(v) - i(u)\|_{L^2(D)} \to 0$$

as  $n \to \infty$ .

(ii) Lower semicontinuity:

$$\sup_{u\in W^s\cap B}\inf_{v\in W^s_n\cap B_n}\|i_n(v)-i(u)\|_{L^2(D)}\to 0$$

as  $n \to \infty$ .

In the remainder, we show the existence of local stable and local unstable invariant manifolds for the perturbed problem (3.18) in Theorem 3.3.3 and Theorem 3.3.4. The assertion on upper and lower semicontinuity will be proved in Section 3.5 for unstable invariant manifolds and in Section 3.6 for stable invariant manifolds.

By our assumptions in Theorem 3.3.3 or Theorem 3.3.4, we have from Theorem 2.4.9 that for every  $u_{0,n} \in L^2(\Omega_n)$  and  $u_0 \in L^2(\Omega)$  with  $ri_n u_{0,n} \rightharpoonup u_0$  in  $L^2(\Omega)$  weakly, the perturbed equation (3.18) has a unique maximal solution  $u_n \in C([0, t_n^+(u_{0,n})), L^2(\Omega_n))$ and  $i_n u_n(t) \rightarrow iu(t)$  in  $L^2(D)$  uniformly with respect to  $t \in (0, t_0]$  for all  $t_0 \in (0, t^+(u_0))$ . In terms of semiflows, the above can be stated as

$$i_n \Phi_{t,n}(u_{0,n}) \to i \Phi_t(u_0) \tag{3.20}$$

in  $L^2(D)$  as  $n \to \infty$  uniformly with respect to  $t \in (0, t_0]$  for all  $t_0 \in (0, t^+(u_0))$ . In addition, if the initial condition  $i_n u_{0,n}$  converges strongly to  $iu_0$  in  $L^2(D)$ , then (3.20) holds uniformly with respect to  $t \in [0, t_0]$  for all  $t_0 \in (0, t^+(u_0))$ . Recall also that our assumptions imply convergence of pseudo resolvents in the operator norm (Theorem 2.4.11) and convergence of spectral projections (Corollary 2.4.12).

We decompose  $\sigma(-A_n) = \sigma_n^s \cup \sigma_n^c \cup \sigma_n^u$  where  $\sigma_n^s, \sigma_n^c$  and  $\sigma_n^u$  are sets defined similarly to (3.4). By hyperbolicity assumption that  $\sigma_c = \emptyset$  (Assumption 3.3.1 (ii)) and Corollary 2.4.12, we have that  $\Gamma^c$  and  $\Gamma^u$  separate  $\sigma_n^c$  and  $\sigma_n^u$  respectively from the remaining of spectrum for *n* sufficiently large. In fact, the convergence of spectrum in Corollary 2.4.12 implies that  $\sigma_n^c = \emptyset$  for all *n* sufficiently large. Hence, the fixed point  $0 \in L^2(\Omega_n)$ of (3.18) is hyperbolic for all *n* sufficiently large. We define

$$P_n^c = \frac{1}{2\pi i} \int_{\Gamma^c} (\lambda + A_n)^{-1} d\lambda = 0$$
$$P_n^u = \frac{1}{2\pi i} \int_{\Gamma^u} (\lambda + A_n)^{-1} d\lambda$$
$$P_n^s = 1 - P_n^c - P_n^u.$$

Then

$$i_n P_n^c r_n \to i P^c r$$
 and  $i_n P_n^u r_n \to i P^u r$  (3.21)

in  $\mathscr{L}(L^2(D))$  as  $n \to \infty$ . Denote by  $X_n^s, X_n^c$  and  $X_n^u$  the images of  $1 - P_n^c - P_n^u, P_n^c$  and  $P_n^u$ , respectively. For *n* sufficiently large, we can decompose

$$L^{2}(\Omega_{n}) = X_{n}^{s} \oplus X_{n}^{c} \oplus X_{n}^{u}.$$

$$(3.22)$$

Indeed, from the above consideration we have that  $X_n^c = \{0\}$  and  $X_n^u$  is a finite dimensional subspace with  $\dim(X_n^u) = \dim(X^u)$  for all *n* sufficiently large.

To construct invariant manifolds for the modified system, we decompose  $L^2(\Omega_n)$  as  $X_n^- \oplus X_n^+$  in two different ways as in Section 3.2.2. In particular,  $\dim(X_n^+) = \dim(X^+) < \infty$  for *n* sufficiently large and

$$i_n P_n^+ r_n \to i P^+ r \tag{3.23}$$

in  $\mathscr{L}(L^2(D))$ . Moreover, if  $u = v \oplus w \in L^2(\Omega_n)$  where  $v \in X_n^-$  and  $w \in X_n^+$  then

$$\frac{1}{\|P_n^-\| + \|P_n^+\|} (\|v\|_{L^2(\Omega_n)} + \|w\|_{L^2(\Omega_n)}) \le \|u\|_{L^2(\Omega_n)} \le (\|v\|_{L^2(\Omega_n)} + \|w\|_{L^2(\Omega_n)}).$$
(3.24)

By Assumption 3.3.1 (i), we can choose the parameters  $\alpha$  and  $\beta$  for the restriction of semigroup  $S_n(t)$  to  $X_n^-$  and  $X_n^+$  uniformly with respect to  $n \in \mathbb{N}$  so that  $S_n^-(t)$  and  $S_n^+(t)$  satisfy similar conditions as (3.5) and (3.6), respectively. We see that

$$\|v\|_{X_n^-} := \sup_{t \ge 0} e^{-\alpha t} \|S_n^-(t)v\|_{L^2(\Omega_n)},$$
(3.25)

for  $v \in X_n^-$  and

$$\|w\|_{X_n^+} := \sup_{t \le 0} e^{-\beta t} \|S_n^+(t)w\|_{L^2(\Omega_n)},$$
(3.26)

for  $w \in X_n^+$  give equivalent norms on  $X_n^-$  and  $X_n^+$ , respectively. Indeed, we have

$$\|v\|_{L^{2}(\Omega_{n})} \leq \|v\|_{X_{n}^{-}} \leq M_{1}\|v\|_{L^{2}(\Omega_{n})}$$
(3.27)

for all  $v \in X_n^-$  and

$$\|w\|_{L^{2}(\Omega_{n})} \leq \|w\|_{X_{n}^{+}} \leq M_{2}\|w\|_{L^{2}(\Omega_{n})}$$
(3.28)

for all  $w \in X_n^+$  where  $M_1$  and  $M_2$  can be chosen uniformly with respect to  $n \in \mathbb{N}$ .

By Assumption 3.3.2 (i), there exists  $\delta > 0$  independent of n such that the modification  $\tilde{f}_n$  of  $f_n$  has a Lipschitz constant  $\varepsilon < (\beta - \alpha)/4$  and the modified system agrees with the original system on  $B_n := r_n(B)$  where  $B := B_{L^2(D)}(0, \delta)$  for all  $n \in \mathbb{N}$ . Therefore, we can construct the stable and unstable invariant manifold for the modified system by using uniform parameters  $\gamma, \mu$  and  $\nu$  for all  $n \in \mathbb{N}$ . By Theorem 3.2.7, there exists a product neighbourhood  $U_n \subset B_n$  such that a local stable invariant manifold is  $W_n^s(U_n) = \operatorname{graph}(h_n^-) \cap U_n$ . Since the parameters  $\alpha$  and  $\beta$  are chosen uniformly for the renorming of  $X_n^-$  and  $X_n^+$  respectively, we can choose  $U_n \subset B_n$  to be a product neighbourhood  $V_{1,n} \times V_{2,n}$  where  $V_{1,n} \subset X_n^-$  is a ball of radius  $\delta_1$  and  $V_{2,n} \subset X_n^+$  is a ball of radius  $\delta_2$  with  $\delta_1 < \delta_2$  for all  $n \in \mathbb{N}$ . Without loss of generality we may choose  $\delta$ smaller so that the modified system agrees with the original system on  $\overline{B}_n$  for all  $n \in \mathbb{N}$ . Similarly, by Theorem 3.2.8, there exists a product neighbourhood  $\tilde{U}_n \subset B_n$  such that a local unstable invariant manifold is  $W_n^u(\tilde{U}_n) = \operatorname{graph}(h_n^+) \cap \tilde{U}_n$ . Since the parameters  $\alpha$  and  $\beta$  are chosen uniformly for the renorming of  $X_n^-$  and  $X_n^+$  respectively, we can choose  $\tilde{U}_n \subset B_n$  to be a product neighbourhood  $\tilde{V}_{1,n} \times \tilde{V}_{2,n}$  where  $\tilde{V}_{1,n} \subset X_n^-$  is a ball of radius  $\delta_1$  and  $\tilde{V}_{2,n} \subset X_n^+$  is a ball of radius  $\tilde{\delta}_2$  with  $\tilde{\delta}_1 > \tilde{\delta}_2$  for all  $n \in \mathbb{N}$ . Again we may choose  $\delta$  smaller so that the modified system agrees with the original system on  $\overline{B}_n$  for all  $n \in \mathbb{N}$ . Therefore, we have established the existence of local unstable manifolds and local stable manifolds for the perturbed problem (3.18).

We can assume that the choice of neighbourhoods considered above applies to the limit problem (3.2) (by possibly shrinking  $\delta$ ). Therefore, to prove Theorem 3.3.3 and Theorem 3.3.4, it remains to verify the continuity under domain perturbation (upper and lower semicontinuity) of local stable and local unstable invariant manifolds inside some ball  $B_n = B_{L^2(\Omega_n)}(0, \hat{\delta})$  contained in  $V_{1,n} \times V_{2,n}$  or  $\tilde{V}_{1,n} \times \tilde{V}_{2,n}$ .

# 3.4 Some technical results towards the proof of semicontinuity

In this section, we give some technical results required to prove upper and lower semicontinuity in Theorem 3.3.3 and Theorem 3.3.4. In particular, we prove some convergence result for a bounded sequence  $(w_n)_{n \in \mathbb{N}}$  with  $w_n \in X_n^+$  for each  $n \in \mathbb{N}$ . Moreover, we give a characterization of upper and lower semicontinuity.

#### 3.4.1 Convergence of sequence in finite dimensional subspaces

We first give a direct consequence of convergence of spectral projections in (3.21).

**Lemma 3.4.1.** Let  $(\phi_n)_{n\in\mathbb{N}}$  be a sequence with  $\phi_n \in L^2(\Omega_n)$  for each  $n \in \mathbb{N}$  and  $\phi \in L^2(\Omega)$ . We decompose  $\phi_n := \phi_n^s \oplus \phi_n^c \oplus \phi_n^u$  corresponding to the decomposition (3.22). Similarly, we decompose  $\phi := \phi^s \oplus \phi^c \oplus \phi^u$ .

(i) If  $i_n\phi_n \to i\phi$  in  $L^2(D)$  strongly, then  $i_n\phi_n^* \to i\phi^*$  in  $L^2(D)$  strongly for \* = s, c, u.

- (ii) If  $i_n\phi_n \rightharpoonup i\phi$  in  $L^2(D)$  weakly, then  $i_n\phi_n^* \rightharpoonup i\phi^*$  in  $L^2(D)$  weakly for \* = s, c, u.
- (iii) If  $ri_n\phi_n \to \phi$  in  $L^2(\Omega)$  strongly and  $||i_n\phi_n||_{L^2(D)}$  is uniformly bounded, then  $ri_n\phi_n^* \to \phi^*$  in  $L^2(\Omega)$  strongly for \* = s, c, u.
- (iv) If  $ri_n\phi_n \rightarrow \phi$  in  $L^2(\Omega)$  weakly and  $||i_n\phi_n||_{L^2(D)}$  is uniformly bounded, then  $ri_n\phi_n^* \rightarrow \phi^*$  in  $L^2(\Omega)$  weakly for \* = s, c, u.

*Proof.* Suppose that  $i_n \phi_n \to i \phi$  in  $L^2(D)$  strongly. By (3.21), we have

$$\begin{aligned} \|i_n\phi_n^* - i\phi^*\|_{L^2(D)} &\leq \|i_nP_n^*r_ni_n\phi_n - iP^*ri_n\phi_n\|_{L^2(D)} \\ &+ \|iP^*ri_n\phi_n - iP^*ri\phi\|_{L^2(D)} \\ &\leq \|i_nP_n^*r_n - iP^*r\|\|i_n\phi_n\|_{L^2(D)} \\ &+ \|iP^*r\|\|i_n\phi_n - i\phi\|_{L^2(D)} \\ &\to 0 \end{aligned}$$

as  $n \to \infty$  for \* = c, u. Since  $\phi_n^s = (1 - P_n^c - P_n^u)\phi_n$  and  $\phi^s = (1 - P^c - P^u)\phi$ , we obtain the convergence  $i_n \phi_n^s \to i \phi^s$ . This proves assertion (i). The statement (iii) follows from a similar argument above.

Suppose now that  $i_n \phi_n \rightarrow i \phi$  in  $L^2(D)$  weakly. By (3.21) and the weak convergence of  $i_n \phi_n$ , we have for  $\chi \in L^2(D)$  and \* = c, u

$$\begin{aligned} \left| (i_{n}\phi_{n}^{*}) - i\phi^{*})|\chi\rangle_{L^{2}(D)} \right| &= \left| (i_{n}P_{n}^{*}\phi_{n} - iP^{*}\phi|\chi)_{L^{2}(D)} \right| \\ &\leq \left| (i_{n}P_{n}^{*}\phi_{n} - iP^{*}ri_{n}\phi_{n}|\chi\rangle_{L^{2}(D)} \right| \\ &+ \left| (iP^{*}ri_{n}\phi_{n} - iP^{*}\phi|\chi)_{L^{2}(D)} \right| \\ &\leq \|i_{n}P_{n}^{*}r_{n} - iP^{*}r\|\|i_{n}\phi_{n}\|_{L^{2}(D)}\|\chi\|_{L^{2}(D)} \\ &+ \left| (i_{n}\phi_{n} - i\phi|(iP^{*}r)'\chi)_{L^{2}(D)} \right| \\ &\to 0 \end{aligned}$$

as  $n \to \infty$  where  $(iP^*r)'$  denotes the adjoint operator of  $iP^*r$ . This implies  $i_n(\phi_n^*) \rightharpoonup i(\phi^*)$  in  $L^2(D)$  weakly for \* = c, u. Since  $\phi_n^s = (1 - P_n^c - P_n^u)\phi_n$  and  $\phi^s = (1 - P^c - P^u)\phi$ , we obtain the weak convergence  $i_n(\phi_n^s) \rightharpoonup i(\phi^s)$  in  $L^2(D)$ . This proves assertion (ii). The statement (iv) can be proved in a similar fashion.

Note that if  $i_n \phi_n$  is uniformly bounded in  $L^2(D)$ , then there exists a weak convergent subsequence in  $L^2(D)$ . In general, we do not know whether the weak limit is zero almost everywhere on  $D \setminus \Omega$ , that is of the form  $i\phi$  for some  $\phi \in L^2(\Omega)$ . This is the reason we include both assertions (ii) and (iv) in Lemma 3.4.1 above. Remark 3.4.2. The convergence of  $i_n(\phi_n^s) \to i(\phi^s)$  in Lemma 3.4.1 (i) is different to convergence of the projections  $i_n(1 - P_n^c - P_n^u)r_n \to i(1 - P^c - P^u)r$  in  $\mathscr{L}(L^2(D))$ . For example, consider a square domain  $\Omega$  in  $\mathbb{R}^2$  perturbed by attaching "fingers" to one of the sides as previously seen in Figure 1.2. If we increase the number of fingers so that the measure remains the same (by letting their width go to zero). Then  $|\Omega_n \setminus \Omega|$ is a positive constant for all  $n \in \mathbb{N}$ . Recall that  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco. Let  $f \in L^2(D)$  be the constant function 1. By (3.21), we have that  $i_n P_n^c r_n(f) \to i P^c r(f)$  and  $i_n P_n^u r_n(f) \to i P^u r(f)$  in  $L^2(D)$ . If  $i_n(1 - P_n^c - P_n^u)r_n(f) \to$  $i(1 - P^c - P^u)r(f)$  in  $L^2(D)$ , then  $i_n r_n(f) \to ir(f)$  in  $L^2(D)$ . This cannot be true because  $\|i_n r_n(f) - ir(f)\|_{L^2(D)} = |\Omega_n \setminus \Omega| > 0$  for all  $n \in \mathbb{N}$ . Hence,  $i_n(1 - P_n^c - P_n^u)r_n$  does not converge to  $i(1 - P^c - P^u)r$  in  $\mathscr{L}(L^2(D))$ . Note that if we impose the assumption that Lebesgue measure of the domain converges, that is  $|\Omega_n| \to |\Omega|$  as  $n \to \infty$ , then we obtain convergence of the projections  $i_n(1 - P_n^c - P_n^u)r_n \to i(1 - P^c - P^u)r$  in  $\mathscr{L}(L^2(D))$ .

In the next few results, we consider an arbitrary finite dimensional subspace of  $L^2(\Omega_n)$ .

**Lemma 3.4.3.** Let m be a positive integer. Suppose  $V_n$  is an m-dimensional subspace of  $L^2(\Omega_n)$  with a basis  $\{f_{1,n}, f_{2,n}, \ldots, f_{m,n}\}$  for each  $n \in \mathbb{N}$ , and V is an m-dimensional subspace of  $L^2(\Omega)$  with a basis  $\{f_1, f_2, \ldots, f_m\}$ . If  $i_n(f_{j,n}) \to i(f_j)$  in  $L^2(D)$  as  $n \to \infty$ for all  $j = 1, \ldots, m$ , then there exists  $\hat{c} > 0$  such that

$$c_n := \inf \left\{ \left\| \sum_{j=1}^m \xi_j f_{j,n} \right\|_{L^2(\Omega_n)} : \xi = (\xi_1, \dots, \xi_m) \in \mathbb{R}^m, |\xi| = 1 \right\} \ge \hat{c},$$

for all  $n \in \mathbb{N}$ .

*Proof.* Let  $\xi = (\xi_1, \ldots, \xi_m) \in \mathbb{R}^m$  with  $|\xi| = 1$ . By convergence of the bases, we get

$$\left\|\sum_{j=1}^{m} \xi_{j} i_{n}(f_{j,n}) - \sum_{j=1}^{m} \xi_{j} i(f_{j})\right\|_{L^{2}(D)} \leq \sum_{j=1}^{m} |\xi_{j}| \|i_{n}(f_{j,n}) - i(f_{j})\|_{L^{2}(D)}$$
$$\leq \sum_{j=1}^{m} \|i_{n}(f_{j,n}) - i(f_{j})\|_{L^{2}(D)}$$
$$\to 0$$

as  $n \to \infty$ . Notice that the above convergence does not depend on  $\xi$ . This means  $\|\sum_{j=1}^{m} \xi_j f_{j,n}\|_{L^2(\Omega_n)} \to \|\sum_{j=1}^{m} \xi_j f_j\|_{L^2(\Omega)}$  uniformly with respect to  $\xi \in \mathbb{R}^m$  with  $|\xi| = 1$ . Let

$$c := \inf \left\{ \left\| \sum_{j=1}^{m} \xi_j f_j \right\|_{L^2(\Omega)} : \xi = (\xi_1, \dots, \xi_m) \in \mathbb{R}^m, |\xi| = 1 \right\}.$$

In particular, choosing  $\zeta > 0$  such that  $c - \zeta > 0$ , there exists  $N_0 \in \mathbb{N}$  (independent of  $\xi \in \mathbb{R}^m$  with  $|\xi| = 1$ ) such that

$$\left\|\sum_{j=1}^{m}\xi_{j}f_{j,n}\right\|_{L^{2}(\Omega_{n})} \geq \left\|\sum_{j=1}^{m}\xi_{j}f_{j}\right\|_{L^{2}(\Omega)} - \zeta,$$

for all  $n > N_0$  and for all  $\xi \in \mathbb{R}^m$  with  $|\xi| = 1$ . Since  $\|\sum_{j=1}^m \xi_j f_j\|_{L^2(\Omega)} \ge c$ , it follows that

$$\left\|\sum_{j=1}^{m} \xi_{j} f_{j,n}\right\|_{L^{2}(\Omega_{n})} \ge c - \zeta,$$
(3.29)

for all  $n > N_0$  and for all  $\xi \in \mathbb{R}^m$  with  $|\xi| = 1$ . Taking the infimum over  $\xi \in \mathbb{R}^m$  with  $|\xi| = 1$ , we obtian

$$c_n \ge c - \zeta > 0,$$

for all  $n \ge N_0$ . Finally, taking  $\hat{c} := \min\{c_1, \ldots, c_{N_0}, c - \zeta\}$ , the lemma follows.  $\Box$ 

An immediate application of Lemma 3.4.3 is the following result.

**Corollary 3.4.4.** Assume that  $V_n$  and V are as in Lemma 3.4.3 and that the convergence of bases  $i_n(f_{j,n}) \to i(f_j)$  in  $L^2(D)$  as  $n \to \infty$  holds for all  $j = 1, \ldots, m$ . Let  $u_n$  be a sequence such that  $u_n \in V_n$  for each  $n \in \mathbb{N}$ . If  $||u_n||_{L^2(\Omega_n)}$  is uniformly bounded then there exists a subsequence  $u_{n_k}$  such that  $i_{n_k}(u_{n_k}) \to i(u)$  in  $L^2(D)$  with a limit  $u \in V$ .

*Proof.* For each  $n \in \mathbb{N}$ , we write  $u_n = \sum_{j=1}^m \xi_{j,n} f_{j,n}$ . By a standard argument in the proof of equivalence of norms for finite dimensional spaces,

$$\sum_{j=1}^{m} |\xi_{j,n}| \le \frac{m}{c_n} ||u_n||_{L^2(\Omega_n)},$$

for all  $n \in \mathbb{N}$ , where  $c_n$  is given in Lemma 3.4.3. It follows from the uniform boundedness of  $||u_n||_{L^2(\Omega_n)}$  and Lemma 3.4.3 that  $\sum_{j=1}^m |\xi_{j,n}|$  is uniform bounded. We can extract a subsequence  $\xi_{j,n_k}$  such that  $\xi_{j,n_k} \to \xi_j$  for all  $j = 1, \ldots, m$ . Hence,

$$i_{n_k}(u_{n_k}) \to i(u) := i\Big(\sum_{j=1}^m \xi_j f_j\Big)$$

in  $L^2(D)$ .

Recall that we have  $\dim(X_n^+) = \dim(X^+) < \infty$  for sufficiently large n. We set  $d := \dim(X^+)$  and fix a certain basis  $\{f_1, f_2, \ldots, f_d\}$  of  $X^+$ . Define

$$f_{j,n} := P_n^+ r_n i(f_j), (3.30)$$

for j = 1, ..., d. Then we obtain a basis of  $X_n^+$  as shown below.

**Theorem 3.4.5.** There exists  $N_0 \in \mathbb{N}$  such that  $\{f_{1,n}, f_{2,n}, \ldots, f_{d,n}\}$  where  $f_{j,n}$  defined by (3.30) is a basis of  $X_n^+$  for each  $n > N_0$ . Moreover,  $i_n(f_{j,n}) \to i(f_j)$  in  $L^2(D)$  as  $n \to \infty$  holds for all  $j = 1, \ldots, d$ .

Proof. The convergence  $i_n(f_{j,n}) \to i(f_j)$  is clear from the definition of  $f_{j,n}$  and (3.23). Since  $X_n^+$  is d-dimensional subspace for all n sufficiently large, it suffices to show that there exists  $N_0 \in \mathbb{N}$  such that  $f_{1,n}, f_{2,n}, \ldots, f_{d,n}$  are linearly independent for each  $n > N_0$ . We prove this by using mathematical induction on m for  $m = 1, \ldots, d$  in the following statement: there exists  $N_m \in \mathbb{N}$  such that  $f_{1,n}, f_{2,n}, \ldots, f_{m,n}$  are linearly independent for each  $n > N_m$ .

The statement is trivial for m = 1. For the induction step, suppose that the statement is true for  $1, \ldots, m$  with m < d, but there is no  $N_{m+1} \in \mathbb{N}$  such that  $f_{1,n}, f_{2,n}, \ldots, f_{m+1,n}$  are linearly independent for each  $n > N_{m+1}$ . This implies that for each  $n \in \mathbb{N}$ , there exists  $l \in \mathbb{N}$  with l > n such that  $f_{1,l}, f_{2,l}, \ldots, f_{m+1,l}$  are linearly dependent. Thus, we can find a subsequence  $n_k$  (choosing  $n_k > N_m$  for all  $k \in \mathbb{N}$ ) such that  $f_{1,n_k}, f_{2,n_k}, \ldots, f_{m+1,n_k}$  are linearly dependent. Since  $f_{1,n_k}, f_{2,n_k}, \ldots, f_{m,n_k}$  are linearly independent, we can write

$$f_{m+1,n_k} = \sum_{j=1}^m \xi_{j,n_k} f_{j,n_k},$$

for all  $k \in \mathbb{N}$ . Since  $i_{n_k}(f_{m+1,n_k}) \to i(f_{m+1})$  in  $L^2(D)$  as  $k \to \infty$ , it follows that  $\|f_{m+1,n_k}\|_{L^2(\Omega_{n_k})}$  is uniformly bounded. Corollary 3.4.4 implies that there exists a subsequence denoted again by  $f_{m+1,n_k}$  such that  $i_{n_k}(f_{m+1,n_k}) \to i(f)$  in  $L^2(D)$  as  $k \to \infty$ , where the limit f belongs to the m-dimensional subspace spanned by  $f_1, f_2, \ldots, f_m$ . By the uniqueness of a limit, we conclude that  $f_{m+1} = f$ . This is a contradiction to the assumption that  $\{f_1, f_2, \ldots, f_d\}$  is a basis of  $X^+$ . Hence, the induction statement is true for m+1 and the theorem is proved.

As a consequence, we obtain the following convergence of a bounded sequence with each term belongs to a sequence of the spaces  $X_n^+$ .

**Corollary 3.4.6.** Let  $(w_n)_{n \in \mathbb{N}}$  be a sequence with  $w_n \in X_n^+$  for each  $n \in \mathbb{N}$ . If  $\|w_n\|_{L^2(\Omega_n)}$  (or  $\|w_n\|_{X_n^+}$ ) is uniformly bounded, then there exists a subsequence  $w_{n_k}$  such that

$$i_{n_k}(w_{n_k}) \to i(w)$$

in  $L^2(D)$  with the limit  $w \in X^+$ .

*Proof.* Note that  $||w_n||_{X_n^+} < C$  for all  $n \in \mathbb{N}$  implies  $||w_n||_{L^2(\Omega_n)} < C$  for all  $n \in \mathbb{N}$  (see (3.27)). The result follows immediately from Corollary 3.4.4 and Theorem 3.4.5.

Remark 3.4.7. The above result implies that there exists a subsequence  $w_{n_k}$  such that  $\|w_{n_k}\|_{L^2(\Omega_n)} \to \|w\|_{L^2(\Omega)}$  but does not implies  $\|w_{n_k}\|_{X_n^+} \to \|w\|_{X^+}$  as degenerate semigroup only converges uniformly on compact subsets of  $(0, \infty)$ .

We note that the convergence of components in finite dimensional subspaces in Lemma 3.4.1 (ii) is indeed a strong convergence as stated below.

**Lemma 3.4.8.** Suppose that the assumptions in Lemma 3.4.1 (ii) are satisfied. If  $i_n\phi_n \rightarrow i\phi$  in  $L^2(D)$  weakly, then  $i_n\phi_n^s \rightarrow i\phi^s$  in  $L^2(D)$  weakly and  $i_n\phi_n^* \rightarrow i\phi^*$  in  $L^2(D)$  strongly for \* = c, u.

Proof. From Lemma 3.4.1, we have a weak convergence  $i_n \phi_n^* \rightharpoonup i \phi^*$  in  $L^2(D)$  for \* = s, c, u. Since  $X_n^u$  is a finite dimensional subspace, Corollary 3.4.6 implies that for any subsequence  $n_k$  we can extract a further convergent subsequence  $i_{n_k} \phi_{n_k}^u \rightarrow \xi^u$  in  $L^2(D)$  for some  $\xi^u \in L^2(D)$ . The weak convergence above implies that  $\xi^u = i \phi^u$  and hence  $i_{n_k} \phi_{n_k}^u \rightarrow i \phi^u$  in  $L^2(D)$  strongly. Since this is true for any subsequence, we conclude that the whole sequence  $i_n \phi_n^u$  converges to  $i \phi^u$  in  $L^2(D)$  strongly. The same argument shows that  $i_n(x_n^c) \rightarrow i(x^c)$  in  $L^2(D)$ .

#### 3.4.2 Characterisation of upper and lower semicontinuity

We give some equivalent statements for upper and lower semicontinuity mentioned in Theorem 3.3.3 and Theorem 3.3.4. We simplify the notations by considering bounded subsets  $\mathscr{A}_n, \mathscr{A}$  of  $L^2(D)$ .

**Lemma 3.4.9** (Characterisation of upper semicontinuity). *The following statements are equivalent.* 

- (i)  $\sup_{v \in \mathscr{A}_n} \inf_{u \in \mathscr{A}} \|v u\|_{L^2(D)} \to 0 \text{ as } n \to \infty.$
- (ii) For any sequence  $\{v_n\}_{n\in\mathbb{N}}$  with  $v_n \in \mathscr{A}_n$ , we have  $\inf_{u\in\mathscr{A}} ||v_n u||_{L^2(D)} \to 0$  as  $n \to \infty$ .
- (iii) For any sequence  $\{v_n\}_{n\in\mathbb{N}}$  with  $v_n \in \mathscr{A}_n$ , if  $\{v_{n_k}\}_{k\in\mathbb{N}}$  is a subsequence, then there exist a further subsequence (denoted again by  $v_{n_k}$ ) and a sequence  $\{u_{n_k}\}_{k\in\mathbb{N}}$  with  $u_{n_k} \in \mathscr{A}$  such that  $\|v_{n_k} u_{n_k}\|_{L^2(D)} \to 0$  as  $k \to \infty$ .

*Proof.* The statement (i)  $\Rightarrow$  (ii) is clear as

$$\inf_{u \in \mathscr{A}} \|v_n - u\|_{L^2(D)} \le \sup_{v \in \mathscr{A}_n} \inf_{u \in \mathscr{A}} \|v - u\|_{L^2(D)},$$

for any sequence  $\{v_n\}_{n\in\mathbb{N}}$  with  $v_n\in\mathscr{A}_n$ .

For (ii)  $\Rightarrow$  (i), we prove by contrapositive. Suppose that (i) fails. Then

$$\limsup_{n \to \infty} \left\{ \sup_{v \in \mathscr{A}_n} \inf_{u \in \mathscr{A}} \|v - u\|_{L^2(D)} \right\} =: a > 0.$$

Hence, there exists a subsequence  $n_k \to \infty$  such that

$$\sup_{v \in \mathscr{A}_{n_k}} \inf_{u \in \mathscr{A}} \|v - u\|_{L^2(D)} \to a,$$

as  $k \to \infty$ . This implies that there exists  $v_{n_k} \in \mathscr{A}_{n_k}$  such that

$$\inf_{u \in \mathscr{A}} \|v_{n_k} - u\|_{L^2(D)} > a/2,$$

for all  $k \in \mathbb{N}$ . Hence, (ii) fails.

For the statement (ii)  $\Leftrightarrow$  (iii), notice first that  $\inf_{u \in \mathscr{A}} \|v_n - u\|_{L^2(D)} \to 0$  as  $n \to \infty$ if and only if there exists  $u_n \in \mathscr{A}$  such that  $\|v_n - u_n\|_{L^2(D)} \to 0$  as  $n \to \infty$ . To see this, we choose  $u_n \in \mathscr{A}$  such that  $\|v_n - u_n\|_{L^2(D)} < \inf_{u \in \mathscr{A}} \|v_n - u\|_{L^2(D)} + 1/n$  for each  $n \in \mathbb{N}$ . Then the forward implication follows. The backward implication is clear as  $\inf_{u \in \mathscr{A}} \|v_n - u\|_{L^2(D)} < \|v_n - u_n\|_{L^2(D)}$  for all  $u_n \in \mathscr{A}$ . The statement (ii)  $\Leftrightarrow$  (iii) then simply follows from the above and a standard subsequence characterisation of a limit.

By a similar argument, we can state the following lemma.

Lemma 3.4.10 (Characterisation of lower semicontinuity). The following statements are equivalent.

- (i)  $\sup_{u \in \mathscr{A}} \inf_{v \in \mathscr{A}_n} \|v u\|_{L^2(D)} \to 0 \text{ as } n \to \infty.$
- (ii) For any sequence  $\{u_n\}_{n\in\mathbb{N}}$  with  $u_n \in \mathscr{A}$ , we have  $\inf_{v\in\mathscr{A}_n} \|v-u_n\|_{L^2(D)} \to 0$  as  $n \to \infty$ .
- (iii) For any sequence {u<sub>n</sub>}<sub>n∈ℕ</sub> with u<sub>n</sub> ∈ A, if {u<sub>nk</sub>}<sub>k∈ℕ</sub> is a subsequence, then there exist a further subsequence (denoted again by u<sub>nk</sub>) and a sequence {v<sub>nk</sub>}<sub>k∈ℕ</sub> with v<sub>nk</sub> ∈ A<sub>nk</sub> such that ||v<sub>nk</sub> u<sub>nk</sub>||<sub>L<sup>2</sup>(D)</sub> → 0 as k → ∞.

#### 3.5 Convergence of unstable invariant manifolds

In this section, we prove upper and lower semicontinuity of local unstable invariant manifolds. We first show pointwise convergence of global unstable manifolds for the modified systems in Section 3.5.1. Consequently, we prove Theorem 3.3.3 in Section 3.5.2.

#### 3.5.1 Convergence of global unstable manifolds

Let

$$Y_n = \{h \in C(X_n^+, X_n^-) : h(0) = 0 \text{ and } h \text{ is } \nu^{-1}\text{-Lipschitz } \}.$$

Then  $Y_n$  is a complete metric space with the norm

$$\|h\|_{\text{Lip}} = \sup_{w \neq 0} \frac{\|h(w)\|_{X_n^-}}{\|w\|_{X_n^+}}.$$
(3.31)

We define  $T_{t,n}: Y_n \to Y_n$  for  $t \ge 0$  by

 $T_{t,n}(h) = \tilde{h},$ 

where  $\tilde{h} \in Y_n$  such that graph $(\tilde{h}) = \Phi_{t,n}(\operatorname{graph}(h))$ . Fix t > 0 sufficiently large such that

$$K := \nu(\nu - \mu)^{-1} \exp((\alpha - \beta + \varepsilon(2 + \mu + \nu^{-1}))t) < 1.$$
(3.32)

As in Theorem 3.2.5,  $T_{t,n}$  is a contraction on  $Y_n$  with a uniform contraction constant K for all  $n \in \mathbb{N}$ . Moreover,  $W_n^+$  is a graph of the fixed point  $h_n^+$  of  $T_{t,n}$ . To prove convergence of global unstable manifolds, we show that the fixed point  $h_n^+$  of  $T_{t,n}$  converges to the fixed point  $h^+$  of  $T_t$ .

**Lemma 3.5.1.** Suppose that  $\Omega_n$  and  $\Omega$  satisfy Assumption 1.4.1 and  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco. Then for every  $v \in X^-$ , there exists  $v_n \in X_n^-$  such that  $i_n(v_n) \to i(v)$  in  $L^2(D)$ .

*Proof.* Let  $v \in X^- \subset L^2(\Omega)$ . By the density of  $H_0^1(\Omega)$  in  $L^2(\Omega)$  and Mosco convergence, it follows from a standard diagonal procedure that there exists  $\xi_n \in H_0^1(\Omega_n)$  such that

$$i_n(\xi_n) \to i(v)$$

in  $L^2(D)$  as  $n \to \infty$ . By Lemma 3.4.1 (i), we get  $i_n P_n^-(\xi_n) \to iP^-(v) = i(v)$  in  $L^2(D)$  as  $n \to \infty$ . By taking  $v_n := P_n^-(\xi_n)$ , the lemma follows.

Let us define  $h \in Y$  by

$$h(w) := \frac{1}{C}h^{+}(w), \qquad (3.33)$$

for all  $w \in X^+$  where C is a positive constant satisfying

$$\|P^+\|\|1 - P_n^+\|M_1M_2 \le C, (3.34)$$

for all  $n \in \mathbb{N}$ . Note that although  $i_n(1 - P_n^+)r_n$  does not converge to  $i(1 - P^+)r$  in  $\mathscr{L}(L^2(D))$  with respect to the operator norm, we use  $||1 - P_n^+|| \le 1 + ||P_n^+||$  and (3.23) to obtain a bound C above. We will use that both  $||P_n^+||$  and  $||P_n^-|| = ||1 - P_n^+||$  are bounded without further notice.

In the next lemma, we obtain an approximation of h by functions in  $Y_n$ .

**Lemma 3.5.2.** Let h be as in (3.33). There exists a sequence  $\{h_n\}$  with  $h_n \in Y_n$  for each  $n \in \mathbb{N}$  such that

- (i)  $i_n h_n P_n^+ r_n(u) \to ih P^+ r(u)$  in  $L^2(D)$  as  $n \to \infty$  for all  $u \in L^2(D)$
- (ii) for each  $m \in \mathbb{N}$ , we have  $i_n T^m_{t,n}(h_n) P^+_n r_n(u) \to i T^m_t(h) P^+ r(u)$  in  $L^2(D)$  as  $n \to \infty$ for all  $u \in L^2(D)$ .

*Proof.* We construct  $h_n \in Y_n$  as follows. Define  $h_n : X_n^+ \to X_n^-$  by

$$h_n(w) := \frac{1}{C} (1 - P_n^+) r_n i h^+ P^+ r i_n(w), \qquad (3.35)$$

for  $w \in X_n^+$ . It is clear that  $h_n(0) = 0$ . Moreover, for  $w_1, w_2 \in X_n^+$ , it follows from the Lipschitz continuity of  $h^+$  and the choice of C in (3.34) that

$$\begin{split} \|h_{n}(w_{1}) - h_{n}(w_{2})\|_{X_{n}^{-}} \\ &\leq M_{1} \|h_{n}(w_{1}) - h_{n}(w_{2})\|_{L^{2}(\Omega_{n})} \\ &= M_{1} \Big\| \frac{1}{C} (1 - P_{n}^{+}) r_{n} i h^{+} P^{+} r i_{n}(w_{1}) - \frac{1}{C} (1 - P_{n}^{+}) r_{n} i h^{+} P^{+} r i_{n}(w_{2}) \Big\|_{L^{2}(\Omega_{n})} \\ &\leq M_{1} \frac{1}{C} \|1 - P_{n}^{+}\| \|r_{n}\| \|i\| \|h^{+} P^{+} r i_{n}(w_{1}) - h^{+} P^{+} r i_{n}(w_{2})\|_{L^{2}(\Omega)} \\ &\leq M_{1} \frac{1}{C} \|1 - P_{n}^{+}\| \|h^{+} P^{+} r i_{n}(w_{1}) - h^{+} P^{+} r i_{n}(w_{2})\|_{X^{-}} \\ &\leq M_{1} \frac{1}{C} \nu^{-1} \|1 - P_{n}^{+}\| \|P^{+} r i_{n}(w_{1}) - P^{+} r i_{n}(w_{2})\|_{X^{+}} \\ &\leq M_{1} \frac{1}{C} \nu^{-1} M_{2} \|1 - P_{n}^{+}\| \|P^{+} r i_{n}(w_{1}) - P^{+} r i_{n}(w_{2})\|_{L^{2}(\Omega)} \\ &\leq M_{1} \frac{1}{C} \nu^{-1} M_{2} \|1 - P_{n}^{+}\| \|P^{+}\| \|r\| \|i_{n}\| \|w_{1} - w_{2}\|_{L^{2}(\Omega_{n})} \\ &\leq \nu^{-1} \|w_{1} - w_{2}\|_{X_{n}^{+}}. \end{split}$$

Hence,  $h_n$  is  $\nu^{-1}$ -Lipschitz and thus  $h_n \in Y_n$ . Note that we need to be careful about the norm used in the above calculation. In particular, we take care of the equivalence of norms on  $X^-$  and  $X^+$  given in (3.10) and (3.11). This will be applied throughout this chapter without further comment.

We claim that  $h_n$  defined above satisfies the properties (i) and (ii). For (i), let  $u \in L^2(D)$  be arbitrary. By Lemma 3.5.1, there exists  $(v_n)_{n \in \mathbb{N}}$  with  $v_n \in X_n^-$  such that

$$i_n(v_n) \to ih^+ P^+ r(u) \tag{3.36}$$

in  $L^2(D)$  as  $n \to \infty$ . We have from the triangle inequality that

$$\begin{aligned} \|i_{n}h_{n}P_{n}^{+}r_{n}(u) - ihP^{+}r(u)\|_{L^{2}(D)} \\ &= \left\|\frac{1}{C}i_{n}(1-P_{n}^{+})r_{n}ih^{+}P^{+}ri_{n}P_{n}^{+}r_{n}(u) - \frac{1}{C}ih^{+}P^{+}r(u)\right\|_{L^{2}(D)} \\ &\leq \frac{1}{C}\|i_{n}(1-P_{n}^{+})r_{n}ih^{+}P^{+}ri_{n}P_{n}^{+}r_{n}(u) - i_{n}(1-P_{n}^{+})r_{n}ih^{+}P^{+}r(u)\|_{L^{2}(D)} \\ &+ \frac{1}{C}\|i_{n}(1-P_{n}^{+})r_{n}ih^{+}P^{+}r(u) - ih^{+}P^{+}r(u)\|_{L^{2}(D)}. \end{aligned}$$
(3.37)

Using the equivalence of norms on  $X^-$  and  $X^+$ , we can calculate

$$\frac{1}{C} \|i_n(1-P_n^+)r_nih^+P^+ri_nP_n^+r_n(u) - i_n(1-P_n^+)r_nih^+P^+r(u)\|_{L^2(D)} \\
\leq \frac{1}{C} \|i_n\| \|1-P_n^+\| \|i_nh^+P^+ri_nP_n^+r_n(u) - ih^+P^+riP^+r(u)\|_{L^2(D)} \\
\leq \frac{1}{C} \|1-P_n^+\| \|i_nh^+P^+ri_nP_n^+r_n(u) - ih^+P^+riP^+r(u)\|_{X^-} \\
\leq \frac{1}{C} \|1-P_n^+\| \|h^+P^+ri_nP_n^+r_n(u) - h^+P^+riP^+r(u)\|_{X^+} \\
\leq \frac{1}{C} \nu^{-1} \|1-P_n^+\| \|P^+ri_nP_n^+r_n(u) - P^+riP^+r(u)\|_{X^+} \\
\leq \frac{1}{C} \nu^{-1}M_2 \|1-P_n^+\| \|P^+ri_nP_n^+r_n(u) - P^+riP^+r(u)\|_{L^2(D)} \\
\leq \frac{1}{C} \nu^{-1}M_2 \|1-P_n^+\| \|P^+r\| \|i_nP_n^+r_n(u) - iP^+r(u)\|_{L^2(D)} \\
\to 0$$
(3.38)

as  $n \to \infty$ , where we use (3.23) and the boundedness of  $||1 - P_n^+||$  in the last step. For the second term on the right of (3.37), we use (3.36) and  $i_n(1 - P_n^+)r_ni_nv_n = i_nv_n$  to obtain

-1

$$\frac{1}{C} \|i_n(1-P_n^+)r_nih^+P^+r(u) - ih^+P^+r(u)\|_{L^2(D)} 
\leq \frac{1}{C} \|i_n(1-P_n^+)r_nih^+P^+r(u) - i_n(v_n)\|_{L^2(D)} 
+ \frac{1}{C} \|i_n(v_n) - ih^+P^+r(u)\|_{L^2(D)} 
= \frac{1}{C} \|i_n(1-P_n^+)r_nih^+P^+r(u) - i_n(1-P_n^+)r_ni_n(v_n)\|_{L^2(D)} 
+ \frac{1}{C} \|i_n(v_n) - ih^+P^+r(u)\|_{L^2(D)} 
\leq \frac{1}{C} \|i_n(1-P_n^+)r_n\| \|ih^+P^+r(u) - i_n(v_n)\|_{L^2(D)} 
+ \frac{1}{C} \|i_n(v_n) - ih^+P^+r(u)\|_{L^2(D)} 
\rightarrow 0$$
(3.39)

as  $n \to \infty$ . It follows from (3.37) - (3.39) that

$$||i_n h_n P_n^+ r_n(u) - ihP^+ r(u)||_{L^2(D)} \to 0$$

as  $n \to \infty$ . Since the above argument is valid for any  $u \in L^2(D)$ , statement (i) follows.

We next prove (ii) by induction on  $m \in \mathbb{N}$ . By part (i) of this proof, the property (ii) is true for m = 0. For induction step, assume that

$$i_n T^m_{t,n}(h_n) P^+_n r_n(u) \to i T^m_t(h) P^+ r(u)$$

in  $L^2(D)$  as  $n \to \infty$  for all  $u \in L^2(D)$  holds true for  $m = 0, 1, \ldots, k$ . We need to show that

$$i_n T_{t,n}^{k+1}(h_n) P_n^+ r_n(u) \to i T_t^{k+1}(h) P^+ r(u)$$
 (3.40)

in  $L^2(D)$  as  $n \to \infty$  for all  $u \in L^2(D)$ . Let  $u \in L^2(D)$  be arbitrary. We set  $w := P^+r(u) \in X^+$  and  $w_n := P^+_n r_n(u) \in X^+_n$ . It follows from (3.23) that

$$i_n(w_n) \to i(w) \tag{3.41}$$

in  $L^2(D)$  as  $n \to \infty$ . Since graph $(T_t^{k+1}(h)) = \Phi_t(\operatorname{graph}(T_t^k(h)))$ , there exists  $w_0 \in X^+$  such that

$$\Phi_t\left(w_0\oplus T_t^k(h)(w_0)\right)=w\oplus T_t^{k+1}(h)(w).$$

For each  $n \in \mathbb{N}$ , we define  $w_{0,n} := P_n^+ r_n i(w_0)$ . Again by (3.23), we have  $i_n(w_{0,n}) \to i(w_0)$ in  $L^2(D)$  as  $n \to \infty$ . Moreover, by induction hypothesis,

$$i_n T_{t,n}^k(h_n)(w_{0,n}) = i_n T_{t,n}^k(h_n) P_n^+ r_n i(w_0) \to i T_t^k(h) P^+ r i(w_0) = i T_t^k(h)(w_0)$$
in  $L^2(D)$  as  $n \to \infty$ . Hence, it follows from the convergence of solutions under domain perturbation in (3.20) that

$$i_n\Phi_{t,n}\left(w_{0,n}\oplus T_{t,n}^k(h_n)(w_{0,n})\right)\to i\Phi_t\left(w_0\oplus T_t^k(h)(w_0)\right)=i\left(w\oplus T_t^{k+1}(h)(w)\right)$$

in  $L^2(D)$  as  $n \to \infty$ . Since graph $(T_{t,n}^{k+1}(h_n)) = \Phi_{t,n}(\operatorname{graph}(T_{t,n}^k(h_n)))$ , there exists  $\xi_n \in X_n^+$  such that

$$\Phi_{t,n}\left(w_{0,n}\oplus T_{t,n}^{k}(h_{n})(w_{0,n})\right) = \xi_{n}\oplus T_{t,n}^{k+1}(h_{n})(\xi_{n}),$$

for each  $n \in \mathbb{N}$ . Hence,

$$i_n\left(\xi_n \oplus T_{t,n}^{k+1}(h_n)(\xi_n)\right) \to i\left(w \oplus T_t^{k+1}(h)(w)\right)$$
(3.42)

in  $L^2(D)$  as  $n \to \infty$ . By Lemma 3.4.1 (i), it follows from (3.42) that

$$i_n(\xi_n) \to i(w) \tag{3.43}$$

and

$$i_n\left(T_{t,n}^{k+1}(h_n)(\xi_n)\right) \to i\left(T_t^{k+1}(h)(w)\right)$$
(3.44)

in  $L^2(D)$  as  $n \to \infty$ . We obtain from (3.41) and (3.43) that

$$||i_n(\xi_n) - i_n(w_n)||_{L^2(D)} \to 0$$

as  $n \to \infty$ . Since  $T_{t,n}^{k+1}(h_n)$  is  $\nu^{-1}$ -Lipschitz, it follows that

$$\begin{aligned} \left\| T_{t,n}^{k+1}(h_n)(\xi_n) - T_{t,n}^{k+1}(h_n)(w_n) \right\|_{L^2(\Omega_n)} &\leq \left\| T_{t,n}^{k+1}(h_n)(\xi_n) - T_{t,n}^{k+1}(h_n)(w_n) \right\|_{X_n^-} \\ &\leq \nu^{-1} \|\xi_n - w_n\|_{X_n^+} \\ &\leq \nu^{-1} M_2 \|\xi_n - w_n\|_{L^2(\Omega_n)} \\ &\to 0 \end{aligned}$$
(3.45)

as  $n \to \infty$ . By definitions of  $w_n$  and w together with (3.44) and (3.45), we conclude that

$$\begin{split} \left\| i_n T_{t,n}^{k+1}(h_n) P_n^+ r_n(u) - i T_t^{k+1}(h) P^+ r(u) \right\|_{L^2(D)} \\ &= \left\| i_n T_{t,n}^{k+1}(h_n)(w_n) - i T_t^{k+1}(h)(w) \right\|_{L^2(D)} \\ &\leq \left\| i_n T_{t,n}^{k+1}(h_n)(w_n) - i_n T_{t,n}^{k+1}(h_n)(\xi_n) \right\|_{L^2(D)} \\ &+ \left\| i_n T_{t,n}^{k+1}(h_n)(\xi_n) - i T_t^{k+1}(h)(w) \right\|_{L^2(D)} \\ &\to 0 \end{split}$$

as  $n \to \infty$ . As  $u \in L^2(D)$  was arbitrary, we have shown (3.40).

We next recall a standard result on the rate of convergence to the fixed point of a contraction mapping below (see for example [74, Remark 1.2.3 (ii)]).

**Lemma 3.5.3.** Let (X, d) be a complete metric space and  $T : X \to X$  be a contraction mapping with a contraction constant  $k \in [0, 1)$ , that is,

$$d(Tx_1, Tx_2) \le kd(x_1, x_2),$$

for all  $x_1, x_2 \in X$ . Denote by  $x^*$  the unique fixed point of T. Then for any  $x \in X$ ,

$$d(x^*, T^n x) \le \frac{k^n}{1-k} d(x, Tx),$$
(3.46)

for all  $n \in \mathbb{N}$ .

We prove the pointwise convergence of global unstable invariant manifolds in the following theorem.

**Theorem 3.5.4.** Assume that all assumptions in Theorem 3.3.3 are satisfied and  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco. Then we have

$$i_n h_n^+ P_n^+ r_n(u) \to i h^+ P^+ r(u)$$

in  $L^2(D)$  as  $n \to \infty$  for all  $u \in L^2(D)$ .

*Proof.* Fix  $u \in L^2(D)$  and let  $\zeta > 0$  be arbitrary. We can choose  $m_0 \in \mathbb{N}$  independent of n such that the contraction constant K in (3.32) satisfies

$$\frac{K^{m_0}}{1-K} 2\nu^{-1} \|P_n^+ r_n(u)\|_{X_n^+} \le \frac{\zeta}{3},\tag{3.47}$$

for all  $n \in \mathbb{N}$  and

$$\frac{K^{m_0}}{1-K} 2\nu^{-1} \|P^+ r(u)\|_{X^+} \le \frac{\zeta}{3}.$$
(3.48)

We take  $h_n \in Y_n$  and  $h \in Y$  as in Lemma 3.5.2. Then by the definition of Lip-norm on

Y and  $Y_n$  (see (3.16) and (3.31), respectively), we see that

$$\begin{split} \|i_{n}h_{n}^{+}P_{n}^{+}r_{n}(u) - ih^{+}P^{+}r(u)\|_{L^{2}(D)} \\ &\leq \|i_{n}h_{n}^{+}P_{n}^{+}r_{n}(u) - i_{n}T_{t,n}^{m_{0}}(h_{n})P_{n}^{+}r_{n}(u)\|_{L^{2}(D)} \\ &+ \|i_{n}T_{t,n}^{m_{0}}(h_{n})P_{n}^{+}r_{n}(u) - iT_{t}^{m_{0}}(h)P^{+}r(u)\|_{L^{2}(D)} \\ &+ \|iT_{t}^{m_{0}}(h)P^{+}r(u) - ih^{+}P^{+}r(u)\|_{L^{2}(D)} \\ &\leq \|h_{n}^{+}P_{n}^{+}r_{n}(u) - T_{t,n}^{m_{0}}(h_{n})P_{n}^{+}r_{n}(u)\|_{X_{n}^{-}} \\ &+ \|i_{n}T_{t,n}^{m_{0}}(h_{n})P_{n}^{+}r_{n}(u) - iT_{t}^{m_{0}}(h)P^{+}r(u)\|_{L^{2}(D)} \\ &+ \|T_{t}^{m_{0}}(h)P^{+}r(u) - h^{+}P^{+}r(u)\|_{X_{n}^{-}} \\ &\leq \|h_{n}^{+} - T_{t,n}^{m_{0}}(h_{n})\|_{\mathrm{Lip}}\|P_{n}^{+}r_{n}(u)\|_{X_{n}^{+}} \\ &+ \|i_{n}T_{t,n}^{m_{0}}(h_{n})P_{n}^{+}r_{n}(u) - iT_{t}^{m_{0}}(h)P^{+}r(u)\|_{L^{2}(D)} \\ &+ \|T_{t}^{m_{0}}(h) - h^{+}\|_{\mathrm{Lip}}\|P^{+}r(u)\|_{X^{+}}, \end{split}$$

for all  $n \in \mathbb{N}$ . By Lemma 3.5.3, we have

$$\|h^{+} - T_{t}^{m_{0}}(h)\|_{\text{Lip}} \leq \frac{K^{m_{0}}}{1 - K} \|h - T_{t}(h)\|_{\text{Lip}} \leq \frac{K^{m_{0}}}{1 - K} 2\nu^{-1}.$$
 (3.50)

Similarly,

$$\|h_n^+ - T_{t,n}^{m_0}(h_n)\|_{\text{Lip}} \le \frac{K^{m_0}}{1-K} \|h_n - T_{t,n}(h_n)\|_{\text{Lip}} \le \frac{K^{m_0}}{1-K} 2\nu^{-1},$$
(3.51)

for all  $n \in \mathbb{N}$ . Moreover, Lemma 3.5.2 (ii) implies that there exists  $N_0 \in \mathbb{N}$  such that

$$\left\|i_n T_{t,n}^{m_0}(h_n) P_n^+ r_n(u) - i T_t^{m_0}(h) P^+ r(u)\right\|_{L^2(D)} \le \frac{\zeta}{3},\tag{3.52}$$

for all  $n > N_0$ . It follows from (3.49) – (3.52) that

$$\begin{aligned} \left\| i_n h_n^+ P_n^+ r_n(u) - i h^+ P^+ r(u) \right\|_{L^2(D)} &\leq \frac{K^{m_0}}{1 - K} 2\nu^{-1} \|P_n^+ r_n(u)\|_{X_n^+} + \frac{\zeta}{3} \\ &+ \frac{K^{m_0}}{1 - K} 2\nu^{-1} \|P^+ r(u)\|_{X^+}, \end{aligned}$$

for all  $n > N_0$ . By our choice of  $m_0$  in (3.47) and (3.48), we conclude that

$$\|i_n h_n^+ P_n^+ r_n(u) - ih^+ P^+ r(u)\|_{L^2(D)} \le \zeta,$$

for all  $n > N_0$ . As  $\zeta > 0$  was arbitrary, we get

$$i_n h_n^+ P_n^+ r_n(u) \to i h^+ P^+ r(u)$$

in  $L^2(D)$  as  $n \to \infty$ . Since this argument works for any  $u \in L^2(D)$ , the statement of the theorem follows.

#### 3.5.2 Upper and lower semicontinuity of local unstable manifolds

We are now in the position to prove Theorem 3.3.3.

Proof of Theorem 3.3.3 (ii). As discussed at the end of Section 3.3, there exist  $\delta_1$  and  $\delta_2$  such that  $W_n^u = W_n^u(U_n)$  is a local unstable invariant manifold where  $U_n = V_{1,n} \times V_{2,n}$  with  $V_{1,n}$  is a ball of radius  $\delta_1$  in  $X_n^-$  and  $V_{2,n}$  is a ball of radius  $\delta_2$  in  $X_n^+$  for all  $n \in \mathbb{N}$ . Moreover, a similar statement holds for the unperturbed problem. By the equivalence of norms on  $X_n^-$  and  $X_n^+$  with uniform parameters  $\alpha$  and  $\beta$ , we can chose  $\delta > 0$  such that  $B_n := B_{L^2(\Omega_n)}(0, \delta) \subset V_{1,n} \times V_{2,n}$  for all  $n \in \mathbb{N}$  and  $B := B_{L^2(\Omega)}(0, \delta) \subset V_1 \times V_2$ .

To prove the lower semicontinuity, we show that for every  $\zeta > 0$ , there exists  $N_0 \in \mathbb{N}$ independent of  $u \in \operatorname{graph}(h^+) \cap B$  such that

$$\inf_{v \in \operatorname{graph}(h_n^+) \cap B_n} \|i(u) - i_n(v)\|_{L^2(D)} < \zeta,$$

for all  $n > N_0$  and for all  $u \in \operatorname{graph}(h^+) \cap B$ . Let  $\zeta > 0$  be arbitrary. By the Lipschitz continuity of  $h^+: X^+ \to X^-$  (taking (3.10) and (3.11) into account), we have that for every  $w_0 \in X^+$ , there exists  $\rho > 0$  such that

$$\|(w \oplus h^+(w)) - (w_0 \oplus h^+(w_0))\|_{L^2(\Omega)} < \frac{\zeta}{2},$$
(3.53)

for all  $w \in B_{X^+}(w_0, \rho) := \{w \in X^+ : \|w - w_0\|_{L^2(\Omega)} < \rho\}$ . Note that  $\rho$  is independent of  $w_0 \in X^+$ . We set

$$W := P^+ (graph(h^+) \cap B) = \{ w \in X^+ : w \oplus h^+(w) \in B \}.$$

Since dim $(X^+) < \infty$ , the set  $\overline{W}$  is compact. If we take the open cover  $\{B_{X^+}(w,\rho) : w \in W\}$  of  $\overline{W}$ , then there exists a finite subcover

$$\{B_{X^+}(w_k,\rho): w_k \in W, k = 1, \dots, m\}.$$

Hence,

$$W \subset \bigcup_{k=1}^{m} B_{X^+}(w_k, \rho).$$
(3.54)

Denoted by  $\Delta := \min\{\delta - \|w_k \oplus h^+(w_k)\|_{L^2(\Omega)} : k = 1, \ldots, m\}$ . Setting  $w_{k,n} := P_n^+ r_n i(w_k) \in X_n^+$  for  $n \in \mathbb{N}$  and  $k = 1, \ldots, m$ . We have from (3.23) that  $i_n(w_{k,n}) \to i(w_k)$  in  $L^2(D)$  as  $n \to \infty$  for each  $k = 1, \ldots, m$ . Moreover, by Theorem 3.5.4  $i_n h_n^+(w_{k,n}) \to ih^+(w_k)$  in  $L^2(D)$  as  $n \to \infty$  for each  $k = 1, \ldots, m$ . Hence, for each  $k = 1, \ldots, m$  we can find  $N_k \in \mathbb{N}$  such that

$$\|i_n(w_{k,n} \oplus h_n^+(w_{k,n})) - i(w_k \oplus h^+(w_k))\|_{L^2(D)} < \min\left\{\frac{\zeta}{2}, \Delta\right\},\$$

for all  $n > N_k$ . Setting  $N_0 = \max\{N_k : k = 1, \dots, m\}$ , we get

$$\|i_n(w_{k,n} \oplus h_n^+(w_{k,n})) - i(w_k \oplus h^+(w_k))\|_{L^2(D)} < \min\left\{\frac{\zeta}{2}, \Delta\right\},$$
(3.55)

for all  $n > N_0$  and for all k = 1, ..., m. Using (3.55), we obtain

$$\begin{split} \|w_{k,n} \oplus h_n^+(w_{k,n})\|_{L^2(\Omega_n)} &\leq \|i_n(w_{k,n} \oplus h_n^+(w_{k,n})) - i(w_k \oplus h^+(w_k))\|_{L^2(D)} \\ &+ \|w_k \oplus h_n^+(w_k)\|_{L^2(\Omega)} \\ &< \|w_k \oplus h_n^+(w_k)\|_{L^2(\Omega)} + \Delta \\ &\leq \|w_k \oplus h_n^+(w_k)\|_{L^2(\Omega)} + (\delta - \|w_k \oplus h_n^+(w_k)\|_{L^2(\Omega)}) \\ &= \delta, \end{split}$$

for all  $n > N_0$  and for all k = 1, ..., m. Hence,  $w_{k,n} \oplus h_n^+(w_{k,n}) \in \operatorname{graph}(h_n^+) \cap B_n$  for all  $n > N_0$  and for all k = 1, ..., m. Let u be in  $\operatorname{graph}(h^+) \cap B$  and write  $u = w \oplus h^+(w)$  for some  $w \in W$ . By (3.54), there exists  $k \in \{1, ..., m\}$  such that  $w \in B_{X^+}(w_k, \rho)$ . It follows from (3.53) and (3.55) that

$$\begin{split} \|i_{n}(w_{k,n} \oplus h_{n}^{+}(w_{k,n})) - i(w \oplus h^{+}(w))\|_{L^{2}(D)} \\ &\leq \|i_{n}(w_{k,n} \oplus h_{n}^{+}(w_{k,n})) - i(w_{k} \oplus h^{+}(w_{k}))\|_{L^{2}(D)} \\ &+ \|i(w_{k} \oplus h^{+}(w_{k})) - i(w \oplus h^{+}(w))\|_{L^{2}(D)} \\ &< \frac{\zeta}{2} + \frac{\zeta}{2} \\ &= \zeta, \end{split}$$

for all  $n > N_0$ . Since  $w_{k,n} \oplus h_n^+(w_{k,n}) \in \operatorname{graph}(h_n^+) \cap B_n$  for all  $n > N_0$ , we get

$$\inf_{v\in \operatorname{graph}(h_n^+)\cap B_n} \|i(u)-i_n(v)\|_{L^2(D)} < \zeta,$$

for all  $n > N_0$ . The above estimate holds for every  $u = w \oplus h^+(w) \in \operatorname{graph}(h^+) \cap B$ and notice that  $N_0$  is independent of u. Hence,

$$\inf_{v \in \operatorname{graph}(h_n^+) \cap B_n} \|i(u) - i_n(v)\|_{L^2(D)} < \zeta,$$

for all  $n > N_0$  and for all  $u \in \operatorname{graph}(h^+) \cap B$ . As  $\zeta > 0$  was arbitrary, we obtain the lower semicontinuity.

Using our characterisation in Lemma 3.4.9, we can show the upper semicontinuity of unstable invariant manifolds.

Proof of Theorem 3.3.3 (i). We consider the same neighbourhood  $B_n$  and B as in the proof above. By Lemma 3.4.9, we need to show that for any sequence  $\{\xi_n\}_{n\in\mathbb{N}}$  with  $\xi_n \in \operatorname{graph}(h_n^+) \cap B_n$ , if  $\{\xi_{n_k}\}_{k\in\mathbb{N}}$  is a subsequence then there exist a further subsequence (denoted again by  $\xi_{n_k}$ ) and a sequence  $\{u_{n_k}\}_{k\in\mathbb{N}}$  with  $u_{n_k} \in \operatorname{graph}(h^+) \cap B$  such that  $\|i_{n_k}(\xi_{n_k}) - i(u_{n_k})\|_{L^2(D)} \to 0$  as  $k \to \infty$ .

Let  $\{\xi_n\}_{n\in\mathbb{N}}$  be a sequence with  $\xi_n \in \operatorname{graph}(h_n^+) \cap B_n$  and  $(\xi_{n_k})_{k\in\mathbb{N}}$  be an arbitrary subsequence. We write  $\xi_{n_k} := w_{n_k} \oplus h_{n_k}^+(w_{n_k})$  for some  $w_{n_k} \in X_{n_k}^+$ . Since  $\|\xi_{n_k}\|_{L^2(\Omega_{n_k})} =$  $\|w_{n_k} \oplus h_{n_k}^+(w_{n_k})\|_{L^2(\Omega_{n_k})} < \delta$  for all  $k \in \mathbb{N}$ , we can apply Corollary 3.4.6 to extract a subsequence from  $w_{n_k}$  (indexed again by  $n_k$ ) such that

$$i_{n_k}(w_{n_k}) \to i(w)$$

in  $L^2(D)$  with the limit  $w \in X^+$ . Hence, by the Lipschitz continuity of  $h_n^+$  and Theorem 3.5.4, we get

$$\begin{aligned} \|i_{n_k}h_{n_k}^+(w_{n_k}) - ih^+(w)\|_{L^2(D)} &\leq \|i_{n_k}h_{n_k}^+(w_{n_k}) - i_{n_k}h_{n_k}^+P_{n_k}^+r_{n_k}i(w)\|_{L^2(D)} \\ &+ \|i_{n_k}h_{n_k}^+P_{n_k}^+r_{n_k}i(w) - ih^+(w)\|_{L^2(D)} \\ &\to 0 \end{aligned}$$

as  $k \to \infty$ . If we set  $u := w \oplus h^+(w) \in \operatorname{graph}(h^+)$ , then  $i_{n_k}(\xi_{n_k}) \to i(u)$  in  $L^2(D)$ as  $k \to \infty$ . Since  $\|i_{n_k}(\xi_{n_k})\|_{L^2(D)} < \delta$  for all  $k \in \mathbb{N}$ , we get  $\|i(u)\|_{L^2(D)} \leq \delta$ . Hence,  $u \in \operatorname{graph}(h^+) \cap \overline{B} = \overline{\operatorname{graph}(h^+) \cap B}$ . We can find  $u_{n_k} \in \operatorname{graph}(h^+) \cap B$  such that  $u_{n_k} \to u$  in  $L^2(\Omega)$  as  $k \to \infty$ . Therefore,

$$\|i_{n_k}(\xi_{n_k}) - i(u_{n_k})\|_{L^2(D)} \le \|i_{n_k}(\xi_{n_k}) - i(u)\|_{L^2(D)} + \|i(u) - i(u_{n_k})\|_{L^2(D)} \to 0$$

as  $k \to \infty$  and we obtain the required subsequence.

### **3.6** Convergence of stable invariant manifolds

Recall that the local stable manifold is a graph of Lipschitz function  $h^-: X^- \to X^+$ inside a suitable product neighbourhood of  $0 \in L^2(\Omega)$  determined by the modification in the construction (Theorem 3.2.7). In this section, we prove the upper and lower semicontinuity of local stable invariant manifolds with the following modification.

Fix the renorming of  $X_n^-$ ,  $X_n^+$ ,  $X^-$  and  $X^+$  by (3.25), (3.26), (3.8) and (3.9) respectively using the same parameters  $\alpha$  and  $\beta$  for all  $n \in \mathbb{N}$ . By shrinking the neighbourhood (choosing a smaller Lipschitz constant  $\varepsilon$  for the nonlinear terms  $f_n$  and f), we can make the following assumption.

Assumption 3.6.1. We assume that

$$0 < \mu_0 < \inf\left\{\frac{1}{2(\|P^+\| + \|P^-\|)}, \frac{1}{2(\|P_n^+\| + \|P_n^-\|)} : n \in \mathbb{N}\right\}$$
(3.56)

and

$$\mu := \frac{\mu_0}{M_1 M_2} \tag{3.57}$$

are parameters such that both  $\mu_0$  and  $\mu$  satisfy the conditions for  $\mu$  in (3.14) and (3.15).

We denote the Lipschitz functions for the modification  $\mu_0$  by  $\hat{h}^-$  and for the modification  $\mu$  by  $h^-$ . Let U be a smaller product neighbourhood of 0 in  $L^2(\Omega)$  such that both modifications agree. Hence, the local stable manifold is  $W^s(U) := \operatorname{graph}(h^-) \cap U =$  $\operatorname{graph}(\hat{h}^-) \cap U$ . Similarly, for each  $n \in \mathbb{N}$ , we denote the Lipschitz functions for the modification  $\mu_0$  by  $\hat{h}_n^-$  and for the modification  $\mu$  by  $h_n^-$ . As discussed at the end of Section 3.3, we can take a uniform product neighbourhood  $U_n$  of 0 in  $L^2(\Omega_n)$  such that both modifications agree. Hence, the local stable manifold is  $W_n^s(U_n) := \operatorname{graph}(h_n^-) \cap U_n =$  $\operatorname{graph}(\hat{h}_n^-) \cap U_n$ . We choose  $\delta > 0$  so that  $\overline{B} \subset U$  and  $\overline{B}_n \subset U_n$ , where  $B := B_{L^2(\Omega)}(0, \delta)$ and  $B_n := B_{L^2(\Omega_n)}(0, \delta)$ . Hence,  $h^-(v) = \hat{h}^-(v)$  on  $\overline{B}$  and  $h_n^-(v) = \hat{h}_n^-(v)$  on  $\overline{B}_n$ . We prove Theorem 3.3.4 by taking the balls of radius  $\delta$  chosen above.

**Lemma 3.6.2.** Let  $\delta > 0$  and  $\zeta_n > 0$  be a sequence with  $\zeta_n \to 0$  as  $n \to \infty$ .

(i) If  $z_n = y_n \oplus h^-(y_n)$  is a sequence in graph $(h^-)$  such that  $z_n \in B_{L^2(\Omega)}(0, \delta + \zeta_n)$  for each  $n \in \mathbb{N}$ , then there exist a subsequence  $z_{n_k}$  and a sequence  $u_{n_k}$  in graph $(h^-) \cap B$ such that

$$||z_{n_k} - u_{n_k}||_{L^2(\Omega)} \to 0$$

as  $k \to \infty$ .

(ii) If  $z_n = y_n \oplus h^-(y_n)$  is a sequence with  $z_n \in \operatorname{graph}(h_n^-) \cap B_{L^2(\Omega_n)}(0, \delta + \zeta_n)$  for each  $n \in \mathbb{N}$ , then there exist a subsequence  $z_{n_k}$  and a sequence  $u_{n_k}$  with  $u_{n_k} \in \operatorname{graph}(h_{n_k}^-) \cap B_{n_k}$  for each  $k \in \mathbb{N}$  such that

$$||z_{n_k} - u_{n_k}||_{L^2(\Omega_{n_k})} \to 0$$

as  $k \to \infty$ .

*Proof.* For assertion (i), using (3.56) we can fix b > 0 such that

$$b > \frac{1}{(\|P^+\| + \|P^-\|)^{-1} - 2\mu_0}.$$
(3.58)

Since  $\zeta_n \to 0$ , we can find  $N_0 \in \mathbb{N}$  such that  $\zeta_n < \delta/b$  for all  $n > N_0$ . We extract a subsequence  $\zeta_{n_k}$  so that  $\zeta_{n_k} < \delta/b$  for all  $k \in \mathbb{N}$ . Define

$$a_{n_k} := 1 - \frac{b\zeta_{n_k}}{\|y_{n_k}\|_{L^2(\Omega)} + \|h^-(y_{n_k})\|_{L^2(\Omega)}},$$
(3.59)

for each  $k \in \mathbb{N}$ . By our assumptions,  $\|z_{n_k}\|_{L^2(\Omega)} = \|y_{n_k} \oplus h^-(y_{n_k})\|_{L^2(\Omega)} < \delta + \zeta_{n_k}$  for all  $k \in \mathbb{N}$ . If  $\|z_{n_k}\|_{L^2(\Omega)} \ge \delta$ , then

$$\|y_{n_k}\|_{L^2(\Omega)} + \|h^-(y_{n_k})\|_{L^2(\Omega)} \ge \|y_{n_k} \oplus h^-(y_{n_k})\|_{L^2(\Omega)} \ge \delta.$$

Since  $\zeta_{n_k} < \delta/b$ , if  $||z_{n_k}||_{L^2(\Omega)} \ge \delta$  we have that

$$\frac{b\zeta_{n_k}}{\|y_{n_k}\|_{L^2(\Omega)} + \|h^-(y_{n_k})\|_{L^2(\Omega)}} < \frac{b(\delta/b)}{\delta} = 1$$

It follows from (3.59) that  $0 < a_{n_k} \leq 1$  if  $||z_{n_k}||_{L^2(\Omega)} \geq \delta$ . For each  $k \in \mathbb{N}$ , we define  $u_{n_k} \in \operatorname{graph}(h^-)$  by

$$u_{n_k} := \begin{cases} z_{n_k} & \text{if } \|z_{n_k}\|_{L^2(\Omega)} < \delta \\ a_{n_k} y_{n_k} \oplus h^-(a_{n_k} y_{n_k}) & \text{if } \|z_{n_k}\|_{L^2(\Omega)} \ge \delta. \end{cases}$$
(3.60)

Clearly,  $||z_{n_k} - u_{n_k}||_{L^2(\Omega)} = 0$  if  $||z_{n_k}||_{L^2(\Omega)} < \delta$ . Moreover, if  $||z_{n_k}||_{L^2(\Omega)} \ge \delta$ , then

$$\begin{split} \|z_{n_{k}} - u_{n_{k}}\|_{L^{2}(\Omega)} &= \|(y_{n_{k}} \oplus h^{-}(y_{n_{k}})) - (a_{n_{k}}y_{n_{k}} \oplus h^{-}(a_{n_{k}}y_{n_{k}}))\|_{L^{2}(\Omega)} \\ &\leq \|y_{n_{k}} - a_{n_{k}}y_{n_{k}}\|_{L^{2}(\Omega)} + \|h^{-}(y_{n_{k}}) - h^{-}(a_{n_{k}}y_{n_{k}})\|_{L^{2}(\Omega)} \\ &\leq \|y_{n_{k}} - a_{n_{k}}y_{n_{k}}\|_{L^{2}(\Omega)} + \|h^{-}(y_{n_{k}}) - h^{-}(a_{n_{k}}y_{n_{k}})\|_{X^{+}} \\ &\leq \|y_{n_{k}} - a_{n_{k}}y_{n_{k}}\|_{L^{2}(\Omega)} + \mu\|y_{n_{k}} - a_{n_{k}}y_{n_{k}}\|_{X^{-}} \\ &\leq (1 + \mu M_{1})\|y_{n_{k}} - a_{n_{k}}y_{n_{k}}\|_{L^{2}(\Omega)} \\ &\leq (1 + \mu M_{1})|1 - a_{n_{k}}| \|y_{n_{k}}\|_{L^{2}(\Omega)} \\ &\leq (1 + \mu M_{1})\frac{b\zeta_{n_{k}}}{\|y_{n_{k}}\|_{L^{2}(\Omega)} + \|h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)}}\|y_{n_{k}}\|_{L^{2}(\Omega)} \\ &\leq (1 + \mu M_{1})b\zeta_{n_{k}}. \end{split}$$

Hence,  $||z_{n_k} - u_{n_k}||_{L^2(\Omega)} \le (1 + \mu M_1) b\zeta_{n_k}$  for all  $k \in \mathbb{N}$ . As  $\zeta_{n_k} \to 0$ , we conclude that

$$\|z_{n_k} - u_{n_k}\|_{L^2(\Omega)} \to 0 \tag{3.61}$$

as  $k \to \infty$ . It remains to show that  $u_{n_k} \in B_{L^2(\Omega)}(0, \delta)$  for all  $k \in \mathbb{N}$ . If  $||z_{n_k}||_{L^2(\Omega)} < \delta$ ,

then  $u_{n_k} \in B_{L^2(\Omega)}(0, \delta)$ . If  $||z_{n_k}||_{L^2(\Omega)} \ge \delta$ , we can write

$$\begin{aligned} \|u_{n_{k}}\|_{L^{2}(\Omega)} &\leq \|u_{n_{k}} - a_{n_{k}} z_{n_{k}}\|_{L^{2}(\Omega)} + \|a_{n_{k}} z_{n_{k}}\|_{L^{2}(\Omega)} \\ &= \|(a_{n_{k}} y_{n_{k}} \oplus h^{-}(a_{n_{k}} y_{n_{k}})) - a_{n_{k}}(y_{n_{k}} \oplus h^{-}(y_{n_{k}}))\|_{L^{2}(\Omega)} \\ &+ \|a_{n_{k}} z_{n_{k}}\|_{L^{2}(\Omega)} \\ &\leq \|h^{-}(a_{n_{k}} y_{n_{k}}) - a_{n_{k}}h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)} + \|a_{n_{k}} z_{n_{k}}\|_{L^{2}(\Omega)} \\ &\leq \|h^{-}(a_{n_{k}} y_{n_{k}}) - h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)} + \|h^{-}(y_{n_{k}}) - a_{n_{k}}h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)} \\ &+ \|a_{n_{k}} z_{n_{k}}\|_{L^{2}(\Omega)}. \end{aligned}$$

$$(3.62)$$

Now, if  $||z_{n_k}||_{L^2(\Omega)} \ge \delta$ , then by the Lipschitz continuity of  $h^-$  and (3.57)

$$\|h^{-}(a_{n_{k}}y_{n_{k}}) - h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)} \leq \|h^{-}(a_{n_{k}}y_{n_{k}}) - h^{-}(y_{n_{k}})\|_{X^{+}}$$

$$\leq \mu \|a_{n_{k}}y_{n_{k}} - y_{n_{k}}\|_{X^{-}}$$

$$\leq \mu M_{1}|a_{n_{k}} - 1| \|y_{n_{k}}\|_{L^{2}(\Omega)}$$

$$= \frac{\mu_{0}}{M_{1}M_{2}}M_{1}\frac{b\zeta_{n_{k}}\|y_{n_{k}}\|_{L^{2}(\Omega)}}{\|y_{n_{k}}\|_{L^{2}(\Omega)} + \|h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)}}$$

$$\leq \mu_{0}b\zeta_{n_{k}}.$$
(3.63)

Similarly, if  $||z_{n_k}||_{L^2(\Omega)} \ge \delta$ , then

$$\|h^{-}(y_{n_{k}}) - a_{n_{k}}h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)} \leq |1 - a_{n_{k}}| \|h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)}$$

$$\leq |1 - a_{n_{k}}| \|h^{-}(y_{n_{k}})\|_{X^{+}}$$

$$\leq \mu |1 - a_{n_{k}}| \|y_{n_{k}}\|_{X^{-}}$$

$$\leq \mu M_{1}|1 - a_{n_{k}}| \|y_{n_{k}}\|_{L^{2}(\Omega)}$$

$$= \frac{\mu_{0}}{M_{1}M_{2}}M_{1}\frac{b\zeta_{n_{k}}\|y_{n_{k}}\|_{L^{2}(\Omega)}}{\|y_{n_{k}}\|_{L^{2}(\Omega)} + \|h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)}}$$

$$\leq \mu_{0}b\zeta_{n_{k}}.$$
(3.64)

Since  $||z_{n_k}||_{L^2(\Omega)} \ge (||y_{n_k}||_{L^2(\Omega)} + ||h^-(y_{n_k})||_{L^2(\Omega)})/(||P^+|| + ||P^-||)$ , it follows that

$$\frac{b\zeta_{n_k}\|z_{n_k}\|_{L^2(\Omega)}}{\|y_{n_k}\|_{L^2(\Omega)} + \|h^-(y_{n_k})\|_{L^2(\Omega)}} \ge \frac{b\zeta_{n_k}}{\|P^+\| + \|P^-\|}.$$

Hence, if  $||z_{n_k}||_{L^2(\Omega)} \ge \delta$ , then

$$\begin{aligned} \|a_{n_{k}}z_{n_{k}}\|_{L^{2}(\Omega)} &= |a_{n_{k}}|\|z_{n_{k}}\|_{L^{2}(\Omega)} \\ &= a_{n_{k}}\|z_{n_{k}}\|_{L^{2}(\Omega)} \\ &= \left(1 - \frac{b\zeta_{n_{k}}}{\|y_{n_{k}}\|_{L^{2}(\Omega)} + \|h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)}}\right) \|z_{n_{k}}\|_{L^{2}(\Omega)} \\ &= \|z_{n_{k}}\|_{L^{2}(\Omega)} - \frac{b\zeta_{n_{k}}\|z_{n_{k}}\|_{L^{2}(\Omega)}}{\|y_{n_{k}}\|_{L^{2}(\Omega)} + \|h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)}} \\ &\leq \|z_{n_{k}}\|_{L^{2}(\Omega)} - \frac{b\zeta_{n_{k}}}{\|P^{+}\| + \|P^{-}\|} \\ &< \delta + \zeta_{n_{k}} - \frac{b\zeta_{n_{k}}}{\|P^{+}\| + \|P^{-}\|}. \end{aligned}$$

$$(3.65)$$

Therefore, by (3.62) – (3.65), if  $||z_{n_k}||_{L^2(\Omega)} \ge \delta$ , then

$$\|u_{n_k}\|_{L^2(\Omega)} < \mu_0 b \zeta_{n_k} + \mu_0 b \zeta_{n_k} + \delta + \zeta_{n_k} - \frac{b \zeta_{n_k}}{\|P^+\| + \|P^-\|}$$
  
=  $\delta + \left(2\mu_0 b - \frac{b}{\|P^+\| + \|P^-\|} + 1\right) \zeta_{n_k}.$  (3.66)

By the choice of b in (3.58), we get

$$2\mu_0 b - \frac{b}{\|P^+\| + \|P^-\|} + 1 = \left(2\mu_0 - \frac{1}{\|P^+\| + \|P^-\|}\right)b + 1$$
$$= -\left((\|P^+\| + \|P^-\|)^{-1} - 2\mu_0\right)b + 1$$
$$< -1 + 1$$
$$= 0.$$

It follows from (3.66) that  $||u_{n_k}||_{L^2(\Omega)} < \delta$  if  $||z_{n_k}||_{L^2(\Omega)} \ge \delta$ . Hence, we conclude that  $u_{n_k} \in \operatorname{graph}(h^-) \cap B_{L^2(\Omega)}(0, \delta)$  for all  $k \in \mathbb{N}$  and statement (i) follows.

Statement (ii) can be proved similarly. The only difference is that the sequence  $z_n$  belongs to different spaces  $L^2(\Omega_n)$  for each  $n \in \mathbb{N}$ . We only need to adjust the proof in part (i) and keep track of the dependence on n. In particular, we replace (3.58) by

$$b > \sup_{n \in \mathbb{N}} \left\{ \frac{1}{(\|P_n^+\| + \|P_n^-\|)^{-1} - 2\mu_0} \right\} > 0$$

and (3.59) by

$$a_{n_k} := 1 - \frac{b\zeta_{n_k}}{\|y_{n_k}\|_{L^2(\Omega_{n_k})} + \|h^-(y_{n_k})\|_{L^2(\Omega_{n_k})}},$$

for each  $k \in \mathbb{N}$ .

We now show the upper semicontinuity of local stable invariant manifolds.

Proof of Theorem 3.3.4 (i). By Lemma 3.4.9, we need to show that for any sequence  $\{\xi_n\}_{n\in\mathbb{N}}$  with  $\xi_n \in \operatorname{graph}(h_n^-) \cap B_n$ , if  $\{\xi_{n_k}\}_{k\in\mathbb{N}}$  is a subsequence then there exist a further subsequence (denoted again by  $\xi_{n_k}$ ) and a sequence  $\{u_{n_k}\}_{k\in\mathbb{N}}$  with  $u_{n_k} \in \operatorname{graph}(h^-) \cap B$  such that  $\|i_{n_k}(\xi_{n_k}) - i(u_{n_k})\|_{L^2(D)} \to 0$  as  $k \to \infty$ .

Let  $\{\xi_n\}_{n\in\mathbb{N}}$  be a sequence with  $\xi_n \in \operatorname{graph}(h_n^-) \cap B_n$  and  $(\xi_{n_k})_{k\in\mathbb{N}}$  be an arbitrary subsequence. We write  $\xi_{n_k} := v_{n_k} \oplus h_{n_k}^-(v_{n_k})$  for some  $v_{n_k} \in X_{n_k}^-$ . Since  $\|\xi_{n_k}\|_{L^2(\Omega_{n_k})} =$  $\|v_{n_k} \oplus h_{n_k}^-(v_{n_k})\|_{L^2(\Omega_{n_k})} < \delta$  for all  $k \in \mathbb{N}$ , we can extract a subsequence of  $v_{n_k}$  (indexed again by  $n_k$ ) such that

$$i_{n_k}(v_{n_k}) \rightharpoonup v$$

in  $L^2(D)$  as  $k \to \infty$ . By the assumption that  $|\Omega_n| \to |\Omega|$ , we conclude that v = 0 almost everywhere in  $D \setminus \Omega$ , that is,  $v \in L^2(\Omega)$ . Hence,

$$i_{n_k}(v_{n_k}) \rightharpoonup i(v) \tag{3.67}$$

in  $L^2(D)$  as  $k \to \infty$ . Moreover, by the convergence of  $i_n P_n^- r_n \to i P^- r$  in  $\mathscr{L}(L^2(D))$ (see Remark 3.4.2) and the weak convergence of  $v_{n_k}$ , it follows that

$$\begin{aligned} \left| \left( i_{n_k}(v_{n_k}) - iP^-(v) | \phi \right)_{L^2(D)} \right| &\leq \left| \left( i_{n_k}(v_{n_k}) - iP^-ri_{n_k}(v_{n_k}) | \phi \right)_{L^2(D)} \right| \\ &+ \left| \left( iP^-ri_{n_k}(v_{n_k}) - iP^-(v) | \phi \right)_{L^2(D)} \right| \\ &\leq \| i_{n_k} P_{n_k}^- r_{n_k} - iP^-r\| \| i_{n_k}(v_{n_k}) \|_{L^2(D)} \| \phi \|_{L^2(D)} \\ &+ \left| \left( i_{n_k}(v_{n_k}) - i(v) | (iP^-r)^* \phi \right)_{L^2(D)} \right| \\ &\to 0 \end{aligned}$$

as  $k \to \infty$  for all  $\phi \in L^2(D)$ , where  $(iP^-r)^*$  is the adjoint operator of  $iP^-r$ . This means  $i_{n_k}(v_{n_k}) \rightharpoonup iP^-(v)$  in  $L^2(D)$  as  $k \to \infty$ . By the uniqueness of weak limit,  $v = P^-(v)$  and hence  $v \in X^-$ . Since  $\|i_{n_k}h_{n_k}^-(v_{n_k})\|_{L^2(D)}$  is uniformly bounded, we can apply Corollary 3.4.6 to extract a further subsequence (indexed again by  $n_k$ ) such that

$$i_{n_k} h_{n_k}^-(v_{n_k}) \to i(w)$$
 (3.68)

in  $L^2(D)$  as  $k \to \infty$  with the limit  $w \in X^+$ . Thus, we get

$$i_{n_k}(v_{n_k} \oplus h_{n_k}^-(v_{n_k})) \rightharpoonup i(v \oplus w) \tag{3.69}$$

in  $L^2(D)$  as  $k \to \infty$ . By a standard property of weak convergence,

$$\|i(v \oplus w)\|_{L^{2}(D)} \leq \liminf_{k \to \infty} \|i_{n_{k}}(v_{n_{k}} \oplus h_{n_{k}}^{-}(v_{n_{k}}))\|_{L^{2}(D)}$$
  
$$\leq \limsup_{k \to \infty} \|i_{n_{k}}(v_{n_{k}} \oplus h_{n_{k}}^{-}(v_{n_{k}}))\|_{L^{2}(D)}$$
  
$$\leq \delta.$$
(3.70)

Hence,  $u := v \oplus w$  belongs to  $\overline{B}$ . Applying domain perturbation result for solutions of semilinear equations in (3.20), we get from (3.69) and globally Lipschitz assumption for the modified function  $\tilde{f}$  that

$$i_{n_k}\Phi_{t,n_k}(v_{n_k}\oplus h_{n_k}^-(v_{n_k}))\to i\Phi_t(v\oplus w)$$

in  $L^2(D)$  as  $k \to \infty$  for all t > 0. Lemma 3.4.1 (i) implies that

$$i_{n_k}P^-_{n_k}\Phi_{t,n_k}(v_{n_k}\oplus h^-_{n_k}(v_{n_k}))\to iP^-\Phi_t(v\oplus w)$$

and

$$i_{n_k}P^+_{n_k}\Phi_{t,n_k}(v_{n_k}\oplus h^-_{n_k}(v_{n_k}))\to iP^+\Phi_t(v\oplus w)$$

in  $L^2(D)$  as  $k \to \infty$  for all t > 0. By the construction of  $h_{n_k}^-(v_{n_k})$  (see Theorem 3.2.4), we have that

$$\|P_{n_k}^+\Phi_{t,n_k}(v_{n_k}\oplus h_{n_k}^-(v_{n_k}))\|_{X_{n_k}^+} \le \mu \|P_{n_k}^-\Phi_{t,n_k}(v_{n_k}\oplus h_{n_k}^-(v_{n_k}))\|_{X_{n_k}^-},$$

for all  $t \ge 0$ . Taking the norms on  $X_n^+$  and  $X_n^-$  in (3.27) and (3.28) into account, the above implies

$$\|P_{n_k}^+\Phi_{t,n_k}(v_{n_k}\oplus h_{n_k}^-(v_{n_k}))\|_{L^2(\Omega_{n_k})} \le \mu M_1 \|P_{n_k}^-\Phi_{t,n_k}(v_{n_k}\oplus h_{n_k}^-(v_{n_k}))\|_{L^2(\Omega_{n_k})},$$

for all  $t \ge 0$ . Passing to the limit as  $k \to \infty$ , we obtain

$$\|P^{+}\Phi_{t}(v\oplus w)\|_{L^{2}(\Omega)} \leq \mu M_{1}\|P^{-}\Phi_{t}(v\oplus w)\|_{L^{2}(\Omega)}$$

for all t > 0. By the assumptions on  $\mu_0$  and  $\mu$  in (3.56) and (3.57), and the equivalence of norms on  $X^-$  and  $X^+$  (see (3.10) and (3.11)), it follows that

$$\|P^{+}\Phi_{t}(v\oplus w)\|_{X^{+}} \leq \mu M_{1}M_{2}\|P^{-}\Phi_{t}(v\oplus w)\|_{X^{-}} = \mu_{0}\|P^{-}\Phi_{t}(v\oplus w)\|_{X^{-}}, \quad (3.71)$$

for all t > 0. We claim that  $||w||_{X^+} \le \mu_0 ||v||_{X^-}$ . If  $||w||_{X^+} > \mu_0 ||v||_{X^-}$ , that is  $v \oplus w$  is in the interior of the cone  $K_{\mu_0}$  defined by (3.13), we can find a product neighbourhood U(v, w) of  $v \oplus w$  such that  $U(v, w) \subset \operatorname{Int}(K_{\mu_0})$ . Since the solution of parabolic equation with the initial condition  $v \oplus w$  is continuous, there exists  $t_0 > 0$  such that  $\Phi_t(v \oplus w) \in$ U(v, w) for  $0 \leq t \leq t_0$ . This implies that  $\|P^+\Phi_t(v \oplus w)\|_{X^+} > \mu_0\|P^-\Phi_t(v \oplus w)\|_{X^-}$ for  $0 \leq t \leq t_0$ , which is a contradiction to (3.71). Hence, by the definition of  $\hat{h}^-$  (a modification with the cone  $K_{\mu_0}$ ), we conclude that  $w = \hat{h}^-(v)$ . As both modification agree on  $\bar{B}$ , we have  $w = h^-(v)$ . Therefore, (3.68) implies

$$i_{n_k}h_{n_k}^-(v_{n_k}) \to ih^-(v)$$
 (3.72)

in  $L^2(D)$  as  $k \to \infty$ .

The remainder of this proof deals with the existence of the required sequence  $u_{n_k} \in \operatorname{graph}(h^-) \cap B$ . At this stage, we keep the index of our subsequence as in the previous part. We define

$$y_{n_k} := P^- ri_{n_k}(v_{n_k}) \in X^-,$$

for each  $k \in \mathbb{N}$ . By the convergence of  $i_n P_n^- r_n \to i P^- r$  in  $\mathscr{L}(L^2(D))$  (see Remark 3.4.2) and the boundedness of  $||i_{n_k}(v_{n_k})||_{L^2(D)}$ , we get

$$\|i(y_{n_k}) - i_{n_k}(v_{n_k})\|_{L^2(D)} = \|iP^- r i_{n_k}(v_{n_k}) - i_{n_k}(v_{n_k})\|_{L^2(D)}$$
  

$$\leq \|iP^- r - i_{n_k}P^-_{n_k}r_{n_k}\|\|i_{n_k}(v_{n_k})\|_{L^2(D)} \qquad (3.73)$$
  

$$\to 0$$

as  $k \to \infty$ . In particular,  $\|y_{n_k}\|_{L^2(\Omega)}$  is uniformly bounded. Moreover, by (3.73) and the weak convergence of  $v_{n_k}$ ,

$$\begin{aligned} \left| (i(y_{n_k}) - i(v)|\phi)_{L^2(D)} \right| &\leq \left| (i(y_{n_k}) - i_{n_k}(v_{n_k})|\phi)_{L^2(D)} \right| \\ &+ \left| (i_{n_k}(v_{n_k}) - i(v)|\phi)_{L^2(D)} \right| \\ &\leq \|i(y_{n_k}) - i_{n_k}(v_{n_k})\|_{L^2(D)} \|\phi\|_{L^2(D)} \\ &+ \left| (i_{n_k}(v_{n_k}) - i(v)|\phi)_{L^2(D)} \right| \\ &\to 0 \end{aligned}$$

as  $k \to \infty$  for all  $\phi \in L^2(D)$ . Hence,

$$i(y_{n_k}) \rightharpoonup i(v) \tag{3.74}$$

in  $L^2(D)$  as  $k \to \infty$ . By the Lipschitz continuity of  $h^-$  together with (3.10) and (3.11),

$$\|h^{-}(y_{n_{k}})\|_{L^{2}(\Omega)} \leq \|h^{-}(y_{n_{k}})\|_{X^{+}} \leq \mu \|y_{n_{k}}\|_{X^{-}} \leq \mu M_{1} \|y_{n_{k}}\|_{L^{2}(\Omega)} < \infty$$

for all  $k \in \mathbb{N}$ . Since  $X^+$  is a finite dimensional space, we can extract a further subsequence (indexed again by  $n_k$ ) such that

$$ih^-(y_{n_k}) \to i(\tilde{w})$$
 (3.75)

in  $L^2(D)$  as  $k \to \infty$  with the limit  $\tilde{w} \in X^+$ . Therefore,

$$i(y_{n_k} \oplus h^-(y_{n_k})) \rightharpoonup i(v \oplus \tilde{w}) \tag{3.76}$$

in  $L^2(D)$  as  $k \to \infty$ . By the convergence of solutions under domain perturbation in (3.20) (with  $\Omega_n = \Omega$  for all  $n \in \mathbb{N}$ ), we get from (3.76) that

$$i\Phi_t(y_{n_k}\oplus h^-(y_{n_k}))\to i\Phi_t(v\oplus \tilde{w})$$

in  $L^2(D)$  as  $k \to \infty$  for all t > 0. Hence,

$$P^{-}\Phi_t(y_{n_k} \oplus h^{-}(y_{n_k})) \to P^{-}\Phi_t(v \oplus \tilde{w})$$

and

$$P^+\Phi_t(y_{n_k}\oplus h^-(y_{n_k}))\to P^+\Phi_t(v\oplus\tilde{w})$$

in  $L^2(\Omega)$  as  $k \to \infty$  for all t > 0. Since these sequences are in the fixed spaces  $X^-$  and  $X^+$  respectively, (3.10) and (3.11) imply that they converge under  $\|\cdot\|_{X^-}$  and  $\|\cdot\|_{X^+}$ , respectively. By the construction of  $h^-(y_{n_k})$  (see Theorem 3.2.4), we have that

$$\|P^{+}\Phi_{t}(y_{n_{k}}\oplus h^{-}(y_{n_{k}}))\|_{X^{+}} \leq \mu \|P^{-}\Phi_{t}(y_{n_{k}}\oplus h^{-}(y_{n_{k}}))\|_{X^{-}},$$

for all  $t \ge 0$ . Passing to the limit as  $k \to \infty$ , we obtain

$$\|P^{+}\Phi_{t}(v\oplus\tilde{w})\|_{X^{+}} = \mu \|P^{-}\Phi_{t}(v\oplus\tilde{w})\|_{X^{-}}$$
(3.77)

for all t > 0. By a similar argument appeared after (3.71), we conclude that  $\|\tilde{w}\|_{X^+} \le \mu \|v\|_{X^-}$ . Hence,  $\tilde{w}$  agrees with  $w = h^-(v)$ . Therefore, (3.75) implies

$$ih^-(y_{n_k}) \to ih^-(v) \tag{3.78}$$

in  $L^2(D)$  as  $k \to \infty$ . Recall that  $\xi_{n_k} = v_{n_k} \oplus h_{n_k}^-(v_{n_k})$ . If we set  $z_{n_k} := y_{n_k} \oplus h^-(y_{n_k}) \in$ graph $(h^-)$ , then by (3.72), (3.73) and (3.78)

$$\begin{aligned} \|i_{n_{k}}(\xi_{n_{k}}) - i(z_{n_{k}})\|_{L^{2}(D)} \\ &= \|i_{n_{k}}(v_{n_{k}} \oplus h_{n_{k}}^{-}(v_{n_{k}})) - i(y_{n_{k}} \oplus h^{-}(y_{n_{k}}))\|_{L^{2}(D)} \\ &\leq \|i_{n_{k}}(v_{n_{k}}) - i(y_{n_{k}})\|_{L^{2}(D)} + \|i_{n_{k}}h_{n_{k}}^{-}(v_{n_{k}}) - ih^{-}(y_{n_{k}})\|_{L^{2}(D)} \\ &\leq \|i_{n_{k}}(v_{n_{k}}) - i(y_{n_{k}})\|_{L^{2}(D)} + \|i_{n_{k}}h_{n_{k}}^{-}(v_{n_{k}}) - ih^{-}(v)\|_{L^{2}(D)} \\ &+ \|ih^{-}(v) - ih^{-}(y_{n_{k}})\|_{L^{2}(D)} \\ &\rightarrow 0 \end{aligned}$$

$$(3.79)$$

as  $k \to \infty$ . Therefore, we can extract a further subsequence (indexed again by  $n_k$ ) and  $\zeta_{n_k} > 0$  with  $\zeta_{n_k} \to 0$  as  $k \to \infty$  such that

$$\|i_{n_k}(\xi_{n_k}) - i(z_{n_k})\|_{L^2(D)} < \zeta_{n_k}$$

for all  $k \in \mathbb{N}$ . It follows that

$$\|i(z_{n_k})\|_{L^2(D)} \le \|i_{n_k}(\xi_{n_k})\|_{L^2(D)} + \zeta_{n_k} < \delta + \zeta_{n_k}$$

for all  $k \in \mathbb{N}$ , that is,  $z_{n_k} \in \operatorname{graph}(h^-) \cap B_{L^2(\Omega)}(0, \delta + \zeta_{n_k})$  for all  $k \in \mathbb{N}$ . We can apply Lemma 3.6.2 (i) to obtain a subsequence (indexed again by  $n_k$ )  $z_{n_k}$  and a sequence  $u_{n_k} \in \operatorname{graph}(h^-) \cap B$  such that  $||z_{n_k} - u_{n_k}||_{L^2(\Omega)} \to 0$  as  $k \to \infty$ . It follows from (3.79) that

$$\|i_{n_k}(\xi_{n_k}) - i(u_{n_k})\|_{L^2(D)} \le \|i_{n_k}(\xi_{n_k}) - i(z_{n_k})\|_{L^2(D)} + \|i(z_{n_k}) - i(u_{n_k})\|_{L^2(D)}$$
  
$$\to 0$$

as  $k \to \infty$ . Hence, we obtain the required sequence  $u_{n_k}$ . Since we start with an arbitrary sequence  $\xi_n \in \operatorname{graph}(h_n^-) \cap B_n$ , the assertion of Theorem 3.3.4 (i) follows.

The lower semicontinuity of local stable invariant manifolds can be obtained by a similar fashion.

*Proof of Theorem 3.3.4 (ii).* The statement follows by a similar argument to the proof of Theorem 3.3.4 (i). We use Lemma 3.4.10 and Lemma 3.6.2 (ii) instead of Lemma 3.4.9 and 3.6.2 (i).  $\Box$ 

## Chapter 4

## Persistence of bounded entire solutions of parabolic equations under domain perturbation

In this chapter, we investigate domain perturbation for a class of bounded entire solutions (solutions that are defined for all time  $t \in (-\infty, \infty)$  and are bounded in a suitable function space) of parabolic equations. The existence of bounded entire solutions is commonly obtained from an *exponential dichotomy* assumption (see Definition 4.1.2 below). A study of persistence of bounded entire solutions under perturbation has also attracted interests. For example, the work of [83] deals with persistence of bounded solutions of parabolic equations under a non-autonomous perturbation of the inhomogeneous terms. Many other authors focus their studies on homoclinic and heteroclinic solutions to understand some complicated dynamical behaviour. For results on homoclinic solutions, we refer to [17] and [82]. The persistence of heteroclinic orbits of semilinear heat equation under approximation of a (non- $C^1$ ) nonlinearity by  $C^1$  functions is obtained in [33] to calculate the Conley index. A usual technique to deal with perturbation of the nonlinearity (or inhomogeneous terms) is the use of Liapunov-Schmidt decomposition method and the implicit function theorem to derive a bifurcation equation (see [50]). When dealing with domain perturbation, a standard implicit function theorem cannot be applied because the differential operators vary upon domains.

In the literature, domain perturbation for bounded entire solutions seems only be treated for particular types of solutions including stationary (equilibrium) solutions in [51] and periodic solutions in [35, Section 5 and 7]. For periodic solutions, convergence of *T*-periodic solutions under domain perturbation can be obtained by applying domain perturbation results for initial value problems (see Chapter 1) and considering the *Time-T-operator*, that is the operator U(T) mapping  $u_0$  to the solution of the abstract homogeneous parabolic equation ((2.2) with s = 0) at time *T*.

The purpose of this chapter is to study domain perturbation for general bounded entire solutions of linear and semilinear parabolic equations. We restrict our attention to autonomous parabolic equations under Dirichlet boundary condition. In Section 4.1, we recall the existence and uniqueness of bounded entire solutions of linear parabolic equations under an exponential dichotomy assumption. Persistence of bounded entire solutions is studied in Section 4.2 and Section 4.3 for linear parabolic equations under  $L^2$  and  $L^p$ -norms respectively. We consider semilinear equations in Section 4.4. The technique in this chapter is not applicable to domain perturbation for homoclinic and heteroclinic orbits. The problem is rather difficult but it is an interesting topic for future research.

### 4.1 Bounded entire solutions of linear parabolic equations

Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$ . We consider the following autonomous parabolic equations subject to Dirichlet boundary condition on the whole real line

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}u = g(x, t) & \text{in } \Omega \times (-\infty, \infty) \\ u = 0 & \text{on } \partial \Omega \times (-\infty, \infty), \end{cases}$$
(4.1)

where  $g \in C(\mathbb{R}^N \times \mathbb{R})$  and  $\mathcal{A}$  is an elliptic operator defined by (2.31). As usual, we write (4.1) in the abstract form in  $L^2(\Omega)$  as

$$u'(t) + Au = g(t), \quad t \in (-\infty, \infty), \tag{4.2}$$

where A is the maximal restriction operator of  $A_{\Omega}$  given in (2.37) on  $H = L^2(\Omega)$ . In particular, we have from Section 2.4.2 that -A is the generator of a strongly continuous analytic semigroup  $S(t), t \ge 0$  on  $L^2(\Omega)$ .

Let  $J \subset \mathbb{R}$  and X be a Banach space, we write

$$BC(J,X) := \Big\{ v : J \to X : v \text{ is continuous and } \sup_{t \in J} \|v(t)\|_X < \infty \Big\}.$$

It is well known that BC(J, X) is a Banach space under the norm  $\|\cdot\|_{\infty}$  given by

$$||v||_{\infty} := \sup_{t \in J} ||v(t)||_X.$$

**Definition 4.1.1.** A function  $u \in BC(\mathbb{R}, L^2(\Omega))$  is called a (mild) *bounded entire* solution of (4.2) if for all  $t \ge s$ 

$$u(t) = S(t-s)u(s) + \int_s^t S(t-\tau)g(\tau)d\tau$$

holds for any  $s \in \mathbb{R}$ .

To obtain existence of bounded entire solutions, we require the notion of *exponential dichotomy*. We give a definition below for a more general case of non-autonomous problems.

**Definition 4.1.2** (Exponential dichotomy (see e.g. [52, Section 7.6])). We say that the equation u'(t) + A(t)u(t) = 0 has an exponential dichotomy on interval  $J \subset \mathbb{R}$  with exponent  $\beta > 0$  and bound M if there exist projections  $P(t), t \in J$  strongly continuous in t such that the evolution systems  $U(t, s) \in \mathscr{L}(H), t \geq s$  satisfies

- (i) U(t,s)P(s) = P(t)U(t,s) for all  $t \ge s$  in J;
- (ii) The restriction  $U(t,s)|_{\mathscr{R}(P(s))}$  is an isomorphism from  $\mathscr{R}(P(s))$  onto  $\mathscr{R}(P(t))$  for all  $t \ge s$  in J where  $\mathscr{R}(P(s))$  denotes the range of P(s). We define U(s,t) as the inverse map from  $\mathscr{R}(P(t))$  to  $\mathscr{R}(P(s))$ ;
- (iii)  $||U(t,s)(I P(s))|| \le Me^{-\beta(t-s)}$  for all  $t \ge s$  in J;
- (iv)  $||U(t,s)P(s)|| \le Me^{\beta(t-s)}$  for all  $s \ge t$  in J.

Note that for autonomous problems, evolution systems are just semigroups  $e^{-A(t-s)}$ for  $t \ge s$ . In this case, the projections P(t) is replaced by a constant projection P.

Example 4.1.3. Consider the autonomous parabolic equation u'(t) + Au(t) = 0 in a Banach space  $X := L^2(\Omega)$  where -A is a generator of a strongly continuous analytic semigroup  $S(t), t \ge 0$ . The evolution systems are U(t,s) := S(t-s) for  $t \ge s$ . The equation u'(t) + Au(t) = 0 has an exponential dichotomy on  $\mathbb{R}$  if the spectrum  $\sigma(-A)$ can be decomposed as  $\sigma(-A) = \sigma^+ \cup \sigma^-$  where

$$\sigma^{+} := \{\lambda \in \sigma(-A) : \operatorname{Re}(\lambda) > 0\}$$
  
$$\sigma^{-} := \{\lambda \in \sigma(-A) : \operatorname{Re}(\lambda) < 0\}.$$
(4.3)

Indeed, the exponent  $\beta$  can be determined by the value of  $\beta$  such that  $\sigma(A)$  is disjoint from the strip  $\{\lambda : -\beta \leq \operatorname{Re}(\lambda) \leq \beta\}$ . As seen in Section 3.2.2,  $\sigma^+$  is a finite set. Let  $P^+$  be the spectral projection defined by

$$P^{+} = \frac{1}{2\pi i} \int_{\Gamma} (\lambda + A)^{-1} d\lambda, \qquad (4.4)$$

where  $\Gamma$  is a rectifiable closed curve separating  $\sigma^+$  from  $\sigma^-$ . We can take constant dichotomy projections  $P(t) := P^+$  and  $I - P(t) := I - P^+$  for all  $t \in \mathbb{R}$ .

In fact, the above conditions imply exponential dichotomy in a fractional power space (see examples in [52]).

We make the following assumption.

Assumption 4.1.4. We assume that the corresponding homogeneous equation of (4.2), that is u'(t) + Au(t) = 0 has an exponential dichotomy on  $\mathbb{R}$  with exponent  $\beta > 0$  and bound M.

We next prove the existence and uniqueness of bounded entire solutions of (4.2).

**Theorem 4.1.5** ([69, Theorem 4]). Suppose that Assumption 4.1.4 is satisfied. For every  $g \in BC(\mathbb{R}, L^2(\Omega))$ , there exists a unique bounded entire solution  $u \in BC(\mathbb{R}, L^2(\Omega))$  of (4.2). Moreover, u can be represented by

$$u(t) = \int_{-\infty}^{t} S(t-\tau)(I-P)g(\tau)d\tau - \int_{t}^{\infty} S(t-\tau)Pg(\tau)d\tau, \qquad (4.5)$$

for all  $t \in \mathbb{R}$ .

*Proof.* Let  $g \in BC(\mathbb{R}, L^2(\Omega))$  be arbitrary. Consider u(t) defined by (4.5). Applying the estimates (iii) and (iv) in the definition of exponential dichotomy, we see that

$$\|u(t)\|_{L^{2}(\Omega)} \leq \int_{-\infty}^{t} M e^{-\beta(t-\tau)} \|g\|_{\infty} d\tau + \int_{t}^{\infty} M e^{\beta(t-\tau)} \|g\|_{\infty} d\tau \leq \frac{2M}{\beta} \|g\|_{\infty},$$

for all  $t \in \mathbb{R}$ . By the change of variables, we can write (4.5) as

$$u(t) = \int_0^\infty S(\tau)(I - P)g(t - \tau)d\tau - \int_{-\infty}^0 S(\tau)Pg(t - \tau)d\tau,$$
 (4.6)

for all  $t \in \mathbb{R}$ . From (4.6), it is easy to see (using the dominated convergence theorem) that u is continuous on  $\mathbb{R}$ . Hence,  $u \in BC(\mathbb{R}, L^2(\Omega))$ . For  $t \ge s$ , we have

$$\begin{split} u(t) - S(t-s)u(s) &= \int_{-\infty}^{t} S(t-\tau)(I-P)g(\tau)d\tau - \int_{t}^{\infty} S(t-\tau)Pg(\tau)d\tau \\ &- S(t-s)\int_{-\infty}^{s} S(s-\tau)(I-P)g(\tau)d\tau \\ &+ S(t-s)\int_{s}^{\infty} S(s-\tau)Pg(\tau)d\tau \\ &= \int_{s}^{t} S(t-\tau)(I-P)g(\tau)d\tau + \int_{s}^{t} S(t-\tau)Pg(\tau)d\tau \\ &= \int_{s}^{t} S(t-\tau)g(\tau)dt. \end{split}$$

Therefore, u defined by (4.5) is a bounded entire solution of (4.2).

To prove the uniqueness, let u and v be bounded entire solutions of (4.2). Setting w := u - v, we have w is a bounded entire solution of the homogeneous equation w'(t) + Aw(t) = 0. Hence, for any  $t \in \mathbb{R}$ , we have w(t) = S(s)w(t-s) for  $s \ge 0$ . By the exponential dichotomy,

$$\|(I-P)w(t)\|_{L^{2}(\Omega)} \leq \|S(s)(I-P)\|\|w(t-s)\|_{L^{2}(\Omega)} \leq Me^{-\beta s}\|w\|_{\infty} \to 0$$

as  $s \to \infty$ . Hence, (I - P)w(t) = 0 and Pw(t) = w(t) for all  $t \in \mathbb{R}$ . Similarly, for any  $t \in \mathbb{R}$ , we have Pw(t) = S(-s)Pw(t+s) for  $s \ge 0$ . Thus,

$$||Pw(t)||_{L^2(\Omega)} \le ||S(-s)P|| ||Pw(t+s)|| \le Me^{-\beta s} ||w||_{\infty} \to 0$$

as  $s \to \infty$ . Hence, we have Pw(t) = 0 for all  $t \in \mathbb{R}$ . This implies w(t) = 0 for all  $t \in \mathbb{R}$ and thus u = v.

It follows from the above theorem that the operator

$$L^{-1}: BC(\mathbb{R}, L^2(\Omega)) \to BC(\mathbb{R}, L^2(\Omega))$$

given by

$$L^{-1}g(t) := \int_{-\infty}^{t} S(t-\tau)(I-P)g(\tau)d\tau - \int_{t}^{\infty} S(t-\tau)Pg(\tau)d\tau,$$
(4.7)

for  $t \geq 0$  is well defined. Moreover,  $L^{-1}$  is bounded and linear with the operator norm  $||L^{-1}|| \leq 2M/\beta$ . Hence,  $L^{-1}$  is continuous under the topology induced by the norm of  $BC(\mathbb{R}, L^2(\Omega))$ . We also remark the continuity under the topology of uniform convergence on compact subsets below.

**Proposition 4.1.6.** Suppose that  $g_n, g \in BC(\mathbb{R}, L^2(\Omega))$  are functions such that  $g_n(t) \to g(t)$  in  $L^2(\Omega)$  uniformly with respect to t on compact subsets of  $\mathbb{R}$  and  $||g_n||_{\infty}$  is uniformly bounded. Then  $L^{-1}g_n(t) \to L^{-1}g(t)$  in  $L^2(\Omega)$  uniformly with respect to t on compact subsets of  $\mathbb{R}$ .

Proof. It is easy to see from the dominated convergence theorem and the representation of bounded entire solution in (4.5) that  $L^{-1}g_n(t) \to L^{-1}g(t)$  in  $L^2(\Omega)$  for all  $t \in \mathbb{R}$ . Let  $J \subset \mathbb{R}$  be a compact set. We show that  $L^{-1}g_n(t) \to L^{-1}g(t)$  in  $L^2(\Omega)$  uniformly on J.

Let  $\varepsilon > 0$  be arbitrary. By the estimates (iii) and (iv) in the definition of exponential dichotomy (Definition 4.1.2) and the uniform boundedness of  $||g_n||_{\infty}$ , we can choose

T > 0 depending only on  $\varepsilon$  such that

$$\int_{T}^{\infty} \|S(\tau)(I-P)(g_n(t-\tau) - g(t-\tau))\|_{L^2(\Omega)} d\tau \leq \frac{\varepsilon}{4}$$

$$\int_{-\infty}^{-T} \|S(\tau)P(g_n(t-\tau) - g(t-\tau))\|_{L^2(\Omega)} d\tau \leq \frac{\varepsilon}{4},$$
(4.8)

for all  $n \in \mathbb{N}$ . Using the representation (4.6) of bounded entire solutions and (4.8), we get from the exponential dichotomy that

$$\begin{split} \|L^{-1}g_{n}(t) - L^{-1}g(t)\|_{L^{2}(\Omega)} &\leq \int_{0}^{\infty} \|S(\tau)(I-P)\big(g_{n}(t-\tau) - g(t-\tau)\big)\|_{L^{2}(\Omega)}d\tau \\ &+ \int_{-\infty}^{0} \|S(\tau)P\big(g_{n}(t-\tau) - g(t-\tau)\big)\|_{L^{2}(\Omega)}d\tau \\ &\leq \int_{0}^{T} \|S(\tau)(I-P)\big(g_{n}(t-\tau) - g(t-\tau)\big)\|_{L^{2}(\Omega)}d\tau + \frac{\varepsilon}{2} \quad (4.9) \\ &\leq \sup_{\tau \in [0,T]} \|g_{n}(t-\tau) - g(t-\tau)\|_{L^{2}(\Omega)} \int_{0}^{T} Me^{-\beta\tau}d\tau \\ &+ \sup_{\tau \in [-T,0]} \|g_{n}(t-\tau) - g(t-\tau)\|_{L^{2}(\Omega)} \int_{-T}^{0} Me^{\beta\tau}d\tau \\ &+ \frac{\varepsilon}{2}, \end{split}$$

for all  $n \in \mathbb{N}$ . Since J and [-T, T] are compact, we have  $\{t - \tau : t \in J, \tau \in [-T, T]\}$  is a compact subset of  $\mathbb{R}$ . By our assumption,  $g_n(t - \tau) \to g(t - \tau)$  in  $L^2(\Omega)$  uniformly with respect to  $(t, \tau) \in J \times [-T, T]$ . Hence, there exists  $N_0 \in \mathbb{N}$  such that

$$\sup_{t\in J} \sup_{\tau\in[-T,T]} \|g_n(t-\tau) - g(t-\tau)\|_{L^2(\Omega)} \frac{M}{\beta} (1-e^{-\beta T}) \le \frac{\varepsilon}{4},$$

for all  $n > N_0$ . It follows from (4.9) that for every  $t \in J$ ,

$$||L^{-1}g_n(t) - L^{-1}g(t)||_{L^2(\Omega)} \le \varepsilon,$$

for all  $n > N_0$ . As  $\varepsilon > 0$  was arbitrary, this proves the required uniform convergence.  $\Box$ 

We show below that  $L^{-1}$  leaves some subspaces of  $BC(\mathbb{R}, L^2(\Omega))$  invariant. We

denote the following subspaces of  $BC(\mathbb{R}, L^2(\Omega))$  by

$$BUC(\mathbb{R}, L^{2}(\Omega)) := \{ f \in BC(\mathbb{R}, L^{2}(\Omega)) : f \text{ is uniformly continuous on } \mathbb{R} \}$$
$$AP(\mathbb{R}, L^{2}(\Omega)) := \{ f \in BC(\mathbb{R}, L^{2}(\Omega)) : f \text{ is almost periodic} \}$$
$$P(\omega, L^{2}(\Omega)) := \{ f \in BC(\mathbb{R}, L^{2}(\Omega)) : f \text{ is } \omega \text{-periodic} \}$$
$$C_{1}(\mathbb{R}, L^{2}(\Omega)) := \{ f \in BC(\mathbb{R}, L^{2}(\Omega)) : \lim_{t \to \pm \infty} f(t) \text{ exists} \}$$
$$C_{0}(\mathbb{R}, L^{2}(\Omega)) := \{ f \in BC(\mathbb{R}, L^{2}(\Omega)) : \lim_{t \to \pm \infty} f(t) = 0 \}.$$

Recall that a function  $f : \mathbb{R} \to L^2(\Omega)$  is almost periodic if  $\{f_\tau\}_{\tau \in \mathbb{R}}$  is relatively compact in  $BC(\mathbb{R}, L^2(\Omega))$ , where

$$f_{\tau}(\cdot) := f(\tau + \cdot) \tag{4.10}$$

is the  $\tau$ -translation of f. This condition is called Bochner's characterisation of almost periodicity. An equivalent definition can be given in terms of relative dense sets in  $\mathbb{R}$ as follows: f is continuous and for every  $\varepsilon > 0$  there exists  $\ell(\varepsilon) > 0$  such that every interval J of length  $\ell(\varepsilon)$  contains  $\tau$  such that  $\|f(t+\tau) - f(t)\|_{L^2(\Omega)} < \varepsilon$  for all  $t \in \mathbb{R}$ . We refer to [28] or [80] for the theory of almost periodic functions.

**Proposition 4.1.7** ([69, Proposition 3]). The operator  $L^{-1}$  has the following properties.

- (i)  $L^{-1}g_{\tau} = (L^{-1}g)_{\tau}$  for all  $\tau \in \mathbb{R}$ .
- (ii) The following spaces BUC(ℝ, L<sup>2</sup>(Ω)), AP(ℝ, L<sup>2</sup>(Ω)), P(ω, L<sup>2</sup>(Ω)), C<sub>1</sub>(ℝ, L<sup>2</sup>(Ω)) and C<sub>0</sub>(ℝ, L<sup>2</sup>(Ω)) are invariant under L<sup>-1</sup>.

Proof. The assertion (i) is almost trivial. Let  $g \in BC(\mathbb{R}, L^2(\Omega))$ . Then  $u(t) := (L^{-1}g)(t+\tau)$  is a bounded entire solution of (4.2) with g(t) replaced by  $g_{\tau}(t) = g(t+\tau)$ . By the uniqueness of bounded entire solutions, the statement follows.

We next show the invariance of the above subspaces. It is evident by (i) that  $P(\omega, L^2(\Omega))$  is invariant under  $L^{-1}$ . Moreover, if  $g \in AP(\mathbb{R}, L^2(\Omega))$ , then by definition,  $\{g_{\tau}\}_{\tau \in \mathbb{R}}$  is relatively compact in  $BC(\mathbb{R}, L^2(\Omega))$ . By (i) again, we see that  $\{(L^{-1}g)_{\tau}\}_{\tau \in \mathbb{R}} = \{L^{-1}g_{\tau}\}_{\tau \in \mathbb{R}}$  is relatively compact in  $BC(\mathbb{R}, L^2(\Omega))$  because  $L^{-1}$  is a bounded linear map. This proves the invariance of  $AP(\mathbb{R}, L^2(\Omega))$ . In addition, for any  $g \in BC(\mathbb{R}, L^2(\Omega))$ , we see from (i) that

$$\|(L^{-1}g)_{\tau} - L^{-1}g\|_{\infty} = \|L^{-1}g_{\tau} - L^{-1}g\|_{\infty}$$
$$\leq \|L^{-1}\|\|g_{\tau} - g\|_{\infty},$$

for all  $\tau \in \mathbb{R}$ . Hence,  $BUC(\mathbb{R}, L^2(\Omega))$  is also invariant under  $L^{-1}$ .

Now, let g be a function in  $C_1(\mathbb{R}, L^2(\Omega))$ . We denote by  $g(\infty) := \lim_{t\to\infty} g(t)$ . We define the constant function f by  $f(t) := g(\infty)$  for all  $t \in \mathbb{R}$ . Then  $g_{\tau}(t)$  converges to f(t) in  $L^2(\Omega)$  as  $\tau \to \infty$  uniformly on compact subsets of  $\mathbb{R}$ . By Proposition 4.1.6, we have

$$(L^{-1}g)_{\tau} = L^{-1}g_{\tau} \to L^{-1}f$$

in  $L^2(\Omega)$  as  $\tau \to \infty$  uniformly on compact subsets of  $\mathbb{R}$ . In particular, this implies

$$L^{-1}g(\tau) = (L^{-1}g)_{\tau}(0) \to L^{-1}f(0) = A^{-1}g(\infty)$$

in  $L^2(\Omega)$  as  $\tau \to \infty$ . By a similar argument, we see that the limit of  $L^{-1}g(\tau)$  as  $\tau \to -\infty$  exists. Hence,  $C_1(\mathbb{R}, L^2(\Omega))$  is invariant under  $L^{-1}$ . It is now clear for the invariance of  $C_0(\mathbb{R}, L^2(\Omega))$ .

For the invariance of  $C_0(\mathbb{R}, L^2(\Omega))$  under  $L^{-1}$ , the proof above does not answer how fast  $L^{-1}g(t)$  converges to zero as  $t \to \pm \infty$ . We give an alternative (direct) proof below to keep track of how large t is required for making  $L^{-1}g(t)$  closed to zero.

**Proposition 4.1.8.** The operator  $L^{-1}$  maps  $C_0(\mathbb{R}, L^2(\Omega))$  into itself.

*Proof.* Suppose  $g \in C_0(\mathbb{R}, L^2(\Omega))$ . Let  $\varepsilon > 0$  be arbitrary. Then there exists T > 0 such that

$$\|g(t)\|_{L^2(\Omega)} \le \frac{\beta}{4M}\varepsilon_{t}$$

if |t| > T. Fix now t > T. If  $\tau \in (-\infty, t - T)$ , then  $t - \tau > T$ . Hence,  $||g(t - \tau)||_{L^2(\Omega)} \le \frac{\beta}{4M}\varepsilon$  for all  $\tau \in (-\infty, t - T)$ . Therefore,

$$\int_{0}^{\infty} \|S(\tau)(I-P)g(t-\tau)\|_{L^{2}(\Omega)} d\tau = \int_{0}^{t-T} \|S(\tau)(I-P)g(t-\tau)\|_{L^{2}(\Omega)} d\tau$$
$$+ \int_{t-T}^{\infty} \|S(\tau)(I-P)g(t-\tau)\|_{L^{2}(\Omega)} d\tau$$
$$\leq \frac{\beta}{4M} \varepsilon \int_{0}^{t-T} M e^{-\beta\tau} d\tau + \|g\|_{\infty} \int_{t-T}^{\infty} M e^{-\beta\tau} d\tau$$
$$\leq \frac{\varepsilon}{4} + \|g\|_{\infty} \frac{M}{\beta} e^{-\beta(t-T)}.$$
(4.11)

Moreover,

$$\int_{-\infty}^{0} \|S(\tau)Pg(t-\tau)\|_{L^{2}(\Omega)} d\tau \leq \frac{\beta}{4M} \varepsilon \int_{-\infty}^{0} M e^{\beta\tau} d\tau \leq \frac{\varepsilon}{4}.$$
 (4.12)

Combining (4.11) and (4.12), we have

$$\begin{split} \|L^{-1}g(t)\|_{L^{2}(\Omega)} &\leq \int_{0}^{\infty} \|S(\tau)(I-P)g(t-\tau)\|_{L^{2}(\Omega)} d\tau + \int_{-\infty}^{0} \|S(\tau)Pg(t-\tau)\|_{L^{2}(\Omega)} d\tau \\ &\leq \frac{\varepsilon}{2} + \|g\|_{\infty} \frac{M}{\beta} e^{-\beta(t-T)}. \end{split}$$

$$(4.13)$$

Since the above argument works for any fixed t > T, we have that (4.13) holds for all t > T. Choosing  $\tilde{T} > T$  so that

$$\|g\|_{\infty} \frac{M}{\beta} e^{-\beta(\tilde{T}-T)} < \frac{\varepsilon}{2},$$

it follows that  $\|L^{-1}g(t)\|_{L^2(\Omega)} < \varepsilon$  if  $t > \tilde{T}$ . Since  $\varepsilon > 0$  was arbitrary, this implies  $L^{-1}g(t) \to 0$  in  $L^2(\Omega)$  as  $t \to \infty$ .

To get  $L^{-1}g(t) \to 0$  in  $L^2(\Omega)$  as  $t \to -\infty$ , we instead consider t < -T and follow a similar argument as above. Then we estimate  $\|L^{-1}g(t)\|_{L^2(\Omega)}$  by splitting it into

$$\int_{-\infty}^{t+T} \|S(\tau)Pg(t-\tau)\|_{L^{2}(\Omega)} d\tau + \int_{t+T}^{0} \|S(\tau)Pg(t-\tau)\|_{L^{2}(\Omega)} d\tau + \int_{0}^{\infty} \|S(\tau)(I-P)g(t-\tau)\|_{L^{2}(\Omega)} d\tau.$$

# 4.2 Convergence of bounded entire solutions of linear equations: $L^2$ -Theory

Let  $\Omega_n$  and  $\Omega$  be bounded open sets in  $\mathbb{R}^N$  satisfying Assumption 1.4.1. We consider the perturbation of (4.1) given by

$$\begin{cases} \frac{\partial u}{\partial t} + \mathcal{A}_n u = g_n(x, t) & \text{in } \Omega_n \times (-\infty, \infty) \\ u = 0 & \text{on } \partial \Omega_n \times (-\infty, \infty), \end{cases}$$
(4.14)

where  $g_n \in C(\mathbb{R}^N \times \mathbb{R})$  and  $\mathcal{A}_n$  is an elliptic operator defined by (2.40). As previously seen in Section 2.4.2, we can write (4.14) in the abstract form in  $L^2(\Omega_n)$  as

$$u'(t) + A_n u = g_n(t), \quad t \in (-\infty, \infty).$$
 (4.15)

The operator  $-A_n$  is the generator of a strongly continuous analytic semigroup  $S_n(t), t \ge 0$  on  $L^2(\Omega_n)$ .

As before, we use the inclusion operators  $i_n$  and i defined by (1.27) and (1.28) respectively to handle domain perturbation. We also use the restriction operators  $r_n$  and r defined by (1.29) and (1.30), respectively. We make the following assumptions for the unperturbed equation.

Assumption 4.2.1. We assume that  $\sigma(-A)$  can be decomposed as  $\sigma(-A) = \sigma^+ \cup \sigma^$ where  $\sigma^+$  and  $\sigma^-$  are given as in (4.3).

We see in Example 4.1.3 that Assumption 4.2.1 implies an exponential dichotomy of u'(t) + Au(t) = 0 with the projection  $P = P^+$  given by (4.4). We assume the conditions below for our perturbation.

Assumption 4.2.2. We assume that

- (i) The coefficients of the operator  $\mathcal{A}_n$  converge to the corresponding coefficients of  $\mathcal{A}$  as  $n \to \infty$  (see Assumption 2.4.5);
- (ii)  $H_0^1(\Omega_n)$  converges to  $H_0^1(\Omega)$  in the sense of Mosco.

**Lemma 4.2.3.** Suppose that Assumption 4.2.1 and Assumption 4.2.2 are satisfied. Then  $u'(t) + A_n u(t) = 0$  has an exponential dichotomy on  $\mathbb{R}$  with uniform exponent  $\beta$ and uniform bound M for all n sufficiently large. In particular, for n sufficiently large the dichotomy projections  $P_n$  can be taken as the spectral projection

$$P_n := P_n^+ = \frac{1}{2\pi i} \int_{\Gamma} (\lambda + A_n)^{-1} d\lambda, \qquad (4.16)$$

where  $\Gamma$  is a rectifiable closed curve separating  $\sigma^+$  from  $\sigma^-$ . Moreover,  $i_n P_n r_n \to i Pr$ in  $\mathscr{L}(L^2(D))$  as  $n \to \infty$ .

*Proof.* The assertion directly follows from Corollary 2.4.12 by noting that  $\sigma_n$  is separated by the strip  $\{\lambda : -\beta \leq \operatorname{Re}(\lambda) \leq \beta\}$  with the uniform parameter  $\beta$  for n sufficiently large.

We prove the convergence of solutions in the stable and unstable directions below.

**Lemma 4.2.4.** Suppose that Assumption 4.2.1 and Assumption 4.2.2 are satisfied. Let  $u_{0,n} \in L^2(\Omega_n)$  and  $u_0 \in L^2(\Omega)$  be such that  $ri_n u_{0,n} \rightharpoonup u_0$  in  $L^2(\Omega)$  weakly and  $\|i_n u_{0,n}\|_{L^2(D)}$  is uniformly bounded. Then

(i)  $i_n S_n(t)(I - P_n)u_{0,n} \to iS(t)(I - P)u_0$  in  $L^2(D)$  for all t > 0 and uniformly on compact subsets of  $(0, \infty)$ .

(ii)  $i_n S_n(t) P_n u_{0,n} \to i S(t) P u_0$  in  $L^2(D)$  for all  $t \leq 0$  and uniformly on compact subsets of  $(-\infty, 0]$ .

Moreover, if  $i_n u_{0,n} \to i u_0$  in  $L^2(D)$  strongly, then the convergence in assertion (i) holds uniformly on compact subsets of  $[0, \infty)$ .

Proof. It follows from Lemma 3.4.1 (iv) that  $ri_n(I - P_n)u_{0,n} \rightarrow (I - P)u_0$  in  $L^2(\Omega)$  weakly. Applying domain perturbation results for initial value problems in Theorem 1.4.9 (with homogeneous data  $f_n, f = 0$ ), assertion (i) follows. Taking Lemma 3.4.1 (i) into account, the additional statement regarding to uniform convergence on compact subset of  $[0, \infty)$  holds.

We next prove assertion (ii). By Lemma 3.4.1 (iv), we have  $ri_nP_nu_{0n} \rightarrow Pu_0$  in  $L^2(\Omega)$  weakly. By uniform boundedness of  $i_nu_{0,n}$  in  $L^2(D)$ , we have  $||P_nu_{0,n}||_{L^2(\Omega_n)}$  is also uniformly bounded. Applying Corollary 3.4.6, we obtain a subsequence such that  $i_{n_k}P_{n_k}u_{0,n_k} \rightarrow iw$  in  $L^2(D)$  for some  $w \in X^+ := P(L^2(\Omega))$ . In particular,  $ri_{n_k}P_{n_k}u_{0,n_k} \rightarrow w$  in  $L^2(\Omega)$ . Hence, we conclude that  $w = Pu_0$  and the whole sequence quence

$$i_n P_n u_{0,n} \to i P u_0 \tag{4.17}$$

in  $L^2(D)$  strongly. Let now T > 0 be arbitrary. By Lemma 4.2.3, we have

$$i_n P_n r_n i S(-T) P u_0 \to i PriS(-T) P u_0 = i S(-T) P u_0 \tag{4.18}$$

in  $L^2(D)$ . Applying domain perturbation results in Theorem 1.4.9 (for homogeneous equations), we get

$$i_n S_n(T) P_n r_n i S(-T) P u_0 \to i S(T) S(-T) P u_0 = i P u_0$$

$$\tag{4.19}$$

in  $L^2(D)$ . On the other hand, using (4.17) we also get

$$i_n S_n(T) P_n S_n(-T) P_n u_{0,n} = i_n P_n u_{0,n} \to i P u_0$$
(4.20)

in  $L^2(D)$ . It follows from (4.19) and (4.20) that

$$i_n S_n(T) P_n S_n(-T) P_n u_{0,n} - i_n S_n(T) P_n r_n i S(-T) P u_0 \to 0$$

in  $L^2(D)$ . In particular,

$$\left\| S_n(T) P_n \left( S_n(-T) P_n u_{0,n} - P_n r_n i S(-T) P u_0 \right) \right\|_{L^2(\Omega_n)} \to 0$$
(4.21)

as  $n \to \infty$ . By the exponential dichotomy for the perturbed equations (see Lemma 4.2.3) and (4.21), we see that

$$\begin{aligned} \|S_{n}(-T)P_{n}u_{0,n} - P_{n}r_{n}iS(-T)Pu_{0}\|_{L^{2}(\Omega_{n})} \\ &\leq \|S_{n}(-T)P_{n}\| \left\|S_{n}(T)P_{n}\left(S_{n}(-T)P_{n}u_{0,n} - P_{n}r_{n}iS(-T)Pu_{0}\right)\right\|_{L^{2}(\Omega_{n})} \\ &\leq Me^{-\beta T} \left\|S_{n}(T)P_{n}\left(S_{n}(-T)P_{n}u_{0,n} - P_{n}r_{n}iS(-T)Pu_{0}\right)\right\|_{L^{2}(\Omega_{n})} \\ &\to 0 \end{aligned}$$
(4.22)

as  $n \to \infty$ . Combining (4.22) and (4.18), we conclude that

$$i_n S_n(-T) P_n u_{0,n} \to i S(-T) P u_0 \tag{4.23}$$

in  $L^2(D)$  as  $n \to \infty$ . As T > 0 was arbitrary and (4.17) holds, we have

$$i_n S_n(t) P_n u_{0,n} \to i S(t) P u_0 \tag{4.24}$$

in  $L^2(\Omega)$  for all  $t \leq 0$ . The uniform convergence of (4.24) with respect to t in compact subsets of  $(-\infty, 0]$  immediately follows from Theorem 1.4.9 (by applying to the forward problems with initial data  $S_n(-T)P_nu_{0,n}$  for T > 0).

We next prove our first result on convergence of bounded entire solutions under domain perturbation.

**Theorem 4.2.5.** Suppose that Assumption 4.2.1 and Assumption 4.2.2 are satisfied. Let  $g_n$  and g be functions in  $BC(\mathbb{R}, L^2(\Omega_n))$  and  $BC(\mathbb{R}, L^2(\Omega))$ , respectively. Assume that  $||g_n||_{\infty}$  is uniformly bounded. If  $ri_ng_n(t) \rightarrow g(t)$  weakly in  $L^2(\Omega)$  uniformly with respect to  $t \in \mathbb{R}$ , that is

$$\sup_{t \in \mathbb{R}} (ri_n g_n(t) - g(t)|v)_{L^2(\Omega)} \to 0, \qquad (4.25)$$

for all  $v \in L^2(\Omega)$ , then  $i_n L_n^{-1} g_n \to i L^{-1} g$  in  $BC(\mathbb{R}, L^2(D))$ .

*Proof.* Using the representation (4.6) of bounded entire solutions in Theorem 4.1.5, we have

$$L_n^{-1}g_n(t) = \int_0^\infty S_n(\tau)(I - P_n)g_n(t - \tau)d\tau - \int_{-\infty}^0 S_n(\tau)P_ng_n(t - \tau)d\tau$$
$$L^{-1}g(t) = \int_0^\infty S(\tau)(I - P)g(t - \tau)d\tau - \int_{-\infty}^0 S(\tau)Pg(t - \tau)d\tau,$$

for all  $t \in \mathbb{R}$ . To show the uniform convergence of  $i_n L_n^{-1} g_n(t) \to i L^{-1} g(t)$  in  $L^2(D)$ with respect to  $t \in \mathbb{R}$ , it suffices to show that for every sequence  $\{t_n\}_{n \in \mathbb{N}}$  with  $t_n \in \mathbb{R}$ , we have

$$\int_0^\infty \|i_n S_n(\tau)(I-P_n)g_n(t_n-\tau) - iS(\tau)(I-P)g(t_n-\tau)\|_{L^2(D)}d\tau \to 0$$
$$\int_{-\infty}^0 \|i_n S_n(\tau)P_ng_n(t_n-\tau) - iS(\tau)Pg(t_n-\tau)\|_{L^2(D)}d\tau \to 0$$

as  $n \to \infty$ .

Let  $\{t_n\}_{n\in\mathbb{N}}$  be an arbitrary sequence in  $\mathbb{R}$ . We have from (4.25) that

$$(ri_n g_n(t_n - \tau) - g(t_n - \tau)|v)_{L^2(D)} \to 0, \qquad (4.26)$$

for all  $v \in L^2(\Omega)$  and for all  $\tau \in \mathbb{R}$ . This implies

$$ri_n g_n(t_n - \tau) - g(t_n - \tau) \rightharpoonup 0 \tag{4.27}$$

in  $L^2(\Omega)$  weakly for all  $\tau \in \mathbb{R}$ . By the convergence of spectral projections  $i_n P_n r_n \to i Pr$ (see Lemma 4.2.3) together with (4.27) and the uniform boundedness of  $||g_n||_{\infty}$ , it follows that

$$\begin{aligned} (ri_n P_n g_n(t_n - \tau) - Pg(t_n - \tau) | v)_{L^2(\Omega)} \\ &\leq (ri_n P_n g_n(t_n - \tau) - Pri_n g_n(t_n - \tau) | v)_{L^2(\Omega)} \\ &+ (Pri_n g_n(t_n - \tau) - Pg(t_n - \tau) | v)_{L^2(\Omega)} \\ &\leq \|r\| \|i_n P_n r_n - iPr\| \|i_n g_n(t_n - \tau)\|_{L^2(D)} \|v\|_{L^2(\Omega)} \\ &\quad (ri_n g_n(t_n - \tau) - g(t_n - \tau) | P'v)_{L^2(\Omega)} \\ &\rightarrow 0, \end{aligned}$$

for all  $v \in L^2(\Omega)$  and for all  $\tau \in \mathbb{R}$  where P' is the adjoint operator of P. This implies

$$ri_n P_n g_n(t_n - \tau) - Pg(t_n - \tau) \rightharpoonup 0 \tag{4.28}$$

in  $L^2(\Omega)$  weakly for all  $\tau \in \mathbb{R}$ . By (4.27), we also have that

$$ri_n(I - P_n)g_n(t_n - \tau) - (I - P)g(t_n - \tau) \rightharpoonup 0$$

$$(4.29)$$

in  $L^2(\Omega)$  weakly for all  $\tau \in \mathbb{R}$ .

We fix now an arbitrary  $\tau \in (0, \infty)$ . By the boundedness of  $||(I-P)g(t_n-\tau)||_{L^2(\Omega)}$ , we have that for any subsequence  $||(I-P)g(t_{n_k}-\tau)||_{L^2(\Omega)}$ , there exists a further weak convergent subsequence (indexed again by  $n_k$ ) such that

$$(I-P)g(t_{n_k}-\tau) \rightharpoonup \xi_{\tau}$$

in  $L^2(\Omega)$  for some  $\xi_{\tau} \in L^2(\Omega)$ . We get from (4.29) that

$$ri_{n_k}(I - P_{n_k})g_{n_k}(t_{n_k} - \tau) \rightharpoonup \xi_{\tau}$$

in  $L^2(\Omega)$ . By Lemma 4.2.4 (i), we have

- -

$$i_{n_k}S_{n_k}(t)(I-P_{n_k})g_{n_k}(t_{n_k}-\tau) \to iS(t)(I-P)\xi_{\tau}$$
$$iS(t)(I-P)g(t_{n_k}-\tau) \to iS(t)(I-P)\xi_{\tau}$$

in  $L^2(D)$  for all  $t \in (0, \infty)$ . In particular, taking  $t = \tau$  we have

$$\|i_{n_k}S_{n_k}(\tau)(I-P_{n_k})g_{n_k}(t_{n_k}-\tau) - iS(\tau)(I-P)g(t_{n_k}-\tau)\|_{L^2(D)} \to 0.$$

Since the above argument holds for every subsequence, we conclude that

$$\|i_n S_n(\tau)(I - P_n)g_n(t_n - \tau) - iS(\tau)(I - P)g(t_n - \tau)\|_{L^2(D)} \to 0.$$
(4.30)

As  $\tau \in (0, \infty)$  was arbitrary, the above argument implies pointwise convergence of (4.30) for each  $\tau \in (0, \infty)$ . By the uniform boundedness of  $||g_n||_{\infty}$  and the exponential dichotomy, we see that

$$\int_0^\infty \|i_n S_n(\tau)(I-P_n)g_n(t_n-\tau) - iS(\tau)(I-P)g(t_n-\tau)\|_{L^2(D)}d\tau$$
$$\leq (\|g_n\|_\infty + \|g\|_\infty) \int_0^\infty M e^{-\beta\tau} d\tau$$
$$< \infty.$$

Using (4.30) and applying the dominated convergence theorem, we get

$$\int_0^\infty \|i_n S_n(\tau)(I - P_n)g_n(t_n - \tau) - iS(\tau)(I - P)g(t_n - \tau)\|_{L^2(D)}d\tau \to 0$$

as 
$$n \to \infty$$
.

By a similar argument as above (using (4.28) and Lemma 4.2.4 (ii) instead of (4.29) and Lemma 4.2.4 (i), respectively), we have pointwise convergence

$$\|i_n S_n(\tau) P_n g_n(t_n - \tau) - i S(\tau) P g(t_n - \tau)\|_{L^2(D)} \to 0,$$
(4.31)

for each  $\tau \in (-\infty, 0)$ . Applying the dominated convergence theorem, we get

$$\int_{-\infty}^{0} \|i_n S_n(\tau) P_n g_n(t_n - \tau) - iS(\tau) Pg(t_n - \tau)\|_{L^2(D)} d\tau \to 0$$

as  $n \to \infty$ .

Since a sequence  $\{t_n\}_{n\in\mathbb{N}}$  was arbitrary, we conclude that  $i_n L_n^{-1} g_n(t) \to i L^{-1} g(t)$  in  $L^2(D)$  uniformly on  $\mathbb{R}$  as required.

In the next convergence result, we assume only weak-\* convergence of the inhomogeneous terms in  $L^{\infty}(\mathbb{R}, L^2(\Omega))$ .

**Theorem 4.2.6.** Suppose that Assumption 4.2.1 and Assumption 4.2.2 are satisfied. Let  $g_n$  and g be functions in  $BC(\mathbb{R}, L^2(\Omega_n))$  and  $BC(\mathbb{R}, L^2(\Omega))$ , respectively. Assume that  $||g_n||_{\infty}$  is uniformly bounded. If  $ri_ng_n \stackrel{*}{\rightharpoonup} g$  in  $L^{\infty}(\mathbb{R}, L^2(\Omega))$ , then  $i_nL_n^{-1}g_n$  converges to  $iL^{-1}g$  in  $BC(\mathbb{R}, L^2(D))$  under the topology of uniform convergence on compact sets.

Proof. By the uniform boundedness of  $||g_n||_{\infty}$ , we have that  $i_n L_n^{-1} g_n$  is also uniformly bounded in  $L^{\infty}(\mathbb{R}, L^2(D))$ . As  $\mathbb{R}$  is separable, we can choose a dense sequence  $\{t_1, t_2, \ldots\}$ in  $\mathbb{R}$  (e.g. a sequence of rational numbers). Since  $i_n L_n^{-1} g_n(t_1)$  is bounded in  $L^2(D)$ , we can extract a subsequence  $\{i_{1,n} L_{1,n}^{-1} g_{1,n}(t_1)\}_{n \in \mathbb{N}}$  of  $\{i_n L_n^{-1} g_n(t_1)\}_{n \in \mathbb{N}}$  so that it converges in  $L^2(D)$  weakly. Similarly, by the boundedness of  $i_{1,n} L_{1,n}^{-1} g_{1,n}(t_2)$ , we can extract a subsequence  $\{i_{2,n} L_{2,n}^{-1} g_{2,n}(t_2)\}_{n \in \mathbb{N}}$  of  $\{i_{1,n} L_{1,n}^{-1} g_{1,n}(t_2)\}_{n \in \mathbb{N}}$  so that it converges in  $L^2(D)$ weakly. Inductively, we obtain for every  $k \geq 2$  a subsequence  $\{i_{k,n} L_{k,n}^{-1} g_{k,n}(t_k)\}_{n \in \mathbb{N}}$ of  $\{i_{(k-1),n} L_{(k-1),n}^{-1} g_{(k-1),n}(t_k)\}_{n \in \mathbb{N}}$  which converges in  $L^2(D)$  weakly as  $n \to \infty$ . We denote the weak limit of  $i_{k,n} L_{k,n}^{-1} g_{k,n}(t_k)$  in  $L^2(D)$  by  $\psi_{t_k}$  for each  $k \in \mathbb{N}$ . In particular,

$$ri_{k,n}L_{k,n}^{-1}g_{k,n}(t_k) \rightharpoonup r\psi_{t_k} =: \phi_{t_k}$$

in  $L^2(\Omega)$  as  $n \to \infty$  for all  $k \in \mathbb{N}$ . Consider a diagonal subsequence  $\{i_{n,n}L_{n,n}^{-1}g_{n,n}\}_{n\in\mathbb{N}}$ . By a standard diagonal argument,  $ri_{n,n}L_{n,n}^{-1}g_{n,n}(t_k) \rightharpoonup \phi_{t_k}$  in  $L^2(\Omega)$  weakly for every  $k \ge 1$ . In other words, the diagonal subsequence converges weakly on a dense subset of  $\mathbb{R}$ .

By the weak \*-convergence of  $ri_ng_n \stackrel{*}{\rightharpoonup} g$  in  $L^{\infty}(\mathbb{R}, L^2(\Omega))$ , we have  $ri_ng_n \rightharpoonup g$  in  $L^2(J, L^2(\Omega))$  for every compact interval  $J \subset \mathbb{R}$ . To see this, let  $v \in L^2(J, L^2(\Omega))$  be arbitrary and denote by  $\tilde{v}$  the extension of v by zero on  $\mathbb{R}\setminus \bar{J}$ . We have  $\tilde{v} \in L^1((\mathbb{R}, L^2(\Omega)))$ . It follows from the definition of weak \*-convergence that

$$\int_{J} (ri_n g_n(t)|v(t))_{L^2(\Omega)} dt = \int_{-\infty}^{\infty} (ri_n g_n(t)|\tilde{v}(t))_{L^2(\Omega)} dt$$
$$\rightarrow \int_{-\infty}^{\infty} (g(t)|\tilde{v}(t))_{L^2(\Omega)} dt = \int_{J} (g(t)|v(t))_{L^2(\Omega)} dt.$$

This proves the required weak convergence in  $L^2(J, L^2(\Omega))$ . We note that for every  $k \ge 1$ , the restriction of  $L_{n,n}^{-1}g_{n,n}$  on  $(t_k, \infty)$  satisfies the following initial value problem

$$\begin{cases} \dot{v}(t) + A_{n,n}v(t) = g_{n,n}(t) & \text{for } t \in (t_k, \infty) \\ v(t_k) = L_{n,n}^{-1}g_{n,n}(t_k). \end{cases}$$
(4.32)

Since  $ri_ng_n \rightharpoonup g$  in  $L^2(J, L^2(\Omega))$  for any  $J \subset \mathbb{R}$  and  $ri_{n,n}L_{n,n}^{-1}g_{n,n}(t_k) \rightharpoonup \phi_{t_k}$  in  $L^2(\Omega)$ , we conclude from domain perturbation results in Theorem 1.4.9 that for every  $k \ge 1$ , the restriction  $i_{n,n}L_{n,n}^{-1}g_{n,n}|_{(t_k,\infty)}$  converges to the unique solution of the following problem

$$\begin{cases} \dot{v}(t) + Av(t) = g(t) & \text{for } t \in (t_k, \infty) \\ v(t_k) = \phi_{t_k} \end{cases}$$

$$(4.33)$$

uniformly on compact subsets of  $(t_k, \infty)$ . Since for every  $t \in \mathbb{R}$  there exists  $t_k$  such that  $t_k < t$  (from the density of  $\{t_k, k \in \mathbb{N}\}$  in  $\mathbb{R}$ ), we conclude that  $i_{n,n}L_{n,n}^{-1}g_{n,n}(t)$  converges in  $L^2(D)$  strongly for all  $t \in \mathbb{R}$ . We denote the pointwise limit of  $i_{n,n}L_{n,n}^{-1}g_{n,n}(t)$  in  $L^2(D)$  by  $i\phi_t$  for each  $t \in \mathbb{R}$  where  $\phi_t \in L^2(\Omega)$ . Note that for  $k \ge 1$  the strong limit agrees with the weak limit  $\phi_{t_k}$  considered previously. Since  $\sup_{n \in \mathbb{N}} \sup_{t \in \mathbb{R}} \|i_n L_n^{-1}g_n(t)\|_{L^2(D)} < K$  for some K > 0, we see that  $\|i\phi_t\|_{L^2(D)} \le K$  for all  $t \in \mathbb{R}$ . Setting  $\phi(t) := \phi_t$  for  $t \in \mathbb{R}$ , we have  $\phi \in L^{\infty}(\mathbb{R}, L^2(\Omega))$ .

Repeating the above argument for an arbitrary  $s \in \mathbb{R}$  instead of  $t_k, k \ge 1$ , we have that the restriction of  $\phi$  on  $(s, \infty)$  satisfies

$$\begin{cases} \dot{v}(t) + Av(t) = g(t) & \text{for } t \in (s, \infty) \\ v(s) = \phi(s). \end{cases}$$

In particular, we have  $\phi|_{(s,\infty)} \in C((s,\infty), L^2(\Omega))$ . Since this is true for any  $s \in \mathbb{R}$ , we conclude that  $\phi \in BC(\mathbb{R}, L^2(\Omega))$  and  $\phi$  is a bounded entire solution of (4.2). By the uniqueness of bounded entire solution,  $\phi = L^{-1}g$  and the whole sequence converges, that is

$$i_n L_n^{-1} g_n(t) \to i L^{-1} g(t)$$
 (4.34)

in  $L^2(D)$  pointwise for all  $t \in \mathbb{R}$ . The uniform convergence of (4.34) with respect to t in compact subsets of  $\mathbb{R}$  immediately follows from Theorem 1.4.9.

### 4.3 L<sup>p</sup>-Theory for bounded entire solutions of linear parabolic equations

In this section, we prove the convergence of bounded entire solutions under domain perturbation under the  $L^p$ -norm for a certain range of  $p \ge 2$ . The main ingredients for this result are the  $L^2$  results in Section 4.2 and the interpolation arguments. In addition, we prove Hölder continuity of bounded entire solutions with respect to time under a certain  $L^p$  norm. Recall that by our assumptions on the operator A, the semigroup S(t) for Dirichlet problems satisfies the following heat kernel estimate

$$\|S(t-s)\|_{\mathscr{L}(L^{p}(\Omega),L^{q}(\Omega))} \le C(t-s)^{-\frac{N}{2}\left(\frac{1}{p}-\frac{1}{q}\right)},$$
(4.35)

for all  $1 \le p \le q \le \infty$  and  $t \ge s$ , where C is a positive constant independent of  $\Omega$  (see [36, Section 7]).

We first show that the solutions are actually bounded in  $L^p$  for some range of  $p \ge 2$ . For this, we need the following estimates.

**Lemma 4.3.1.** Let  $2 \le p < \infty$  be such that  $\alpha_p := \frac{N}{2}(\frac{1}{2} - \frac{1}{p}) \in [0, 1)$ . Then there exists  $M_1(p), M_2(p) > 0$  such that

(i) 
$$||S(t)(I-P)||_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))} \leq M_{1}(p)t^{-\alpha_{p}}e^{-\frac{\beta}{2}t}$$
 for all  $t > 0$ 

(ii)  $||S(t)P||_{\mathscr{L}(L^2(\Omega),L^p(\Omega))} \leq M_2(p)e^{\beta t}$  for all t < 0.

*Proof.* Let t > 0. It follows from the heat kernel estimate (4.35) and the exponential dichotomy that

$$\begin{split} \|S(t)(I-P)\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))} &\leq \|S(t/2)\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))}\|S(t/2)(I-P)\|_{\mathscr{L}(L^{2}(\Omega))} \\ &\leq C\left(\frac{t}{2}\right)^{-\alpha_{p}}Me^{-\frac{\beta}{2}t} \\ &\leq M_{1}(p)t^{-\alpha_{p}}e^{-\frac{\beta}{2}t}, \end{split}$$

for all t > 0 where  $M_1(p) := CM2^{\alpha_p}$ . Similarly, for t < 0 we have

$$\begin{split} \|S(t)P\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))} &\leq \|S(1)\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))}\|S(t-1)P\|_{\mathscr{L}(L^{2}(\Omega))} \\ &\leq C1^{-\alpha_{p}}Me^{\beta(t-1)} \\ &\leq M_{2}(p)e^{\beta t}, \end{split}$$

for all t < 0 where  $M_2(p) := CMe^{-\beta}$ .

Remark 4.3.2. The range of p satisfying the condition in the above theorem depends on N. In fact, for domains in two, three or four dimensional spaces,  $2 \le p < \infty$  is valid. For  $N \ge 5$ , we require  $2 \le p < \frac{2N}{N-4}$ .

We next show that the bounded entire solutions are bounded with respect to a stronger  $L^p$  norm.

**Theorem 4.3.3.** Let g be a function in  $BC(\mathbb{R}, L^2(\Omega))$ . The bounded entire solution  $u = L^{-1}g$  of (4.2) belongs to  $BC(\mathbb{R}, L^p(\Omega))$  for all p satisfying  $2 \leq p < \infty$  with  $\alpha_p := \frac{N}{2}(\frac{1}{2} - \frac{1}{p}) \in [0, 1)$ .

*Proof.* Using the representation (4.6) of bounded entire solutions, we show that

$$\int_0^\infty \|S(\tau)(I-P)g(t-\tau)\|_{L^p(\Omega)} d\tau - \int_{-\infty}^0 \|S(\tau)Pg(t-\tau)\|_{L^p(\Omega)} d\tau$$

is uniformly bounded for all  $t \in \mathbb{R}$ . By Lemma 4.3.1 (i), we see that

$$\begin{split} \int_0^\infty \|S(\tau)(I-P)g(t-\tau)\|_{L^p(\Omega)}d\tau \\ &\leq \int_0^\infty \|S(\tau)(I-P)\|_{\mathscr{L}(L^2(\Omega),L^p(\Omega))}\|g(t-\tau)\|_{L^2(\Omega)}d\tau \\ &\leq \|g\|_\infty \int_0^\infty M_1(p)\tau^{-\alpha_p}e^{-\frac{\beta}{2}\tau}d\tau \\ &= \|g\|_\infty M_1(p)\int_0^\infty \left(\frac{2\tau}{\beta}\right)^{-\alpha_p}e^{-\tau}\frac{2}{\beta}d\tau \\ &= \|g\|_\infty M_1(p)\left(\frac{2}{\beta}\right)^{1-\alpha_p}\int_0^\infty \tau^{-\alpha_p}e^{-\tau}d\tau \\ &= \|g\|_\infty M_1(p)\left(\frac{2}{\beta}\right)^{1-\alpha_p}\Gamma(1-\alpha_p), \end{split}$$

for all  $t \in \mathbb{R}$ . By Lemma 4.3.1 (ii), we have

$$\begin{split} \int_{-\infty}^{0} \|S(\tau)Pg(t-\tau)\|_{L^{p}(\Omega)}d\tau &\leq \int_{-\infty}^{0} \|S(\tau)P\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))}\|g(t-\tau)\|_{L^{2}(\Omega)}d\tau \\ &\leq \|g\|_{\infty} \int_{-\infty}^{0} M_{2}(p)e^{\beta t}d\tau \\ &= \|g\|_{\infty} \frac{M_{2}(p)}{\beta}, \end{split}$$

for all  $t \in \mathbb{R}$ .

As a direct application of the convergence results in Section 4.2, we state the following theorems.

**Theorem 4.3.4.** Suppose that all assumptions in Theorem 4.2.5 are satisfied. Then  $i_n L_n^{-1} g_n \to i L^{-1} g$  in  $BC(\mathbb{R}, L^p(D))$  for all p satisfying  $2 \le p < \infty$  and  $\frac{N}{2}(\frac{1}{2} - \frac{1}{p}) \in [0, 1)$ . *Proof.* Taking Remark 4.3.2 into account, we fix p such that  $2 \le p < \infty$  for N = 2, 3, 4 and  $2 \le p < \frac{2N}{N-4}$  for  $N \ge 5$ . There exists q such that  $p < q < \infty$  for N = 2, 3, 4

or  $p \leq q < \frac{2N}{N-4}$  for  $N \geq 5$ . By the uniform estimate of semigroups  $S_n(t)$  and the uniform exponential dichotomy, we see from Theorem 4.3.3 that  $\|i_n L_n^{-1} g_n\|_{BC(\mathbb{R}, L^q(D))}$  and  $\|iL^{-1}g\|_{BC(\mathbb{R}, L^q(D))}$  are uniformly bounded for all  $n \in \mathbb{N}$ . Hence, by Theorem 4.2.5 and interpolation inequality with  $\theta := \frac{(p-2)q}{(q-2)p}$ ,

$$\begin{aligned} \|i_n L_n^{-1} g_n(t) - iL^{-1} g(t)\|_{L^p(D)} \\ &\leq \|i_n L_n^{-1} g_n(t) - iL^{-1} g(t)\|_{L^q(D)}^{\theta} \|i_n L_n^{-1} g_n(t) - iL^{-1} g(t)\|_{L^2(D)}^{1-\theta} \\ &\leq \|i_n L_n^{-1} g_n - iL^{-1} g\|_{BC(\mathbb{R}, L^q(D))}^{\theta} \|i_n L_n^{-1} g_n - iL^{-1} g\|_{BC(\mathbb{R}, L^2(D))}^{1-\theta} \\ &\to 0 \end{aligned}$$

as  $n \to \infty$  uniformly with respect to  $t \in \mathbb{R}$ .

**Theorem 4.3.5.** Suppose that all assumptions in Theorem 4.2.6 are satisfied. Then  $i_n L_n^{-1}g_n$  converges to  $iL^{-1}g$  in  $BC(\mathbb{R}, L^p(D))$  under the topology of uniform convergence on compact subsets for all p satisfying  $2 \le p < \infty$  and  $\frac{N}{2}(\frac{1}{2} - \frac{1}{p}) \in [0, 1)$ .

*Proof.* The result follows from Theorem 4.2.6 and a similar interpolation argument as in the proof of Theorem 4.3.4.  $\Box$ 

In the remainder of this section, we obtain the Hölder continuity of the bounded entire solution of (4.2) with respect to time in  $BC(\mathbb{R}, L^p(\Omega))$ . The result is similar to the Hölder continuity proved in Daners [40] (see Lemma 5.3, Lemma 5.5, Corollary 5.6) for evolution equations in interpolation spaces, or in Henry [52, Section 3.3] for evolution equations in fractional power spaces. As we work in the  $L^p$  scales, the estimate  $||AS(t)||_{\mathscr{L}(L^p(\Omega))} \leq \frac{c_p}{t}$  for t > 0 leads to a singularity when integrating near t = 0. This difficulty does not occur when working in interpolation spaces or fractional power spaces. This is because we can get a better estimate of  $||AS(t)||_{\alpha,\beta}$  for certain  $\alpha$  and  $\beta$  due to the use of graph norm in the definition of these spaces ( $X_1 = D(A)$  for interpolation spaces,  $X^{\alpha} = D(A^{\alpha})$  for fractional power spaces). However, we exploit the boundedness and use that the solution is defined on  $\mathbb{R}$  to establish Hölder continuity of bounded entire solutions under the  $L^p$  norm in Proposition 4.3.7 below. This result seems to be interesting and its proof is a little more involved than those in [40] and [52]. We first give the following lemma.

**Lemma 4.3.6.** For every  $2 satisfying <math>\alpha_p := \frac{N}{2}(\frac{1}{2} - \frac{1}{p}) \in (0, 1)$ , there exists a constant  $C_p > 0$  depending only on p such that

$$\|S(t) - S(s)\|_{\mathscr{L}(L^2(\Omega), L^p(\Omega))} \le \frac{C_p}{s} |t - s|^{1 - \alpha_p},$$

for all  $0 < s \leq t$ .

*Proof.* Let 0 < s < t be arbitrary. Using a standard heat kernel estimate (4.35) and the estimate  $||AS(t)||_{\mathscr{L}(L^p(\Omega))} \leq \frac{c_p}{t}$  for t > 0 (see e.g. [40, p.21] or [66, Chapter 2, Theorem 5.2 (d)]), we compute

$$\begin{split} \|S(t)x - S(s)x\|_{L^{p}(\Omega)} &\leq \int_{s}^{t} \left\| \frac{d}{d\tau} S(\tau)x \right\|_{L^{p}(\Omega)} d\tau \\ &= \int_{s}^{t} \|AS(\tau)x\|_{L^{p}(\Omega)} d\tau \\ &\leq \int_{s}^{t} \|AS(s)\|_{\mathscr{L}(L^{p}(\Omega))} \|A(\tau - s)\|_{\mathscr{L}(L^{2}(\Omega), L^{p}(\Omega))} \|x\|_{L^{2}(\Omega)} d\tau \\ &\leq \|x\|_{L^{2}(\Omega)} \frac{c_{p}}{s} \int_{s}^{t} C(\tau - s)^{-\alpha_{p}} d\tau \\ &= \|x\|_{L^{2}(\Omega)} \frac{c_{p}}{s} \frac{C}{1 - \alpha_{p}} (t - s)^{1 - \alpha_{p}} \\ &\leq \frac{C_{p}}{s} (t - s)^{1 - \alpha_{p}} \|x\|_{L^{2}(\Omega)}, \end{split}$$

for all  $x \in L^2(\Omega)$  where we set  $C_p := \frac{c_p C}{1-\alpha_p}$ . This implies the assertion of the lemma.  $\Box$ 

**Proposition 4.3.7.** Let g be a function in  $BC(\mathbb{R}, L^2(\Omega))$ . For every p, q with  $2 and <math>\alpha_p := \frac{N}{2}(\frac{1}{2} - \frac{1}{p}), \alpha_q := \frac{N}{2}(\frac{1}{2} - \frac{1}{q}) \in (0, 1)$ , there exists a constant C > 0 depending on  $||g||_{\infty}, p, q$ , the bound M and the exponent  $\beta$  of the exponential dichotomy such that the bounded entire solution  $u = L^{-1}g$  of (4.2) satisfies

$$||u(t_1) - u(t_2)||_{L^p(\Omega)} \le C|t_1 - t_2|^{1-\alpha_q},$$

for all  $t_1, t_2 \in \mathbb{R}$ .

*Proof.* If  $|t_1 - t_2| \ge 1$ , then we see from the proof of Theorem 4.3.3 that

$$\begin{aligned} |u(t_{1}) - u(t_{2})||_{L^{p}(\Omega)} &\leq 2 \sup_{t \in \mathbb{R}} ||u(t)||_{L^{p}(\Omega)} \\ &\leq 2 ||g||_{\infty} \left( M_{1}(p) \left(\frac{2}{\beta}\right)^{1-\alpha_{p}} \Gamma(1-\alpha_{p}) + \frac{M_{2}(p)}{\beta} \right) \\ &\leq 2 ||g||_{\infty} \left( M_{1}(p) \left(\frac{2}{\beta}\right)^{1-\alpha_{p}} \Gamma(1-\alpha_{p}) + \frac{M_{2}(p)}{\beta} \right) |t_{1} - t_{2}|^{1-\alpha_{q}} \\ &\leq C_{1} |t_{1} - t_{2}|^{1-\alpha_{q}}, \end{aligned}$$

$$(4.36)$$

for some constant  $C_1 > 0$  depending on the parameters listed in the theorem (excluding q).

Let now  $t_1, t_2 \in \mathbb{R}$  with  $t_1 < t_2$  and  $|t_1 - t_2| < 1$ . We set  $\delta := t_2 - t_1 > 0$  and take  $s := t_1 - 1$ . It is clear that

$$\delta < t_1 - s = 1. \tag{4.37}$$

By the variation of constants formula in the definition of bounded entire solution (Definition 4.1.1), we see that

$$\begin{aligned} \|u(t_{1}) - u(t_{2})\|_{L^{p}(\Omega)} \\ &\leq \|S(t_{1} - s) - S(t_{2} - s)\|_{\mathscr{L}(L^{2}(\Omega), L^{p}(\Omega))} \|u(s)\|_{L^{2}(\Omega)} \\ &+ \int_{s}^{t_{1}} \|S(t_{1} - \tau) - S(t_{2} - \tau)\|_{\mathscr{L}(L^{2}(\Omega), L^{p}(\Omega))} \|g(\tau)\|_{L^{2}(\Omega)} d\tau \\ &+ \int_{t_{1}}^{t_{2}} \|S(t_{2} - \tau)\|_{\mathscr{L}(L^{2}(\Omega), L^{p}(\Omega))} \|g(\tau)\|_{L^{2}(\Omega)} d\tau. \end{aligned}$$

$$(4.38)$$

Applying Lemma 4.3.6 and using  $t_1 - s = 1$ , we have

$$|S(t_{1}-s) - S(t_{2}-s)||_{\mathscr{L}^{2}(\Omega), L^{p}(\Omega))}||u(s)||_{L^{2}(\Omega)} \leq \frac{C_{p}}{t_{1}-s}|t_{1}-t_{2}|^{1-\alpha_{p}}||u||_{\infty}$$

$$\leq C_{p}||u||_{\infty}|t_{1}-t_{2}|^{1-\alpha_{p}}$$

$$\leq C_{p}||L^{-1}|||g||_{\infty}|t_{1}-t_{2}|^{1-\alpha_{p}}.$$
(4.39)

Moreover, a standard heat kernel estimate (4.35) implies

$$\int_{t_1}^{t_2} \|S(t_2 - \tau)\|_{\mathscr{L}(L^2(\Omega), L^p(\Omega))} \|g(\tau)\|_{L^2(\Omega)} d\tau \le \int_{t_1}^{t_2} C(t_2 - \tau)^{-\alpha_p} \|g\|_{\infty} d\tau = \frac{C}{1 - \alpha_p} \|g\|_{\infty} |t_1 - t_2|^{1 - \alpha_p}.$$
(4.40)

The most difficult part is to estimate the second term on the right hand side of (4.38) which involves a singularity near  $\tau = t_1$ . By the change of variables, we can write

$$\begin{split} &\int_{s}^{t_{1}} \|S(t_{1}-\tau) - S(t_{2}-\tau)\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))}\|g(\tau)\|_{L^{2}(\Omega)}d\tau \\ &= \int_{0}^{t_{1}-s} \|S(\tau) - S(t_{2}-t_{1}+\tau)\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))}\|g(t_{1}-\tau)\|_{L^{2}(\Omega)}d\tau \\ &= \int_{0}^{\delta} \|S(\tau) - S(\delta+\tau)\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))}\|g(t_{1}-\tau)\|_{L^{2}(\Omega)}d\tau \\ &+ \int_{\delta}^{1} \|S(\tau) - S(\delta+\tau)\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))}\|g(t_{1}-\tau)\|_{L^{2}(\Omega)}d\tau, \end{split}$$
(4.41)

where we use that  $\delta = t_2 - t_1 < t_1 - s = 1$  from (4.37). Now, by the heat kernel estimate
(4.35), we get

$$\begin{split} &\int_{0}^{\delta} \|S(\tau) - S(\delta + \tau)\|_{\mathscr{L}(L^{2}(\Omega), L^{p}(\Omega))} \|g(t_{1} - \tau)\|_{L^{2}(\Omega)} d\tau \\ &\leq \|g\|_{\infty} \int_{0}^{\delta} C[\tau^{-\alpha_{p}} + (\delta + \tau)^{-\alpha_{p}}] d\tau \\ &= \|g\|_{\infty} \frac{C}{1 - \alpha_{p}} [\delta^{1 - \alpha_{p}} + (2\delta)^{1 - \alpha_{p}} - \delta^{1 - \alpha_{p}}] \\ &= \|g\|_{\infty} \frac{2^{1 - \alpha_{p}} C}{1 - \alpha_{p}} |t_{1} - t_{2}|^{1 - \alpha_{p}}. \end{split}$$
(4.42)

By Lemma 4.3.6, we have

$$\begin{split} &\int_{\delta}^{1} \|S(\tau) - S(\delta + \tau)\|_{\mathscr{L}(L^{2}(\Omega), L^{p}(\Omega))} \|g(t_{1} - \tau)\|_{L^{2}(\Omega)} d\tau \\ &\leq \|g\|_{\infty} \int_{\delta}^{1} \frac{C_{p}}{\tau} \delta^{1 - \alpha_{p}} d\tau \\ &= \|g\|_{\infty} C_{p} |t_{1} - t_{2}|^{1 - \alpha_{q}} \int_{\delta}^{1} \frac{1}{\tau} \delta^{\alpha_{q} - \alpha_{p}} d\tau. \end{split}$$

Note that by our assumption,  $\alpha_q - \alpha_p = \frac{N}{2}(\frac{1}{p} - \frac{1}{q}) \in (0, 1)$  holds. Thus,  $\delta^{\alpha_q - \alpha_p} \leq \tau^{\alpha_q - \alpha_p}$  for all  $\tau \in [\delta, 1]$ . Hence, it follows that

$$\int_{\delta}^{1} \|S(\tau) - S(\delta + \tau)\|_{\mathscr{L}(L^{2}(\Omega), L^{p}(\Omega))} \|g(t_{1} - \tau)\|_{L^{2}(\Omega)} d\tau 
\leq \|g\|_{\infty} C_{p} |t_{1} - t_{2}|^{1 - \alpha_{q}} \int_{\delta}^{1} \tau^{\alpha_{q} - \alpha_{p} - 1} d\tau 
= \|g\|_{\infty} \frac{C_{p}}{\alpha_{q} - \alpha_{p}} |t_{1} - t_{2}|^{1 - \alpha_{q}} \left(1^{\alpha_{q} - \alpha_{p}} - \delta^{\alpha_{q} - \alpha_{p}}\right) 
\leq \|g\|_{\infty} \frac{C_{p}}{\alpha_{q} - \alpha_{p}} |t_{1} - t_{2}|^{1 - \alpha_{q}}.$$
(4.43)

Combining (4.42) and (4.43), we get from (4.41) that

$$\int_{s}^{t_{1}} \|S(t_{1}-\tau) - S(t_{2}-\tau)\|_{\mathscr{L}(L^{2}(\Omega),L^{p}(\Omega))}\|g(\tau)\|_{L^{2}(\Omega)}d\tau 
\leq \|g\|_{\infty} \frac{2^{1-\alpha_{p}}C}{1-\alpha_{p}}|t_{1}-t_{2}|^{1-\alpha_{p}} + \|g\|_{\infty} \frac{C_{p}}{\alpha_{q}-\alpha_{p}}|t_{1}-t_{2}|^{1-\alpha_{q}}.$$
(4.44)

Since  $|t_1 - t_2| < 1$ , we see that

$$|t_1 - t_2|^{1 - \alpha_p} = |t_1 - t_2|^{1 - \alpha_q} |t_1 - t_2|^{\alpha_q - \alpha_p} \le |t_1 - t_2|^{1 - \alpha_q}.$$

Therefore, we obtain from (4.38) - (4.40), and (4.44) that

$$||u(t_1) - u(t_2)||_{L^p(\Omega)} \le C_2 |t_1 - t_2|^{1 - \alpha_q},$$

for all  $t_1, t_2 \in \mathbb{R}$  with  $|t_1 - t_2| < 1$  where  $C_2 > 0$  is a constant depending on the parameters indicated in the theorem. Taking C to be the maximum of  $C_1$  and  $C_2$ , the assertion of the proposition follows.

Remark 4.3.8. Let  $g_n \in BC(\mathbb{R}, L^2(\Omega))$  be such that  $||g_n||_{\infty}$  is uniformly bounded. Using Hölder continuity of bounded entire solutions in Proposition 4.3.7, we have that the family  $\{L^{-1}g_n : n \in \mathbb{N}\}$  is equicontinuous on any compact subset of  $\mathbb{R}$ . Hence, we can simplify the proof of Proposition 4.1.6. Indeed, the uniform convergence of  $L^{-1}g_n(t) \rightarrow$  $L^{-1}g(t)$  with respect to t in compact subsets of  $\mathbb{R}$  follows from the above discussion and Arzelà -Ascoli theorem for vector valued functions (see [43, Proposition 7.3]).

## 4.4 Domain Perturbation for bounded entire solutions of semilinear parabolic equations

In this section, we study the persistence of bounded entire solutions of semilinear parabolic equations under domain perturbation. For a certain class of bounded entire solutions, we prove that the perturbed problem has a bounded entire solution closed to a *known* bounded entire solution of the unperturbed equation. The existence of bounded entire solutions of semilinear equations is generally difficult to prove. One of standard techniques to deal with nonlinear problems is applying degree theory to the equivalent fixed point problem. In particular, the *Leray-Schauder degree* ([43, Chapter 2.8] or [62, Chapter 4]) is commonly used when working in Banach spaces. In [35], persistence result is proved for periodic solutions using Leray-Schauder degree theory. Indeed, the same technique was used in the context of nonlinear elliptic equations in [31]. Since the Leray-Schauder degree is defined for completely continuous (compact) perturbations of the identity, this method relies on compactness of the corresponding fixed point map. In general, for solutions of semilinear problems in  $BC(\mathbb{R}, L^2(\Omega))$ , we do not have such compactness.

For example, consider the autonomous equation

$$u'(t) + Au(t) = f(u(t)) \quad \text{for } t \in \mathbb{R}$$

in  $L^2(\Omega)$ , where A is the operator as in the previous sections and  $f: L^2(\Omega) \to L^2(\Omega)$  is globally Lipschitz continuous with f(0) = 0. It is easy to see that  $u \in BC(\mathbb{R}, L^2(\Omega))$  is a bounded entire solution of the above equation if and only if u is a fixed point of the map  $Q: BC(\mathbb{R}, L^2(\Omega)) \to BC(\mathbb{R}, L^2(\Omega))$  defined by

$$Q(u) := L^{-1}f(u(\cdot)),$$

where  $L^{-1}$  is given by (4.7). Let  $u \in BC(\mathbb{R}, L^2(\Omega))$ . The  $\tau$ -translation  $u_{\tau}$  of u defined by (4.10) belongs to  $BC(\mathbb{R}, L^2(\Omega))$  and  $||u_{\tau}||_{\infty} = ||u||_{\infty}$  for all  $\tau \in \mathbb{R}$ . By Proposition 4.1.7 (i), we have  $Q(u_{\tau}) = (Q(u))_{\tau}$  for all  $\tau \in \mathbb{R}$ . In particular, choosing a non-constant  $u \in C_0(\mathbb{R}, L^2(\Omega)) \subset BC(\mathbb{R}, L^2(\Omega))$  by

$$u(t) = \begin{cases} e^{-\frac{1}{1-t^2}} u_0 & t \in (-1,1) \\ 0 & |t| \ge 1, \end{cases}$$

where  $u_0 \in L^2(\Omega)$ , we have that f(u(t)) = 0 for  $|t| \ge 1$ . The family  $\{Q(u_\tau)\}_{\tau \in \mathbb{R}}$ is the translation of Q(u) containing no convergent subsequence in  $BC(\mathbb{R}, L^2(\Omega))$ . Therefore, Q is not completely continuous.

In this work, we restrict to study the persistence of bounded entire solutions in the class of  $C_0(\mathbb{R}, L^2(\Omega))$  for semilinear parabolic equations

$$u'(t) + Au(t) = f(t, u(t)) \quad \text{for } t \in \mathbb{R}$$

$$(4.45)$$

in  $L^2(\Omega)$ , where  $f : \mathbb{R} \times L^2(\Omega) \to L^2(\Omega)$  is continuous. As in the case of linear problems, we assume that Assumption 4.2.1 is satisfied. Furthermore, we assume that f is of the form  $f(t,\xi) = \phi(t)h(\xi)$  for some continuous functions  $\phi : \mathbb{R} \to \mathbb{R}$  and  $h : L^2(\Omega) \to L^2(\Omega)$ .

Under suitable assumptions, we show that Q is completely continuous as a map on  $C_0(\mathbb{R}, L^2(\Omega))$ . Hence, the Leray-Schauder degree theory can be applied. We discuss persistence of  $C_0(\mathbb{R}, L^2(\Omega))$  solutions in Section 4.4.1 below. It is also worthwhile to remark the case of almost periodic solutions where Q fails to be completely continuous. We comment on this difficulty in Section 4.4.2.

## 4.4.1 Persistence of $C_0(\mathbb{R}, L^2(\Omega))$ solutions

We investigate the persistence of bounded entire solutions of semilinear equations (4.45) with the nonlinear term  $f(\cdot, u(\cdot))$  converges to zero as  $t \to \pm \infty$ .

Assumption 4.4.1. Assume that  $f : \mathbb{R} \times L^2(\Omega) \to L^2(\Omega)$  is of the form  $f(t,\xi) = \phi(t)h(\xi)$ for  $(t,\xi) \in \mathbb{R} \times L^2(\Omega)$  where

(i) 
$$\phi \in C_0(\mathbb{R});$$

(ii)  $h: L^2(\Omega) \to L^2(\Omega)$  is locally Lipschitz, that is for any R > 0, there exists  $k_R > 0$ such that

$$\|h(\xi_1) - h(\xi_2)\|_{L^2(\Omega)} \le k_R \|\xi_1 - \xi_2\|_{L^2(\Omega)},$$

for all  $\xi_1, \xi_2 \in L^2(\Omega)$  with  $\|\xi_1\|_{L^2(\Omega)}, \|\xi_2\|_{L^2(\Omega)} \leq R$ .

Note that we could assume as in Section 2.4 that h is a substitution operator of a function  $\theta : \mathbb{R}^N \times \mathbb{R} \to \mathbb{R}$  satisfying the conditions similar to Assumption 2.4.2 (with g replaced by  $\theta$ ), that is  $h(\xi)(x) = \theta(x, \xi(x))$  a.e.  $x \in \Omega$ .

By the above assumption, it follows that for  $u \in BC(\mathbb{R}, L^2(\Omega))$ , we have

$$\begin{aligned} \|f(t,u(t))\|_{L^{2}(\Omega)} &= \|\phi(t)h(u(t))\|_{L^{2}(\Omega)} \\ &\leq |\phi(t)| \big(\|h(0)\|_{L^{2}(\Omega)} + \|h(u(t)) - h(0)\|_{L^{2}(\Omega)}\big) \\ &\leq |\phi(t)| \big(\|h(0)\|_{L^{2}(\Omega)} + k_{\|u\|_{\infty}} \|u(t)\|_{L^{2}(\Omega)}\big) \\ &\leq |\phi(t)| \big(\|h(0)\|_{L^{2}(\Omega)} + k_{\|u\|_{\infty}} \|u\|_{\infty}\big), \end{aligned}$$

$$(4.46)$$

for all  $t \in \mathbb{R}$ . Since  $\phi$  belongs to  $C_0(\mathbb{R})$ , we have that  $f(\cdot, u(\cdot)) \in C_0(\mathbb{R}, L^2(\Omega))$  for all  $u \in BC(\mathbb{R}, L^2(\Omega))$ . Hence, Proposition 4.1.7 (ii) implies that the map Q defined by

$$Q(u) := L^{-1} f(\cdot, u(\cdot))$$
(4.47)

maps  $C_0(\mathbb{R}, L^2(\Omega))$  into itself, that is  $Q: C_0(\mathbb{R}, L^2(\Omega)) \to C_0(\mathbb{R}, L^2(\Omega))$ . Moreover, u is a solution of (4.45) in  $C_0(\mathbb{R}, L^2(\Omega))$  if and only if u is a fixed point of Q.

Let  $\Omega_n$  and  $\Omega$  be domains satisfying Assumption 1.4.1. We consider the perturbation of (4.45) of the form

$$u'(t) + A_n u(t) = f_n(t, u(t)) \quad \text{for } t \in \mathbb{R}$$

$$(4.48)$$

in  $L^2(\Omega_n)$ , where  $f_n : \mathbb{R} \times L^2(\Omega_n) \to L^2(\Omega_n)$  is continuous. We assume that Assumption 4.2.2 is satisfied. Moreover, we impose the following assumptions on the nonlinearity  $f_n$ .

Assumption 4.4.2. Assume that  $f_n : \mathbb{R} \times L^2(\Omega) \to L^2(\Omega)$  is of the form  $f_n(t,\xi) = \phi_n(t)h_n(\xi)$  for  $(t,\xi) \in \mathbb{R} \times L^2(\Omega_n)$  where

- (i)  $\phi_n \in C_0(\mathbb{R});$
- (ii)  $h_n: L^2(\Omega_n) \to L^2(\Omega_n)$  is locally Lipschitz uniformly with respect to  $n \in \mathbb{N}$ , that is for any R > 0, there exists  $k_R > 0$  independent of  $n \in \mathbb{N}$  such that

$$\|h_n(\xi_1) - h_n(\xi_2)\|_{L^2(\Omega_n)} \le k_R \|\xi_1 - \xi_2\|_{L^2(\Omega_n)}$$

for all  $\xi_1, \xi_2 \in L^2(\Omega_n)$  with  $\|\xi_1\|_{L^2(\Omega_n)}, \|\xi_2\|_{L^2(\Omega_n)} \leq R$  and for all  $n \in \mathbb{N}$ ;

(iii)  $\phi_n \to \phi$  in  $BC(\mathbb{R})$  and  $ri_n h_n(r_n \xi) \rightharpoonup h(r\xi)$  in  $L^2(\Omega)$  weakly for all  $\xi \in L^2(D)$ .

One important example of functions  $f_n$  satisfying Assumption 4.4.2 is the restriction of a fixed semilinear term to the domains  $\Omega_n$ . We discuss this in the example below.

Example 4.4.3. Let  $\phi \in C_0(\mathbb{R})$  and  $\theta : \mathbb{R}^N \times \mathbb{R} \to \mathbb{R}$  with  $\theta(x,0) = 0$  for almost everywhere  $x \in \mathbb{R}^N$ . Denoted by h the substitution operator  $h(\xi)(x) = \theta(x,\xi(x))$ for  $\xi \in L^2(D)$ . We assume that  $h \in C(L^2(D), L^p(D))$  for some p > 2 and h is locally Lipschitz. Note that a sufficient condition for the acting and locally Lipschitz continuity is similar to Assumption 2.4.2 with some modification to achieve the acting from  $L^2(D)$ to  $L^p(D)$  (see [35, Section 6] for the precise conditions and [3, Chapter 3] for general details).

Since  $\theta(x,0) = 0$  for a.e.  $x \in \mathbb{R}^N$ , we see that if  $\xi \in L^2(\Omega)$  we have  $h(i\xi) = 0$ a.e.  $x \in D \setminus \Omega$ . This implies that h maps  $L^2(\Omega)$  into itself if we identify a function in  $L^2(\Omega)$  with the trivial extension  $i\xi \in L^2(D)$ . A similar statement holds for a function in  $L^2(\Omega_n)$ . We take

$$f_n(t,\xi) := \phi(t)r_n h(i_n\xi) \qquad \text{for } (t,\xi) \in \mathbb{R} \times L^2(\Omega_n)$$
$$f(t,\xi) := \phi(t)r h(i\xi) \qquad \text{for } (t,\xi) \in \mathbb{R} \times L^2(\Omega).$$

Recall from Lemma 1.4.8 that Mosco convergence implies  $\operatorname{cap}(K \cap \Omega_n^c) \to 0$  as  $n \to \infty$ for all compact set  $K \subset \Omega$ . Since  $rh(ir\xi) \in L^p(\Omega)$ , we have  $\|rh(ir\xi)\|_{L^2(\Omega \setminus \Omega_n)} \to 0$  for all  $\xi \in L^2(\Omega)$ . It follows that  $rh(i_n r_n \xi) \to rh(ir\xi)$  in  $L^2(\Omega)$ . Hence, condition (iii) of Assumption 4.4.2 is satisfied.

A similar calculation as in (4.46) shows that  $f_n(\cdot, u(\cdot)) \in C_0(\mathbb{R}, L^2(\Omega_n))$  for all  $u \in BC(\mathbb{R}, L^2(\Omega_n))$ . Moreover,  $u_n$  is a solution of (4.48) in  $C_0(\mathbb{R}, L^2(\Omega_n))$  if and only if  $u_n$  is a fixed point of  $Q_n : C_0(\mathbb{R}, L^2(\Omega_n)) \to C_0(\mathbb{R}, L^2(\Omega_n))$  defined by

$$Q_n(u) := L_n^{-1} f_n(\cdot, u(\cdot)).$$
(4.49)

As usual, we need to consider the inclusion and the restriction operators when dealing with domain perturbation. In particular, we consider the maps  $Q_n$  and Q on  $C_0(\mathbb{R}, L^2(D))$  into itself given by

$$Q_n(u) := i_n Q_n(r_n(u)) = i_n L_n^{-1} f_n(\cdot, r_n u(\cdot))$$
  

$$Q(u) := i Q(r(u)) = i L^{-1} f(\cdot, r u(\cdot)),$$
(4.50)

for all  $u \in C_0(\mathbb{R}, L^2(D))$ .

Our next step is to show that  $Q_n$  and Q are completely continuous.

**Lemma 4.4.4.** Let  $\{v_n\}_{n\in\mathbb{N}}$  be a bounded sequence in  $BC(\mathbb{R}, L^2(D))$ . Then for every  $t \in \mathbb{R}$ , the families  $\{\mathcal{Q}_n(v_n)(t)\}_{n\in\mathbb{N}}$  and  $\{\mathcal{Q}(v_n)(t)\}_{n\in\mathbb{N}}$  are relatively compact in  $L^2(D)$ .

*Proof.* Denoted by  $g_n \in C_0(\mathbb{R}, L^2(\Omega_n))$  where  $g_n(t) := f_n(t, r_n v_n(t))$  for all  $t \in \mathbb{R}$ . By the uniform boundedness of  $||v_n||_{\infty}$  and the assumptions on  $f_n$ , there exists C > 0 such that

$$||g_n||_{\infty} = \sup_{t \in \mathbb{R}} |\phi_n(t)| ||h_n(r_n v_n(t))||_{L^2(\Omega_n)} < C,$$
(4.51)

for all  $n \in \mathbb{N}$ . Hence,  $\|\mathcal{Q}_n(v_n)\|_{\infty}$  is uniformly bounded.

Fix now  $t_0 \in \mathbb{R}$ . We can find a subsequence  $\mathcal{Q}_{n_k}(v_{n_k})(t_0-1)$  such that

$$r\mathcal{Q}_{n_k}(v_{n_k})(t_0-1) \rightharpoonup \xi$$

in  $L^2(\Omega)$ . By extracting a further subsequence (indexed again by  $n_k$ ), we can assume that  $ri_{n_k}g_{n_k} \rightharpoonup g$  in  $L^2((t_0-1, t_0+1), L^2(\Omega))$  weakly. Since  $Q_n(u_n) = L_n^{-1}f_n(\cdot, r_nv_n(\cdot))$ , we have that the restriction of  $Q_{n_k}(v_{n_k})$  to  $[t_0-1, t_0+1]$  satisfies

$$\begin{cases} v'(t) + A_{n_k}v(t) = f_{n_k}(t, r_{n_k}v(t)), & t \in (t_0 - 1, t_0 + 1] \\ v(t_0 - 1) = Q_{n_k}(r_{n_k}v_{n_k})(t_0 - 1). \end{cases}$$

By Theorem 1.4.9, we have

$$i_{n_k}Q_{n_k}(r_{n_k}v_{n_k}) = \mathcal{Q}_{n_k}(v_{n_k})(t) \to iv(t)$$

in  $L^2(D)$  on  $t \in (t_0 - 1, t_0]$ , where v is the solution of

$$\begin{cases} v'(t) + Av(t) = g(t), & t \in (t_0 - 1, t_0 + 1] \\ v(t_0 - 1) = \xi. \end{cases}$$

In particular,  $\mathcal{Q}_{n_k}(v_{n_k})(t_0) \to iv(t_0)$  in  $L^2(D)$ . Since this is true for every  $t_0 \in \mathbb{R}$ , we have the relative compactness of  $\{\mathcal{Q}_n(v_n)(t)\}_{n\in\mathbb{N}}$  in  $L^2(D)$  for all  $t\in\mathbb{R}$ . The relative compactness of  $\{\mathcal{Q}(v_n)(t)\}_{n\in\mathbb{N}}$  is a special case of what we have just proved (by taking a sequence of constant domains  $\Omega_n = \Omega$  for all  $n \in \mathbb{N}$ ).

Remark 4.4.5. The assertion of Lemma 4.4.4 remains valid for a more general map  $\mathcal{Q}: BC(\mathbb{R}, L^2(D)) \to BC(\mathbb{R}, L^2(D))$  (not just for the map on  $C_0(\mathbb{R}, L^2(D))$  into itself) as long as

$$||g_n||_{\infty} := \sup_{t \in \mathbb{R}} ||f_n(t, r_n v_n(t))||_{L^2(\Omega_n)}$$

is uniformly bounded, that is (4.51) holds.

**Theorem 4.4.6.** For every bounded sequence  $\{v_n\}_{n\in\mathbb{N}}$  in  $C_0(\mathbb{R}, L^2(D))$ , the sequence  $\mathcal{Q}_n(v_n)$  has a convergence subsequence in  $C_0(\mathbb{R}, L^2(D))$ . In particular,

$$\mathcal{Q}_n, \mathcal{Q}: C_0(\mathbb{R}, L^2(D)) \to C_0(\mathbb{R}, L^2(D))$$

are completely continuous.

Proof. Denoted by  $g_n \in C_0(\mathbb{R}, L^2(\Omega_n))$  where  $g_n(t) := f_n(t, r_n v_n(t))$  for all  $t \in \mathbb{R}$ . We have from (4.51) that  $||g_n||_{\infty}$  is uniformly bounded. Moreover, by Lemma 4.4.4,  $\{Q_n(v_n)(t)\}_{n\in\mathbb{N}}$  is relatively compact in  $L^2(D)$  for all  $t \in \mathbb{R}$ . By our assumptions, we have the uniform exponential dichotomy (see Lemma 4.2.3). Hence, it follows from Theorem 4.3.7 that

$$\|\mathcal{Q}_n(v_n)(t_1) - \mathcal{Q}_n(v_n)(t_2)\|_{L^2(D)} \le C|t_1 - t_2|^{1 - \alpha_q},$$

for all  $t_1, t_2 \in \mathbb{R}$  and for all  $n \in \mathbb{N}$ , where the constant C is independent of  $n \in \mathbb{N}$  and q is chosen so that  $2 and <math>\alpha_p := \frac{N}{2}(\frac{1}{2} - \frac{1}{p}), \alpha_q := \frac{N}{2}(\frac{1}{2} - \frac{1}{q}) \in [0, 1)$ . This means the family  $\{\mathcal{Q}_n(v_n)\}_{n\in\mathbb{N}}$  is equicontinuous on  $\mathbb{R}$ . By Arzelà-Ascoli theorem for vector valued functions (see e.g. [43, Proposition 7.3(b)]), we can extract a subsequence  $\{\mathcal{Q}_{1,n}(v_{1,n})\}_{n\in\mathbb{N}}$  of  $\{\mathcal{Q}_n(v_n)\}_{n\in\mathbb{N}}$  so that  $\mathcal{Q}_{1,n}(v_{1,n})$  converges uniformly on [-1, 1]. Applying Arzelà-Ascoli theorem again, we can extract a further subsequence  $\{\mathcal{Q}_{2,n}(v_{2,n})\}_{n\in\mathbb{N}}$  of  $\{\mathcal{Q}_{1,n}(v_{1,n})\}_{n\in\mathbb{N}}$  so that  $\mathcal{Q}_{2,n}(v_{2,n})$  converges uniformly on [-2, 2]. Applying the above argument inductively, we obtain for every  $k \geq 2$  a subsequence  $\{\mathcal{Q}_{k,n}(v_{k,n})\}_{n\in\mathbb{N}}$  of  $\{\mathcal{Q}_{(k-1),n}(v_{(k-1),n})\}_{n\in\mathbb{N}}$  such that  $\{\mathcal{Q}_{k,n}(v_{k,n})\}_{n\in\mathbb{N}}$  converges uniformly on [-k, k]. By a standard diagonal argument, we see that  $\mathcal{Q}(v_{n,n})(t)$  converges uniformly on any compact sets of  $\mathbb{R}$ . In particular,  $\mathcal{Q}_{n,n}(v_{n,n})(t) \to v_t$  in  $L^2(D)$  for all  $t \in \mathbb{R}$  with the limit  $v_t \in L^2(D)$ . Let  $v(t) := v_t$  for  $t \in \mathbb{R}$ . We have that v is continuous on  $\mathbb{R}$  and sup\_{t\in\mathbb{R}} ||v(t)||\_{L^2(D)} is finite. It remains to show that  $\mathcal{Q}_{n,n}(v_{n,n})$  converges to v uniformly on  $\mathbb{R}$ .

Let  $\varepsilon > 0$  be arbitrary. By a calculation as in (4.46) and the assumption that  $\phi_n \to \phi$ in  $C_0(\mathbb{R})$ , we can choose T > 0 independent of  $n \in \mathbb{N}$  so that

$$||g_n(t)||_{L^2(\Omega_n)} = ||f_n(t, r_n v_n(t))||_{L^2(\Omega_n)} < \frac{\beta}{4M} \varepsilon$$

if |t| > T for all  $n \in \mathbb{N}$ . By the uniform boundedness of  $||g_n||_{\infty}$ , we can find  $\tilde{T} > T$  independent of  $n \in \mathbb{N}$  such that

$$\|g_n\|_{\infty}\frac{M}{\beta}e^{-\beta(\tilde{T}-T)} = \|f_n(\cdot, r_n v_n(\cdot))\|_{\infty}\frac{M}{\beta}e^{-\beta(\tilde{T}-T)} < \frac{\varepsilon}{2},$$

for all  $n \in \mathbb{N}$ . By the argument appeared in the proof of Proposition 4.1.8, the above two inequalities imply  $\|L_n^{-1}f_n(t, v_n(t))\|_{L^2(\Omega)} < \varepsilon$  if  $|t| > \tilde{T}$  for all  $n \in \mathbb{N}$ . Therefore, the limit function v satisfies  $\|v(t)\|_{L^2(\Omega)} \le \varepsilon$  if  $|t| > \tilde{T}$ . Hence,

$$\|\mathcal{Q}_{n,n}(v_{n,n})(t) - v(t)\|_{L^2(\Omega)} \le \varepsilon + \varepsilon = 2\varepsilon \tag{4.52}$$

if  $|t| > \tilde{T}$  for all  $n \in \mathbb{N}$ . As  $\mathcal{Q}_{n,n}(v_{n,n})$  converges uniformly on the compact interval [-k, k] for all  $k \in \mathbb{N}$ , there exists  $N \in \mathbb{N}$  such that

$$\|\mathcal{Q}_{n,n}(v_{n,n})(t) - v(t)\|_{L^2(D)} \le \varepsilon,$$
(4.53)

for all  $t \in [-\tilde{T}, \tilde{T}]$  and for all n > N. Therefore, (4.52) and (4.53) implies

$$\|\mathcal{Q}_{n,n}(v_{n,n})(t) - v(t)\|_{L^2(D)} \le 2\varepsilon,$$

for all  $t \in \mathbb{R}$  and for all n > N. As  $\varepsilon > 0$  was arbitrary, this proves the uniform convergence of  $\mathcal{Q}_{n,n}(v_{n,n})$  on  $\mathbb{R}$ . Since  $C_0(\mathbb{R}, L^2(D))$  is a closed subspace of  $BC(\mathbb{R}, L^2(D))$ , the limit v belongs to  $C_0(\mathbb{R}, L^2(D))$ . This gives the first assertion of the theorem.

The complete continuity of  $\mathcal{Q}$  immediately follows by the same argument above as a special case with  $\Omega_n := \Omega$  for all  $n \in \mathbb{N}$ . Similarly, by choosing a sequence of  $\Omega_n := \Omega_{n_0}$  for all  $n \in \mathbb{N}$  where  $n_0$  is fixed, we obtain complete continuity of  $\mathcal{Q}_{n_0}$ .

If U is a bounded open subset of  $C_0(\mathbb{R}, L^2(D))$  such that  $u \neq \mathcal{Q}(u)$  for all  $u \in \partial U$ , then by complete continuity of  $\mathcal{Q}$ , the Leray-Schauder degree  $\deg(I - \mathcal{Q}, U, 0) \in \mathbb{Z}$ is well-defined. We show that the Leray-Schauder degree is persistent under domain perturbation. The proof is similar to the argument for periodic solutions of parabolic equations in [35, Theorem 7.1] or the argument for  $L^{\infty}$  solutions of elliptic equations in [5, Theorem 8.2]. However, we also consider perturbation of the nonlinear terms. We first give the following lemma.

**Lemma 4.4.7.** Suppose that  $f_n$  and f satisfy Assumptions 4.4.1 and Assumption 4.4.2 respectively. If  $v \in C_0(\mathbb{R}, L^2(D))$ , then

$$ri_n f_n(\cdot, r_n v(\cdot)) \stackrel{*}{\rightharpoonup} f(\cdot, rv(\cdot))$$

in  $L^{\infty}(\mathbb{R}, L^2(\Omega))$  as  $n \to \infty$ .

*Proof.* By a similar calculation as in (4.46), we have

$$||f_n(\cdot, r_n v(\cdot))||_{\infty} \le ||\phi_n||_{\infty} (||h_n(0)||_{L^2(\Omega_n)} + k_{||v||_{\infty}} ||v||_{\infty})$$
$$||f(\cdot, rv(\cdot))||_{\infty} \le ||\phi||_{\infty} (||h(0)||_{L^2(\Omega)} + k_{||v||_{\infty}} ||v||_{\infty}).$$

As  $\phi_n \to \phi$  in  $BC(\mathbb{R})$  and  $ri_nh(0) \to h(0)$  in  $L^2(\Omega)$  (condition (iii) in Assumption 4.4.2), it follows that  $ri_nf_n(\cdot, r_nv(\cdot))$  and  $f(\cdot, rv(\cdot))$  are uniformly bounded in  $L^{\infty}(\mathbb{R}, L^2(\Omega))$ . By Assumption 4.4.2 (iii) again, we see that

$$ri_n f_n(t, r_n v(t)) - f(t, rv(t)) = \phi_n(t) ri_n h_n(r_n v(t)) - \phi(t) h(rv(t))$$
  
=  $\phi_n(t) (ri_n h_n(r_n v(t)) - h(rv(t)))$   
+  $(\phi_n(t) - \phi(t)) h(rv(t))$   
 $\rightarrow 0$  (4.54)

in  $L^2(\Omega)$  weakly as  $n \to \infty$  for all  $t \in \mathbb{R}$ .

Let  $g \in L^1(\mathbb{R}, L^2(\Omega))$  be arbitrary. It follows from (4.54) that

$$\left(ri_n f_n(t, r_n v(t)) - f(t, rv(t)) \mid g(t)\right)_{L^2(\Omega)} \to 0$$

as  $n \to \infty$  for almost every  $t \in \mathbb{R}$ . Applying the dominated convergence theorem, we get

$$\int_{-\infty}^{\infty} \left( ri_n f_n(t, r_n v(t)) - f(t, rv(t)) \mid g(t) \right)_{L^2(\Omega)} dt \to 0$$

as  $n \to \infty$ . Since this is true for any  $g \in L^1(\mathbb{R}, L^2(\Omega))$ , we obtain the required weak-\* convergence in  $L^{\infty}(\mathbb{R}, L^2(\Omega))$ .

**Theorem 4.4.8.** Suppose that all assumptions mentioned in this subsection are satisfied. Let U be a bounded open subset of  $C_0(\mathbb{R}, L^2(D))$  such that  $u \neq \mathcal{Q}(u)$  for all  $u \in \partial U$ . Then  $u \neq \mathcal{Q}_n(u)$  for all  $u \in \partial U$  and

$$\deg(I - \mathcal{Q}_n, U, 0) = \deg(I - \mathcal{Q}, U, 0),$$

for all n sufficiently large.

*Proof.* We apply the homotopy invariance of the Leray-Schauder degree (see [43, section 2.8.3] or [62, Theorem 4.3.4]) to prove the assertion of the theorem. For this, it suffices to show that

$$v \neq \lambda \mathcal{Q}(v) + (1 - \lambda)\mathcal{Q}_n(v)$$

if  $v \in \partial U$  and  $0 \leq \lambda \leq 1$  for all *n* sufficiently large.

By way of contradiction, we assume there exists  $0 \leq \lambda_n \leq 1$  and  $v_n \in \partial U$  such that

$$v_n = \lambda_n \mathcal{Q}(v_n) + (1 - \lambda_n) \mathcal{Q}_n(v_n), \qquad (4.55)$$

for all  $n \in \mathbb{N}$ . We can assume (by possibly selecting a subsequence) that  $\lambda_n$  converges to some  $\lambda$  in [0, 1]. Since  $v_n$  is bounded in  $C_0(\mathbb{R}, L^2(D))$  and  $\mathcal{Q}$  is completely continuous (Lemma 4.4.6), we can extract a subsequence (indexed again by n) such that  $\mathcal{Q}(v_n)$ converges to some limit in  $C_0(\mathbb{R}, L^2(D))$ . Similarly, by Lemma 4.4.6, we extract a further subsequence (still indexed again by n) such that  $\mathcal{Q}_n(v_n)$  converges to some limit in  $C_0(\mathbb{R}, L^2(D))$ . Hence, we get from (4.55) that (for a subsequence)  $v_n \to v$  in  $C_0(\mathbb{R}, L^2(D))$  for some  $v \in C_0(\mathbb{R}, L^2(D))$ . Moreover,  $v \in \partial U$  because  $\partial U$  is closed.

Now by Assumption 4.4.1 and that  $||v_n||_{\infty} < R$  for some R > 0, we get

$$\|f(t, rv_n(t)) - f(t, rv(t))\|_{L^2(\Omega)} \le |\phi(t)|k_R\|rv_n(t) - rv(t)\|_{L^2(\Omega)} \le k_R \|\phi\|_{\infty} \sup_{t \in \mathbb{R}} \|rv_n(t) - rv(t)\|_{L^2(\Omega)},$$
(4.56)

for all  $t \in \mathbb{R}$ . As  $v_n \to v$  in  $C_0(\mathbb{R}, L^2(D))$ , the above implies  $f(\cdot, rv_n(\cdot)) \to f(\cdot, rv(\cdot))$ in  $C_0(\mathbb{R}, L^2(\Omega))$ . Applying Theorem 4.2.5 (with a sequence of domains  $\Omega_n = \Omega$  for all  $n \in \mathbb{N}$ ), we get

$$\mathcal{Q}(v_n) = iL^{-1}f(\cdot, rv_n(\cdot)) \to iL^{-1}f(\cdot, rv(\cdot)) = \mathcal{Q}(v)$$
(4.57)

in  $C_0(\mathbb{R}, L^2(D))$ . By a similar calculation as in (4.56) (using Assumption 4.4.2 instead), we have

$$i_n f_n(\cdot, r_n v_n(\cdot)) - i_n f_n(\cdot, r_n v(\cdot)) \to 0$$

in  $C_0(\mathbb{R}, L^2(D))$  as  $n \to \infty$ . Together with Lemma 4.4.7, it follows that

$$ri_n f_n(\cdot, r_n v_n(\cdot)) \stackrel{*}{\rightharpoonup} f(\cdot, rv(\cdot))$$

in  $L^{\infty}(\mathbb{R}, L^2(\Omega))$  as  $n \to \infty$ . We conclude from Theorem 4.2.6 that

$$\mathcal{Q}_n(v_n) = i_n L_n^{-1} f_n(\cdot, r_n v_n(\cdot)) \to i L^{-1} f(\cdot, rv(\cdot)) = \mathcal{Q}(v)$$

in  $C_0(\mathbb{R}, L^2(D))$  under the topology of uniform convergence on compact subsets. Therefore, we get from (4.55) that  $v = \lambda \mathcal{Q}(v) + (1 - \lambda)\mathcal{Q}(v) = \mathcal{Q}(v)$ . This contradicts our assumption that  $\mathcal{Q}$  has no fixed point on  $\partial U$ . By the homotopy invariance of the Leray-Schauder degree, this implies that

$$\deg(I - \mathcal{Q}_n, U, 0) = \deg(I - \mathcal{Q}, U, 0)$$

for all n large.

We are interested in the persistence of a known solution of the unperturbed problem (4.45). In Particular, if deg $(I - Q, U, 0) \neq 0$ , then (4.45) has a  $C_0(\mathbb{R}, L^2(\Omega))$  solution in U (see [62, Theorem 4.3.2]). We follow [5] to state a sequence of results below.

**Corollary 4.4.9.** Let U be a bounded open subset of  $C_0(\mathbb{R}, L^2(D))$  such that  $u \neq Q(u)$ for all  $u \in \partial U$ . If deg $(I - Q, U, 0) \neq 0$ , then (4.48) has a bounded entire solution in  $U \cap C_0(\mathbb{R}, L^2(\Omega_n))$  for all n sufficiently large.

In addition, if u is an isolated solution of (4.45), then the excision property of the degree implies that for open balls  $B(iu, \varepsilon)$  of radius  $\varepsilon$  and center iu in  $C_0(\mathbb{R}, L^2(D))$  we have deg $(I - \mathcal{Q}, B(u, \varepsilon), 0)$  is constant for small  $\varepsilon > 0$ . The index of u is defined by

$$\operatorname{index}(u) := \lim_{\varepsilon \to 0} \deg(I - \mathcal{Q}, B(iu, \varepsilon), 0).$$

**Theorem 4.4.10.** Suppose that u is an isolated solution of (4.45) with  $index(u) \neq 0$ . For n sufficiently large, there exist solutions  $u_n \in C_0(\mathbb{R}, L^2(\Omega_n))$  of (4.48) such that  $i_n u_n \to iu$  in  $C_0(\mathbb{R}, L^2(D))$  as  $n \to \infty$ .

*Proof.* By assumption there exists  $\varepsilon_0 > 0$  such that

$$\operatorname{index}(u) = \operatorname{deg}(I - \mathcal{Q}, B(u, \varepsilon), 0) \neq 0.$$

for all  $\varepsilon \in (0, \varepsilon_0)$ . Corollary 4.4.9 implies that (4.48) has a solution in  $B(iu, \varepsilon)$  for all  $\varepsilon \in (0, \varepsilon_0)$  if *n* sufficiently large. Suppose the required sequence of solutions  $u_n$  does not exist. Then we can find  $\varepsilon \in (0, \varepsilon_0)$  and a subsequence  $n_k \to \infty$  such that (4.48) (with  $n_k$  in place of *n*) has no solution in  $B(iu, \varepsilon)$  for all  $k \in \mathbb{N}$ . This gives a contradiction.  $\Box$ 

## 4.4.2 Some remarks on almost periodic solutions

The aim of this section is to discuss some difficulties of using the Leray-Schauder degree for almost periodic solutions. We first collect some preliminary results on almost periodic functions below.

**Definition 4.4.11.** Let X and Y be Banach spaces. A continuous function  $g : \mathbb{R} \times X \to Y$  is called *uniformly almost periodic* (with respect to x) if for every  $\varepsilon > 0$  and compact set  $K \subset X$  there exists  $\ell(\varepsilon, K) > 0$  such that every interval J of length  $\ell(\varepsilon, K)$  contains  $\tau$  such that  $||g(t + \tau, x) - g(t, x)||_Y < \varepsilon$  for all  $t \in \mathbb{R}$  and for all  $x \in K$ .

**Lemma 4.4.12.** Let X and Y be Banach spaces. Suppose that  $g : \mathbb{R} \times X \to Y$  is uniformly almost periodic. If  $u : \mathbb{R} \to X$  is almost periodic then the function  $t \mapsto g(t, u(t))$  is also almost periodic, that is  $g(\cdot, u(\cdot)) \in AP(\mathbb{R}, Y)$ .

*Proof.* The proof follows the same argument as either in [80, Theorem I.2.7], [47, Theorem 2.11] or [28, Theorem 2.8] for functions taking values in finite dimensional space

 $(\mathbb{R}^N)$ . The only point we need to be careful is that for almost periodic functions with values in  $\mathbb{R}^N$ , the compactness of  $\overline{\{u(t): t \in \mathbb{R}\}}$  in  $\mathbb{R}^N$  simply follows from the boundedness of |u(t)|. However, we also have the compactness of  $\overline{\{u(t): t \in \mathbb{R}\}}$  in X for  $u \in AP(\mathbb{R}, X)$  from [28, Theorem 6.5]. Hence, we can find a compact set  $K \subset X$  containing the range of u and complete the proof in the same way as  $\mathbb{R}^N$ -valued functions.

**Definition 4.4.13.** A family  $\mathscr{F}$  of functions from  $AP(\mathbb{R}, X)$  is called *equi-almost peri*odic if for every  $\varepsilon > 0$ , there exists  $\ell(\varepsilon) > 0$  such that every compact interval  $J \subset \mathbb{R}$  of length  $\ell(\varepsilon)$  contains  $\tau$  such that

$$\|f(t+\tau) - f(t)\|_X < \varepsilon,$$

for all  $t \in \mathbb{R}$  and for all  $f \in \mathscr{F}$ .

**Lemma 4.4.14** ([28, Theorem 6.10]). Let X be a Banach space. A family  $\mathscr{F}$  of functions from  $AP(\mathbb{R}, X)$  is relatively compact if and only if the following conditions hold:

- (i)  $\mathscr{F}$  is equicontinuous;
- (ii) *F* is equi-almost periodic;
- (iii) for any  $t \in \mathbb{R}$ , the set  $\{f(t) : f \in \mathscr{F}\}$  is relatively compact in X.

Suppose that f satisfies Assumption 4.4.1 with  $\phi \in C_0(\mathbb{R})$  replaced by  $\phi \in AP(\mathbb{R})$ . Then

$$\begin{aligned} \|f(t+\tau,\xi) - f(t,\xi)\|_{L^{2}(\Omega)} &= |\phi(t+\tau) - \phi(t)| \|h(\xi)\|_{L^{2}(\Omega)} \\ &\leq |\phi(t+\tau) - \phi(t)| \big( \|h(0)\|_{L^{2}(\Omega)} + k_{\|\xi\|_{L^{2}(\Omega)}} \|\xi\|_{L^{2}(\Omega)} \big), \end{aligned}$$

$$(4.58)$$

for all  $t, \tau \in \mathbb{R}$  and  $\xi \in L^2(\Omega)$ . Let K be a compact subset of  $L^2(\Omega)$ . We have

$$\|h(0)\|_{L^{2}(\Omega)} + k_{\|\xi\|_{L^{2}(\Omega)}} \|\xi\|_{L^{2}(\Omega)} < C_{K},$$

for all  $\xi \in K$ , where  $C_K > 0$  is a constant depending on the set K. By the almost periodicity of  $\phi$ , we get from (4.58) that f is uniformly almost periodic. As a consequence of Lemma 4.4.12, we have that  $f(\cdot, u(\cdot)) \in AP(\mathbb{R}, L^2(\Omega))$  for all  $u \in AP(\mathbb{R}, L^2(\Omega))$ . Hence, Proposition 4.1.7 (ii) implies that Q defined by (4.47) maps  $AP(\mathbb{R}, L^2(\Omega))$  into itself, that is  $Q : AP(\mathbb{R}, L^2(\Omega)) \to AP(\mathbb{R}, L^2(\Omega))$ . Moreover, u is an almost periodic solutions of (4.45) if and only if it is a fixed point of Q. We show that the technique of using the Leray-Schauder degree cannot be applied as Q is not completely continuous.

Let  $\{v_n\}_{n\in\mathbb{N}}$  be a bounded sequence in  $AP(\mathbb{R}, L^2(\Omega))$ . We set  $g_n(t) := f(t, v_n(t))$ for all  $t \in \mathbb{R}$ . A similar calculation as in (4.51) implies that that  $||g_n||_{\infty}$  is uniformly bounded. By Remark 4.4.5,  $\{Q(v_n)(t)\}_{n\in\mathbb{N}}$  is relatively compact in  $L^2(\Omega)$  for all  $t \in \mathbb{R}$ . In addition, Theorem 4.3.7 implies that  $\{Q(v_n)\}_{n\in\mathbb{N}}$  is equicontinuous on  $\mathbb{R}$ . The above argument shows that the family  $\{Q(v_n)\}_{n\in\mathbb{N}}$  satisfies condition (i) and condition (iii) in Lemma 4.4.14. However, we do not generally have equi-almost periodicity of the family  $\{Q(v_n)\}_{n\in\mathbb{N}}$  when  $\{v_n\}_{n\in\mathbb{N}}$  is a bounded sequence in  $AP(\mathbb{R}, L^2(\Omega))$ . To see this, let  $v_1 \in AP(\mathbb{R}, L^2(\Omega))$  be a *periodic* function with (minimal) period one. For  $n \ge 2$ , we let  $v_n(t) := v_1(t/n)$  for  $t \in \mathbb{R}$ . We have  $v_n \in AP(\mathbb{R}, L^2(\Omega))$  is a periodic function with period n and  $||v_n||_{\infty} = ||v_1||_{\infty}$  for all  $n \in \mathbb{N}$ . If  $\phi$  is also a periodic function, then the substitution  $f(\cdot, v_n(\cdot)) = \phi(\cdot)h(v_n(\cdot))$  is also periodic with larger period as n increases. This means  $\{Q(v_n)\}_{n\in\mathbb{N}}$  cannot be equi-almost periodic. By Lemma 4.4.14, the family  $\{Q(v_n)\}_{n\in\mathbb{N}}$  is not relatively compact in  $AP(\mathbb{R}, L^2(\Omega))$ . Therefore, Q is not completely continuous.

Recall that u is an almost periodic solutions of (4.45) if and only if it is a fixed point of Q. An alternative (rather trivial) approach is using the contraction mapping theorem. This method is used to establish the existence and uniqueness of almost periodic solutions of semilinear non-autonomous problems in [13, Theorem 3.2] under the assumption that the seminlinear term is globally Lipschitz with a sufficiently small Lipschitz constant. Suppose that  $||f(t,\xi_1) - f(t,\xi_2)||_{L^2(\Omega)} \leq k||\xi_1 - \xi_2||_{L^2(\Omega)}$  for all  $\xi_1, \xi_2 \in L^2(\Omega)$ . A straightforward calculation using (4.7) implies that

$$\|Q(u)(t) - Q(v(t))\|_{L^{2}(\Omega)} \le \frac{2M}{\beta}k\|u - v\|_{\infty},$$

for all  $u, v \in AP(\mathbb{R}, L^2(\Omega))$ . Hence, Q is a contraction map if the Lipschitz constant k is small enough. We can prove the convergence of almost periodic solutions under domain perturbation by a similar argument as in Theorem 2.3.3 provided that the family of the maps  $Q_n$  is a uniform contraction and  $Q_n$  converges to Q pointwise. In fact, this method can be applied to any class of bounded entire solutions.

## Bibliography

- R. A. Adams. Sobolev spaces. Academic Press [A subsidiary of Harcourt Brace Jovanovich, Publishers], New York-London, 1975. Pure and Applied Mathematics, Vol. 65.
- [2] H. Amann. Linear and quasilinear parabolic problems. Vol. I, volume 89 of Monographs in Mathematics. Birkhäuser Boston Inc., Boston, MA, 1995. Abstract linear theory.
- [3] J. Appell and P. P. Zabrejko. Nonlinear superposition operators, volume 95 of Cambridge Tracts in Mathematics. Cambridge University Press, Cambridge, 1990.
- [4] W. Arendt. Approximation of degenerate semigroups. Taiwanese J. Math., 5(2):279–295, 2001.
- [5] W. Arendt and D. Daners. Uniform convergence for elliptic problems on varying domains. *Math. Nachr.*, 280(1-2):28–49, 2007.
- [6] J. M. Arrieta. Neumann eigenvalue problems on exterior perturbations of the domain. J. Differential Equations, 118(1):54–103, 1995.
- [7] J. M. Arrieta and S. M. Bruschi. Boundary oscillations and nonlinear boundary conditions. C. R. Math. Acad. Sci. Paris, 343(2):99–104, 2006.
- [8] J. M. Arrieta and S. M. Bruschi. Rapidly varying boundaries in equations with nonlinear boundary conditions. The case of a Lipschitz deformation. *Math. Models Methods Appl. Sci.*, 17(10):1555–1585, 2007.
- [9] J. M. Arrieta and S. M. Bruschi. Very rapidly varying boundaries in equations with nonlinear boundary conditions. The case of a non-uniformly Lipschitz deformation. *Discrete Contin. Dyn. Syst. Ser. B*, 14(2):327–351, 2010.

- [10] J. M. Arrieta and A. N. Carvalho. Spectral convergence and nonlinear dynamics of reaction-diffusion equations under perturbations of the domain. J. Differential Equations, 199(1):143–178, 2004.
- [11] H. Attouch. Variational convergence for functions and operators. Applicable Mathematics Series. Pitman (Advanced Publishing Program), Boston, MA, 1984.
- [12] G. Barbatis, V. I. Burenkov, and P. D. Lamberti. Stability estimates for resolvents, eigenvalues, and eigenfunctions of elliptic operators on variable domains. In Around the research of Vladimir Maz'ya. II, volume 12 of Int. Math. Ser. (N. Y.), pages 23–60. Springer, New York, 2010.
- [13] M. Baroun, S. Boulite, T. Diagana, and L. Maniar. Almost periodic solutions to some semilinear non-autonomous thermoelastic plate equations. J. Math. Anal. Appl., 349(1):74–84, 2009.
- [14] P. W. Bates and C. K. R. T. Jones. Invariant manifolds for semilinear partial differential equations. In *Dynamics reported*, Vol. 2, volume 2 of *Dynam. Report.* Ser. Dynam. Systems Appl., pages 1–38. Wiley, Chichester, 1989.
- [15] P. W. Bates, K. Lu, and C. Zeng. Existence and persistence of invariant manifolds for semiflows in Banach space. *Mem. Amer. Math. Soc.*, 135(645):viii+129, 1998.
- [16] Z. Belhachmi, D. Bucur, and J.-M. Sac-Epee. Finite element approximation of the Neumann eigenvalue problem in domains with multiple cracks. *IMA J. Numer. Anal.*, 26(4):790–810, 2006.
- [17] C. M. Blázquez. Transverse homoclinic orbits in periodically perturbed parabolic equations. Nonlinear Anal., 10(11):1277–1291, 1986.
- [18] H. Brézis. Problèmes unilatéraux. J. Math. Pures Appl. (9), 51:1–168, 1972.
- [19] D. Bucur. Un théorème de caractérisation pour la  $\gamma$ -convergence. C. R. Acad. Sci. Paris Sér. I Math., 323(8):883–888, 1996.
- [20] D. Bucur and G. Buttazzo. Variational methods in shape optimization problems. Progress in Nonlinear Differential Equations and their Applications, 65. Birkhäuser Boston Inc., Boston, MA, 2005.
- [21] D. Bucur and N. Varchon. Boundary variation for a Neumann problem. Ann. Scuola Norm. Sup. Pisa Cl. Sci. (4), 29(4):807–821, 2000.

- [22] D. Bucur and N. Varchon. A duality approach for the boundary variation of Neumann problems. SIAM J. Math. Anal., 34(2):460–477 (electronic), 2002.
- [23] D. Bucur and J.-P. Zolésio. N-dimensional shape optimization under capacitary constraint. J. Differential Equations, 123(2):504–522, 1995.
- [24] V. I. Burenkov and E. B. Davies. Spectral stability of the Neumann Laplacian. J. Differential Equations, 186(2):485–508, 2002.
- [25] V. I. Burenkov and P. D. Lamberti. Spectral stability of general non-negative selfadjoint operators with applications to Neumann-type operators. J. Differential Equations, 233(2):345–379, 2007.
- [26] V. I. Burenkov and P. D. Lamberti. Spectral stability of Dirichlet second order uniformly elliptic operators. J. Differential Equations, 244(7):1712–1740, 2008.
- [27] G. Buttazzo and G. Dal Maso. Shape optimization for Dirichlet problems: relaxed formulation and optimality conditions. *Appl. Math. Optim.*, 23(1):17–49, 1991.
- [28] C. Corduneanu. Almost periodic functions. Interscience Publishers [John Wiley & Sons], New York-London-Sydney, 1968. With the collaboration of N. Gheorghiu and V. Barbu, Translated from the Romanian by Gitta Bernstein and Eugene Tomer, Interscience Tracts in Pure and Applied Mathematics, No. 22.
- [29] R. Courant and D. Hilbert. Methods of mathematical physics. Vol. I. Interscience Publishers, Inc., New York, N.Y., 1953.
- [30] G. dal Maso, F. Ebobisse, and M. Ponsiglione. A stability result for nonlinear Neumann problems under boundary variations. J. Math. Pures Appl. (9), 82(5):503– 532, 2003.
- [31] E. N. Dancer. The effect of domain shape on the number of positive solutions of certain nonlinear equations. J. Differential Equations, 74(1):120–156, 1988.
- [32] E. N. Dancer. The effect of domain shape on the number of positive solutions of certain nonlinear equations. II. J. Differential Equations, 87(2):316–339, 1990.
- [33] E. N. Dancer. A Conley index calculation. Bull. Aust. Math. Soc., 80(3):510–520, 2009.

- [34] E. N. Dancer and D. Daners. Domain perturbation for elliptic equations subject to Robin boundary conditions. J. Differential Equations, 138(1):86–132, 1997.
- [35] D. Daners. Domain perturbation for linear and nonlinear parabolic equations. J. Differential Equations, 129(2):358–402, 1996.
- [36] D. Daners. Heat kernel estimates for operators with boundary conditions. Math. Nachr., 217:13–41, 2000.
- [37] D. Daners. Dirichlet problems on varying domains. J. Differential Equations, 188(2):591–624, 2003.
- [38] D. Daners. Perturbation of semi-linear evolution equations under weak assumptions at initial time. J. Differential Equations, 210(2):352–382, 2005.
- [39] D. Daners. Domain perturbation for linear and semi-linear boundary value problems. In Handbook of differential equations: stationary partial differential equations. Vol. VI, Handb. Differ. Equ., pages 1–81. Elsevier/North-Holland, Amsterdam, 2008.
- [40] D. Daners and P. Koch Medina. Abstract evolution equations, periodic problems and applications, volume 279 of Pitman Research Notes in Mathematics Series. Longman Scientific & Technical, Harlow, 1992.
- [41] R. Dautray and J.-L. Lions. Mathematical analysis and numerical methods for science and technology. Vol. 5. Springer-Verlag, Berlin, 1992. Evolution problems.
  I, With the collaboration of Michel Artola, Michel Cessenat and Hélène Lanchon, Translated from the French by Alan Craig.
- [42] E. A. M. de Abreu and A. N. Carvalho. Attractors for semilinear parabolic problems with Dirichlet boundary conditions in varying domains. *Mat. Contemp.*, 27:37–51, 2004.
- [43] K. Deimling. Nonlinear functional analysis. Springer-Verlag, Berlin, 1985.
- [44] N. Dunford and J. T. Schwartz. *Linear operators. Part I.* Wiley Classics Library. John Wiley & Sons Inc., New York, 1988. General theory, With the assistance of William G. Bade and Robert G. Bartle, Reprint of the 1958 original, A Wiley-Interscience Publication.

- [45] E. Feireisl and H. Petzeltová. On the domain dependence of solutions to the twophase Stefan problem. Appl. Math., 45(2):131–144, 2000.
- [46] N. Fenichel. Persistence and smoothness of invariant manifolds for flows. Indiana Univ. Math. J., 21:193–226, 1971/1972.
- [47] A. M. Fink. Almost periodic differential equations. Lecture Notes in Mathematics, Vol. 377. Springer-Verlag, Berlin, 1974.
- [48] R. D. Grigorieff. Diskret kompakte Einbettungen in Sobolewschen Räumen. Math. Ann., 197:71–85, 1972.
- [49] J. Hadamard. Sur l'équilibre des plaques élastiques circulaires libres ou appuyées et celui de la sphère isotrope. Ann. Sci. École Norm. Sup. (3), 18:313–342, 1901.
- [50] J. K. Hale. Introduction to dynamic bifurcation. In Bifurcation theory and applications (Montecatini, 1983), volume 1057 of Lecture Notes in Math., pages 106–151. Springer, Berlin, 1984.
- [51] J. K. Hale and J. Vegas. A nonlinear parabolic equation with varying domain. Arch. Rational Mech. Anal., 86(2):99–123, 1984.
- [52] D. Henry. Geometric theory of semilinear parabolic equations, volume 840 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 1981.
- [53] D. Henry. Perturbation of the boundary in boundary-value problems of partial differential equations, volume 318 of London Mathematical Society Lecture Note Series. Cambridge University Press, Cambridge, 2005. With editorial assistance from Jack Hale and Antônio Luiz Pereira.
- [54] M. W. Hirsch, C. C. Pugh, and M. Shub. Invariant manifolds. Bull. Amer. Math. Soc., 76:1015–1019, 1970.
- [55] K. Ito and K. Kunisch. Parabolic variational inequalities: the Lagrange multiplier approach. J. Math. Pures Appl. (9), 85(3):415–449, 2006.
- [56] S. Jimbo. The singularly perturbed domain and the characterization for the eigenfunctions with Neumann boundary condition. J. Differential Equations, 77(2):322– 350, 1989.

- [57] T. Kato. Perturbation theory for linear operators. Springer-Verlag, Berlin, second edition, 1976. Grundlehren der Mathematischen Wissenschaften, Band 132.
- [58] M. V. Keldyš. On the solvability and stability of the Dirichlet problem. Uspekhi Matem. Nauk, 8:171–231 (English Translation: Amer. Math. Soc. Translations (2)51 (1966) 173), 1941.
- [59] A. Liapounoff. Problème Général de la Stabilité du Mouvement. Annals of Mathematics Studies, no. 17. Princeton University Press, Princeton, N. J., 1947.
- [60] J.-L. Lions. Quelques méthodes de résolution des problèmes aux limites non linéaires. Dunod, 1969.
- [61] Z. L. Lions. Partial differential inequalities. Uspehi Mat. Nauk, 26(2(158)):205-263, 1971.
- [62] N. G. Lloyd. Degree theory. Cambridge University Press, Cambridge, 1978. Cambridge Tracts in Mathematics, No. 73.
- [63] R. Mañé. Persistent manifolds are normally hyperbolic. Trans. Amer. Math. Soc., 246:261–283, 1978.
- [64] J. Mierczyński and W. Shen. Spectral theory for random and nonautonomous parabolic equations and applications, volume 139 of Chapman & Hall/CRC Monographs and Surveys in Pure and Applied Mathematics. CRC Press, Boca Raton, FL, 2008.
- [65] U. Mosco. Convergence of convex sets and of solutions of variational inequalities. Advances in Math., 3:510–585, 1969.
- [66] A. Pazy. Semigroups of linear operators and applications to partial differential equations, volume 44 of Applied Mathematical Sciences. Springer-Verlag, New York, 1983.
- [67] O. Perron. Die Stabilitätsfrage bei Differentialgleichungen. Math. Z., 32(1):703– 728, 1930.
- [68] M. Prizzi and K. P. Rybakowski. The effect of domain squeezing upon the dynamics of reaction-diffusion equations. J. Differential Equations, 173(2):271–320, 2001.

- [69] J. Prüss. On the spectrum of C<sub>0</sub>-semigroups. Trans. Amer. Math. Soc., 284(2):847– 857, 1984.
- [70] J. Rauch and M. Taylor. Potential and scattering theory on wildly perturbed domains. J. Funct. Anal., 18:27–59, 1975.
- [71] W. Rudin. Functional analysis. McGraw-Hill Book Co., New York, 1973. McGraw-Hill Series in Higher Mathematics.
- [72] G. Savaré and G. Schimperna. Domain perturbations and estimates for the solutions of second order elliptic equations. J. Math. Pures Appl. (9), 81(11):1071–1112, 2002.
- [73] F. Simondon. Domain perturbation for parabolic quasilinear problems. Commun. Appl. Anal., 4(1):1–12, 2000.
- [74] D. R. Smart. Fixed point theorems. Cambridge University Press, London, 1974. Cambridge Tracts in Mathematics, No. 66.
- [75] F. Stummel. Diskrete Konvergenz linearer Operatoren. I. Math. Ann., 190:45–92, 1970/71.
- [76] F. Stummel. Diskrete Konvergenz linearer Operatoren. II. Math. Z., 120:231–264, 1971.
- [77] F. Stummel. Diskrete Konvergenz linearer Operatoren. III. In *Linear operators and approximation (Proc. Conf., Oberwolfach, 1971)*, pages 196–216. Internat. Ser. Numer. Math., Vol. 20. Birkhäuser, Basel, 1972.
- [78] F. Stummel. Perturbation theory for Sobolev spaces. Proc. Roy. Soc. Edinburgh Sect. A, 73:5–49, 1975.
- [79] V. Šverák. On optimal shape design. J. Math. Pures Appl. (9), 72(6):537–551, 1993.
- [80] T. Yoshizawa. Stability theory and the existence of periodic solutions and almost periodic solutions. Springer-Verlag, New York, 1975. Applied Mathematical Sciences, Vol. 14.
- [81] E. Zeidler. Nonlinear functional analysis and its applications. II/A. Springer-Verlag, New York, 1990. Linear monotone operators, Translated from the German by the author and Leo F. Boron.

- [82] W. Zeng. Transversality of homoclinic orbits and exponential dichotomies for parabolic equations. J. Math. Anal. Appl., 216(2):466–480, 1997.
- [83] W. Zhang and I. Stewart. Bounded solutions for non-autonomous parabolic equations. Dynam. Stability Systems, 11(2):109–120, 1996.